Publication No. FHWA-RD-94-042 September 1994



A Simplified Field Method for Capacity Evaluation of Driven Piles



i

U.S. Department of Transportation Federal Highway Administration

Research and Development Turner-Fairbank Highway Research Center 6300 Georgetown Pike McLean, Virginia 22101-2296

> REPRODUCED BY: NTTS. U.S. Department of Commerce National Technical Information Service Springfield, Virginia 22161

FOREWORD

This report presents the results of a study on a simplified field method for the capacity evaluation of driven piles based on dynamic measurements during driving. The simplified method, entitled the Energy Approach is proposed as an alternative to other dynamic analyses that are based on one-dimensional wave equation solutions. Based on the analysis of a large data set of over 120 piles load test to failure with corresponding dynamic measurements, and an additional 403 PDA monitored piles, the authors found the Energy Approach method provided excellent evaluations of pile capacity.

Charles J. Nemmers, P.E.

Charles J. Nemmers, P.E. Director, Office of Engineering and Highway Operations Research and Development

NOTICE

This document is disseminated under the sponsorship of the Department of Transportation in the interest of information exchange. The United States Government assumes no liability for its contents or use thereof. The contents of this report reflect the views of the contractor, who is responsible for the accuracy of the data presented herein. The contents do not cecessarily reflect the official policy of the Department of Transportation. This report does not constitute a standard, specification, or regulation.

The United States Government does not endorse products or manufacturers. Trade or manufacturers' names appear herein only because they are considered essential to the object of this document.

Technical Report Documentation Page

1. Report No. 2. 111 111 111	3. Recipient's Catalog No.
FHWA-RD-94-042	7 - 208490
4. Title and Subtitle	5. Report Date
	September 1994
A SIMPLIFIED FIELD METHOD FOR CAPACITY EVALUATION OF DRIVEN PILES	6. Performing Organization Code
	8. Performing Organization Report No.
Samuel G. Paikowsky, John E. Regan, and John J. McD	Donnell
9. Performing Organization Name and Address	10. Work Unit No. (TRAIS)
University of Massachusetts Lowell	<u>3E3A-0922</u>
Department of Civil Engineering	11. Contract or Grant No.
1 University Avenue	DTFH61-92-C-00038
Lowell, Massachusetts 01854	13. Type of Report and Period Covered
12. Sponsoring Agency Name and Address	Final Report
Highway Operations R&D	May 1992 - June 1993
6300 Georgetown Pike	
McLean, Virginia 22101-2296	14. Sponsoring Agency Code
15. Supplementary Notes	
Contracting Officer's Technical Representative (COTR)	- Carl Ealy, HNR-30
16. Abstract	
done by the pile/soil systems is proposed. This method, plastic load displacement pile-soil relations. Calculated from the measured data, together with the field blow co does not consider the propagation process and is aimed the field.	, entitled the Energy Approach, assumes elasto- transferred energy and maximum pile displacement ount, are used as input parameters. This method at providing a real-time pile-capacity prediction in
Two large data sets were gathered at the University of M 208 dynamic measurement cases on 120 piles monitored failure. The data were obtained from various sources ar systems. The other, PD, contains data on 403 piles mon Dynamics Inc. of Cleveland, Ohio. All cases were exam	Massachusetts at Lowell. One, PD/LT, contains during driving, followed by a static load test to nd reflect variable combinations of soil-pile-driving aitored during driving and was provided by Pile ined and analyzed.
The Energy Approach method was found to provide exc conditions. The method is, therefore, proposed to be us determination. The predictions of this method were fou evaluations than the sophisticated office methods, especi driving. The Energy Approach is, therefore, also propos the office methods.	cellent evaluations of pile capacity under all sed in the field for instantaneous capacity and on the average to provide more accurate ially for records obtained at the end of initial sed to be used as an independent tool to evaluate
17. Key Words	18. Distribution Statement
Driven piles, dynamic analysis of piles, Energy Approach, CAPWAP, TEPWAP, driven-pile capacity.	No restrictions. This document is available to the public through the National Technical Information Service, Springfield, Virginia 22161.
19. Security Classif. (of this report) 20. Security Clas	sit. (of this page) 21. No. of Pages 22. Price
Unclassified Uncla	assified 313
Form DOT F 1700 7 (8-72) Bearraugian of com	

.

PREFACE

This research study presents a simplified field method for the capacity evaluation of driven piles based on dynamic measurements during driving.

Dynamic analyses of piles are methods aimed at the prediction of pile behavior under static loads based on the pile response during installation. These methods are based upon the concept that pile penetration under each blow induces failure of the soil, hence, an instantaneous load test is performed.

The reliability of these analyses is enhanced through data obtained by dynamic measurements during driving. Two methods are currently employed for the analysis of the measured data. Both methods are based on the solution of the one-dimensional wave equation for the stress wave traveling through the pile following the hammer's impact. One, an office analysis, utilizes a numerical solution of a mathematical model for the pile-soil system under measured boundary conditions (e.g., the computer codes CAPWAP or TEPWAP). The other, a field analysis known as the "Case Method," which is based on a simplified closed-form solution and empirical correlations, provides an instantaneous evaluation of the pile capacity following each hammer blow.

Substantial experience suggests the existence of major limitations to the field method. In addition, no large-scale evaluation has been carried out for the office methods since their development.

A simplified method based on energy balance is proposed as an alternative field method. This method, entitled the Energy Approach, assumes elasto-plastic load displacement pile-soil relations. Calculated transferred energy and maximum pile displacement from the measured data, together with the field blow count, are used as input parameters for the Energy Approach.

Two large data sets were gathered at the University of Massachusetts at Lowell. One, PD/LT, contains 208 dynamic measurement cases on 120 piles monitored during driving, followed by a static load test to failure. The data were obtained from various sources and reflect variable combinations of soil-pile-driving systems. The other, PD, contains data on 403 piles monitored during driving and was provided by Pile Dynamics, Inc. of Cleveland, Ohio. All cases were examined and analyzed.

The results of the presented study invalidate the concept of a unique recommended correlation between the viscous damping parameters and soil type in both wave-based analyses. It is shown that energy losses should be attributed more to soil inertia rather than soil damping. As such, energy losses are mostly pile-shape-dependent in addition to the soil type and driving resistance influences.

The Energy Approach method was found to provide excellent evaluations of pile capacity. Therefore, the method is proposed to be used in the field for instantaneous capacity determination. The predictions of this method were found, on the average, to provide more accurate evaluations than the sophisticated office methods, especially for records obtained at the end of initial driving. The Energy Approach is, therefore, also proposed to be used as an independent tool to evaluate the office methods.

Through evaluation of the current dynamic analyses, pointing out their sources of deficiencies and offering an alternative method, this study contributed to the increase in safety and decrease in cost of driven-pile foundation systems.

		SI* (MO	DERN MET	RIC)	CONVER	SION FACTO	RS		
	VPPROXIMATE CO	NVERSIONS TC) SI UNITS			APPROXIMATE COI	NVERSIONS FR	OM SI UNITS	
Symbol	When You Know	Multiply By	To Find	Symbol	Symbol	When You Know	Multiply By	To Find	Symbol
		LENGTH					LENGTH		
.s	inches	25.4	millimeters	E	EE	millimeters	0.039	inches	.9
Ħ	feet	0.305	meters	ε	ε	meters	3.28	feet	Ħ
Ď.E	yards miles	0.914 1.61	meters kilomatars	E	ЕŻ	meters kilometers	1.09 0.621	yards miles	Ъ.Ę
		AREA					AREA		
Ĩ <u>2</u> Ĩ	square inches	645.2	square millimeters	mm²	mm²	square millimeters	0.0016	square inches	in²
¥ :	square teet	0.093	square meters	п2	Ē	square meters	10.764	square feet	ž
ъ.	square yards	0.636	square meters	m.	Ě	square meters	C61.1	square yards	ž Ž
Ē	souare miles	2.59	Nectares sourare kilometere	ha Lm2	km²	souare kilometers	0.386	acres souare miles	ani ^a
		VOLUME					VOLUME	-	
q		1.00			-				
20 L	fluid ounces	29.57	milliters	Ē.	۔ ۲	milliters	0.034	fluid ounces	70 -
981 1	galions	3./85	liters	'	_ `		0.264	gallons	gal
сц.	cubic reet cubic vards	0.765	cubic meters	ÊÊ	ĒĒ	cubic meters cubic meters	1.307	cubic feet cubic vards	<u>م</u> لاً ا
2			cubic meters				100.1		2
NOTE: V	/olumos greater than 100	00 I shall be shown in	ш ³ .						
		MASS					MASS		
02	seouno	28.35	grams		5	grams	0.035	Seonno	02
ہ ا	spunod	0.454	kilograms	, 5 5	2:	kilograms	2.202	spunod	<u>۔</u> و
⊢ -	short tons (2000 lb)	0.907	megagrams (or "metric too")	Mg	Mg (or "t)	megagrams (or "metric ton")	1.103	short tons (2000	 (q
	TEMPER	ATURE (exact)		2		TEMPE	RATURE (exact	(1	
Å.	Fahrenheit temperature	5(F-32)/9 or (F-32)/1.8	Celcius temperature	ပ္	ပ္	Celcius temperature	1.8C + 32	Fahrenheit temperature	ų.
	וררר	IMINATION				F	LUMINATION		
<u>ہ</u> ک	toot-candles	10.76 3.426	lux dala/2	×	: ح	lux coordolo/m2	0.0929	foot-candles	Ļ,
:				COLM	-EQM-		0.2313		=
	FORCE and PI	RESSURE or ST	RESS			FORCE and I	PRESSURE or S	TRESS	
19	poundforce	4.45	newtons	z	z	newtons	0.225	poundiorce	lef I
lbl/in ²	poundiorce per square inch	6.89	kilopascals	kРа	kPa	kilopascals	0.145	poundforce per square inch	bt/in ²
 SI is the syn rounding sh 	nboi for the International ould be made to comply	System of Units. Apr with Section 4 of AS	propriate TM E380					(Revised Septemb	ər 1993)

iv

TABLE OF CONTENTS

1 - INTRODUCTION	1
1.1 OVERVIEW	1
1.2 THE PRESENT RESEARCH STUDY.	2
1.3 CONTRIBUTIONS	3
1.4 MANUSCRIPT LAYOUT	3
2 - BACKGROUND	7
2.1 GENERAL	7
2.2 STATIC ANALYSIS	7
2.3 STATIC LOAD TESTS	8
3 - DVNAMIC ANALYSIS OF PILES	0
3 1 GENERAL	ó
3.2 DYNAMIC FOUATIONS	ó
3.2.1 Review	ó
3.2.2 The Basic Principle	10
3 2.3 Energy Transfer	10
3.3 THE WAVE EQUATION	12
3.3.1 Formulation and Principles	12
3.3.2 Pre-Driving Analysis	16
3.3.3 Post-Driving Analysis - CAPWAP/TEPWAP	16
3.3.4 Wave Equation Analysis - Discussion	16
3.4 FIELD ANALYSIS AND THE PILE-DRIVING ANALYZER	19
3.5 THE CASE METHOD	19
3.5.1 General	19
3.5.2 The Case Method Equation	19
3.5.3 Case Damping Coefficient	20
3.5.4 Case Method Variations	23
(a) The Damping Factor Method, RSP	23
(b) The Maximum Resistance Method, RMX	23
(c) The Minimum Resistance Method, RMN	24
(d) The Unloading Method, RSU	24
(e) The Automatic Method, RAU	24
3.5.5 Evaluation	25
(a) Critical Discussion	25
(b) Review of Existing Experience	25
3.5.6 Capacity Predictions	28
3.5.7 Summary	29
4 . THE ENERGY APPROACH	31
4.1 RACKGROIND	31

4.2 UNDERLYING CONCEPT	32
4.3 THE ENERGY EQUATION	32
4.4 ENERGY LOSSES AND SOIL INERTIA	34
4.4.1 General Considerations	34
4.4.2 Soil Displacement	34
4.4.3 Soil Acceleration	36
4.4.4 Expected Performance	37
5 - DATA BASE BUILDUP	39
5.1 GENERAL	39
5.2 DATA SET PD/LT	39
5.2.1 Static Load Test Analysis	39
(a) Davisson's Criteria	41
(b) The Shape-of-Curve Method	41
(c) The Limited Total Settlement Methods	43
(d) DeBeer's log-log Method	43
(e) The Representive Static Capacity	43
5.2.2 Dynamic Measurements Analysis	43
(a) GROUP 1 - Complete CAPWAP Analyses	45
(b) GROUP 2 - Incomplete CAPWAP Analyses	47
(c) GROUP 3 - TEPWAP Analyses	50
5.3 DATA SET PD	55
6 - DATA SET PD/LT	57
6.1 GENERAL	57
6.2 SITE AND PILE INFORMATION · TABLE 20	57
(a) Columns 1-4	57
(b) Columns 5-8	57
(b) Columns 9 and 10	58
6.3 PILE DRIVING AND DYNAMIC MEASUREMENTS - TABLE 21	58
(a) Columns 1 and 2	58
(b) Columns 3-5	58
(c) Column 6	59
(d) Columns 7-10	59
6.4 PARAMETERS OF DYNAMIC ANALYSES - TABLE 22	60
(a) Columns 1 and 2	60
(b) Column 3	60
(c) Columns 4 and 5	60
(d) Columns 6-9	60
6.5 PILE CAPACITY: STATIC LOAD TEST AND	
DYNAMIC ANALYSES - TABLE 23	60
(a) Columns 1-3	60 60
(a) Columns 1-3	60 60 61

(a) Columns 10-12	
7 - DATA SET PD	6
7.1 PILE/SOIL AND DYNAMIC	MEASUREMENTS OF DATA SET
PD • TABLE 24	
(a) Columns 1 and	1 2 63
(b) Columns 3 and	i 4 63
(c) Columns 5-9	
(d) Columns 10-14	6.
(e) Columns 15 an	
7.2 SIDE/ HP QUAKE AND DA	WIPING PARAMETERS OF DATA
(a) Columns 6 and	17
(a) Columns 8 and	1 9
(0) Condition of and	· · · · · · · · · · · · · · · · · · ·
8 - ANALYSIS OF DATA SET PD/LT	
8.1 OVERVIEW	
8.1.1 Purpose	
8.1.2 Outline	
(a) Damping Para	meters-Soil Type Correlations
(b) Prediction Met	chods-Load Test Capacity
(c) Office Method,	AND COLL TYPE
8.2 DAMPING PARAMETERS A	AND SOIL TIPE
8 2 1 Case Method Damn	ing 6
8.2.2 Smith Damping	6
8.3 DYNAMIC PREDICTIONS-	STATIC CAPACITY GRAPHICAL
CORRELATIONS	
8.3.1 Correlations Break	lown
8.3.2 Pile Type Correlation	o ns
(a) All Piles	
(b) Large Displace	ement Piles
(c) Small Displace	ement Piles
(d) Intermediate (Conclusions
8.3.3 Pile-Soli Type Corre	? ליוויייייייייייייייייייייייייייייייייי
(a) Sanu anu Shi. (b) Clov and Till	/ 7
(c) Rock	7
(d) Intermediate (Conclusions 7
8.3.4 Correlations of Pile	and Soil Type for Different Driving Time 7
(a) All Piles - EO	D
(b) All Piles - BO	R 8
(c) Large Displace	ement Piles - EOD
(d) Large Displace	ement Piles - BOR
	vii

(e) Small Displacement Piles - EOD	81
(f) Small Displacement Piles - BOR	82
(g) Intermediate Conclusions	82
8.4 STATISTICAL ANALYSIS OF DATA SET PD/LT	83
8.4.1 Linear-Regression Analysis	84
8.4.2 Actual Distributions of the K Coefficients and	
their Probabilistic Models	85
8.4.3 Mean and Standard Deviation Analysis	87
8.5 INTERPRETATION OF THE CONTROLLING PARAMETERS	88
8.5.1 Overview	88
8.5.2 Dynamic Predictions - Pile Area Ratio Graphical	
Correlations	88
8.5.3 Dynamic Predictions - Driving Resistance Graphical	
Correlations	88
8.5.4 Dynamic Predictions - Driving Resistance and	
Time of Driving Graphical Correlations	92
(a) All Piles at EOD	92
(b) All Piles at BOR	93
8.5.5 Dynamic Predictions - Driving Resistance and	
Pile-Type Graphical Correlations	93
(a) Small Displacement Piles	93
(b) Large Displacement Piles	94
8.5.6 The Effect of the Combined Major Controlling Parameters	
on the Accuracy of the Dynamic Predictions	95
(a) Breakdown of Combinations	95
(b) Combinations of Pile Type and Driving Resistance	95
(c) Combinations of Pile Type, Driving	
Resistance, and Time of Driving	97
ANALYSIS OF DATA SET PD	181
9.1 INTRODUCTION	181
9.1.1 Purpose	181
9.1.2 Overview	181
(a) Damping Parameters - Soil-Type Correlations	181
(b) Office Method - Field Method Predictions	181
9.2 SMITH DAMPING PARAMETERS AND SOIL-TYPE	
CORRELATIONS	181
9.3 CAPWAP AND THE ENERGY APPROACH CORRELATIONS	182
9.3.1 All Piles - All Soils	182
0.2.2 Lange Dignlagement Biles	100
Jarge Displacement Piles	182
(a) All Cases	182 182
(a) All Cases	182 182 182
(a) All Cases (b) Sand and Silt (c) Clay and Till	182 182 182 183

.

9 -

•

(e) Unknown Soil Type	183
(f) Intermediate Conclusions	183
9.3.3 Small Displacement Piles	183
(a) All Cases	183
(b) Sand and Silt	184
(c) Clay and Till	184
(d) Rock	184
(e) Intermediate Conclusions	184
9.3.4 Miscellaneous Piles	184
(a) All Cases	185
(b) Sand and Silt	185
(c) Clay and Till	185
(d) Rock	185
(e) Unknown Soil Type	185
9.4 STATISTICAL ANALYSIS OF DATA SET PD	185
9.4.1 Linear Regression Analysis	187
9.4.2 Mean and Standard Deviation Analysis	187
9.5 SUMMARY AND CONCLUSIONS	187
10 SUDDIADY CONCLUSIONS AND DECONDUCTIONS	200
10.1 SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS	209
	209
10.2 CUNCLUSIONS	211
10.3 RECOMMENDATIONS	215
10.2.2 The Derfermence of the Office	215
Mothe de (CADUIAD / TEDUIAD)	214
Methods (CAPWAP/TEPWAP)	210
10.3.4 The Completion Detuger the Office	210
Nethods and the Energy Approach	017
Methods and the Energy Approach	217
10.5.5 Factors of Safety and Kisk Analysis	217
(a) General.	21/
(D) Adsolute Salety based on Data Set PD/L1	218
(c) Factor of Salety and Ing Associated Risk	
Based on the Actual Data	243
(a) Factor of Salety and the Associated Kisk Based on the Drobability Distribution Eurotion	224
10.2.6 Decompositions for Implementation	224
TV Recommendations for implementation	223
APPENDIX A - DATA SET PD/LT	235
APPENDIX B - DATA SET PD	263
REFERENCES	285

LIST OF FIGURES

<u>FIGU</u>	RE	<u>PAGE</u>
1	Resistance vs. displacement at the top of the pile	. 11
2	Smith's model simulating the hammer-pile-soil system for use with the one-dimensional wave equation (Smith, 1960)	. 14
3	Soil-pile model (left) and the corresponding elasto-plastic soil resistance-displacement relationship (after Smith, 1960)	. 15
4	Notations used for model of pile and soil in TEPWAP analysis (Paikowsky, 1982)	. 17
5	Flow chart describing the analysis process using TEPWAP (Paikowsky, 1982)	. 18
6	Force and velocity traces showing two impact peaks indicative of driving in soils capable of large deformations	. 21
7	Case damping (J_c) values for capacity prediction of offshore piles in the range of ± 20 percent from load test results (after Paikowsky, 1982)	. 27
8	The proposed way of obtaining the combined quake, Q (soil and pile)	. 35
9	Load-settlement curve of pile-case 95 with the elastic compression line inclined at 20 degrees	. 40
10	Load-settlement curve of pile-case 95 with a scale that does not consider the elastic compression of the pile (following Vesic, 1977)	40
11	Load-settlement curve for pile-case 50 with the elastic compression line inclined at approximately 20 degrees	42
12	Load-settlement data plotted on a logarithmic graph for pile-case 50 to determine the failure load according to DeBeer's method	44
13	Force and velocity (V*EA/C) traces of pile-case 1, a steel HP12x74 that needed a force correction (not to scale)	46

xi Preceding page blank

14	Digitized force and velocity multiplied by the impedance (EA/C) traces for pile-case 192 used for input into INTEGRATE	48
15	INTEGRATE output of pile-case 192 showing the back-calculated Case J_c value and the Energy Approach prediction	49
16	Example of the pile identification information of pile-case 191 used as input for the TEPWAP analysis	51
1 7	Example of the soil and pile properties used along the pile elements of pile-case 191 as input for the TEPWAP analysis	52
18	Measured force and velocity multiplied by the impedance (EA/C) traces of pile-case 191 used by the TEPWAP analysis	53
19	Comparison between measured force near the top of pile-case 191 and the calculated force from TEPWAP analysis	53
20	Summary of the final results from TEPWAP analysis performed on pile-case 191	54
21	Tip soil conditions vs. calculated case damping coefficient (J_c) based on static load test results for 208 PD/LT pile-cases	99
22	Side soil conditions vs. Smith side damping coefficient based on CAPWAP/TEPWAP results	100
23	Tip soil conditions vs. Smith tip damping coefficient based on CAPWAP/TEPWAP results	101
24	Static load test results vs. CAPWAP or TEPWAP predictions for 204 PD/LT pile-cases in all type of soil (AAA)	102
25	Static load test results vs. Energy Approach predictions for 202 PD/LT pile-cases in all types of soil (AAA)	103
26	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 201 PD/LT pile-cases in all types of soil (AAA)	104
27	K _{sw} vs. CAPWAP/TEPWAP predictions for 206 PD/LT pile-cases in all types of soil (AAA)	105
28	K _{sp} vs. Energy Approach predictions for 208 PD/LT pile-cases in all types of soil (AAA)	106

29	Static load test results vs. CAPWAP or TEPWAP predictions for 162 large displacement PD/LT pile-cases in all types of soil (LAA)	107
30	Static load test results vs. Energy Approach predictions for 163 large displacement PD/LT pile-cases in all types of soil (LAA)	108
31	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 161 large displacement PD/LT pile-cases in all types of soil (LAA)	109
32	Static load test results vs. CAPWAP or TEPWAP predictions for 42 small displacement PD/LT pile-cases in all types of soil (SAA)	110
33	Static load test results vs. Energy Approach predictions for 40 small displacement PD/LT pile-cases in all types of soil (SAA)	111
34	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 38 small displacement PD/LT pile-cases in all types of soil (SAA)	112
35	Static load test results vs. CAPWAP or TEPWAP predictions for 139 PD/LT pile-cases in sand and silt (AAS)	113
36	Static load test results vs. Energy Approach predictions for 136 PD/LT pile-cases in sand and silt (AAS)	114
37	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 136 PD/LT pile-cases in sand and silt (AAS)	115
38	Static load test results vs. CAPWAP or TEPWAP predictions for 51 PD/LT pile-cases in clay and till (AAC)	116
39	Static load test results vs. Energy Approach predictions for 53 PD/LT pile-cases in clay and till (AAC)	117
40	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 51 PD/LT pile-cases in clay and till (AAC)	118
41	Static load test results vs. CAPWAP or TEPWAP predictions for 14 PD/LT pile-cases in rock (AAR)	119
42	Static load test results vs. Energy Approach predictions for 14 PD/LT pile-cases in rock (AAR)	120
43	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 14 PD/LT pile-cases in rock (AAR)	12 1

44	Static load test results vs. CAPWAP or TEPWAP predictions for 96 PD/LT pile-cases in all types of soil at EOD (AEA)	122
45	Static load test results vs. Energy Approach predictions for 94 PD/LT pile-cases in all types of soil at EOD (AEA)	123
46	K _{sw} vs. CAPWAP/TEPWAP predictions for 97 PD/LT pile-cases at EOD in all types of soil (AEA)	124
47	K _{sp} vs. Energy Approach predictions for 98 PD/LT pile-cases at EOD in all types of soil (AEA)	125
48	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 94 PD/LT pile-cases in all types of soil at EOD (AEA)	126
49	Static load test results vs. CAPWAP or TEPWAP predictions for 108 PD/LT pile-cases in all types of soil at BOR (ABA)	127
50	Static load test results vs. Energy Approach predictions for 108 PD/LT pile-cases in all types of soil at BOR (ABA)	128
51	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 108 PD/LT pile-cases in all types of soil at BOR (ABA)	129
52	Static load test results vs. CAPWAP or TEPWAP predictions for 68 large displacement PD/LT pile-cases in all types of soil at EOD (LEA)	130
53	Static load test results vs. Energy Approach predictions for 69 large displacement PD/LT pile-cases in all types of soil at EOD (LEA)	131
54	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 68 large displacement PD/LT pile-cases in all types of soil at EOD (LEA)	132
55	Static load test results vs. CAPWAP or TEPWAP predictions for 94 large displacement PD/LT pile-cases in all types of soil at BOR (LBA)	133
56	Static load test results vs. Energy Approach predictions for 94 large displacement PD/LT pile-cases in all types of soil at BOR (LBA)	134

57	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 93 large displacement PD/LT pile-cases in all types of soil at BOR (LBA)	135
58	Static load test results vs. CAPWAP or TEPWAP predictions for 22 small displacement PD/LT pile-cases in all types of soil at EOD (SEA)	136
59	Static load test results vs. Energy Approach predictions for 20 small displacement PD/LT pile-cases in all types of soil at EOD (SEA)	137
60	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 20 small displacement PD/LT pile-cases in all types of soil at EOD (SEA)	138
61	Static load test results vs. CAPWAP or TEPWAP predictions for 12 small displacement PD/LT pile-cases in all types of soil at BOR (SBA)	139
62	Static load test results vs. Energy Approach predictions for 12 small displacement PD/LT pile-cases in all types of soil at BOR (SBA)	1 40
63	CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 12 small displacement PD/LT pile-cases in all types of soil at BOR (SBA)	141
64	Histogram and frequency distribution of K_{sw} for 206 PD/LT pile-cases in all types of soil (AAA)	142
65	Cumulative frequency distribution of K _{sw} for 206 PD/LT pile-cases in all types of soil (AAA)	143
66	Histogram and frequency distribution of K_{sp} for 208 PD/LT pile-cases in all types of soil (AAA)	144
67	Cumulative frequency distribution of K_{sp} for 208 PD/LT pile-cases in all types of soil (AAA)	145
68	Histogram and frequency distribution of K _{ew} for 206 PD/LT pile-cases in all types of soil (AAA)	146
69	K_{sw} vs. the pile area ratio (A_R) for 201 PD/LT pile-cases in all types of soil	147

7 0	K_{sw} vs. the pile area ratio (A _R) for 201 PD/LT pile-cases in all types of soil (logarithmic scale)	148
71	K_{sp} vs. the pile area ratio (A _R) for 203 PD/LT pile-cases in all types of soil	149
72	K_{sp} vs. the pile area ratio (A _R) for 203 PD/LT pile-cases in all types of soil (logarithmic scale)	150
73	K _{sw} vs. blow count (BPI) for 206 PD/LT pile-cases in all types of soil (AAA)	151
74	K _{sp} vs. blow count (BPI) for 208 PD/LT pile-cases in all types of soil (AAA)	152
75	K _{sw} vs. blow count (BPI) for 95 PD/LT pile-cases in all types of soil at EOD (AEA)	153
76	K_{p} vs. blow count (BPI) for 96 PD/LT pile-cases in all types of soil at EOD (AEA)	154
77	K _{sw} vs. blow count (BPI) for 109 PD/LT pile-cases in all types of soil at BOR (ABA)	155
78	K _{sp} vs. blow count (BPI) for 110 PD/LT pile-cases in all types of soil at BOR (ABA)	156
79	K_{sw} vs. blow count (BPI) for 57 PD/LT pile-cases with pile area ratios > 350 in all types of soil	157
80	K_{sp} vs. blow count (BPI) for 57 PD/LT pile-cases with pile area ratios > 350 in all types of soil	158
81	K_{sw} vs. blow count (BPI) for 144 PD/LT pile-cases with pile area ratios < 350 in all types of soil	159
82	K_{sp} vs. blow count (BPI) for 146 PD/LT pile-cases with pile area ratios <350 in all types of soil	160
83	K_{sw} vs. blow count (BPI) for 16 PD/LT pile-cases with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil	161

84	K_{sp} vs. blow count (BPI) for 16 PD/LT pile-cases with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil	162
85	K_{sw} vs. blow count (BPI) for 41 PD/LT pile-cases with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil	163
86	K_{sp} vs. blow count (BPI) for 41 PD/LT pile-cases with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil	164
87	K_{sw} vs. blow count (BPI) for 64 PD/LT pile-cases with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil	165
88	K_{sp} vs. blow count (BPI) for 64 PD/LT pile-cases with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil	166
89	K_{sw} vs. blow count (BPI) for 80 PD/LT pile-cases with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil	1 67
90	K_{sp} vs. blow count (BPI) for 82 PD/LT pile-cases with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil	168
91	K_{sw} vs. blow count (BPI) for 12 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm)	169
92	K_{sp} vs. blow count (BPI) for 12 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm)	170
93	K_{sw} vs. blow count (BPI) for 27 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm)	171
94	K _{sp} vs. blow count (BPI) for 27 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm)	172
95	K _{sw} vs. blow count (BPI) for 36 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm)	173

96	K_{pp} vs. blow count (BPI) for 36 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm)	174
97	K _{sw} vs. blow count (BPI) for 20 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm)	175
98	K _{sp} vs. blow count (BPI) for 21 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm)	176
99	K_{sw} vs. blow count (BPI) for 18 PD/LT pile-cases at BOR with pile area ratios >350 and all blow counts	177
100	K_{rp} vs. blow count (BPI) for 18 PD/LT pile-cases at BOR with pile area ratios >350 and all blow counts	178
101	K _{sw} vs. blow count (BPI) for 88 PD/LT pile-cases at BOR with pile area ratios <350 and all blow counts	179
1 02	K _{sp} vs. blow count (BPI) for 89 PD/LT pile-cases at BOR with pile area ratios <350 and all blow counts	180
1 03	Side soil conditions vs. Smith side damping based on CAPWAP results for 372 PD pile-cases	191
104	Tip soil conditions vs. Smith tip damping based on CAPWAP results for 377 PD pile-cases	192
105	CAPWAP predictions vs. Energy Approach predictions for 398 PD pile-cases in all types of soil	193
1 06	CAPWAP predictions vs. Energy Approach predictions for 238 large displacement PD pile-cases in all types of soil	1 94
107	CAPWAP predictions vs. Energy Approach predictions for 89 large displacement PD pile-cases in sand and silt	195
108	CAPWAP predictions vs. Energy Approach predictions for 50 large displacement PD pile-cases in clay and till	196
10 9	CAPWAP predictions vs. Energy Approach predictions for 76 large displacement PD pile-cases in rock	197
110	CAPWAP predictions vs. Energy Approach predictions for 22 large displacement PD pile-cases in unknown soil types	198

111	CAPWAP predictions vs. Energy Approach predictions for 76 small displacement PD pile-cases in all types of soil	1 99
112	CAPWAP predictions vs. Energy Approach predictions for 26 small displacement PD pile-cases in sand and silt	200
113	CAPWAP predictions vs. Energy Approach predictions for 21 small displacement PD pile-cases in clay and till	201
114	CAPWAP predictions vs. Energy Approach predictions for 29 small displacement PD pile-cases in rock	202
115	CAPWAP predictions vs. Energy Approach predictions for 85 miscellaneous PD pile-cases in all types of soil	203
116	CAPWAP predictions vs. Energy Approach predictions for 40 miscellaneous PD pile-cases in sand and silt	204
117	CAPWAP predictions vs. Energy Approach predictions for 21 miscellaneous PD pile-cases in clay and till	205
118	CAPWAP predictions vs. Energy Approach predictions for 19 miscellaneous PD pile-cases in rock	206
119	CAPWAP predictions vs. Energy Approach predictions for five miscellaneous PD pile-cases in unknown soil types	207
120	Side soil conditions vs. Smith side damping based on CAPWAP/TEPWAP results for 581 pile-cases	212
121	Tip soil conditions vs. Smith tip damping based on CAPWAP/TEPWAP results for 581 pile-cases	213
122	Risk analysis of CAPWAP/TEPWAP predictions for 206 PD/LT pile-cases in all types of soil	228
123	Risk analysis of Energy Approach predictions for 208 PD/LT pile-cases in all types of soil	229
124	Risk analysis of CAPWAP/TEPWAP predictions for 95 PD/LT pile-cases in all types of soil at EOD	230
125	Risk analysis of Energy Approach predictions for 96 PD/LT pile-cases in all types of soil at EOD	231

126	Risk analysis of CAPWAP/TEPWAP predictions for 39 small displacement (A_R >350) PD/LT pile-cases in all types of soil at EOD	232
127	Risk analysis of Energy Approach predictions for 39 small displacement (A _R >350) PD/LT pile-cases in all types of soil at EOD	233

LIST OF TABLES

<u>TABLE</u>

1	Data set PD/LT contributors	5
2	Recommended J_c values according to the soil type at the pile tip	22
3	Subgrouping of the piles in data set PD (indicating the number of piles in each group)	56
4	Breakdown of all PD/LT categories	68
5	Linear-regression analysis of K _{sw} for selected PD/LT pile-cases	70
6	Linear-regression analysis of K _{sp} for selected PD/LT pile-cases	71
7	Linear-regression analysis of K _{ew} for selected PD/LT pile-cases	72
8	Statistical analysis of K coefficients for all PD/LT pile-cases	89
9	Statistical analysis of the area ratio, resistance, and time of driving combination	96
10	Linear-regression analysis of K _{ew} for PD pile-cases	186
11	Statistical analysis of K _{ew} for PD pile-cases	188
1 2	Linear regression summary of selected PD/LT and PD subgroups	189
13	Statistical analysis summary of selected PD/LT and PD subgroups	190
14	Linear regression and statistical analysis of Ksw for selected PD/LT pile-cases	219
15	Linear regression and statistical analysis of Ksp for selected PD/LT pile-cases	220

16	Linear regression and statistical analysis of Kew for selected PD/LT and PD pile-cases	221
17	Linear regression and statistical analysis of Kew for selected PD/LT pile-cases	222
18	Absolute factor of safety based on data set PD/LT	223
1 9	Factor of safety and associated risk	227
20	Site and pile information for PD/LT	235
21	Pile driving and dynamic measurements for PD/LT	242
22	Parameters of dynamic analysis for PD/LT	249
23	Pile capacity based on static load test and dynamic analysis for PD/LT	256
24	Pile/soil and dynamic measurements of data set PD	263
25	Side/tip quake and damping parameters of data set PD	275

.

•

CHAPTER 1 - INTRODUCTION

1.1 OVERVIEW

The study of driven-pile foundations and their behavior under dynamic and static loads dates back to the late 19th century. Until that time, the design of driven piles was mainly based on experience. Dynamic equations were the first attempt at a theoretical assessment of the static capacity of driven piles. The "general" dynamic equation was developed based on the assumption that the pile and the hammer are two rigid bodies and that the calculated resistance is equal to the static capacity of the pile (Poulos and Davis, 1980).

The dynamic analyses are attractive as they attempt to predict the static capacity based on the pile behavior during driving. As such, they utilize data that is readily available during the construction operation. Moreover, they enable "real-time" capacity assessment during installation.

Hence, recent centuries have seen an increasing demand on the foundation engineer to further improve the dynamic methods of analysis. As a result, more research was performed in this area and it was realized that pile driving was not accurately represented by rigid-body mechanics (Newtonian impact), (Cummings, 1940). This realization led to the development of analyses based on wave theory utilizing the one-dimensional wave equation (Smith, 1960).

Stress-wave analyses consider the fact that each hammer blow produces an elastic stress wave that moves down the length of the pile at the speed of sound. This indicates that the entire pile is not stressed simultaneously (rigid-body mechanics), which is one of the basic assumptions of the dynamic equations.

A major improvement was gained with the direct measurement of the pile response under each hammer blow. Early large-scale studies (e.g., Michigan State Highway Commission, 1965; Texas Highway Department, 1973; and Ohio Department of Transportation, 1975) led to the development of an effective and reliable commercial system (Goble *et al.*, 1970, 1975). This system, known as the PDA (Pile-Driving Analyzer), enables complete and relatively easy acquisition of dynamic measurements and their analysis during driving. Similar systems were later developed outside the United States (FPDS-3-TNO, 1993; Reiding *et al.*, 1988; and Iwanowski, 1987).

The obtained dynamic measurements are used in two ways. One is a field analysis known as the Case Method (Goble *et al.*, 1970 and Rausche *et al.*, 1975). This analysis is based on a simplified solution of the wave equation and provides a "real-time" capacity

assessment during driving. The other is an office analysis that is based on the wave equation solution utilizing the force and velocity signals at the point of measurement. Several existing codes are based on this principle, for example, CAse Pile Wave Analysis Program, CAPWAP, (Goble *et al.*, 1970); TEchnion Pile Wave Analysis Program, TEPWAP, (Paikowsky, 1982 and Paikowsky and Whitman, 1989); and TNO (Middendorp and van Weel, 1986).

These analyses enable evaluation of a variety of parameters in addition to the static capacity. These evaluations include extreme stresses, pile-damage assessment, and loadsettlement relations to name a few. These advantages are offset, however, by the time required to produce the results and the cost incurred during this time.

A large-scale assessment (100 or more piles) of the analyses utilizing dynamic measurements has not been carried out since their initiation. Limited studies suggest substantial limitations to the Case Method (e.g., Trow Report, 1978; Paikowsky, 1982; and Thompson and Goble, 1988). Mixed experiences were reported for the office methods. These reports ranged from excellent predictions for very large offshore openpipe piles in sand (Paikowsky, 1982) to poor performance of concrete piles in clay and till (Trow Report, 1978).

Based on the existing experiences, it was clearly evident that in order to improve the state of the art it is necessary: (1) to develop an alternative method for capacity evaluation in the field and (2) to assess the performance of the different dynamic analyses and their underlying assumptions based on accumulating a large data set. Both needs are addressed by the present research.

1.2 THE PRESENT RESEARCH STUDY

The present research study is based on the aforementioned needs and consists of three major parts. The first part (chapter 4) presents an alternative field method known as the Energy Approach. This method combines the basic principle of the energy balance together with data provided through dynamic measurements. The method was first proposed by Paikowsky (1982) based on experience gained during the construction of a large offshore facility. The method was further examined on a limited scale in the Boston area (Paikowsky, 1984, 1990). Preliminary evaluations were carried out by McDonnell (1991) and Paikowsky and Chernauskas (1992).

The second part (chapters 5, 6, and 7) presents the buildup of two large-scale data sets. One data set, PD/LT (Pile Dynamic/Load Test), comprises 208 dynamic measurements on 120 piles monitored during driving, followed by a static load test to failure. All the cases were monitored using the PDA (Pile-Driving Analyzer) and the various data sources are outlined in the following section. The second data set, PD (Pile Dynamic), contains data on 403 piles monitored during driving. This data set was provided exclusively by Pile Dynamics, Inc. of Cleveland, Ohio and was originally presented by McDonnell (1991).

The third part (chapters 8, 9, and 10) presents the analysis and interpretation of the data sets. The field and office methods are examined and analyzed. Possible mechanisms underlying the different methods are suggested and the obtained results are evaluated in light of these proposed mechanisms.

1.3 CONTRIBUTIONS

Advances in geotechnical engineering in general, and foundation engineering in particular, may take place only through ultimate full-scale evaluations. Full-scale observations are difficult to obtain and require collaboration and understanding between the owner (the "client"), the designer, the contractor, and the researcher. In the presented case, such understanding could have taken place through: (1) the vision of the Federal Highway Administration, which realized the need to support and carry out research; (2) the cooperative and research-oriented nature of GRL, Inc. and Pile Dynamics, Inc. of Cleveland, Ohio; and (3) the many contributors outlined below that realized the advantage of sharing their information for the benefit of all. Table 1 outlines the contributors to data set PD/LT. As previously noted, data set PD was provided exclusively by Pile Dynamics, Inc. of Cleveland, Ohio.

As these data sets have been and will continue to be useful to several research areas, the researchers at the University of Massachusetts at Lowell thank all of the contributors for their cooperation in providing their data.

1.4 MANUSCRIPT LAYOUT

The following are short descriptions for each of the following chapters:

- Chapter 2 Provides a brief background of static analyses and static load tests.
- Chapter 3 Details the various dynamic analyses currently employed, including dynamic equations, the Case Method, and CAPWAP/TEPWAP.
- Chapter 4 Develops the proposed Energy Approach.
- Chapter 5 Outlines the buildup and analysis of data sets PD/LT and PD.

- Chapter 6 Outlines the tables presented in appendix A containing data set PD/LT.
- Chapter 7 Outlines the tables presented in appendix B containing data set PD.
- Chapter 8 Discusses and presents the graphical and statistical results obtained from analyzing data set PD/LT.
- Chapter 9 Discusses and presents the graphical and statistical results obtained from analyzing data set PD.
- Chapter 10 Provides summary, conclusions, and recommendations.
- Appendix A Presents data set PD/LT, including pile geometry, subsurface conditions, dynamic measurements, dynamic parameters, static load test results, CAPWAP/TEPWAP capacity predictions, and the Energy Approach predictions.
- Appendix B Presents data set PD, including pile geometry, skin and toe soil, dynamic measurements, and CAPWAP and Energy Approach predictions.

Organization	Persons in charge and/or contact people	Number of cases	Reference
U.S. Federal Highway Administration	Richard Cheney, Jerry DiMaggio, Albert DiMillio, Chris Dumas, and Carl Ealy	126	FHWA Dynamic Monitoring and Pile-Load Test Reports-Project 58: Colorado (1987), lowa (1986), Kentucky (1990), Louisiana (1990), Maine (1990), Minnesota (1991), Missoun (1989), Netraska (1989), Oklahoma (1989), Oregon (1987), Pennsylvania (1991), Vermont (1991), Washington (1984).
Pile Dynamics, Inc. and GRL, Inc.	George Goble, Garland Likins, Frank Rausche, and Mark Svinkin	47	Abe, Likins, and Morgano (1990). Inhouse Reports.
Ontario Ministry of Transportation	Betty Bennet, Murty Devata, John Pertruzziello, and Mark Vasavithisaan	14	Pile-Load Capacity Evaluation HWY 404 Structures-Site 33 (1978), Foundation Evaluation and Design Report-Site 35 Oniano MOT (1983), Thompson and Devata (1980).
The Trow Group Limited	Shaheen Ahmad, Steven Cheng, Tony Maini, and David Thompson	35	The Trow Report (1978), Cheng and Ahmed (1988), Thomsson and Devata (1980), Inhouse Reports, Foundation Evaluation and Design Report-Site 35 Critario MOT (1983),
GZA GeoEnvironmental	William Beloff and Steve Roy	15	Inhouse Reports
Gannet and Flemming	James Langer and John Masland	10	Inhouse Reports
Law Engineering	Kevin Kett	6	Inhouse Reports
STS Consultants	Patrick Hannigan	4	Inhouse Reports
Wagstaff Piling	David Klingberg and Julian Siedel	4	Inhouse Reports
Haley and Aldrich Inc.	Christopher Snow, David Thompson, and James Weaver	6	Inhouse Reports
Florida DOT	William "Bubba" Knight	n "Bubba" Knight 59	
Oklahoma DOT	Steve Jacobi	7	Inhouse Reports
Washington DOT	Ralph Henning	Henning 4 Inhouse Repo	
lowa DOT	Curtis Monk	4	Inhouse Reports
Oregon DOT	Gien Thommen	2	Inhouse Reports
Louisiana DOT	Mark Morvant	3	Inhouse Reports
Anna GeoDynamics, Inc.	Bengt Fellenius	2	Edde and Fellenius (1990)

Table	1.	Data	set	PD/	/LT	contributors.
-------	----	------	-----	-----	-----	---------------

Note: The total number does not add up to 208 pile cases as different sources may have contributed information for the same pile case.

.

.

CHAPTER 2 - BACKGROUND

2.1 GENERAL

The use of driven piles for foundation support for a variety of structures, such as bridges, buildings, towers, and dams, is a practice that dates back to prehistoric times.¹ Piles are used to transfer superstructure loads through soft soil layers and/or water. Pile resistance is developed through the soil, as in the case of friction piles, or from competent underlying soil or rock strata, as in the case of end-bearing piles. Most piles incorporate a combination of both frictional resistance and end-bearing resistance.

In this chapter, the main difficulty with using piles as foundation systems is addressed. Engineers have limited ability to predict the capacity and integrity of driven piles. As a result, high factors of safety are used when designing deep foundations, which add significant costs to projects. Pile capacity may be estimated using static or dynamic analyses and may be confirmed by static load tests. The following includes a brief discussion of static analyses and static load tests. The alternative methods, namely dynamic analyses, are outlined in chapter 3.

2.2 STATIC ANALYSIS

The initial design of pile foundations requires the evaluation of pile capacity via static analysis. The Federal Highway Administration (FHWA) incorporates static formulas (Tomlinson and Nordlund methods) for the analysis of driven piles in their pile analysis program, *SPILE* (DiMaggio, 1991). Static formulas estimate driven-pile capacities on the basis of soil-strength parameters obtained from subsurface exploration programs and from pile-soil interaction relations. The predictions are simply a summation of the estimated point and skin resistance of the pile. For a description of various methods, see, for example, Bowles, 1988. There is, however, a great deal of uncertainty in these analyses and their accuracy is highly questionable.

Briaud *et al.* (1988) examined the capacity predictions from 12 static analyses applied to 100 piles that were statically load tested to failure. They concluded that all methods produced unsatisfactory results, especially in layered soil strata. Similar conclusions were drawn when the best methods were averaged and used to predict the capacity of piles

¹The Neolithic inhabitants of Switzerland supported their homes 12,000 years ago on wooden poles driven in shallow lakes. The ancient Egyptians depicted manpower pile-driving operations and failures. The Romans supported many of their bridges over the Rhine River with driven-timber piles.

driven in varying soil layers.

Unfortunately, the inaccuracy of static analysis results in the use of very high safety factors leading to higher construction costs.

2.3 STATIC LOAD TESTS

Static load testing is the only method available to determine the actual static capacity of piles. This method involves physically loading a pile at specified time intervals (see, for example, ASTM D-1143) and monitoring the settlement of the pile top until failure. The results of these tests are then plotted (load vs. settlement) and the failure load is interpreted using various methods (outlined in chapter 5). These tests are expensive, time-consuming, and, as a result, are not commonly performed.

Static testing is typically carried out as a "proof test" on piles to determine the pile's performance in supporting a service load, usually twice the design load (e.g., Massachusetts Highway Dept. (1989), Virginia DOT (1987), and Alabama State Highway Dept. (1985) State highway codes). It is important to note that the proof test does not provide the ultimate pile capacity and, therefore, does not contribute to the effort of increasing accuracy and reducing foundation costs. Although the test is typically carried out to twice the design load, the actual employed factor of safety may be much higher as the actual pile capacity is unknown. Proof testing is less expensive than loading a pile to failure and is therefore more frequently performed.

In spite of the difficulties in carrying out a load test to failure and the possible inaccuracies of the data (see Fellenius, 1989), it remains as the only means to examine actual pile capacity.

Data set PD/LT, which is presented in chapter 6, contains cases of 120 piles load tested to failure. The interpretation of the test results was carried out using a variety of methods as outlined in section 5.2.1.

CHAPTER 3 - DYNAMIC ANALYSIS OF PILES

3.1 GENERAL

Dynamic analyses of piles are methods that predict pile capacity based on the behavior of the hammer-pile-soil system during driving. Such methods are based on the idea that the driving operation induces failure in the pile-soil system. In other words, pile driving is analogous to a very fast load test under each hammer blow. The pile must, however, experience a minimum permanent displacement, or set (approximately 0.1 inch [2.5 mm]), during each hammer blow to fully mobilize the resistance of the pile-soil system. If there is very little or no permanent downward displacement of the pile tip, then the pile-soil system experiences mostly elastic deformation. As a result, capacity predictions based on measurements taken at this time would not be indicative of the full resistance of the pile-soil system.

There are basically two methods of estimating the capacity of driven piles based on dynamic driving resistance: pile-driving formulas (i.e., dynamic equations) and wave-equation analysis.

3.2 DYNAMIC EQUATIONS

3.2.1 Review

For centuries (Cummings, 1940), quantitative analyses of pile capacity have been performed using dynamic equations. These equations can be categorized into three groups: theoretical equations, empirical equations, and those that consist of a combination of the two. It is important to mention that 45 of the State highway departments in the United States include a dynamic formula in their foundation specifications for the determination of bearing value for single-acting steam/air hammers. Of these 45 States, 30 use the Engineering News Record (ENR) formula and 9 States use other variations of the rational pile formula. In general, all the pile formulas, with the exception of the Gates formula, are derived from the rational pile formula (Bowles, 1988). A reference will be made here only to theoretical equations because:

- Empirical and semi-empirical equations are restricted to the conditions and assumptions of their original data set.
- State highway building codes utilize theoretical equations.

3.2.2 The Basic Principle

The theoretical equations have been formulated around analyses that evaluate the total resistance of the pile, based on the work done by the pile during penetration. Observations of the hammer's ram stroke and the pile set are used in determining this work done by the hammer and the pile. These theoretical equation formulations assume elasto-plastic force-displacement relations (see figure 1). The total work is computed as:

$$W = R_{\mu} \left(S + \frac{Q}{2} \right) \tag{1}$$

where

R_u=yield resistanceS=pile set, denoting the permanent displacement (plastic
deformation) of the pile under each hammer blowQ=quake, denoting the elastic deformation of the pile-soil
system.

In general, dynamic equations are inaccurate (see for example Housel, 1965, 1966; Flaate, 1964; and Olsen and Flaate, 1967) and a high factor of safety (F.S.) is therefore required when using their estimated capacity (e.g., F.S. = 6 for the ENR equation). Dynamic equations are largely inaccurate because:

- Their parameters, such as the efficiency of energy transfer and the pile/soil quake, are crudely approximated.
- Some of the theoretical developments of the rational pile formula, especially those relating the energy transfer mechanism to a Newtonian analysis of ram-pile impact, are theoretically invalid (see, for example, Cummings, 1940 and Taylor, 1948).
- There is no differentiation between static and dynamic soil resistances where it is known that such differences exist, especially in cohesive soils (Taylor, 1948).

3.2.3 Energy Transfer

The theory of energy transfer analysis in many of the dynamic equations assumes that the hammer-pile impact is consistent with Sir Isaac Newton's third law, Conservation of Momentum. Newton's relationship applies to the impact of two free rigid bodies. In the case of dynamic equations, these rigid bodies are considered to be the hammer and the pile. This law of motion states that if no external forces are acting on the two rigid bodies, then the total momentum of the system is conserved. The impulsive forces acting during the impact are actually internal and, therefore, do not affect the total momentum



Figure 1. Resistance vs. displacement at the top of the pile.

of the system (see Cummings, 1940). This is clearly not the case for driven piles, which are elastic rather than rigid and experience end bearing as well as frictional resistance. Newton is reported to have stated that his expression for the impact of two massive bodies did not apply for "bodies ... which suffer some such extension as occurs under the strokes of a hammer" (see Taylor, 1948).

The ENR formula, published in 1888, was originally developed for use with timber piles and a drop hammer (Bowles, 1988). This formula further simplifies the assumptions made by the rational pile formula by equating the efficiency of the ram-pile impact to 1. This oversimplification does not consider three factors:

- Energy losses that occur in the pile-driving system during impact.
- Work used in the elastic compression of the pile and soil.
- Varying efficiencies of the wide range of hammers used today.

These simplifications in the development and use of the ENR formula result in a

necessary safety factor of 6 (Taylor, 1948). Briaud and Tucker (1988) checked the prediction accuracy of the ENR equation in 68 pile cases. The static capacity was determined based on a reference settlement equal to one-tenth of the pile diameter plus the elastic compression of the pile. The mean of the ratio predicted over measured load was 0.82 with a standard deviation of 0.38. Further reference to these results is made in chapter 10. Overall, the low reliability of dynamic equations requires very high factors of safety that make their use extremely uneconomical.

3.3 THE WAVE EQUATION

3.3.1 Formulation and Principles

Issacs (1931) concluded that many pile-driving formulas were incorrectly based on Newtonian mechanics for the pile/hammer impact and he became the first person to suggest the use of an analysis based on the one-dimensional wave equation instead. This proposed solution assumed that the toe of the pile was fixed and that no side resistance existed (Lowery *et al.*, 1969). Fox (1932) proposed an exact solution to Issacs formulation; however, without the aid of computers, many simplified assumptions were necessary because of the complexity of his solution (Smith, 1960).

Stress-wave propagation in a pile during driving can be described by the following onedimensional wave equation (after Paikowsky and Whitman, 1990) modified to include frictional resistance along the pile:

$$E_{\rho} \frac{\partial^2 u}{\partial x^2} - \frac{S_{\rho}}{A_{\rho}} f_s = \rho_{\rho} \frac{\partial^2 u}{\partial t^2}$$
(2)

where

The displacement (u) causes strains in each pile element that can be used to calculate pile stresses as well as the resistance developed in the soil. This displacement can be determined with respect to time and location. The friction stresses (f_s) are generated by the movement of the pile. When the pile is subjected to free-wave motion $(f_s=0)$, the stress propagation equation becomes the familiar one-dimensional wave equation:

t
$$c^2 \frac{\partial^2 u}{\partial x^2} = \frac{\partial^2 u}{\partial t^2}$$
(3)

where

$$c = \sqrt{\frac{E_p}{\rho_p}} \tag{4}$$

Among the assumptions implicit in the development of the one-dimensional wave equation are prismatic shape and homogeneity. Also, it is assumed that under loading, plane parallel cross sections remain plane and parallel and that a uniform distribution of stress exists across each plane. The assumption of uniaxial stress does not include uniaxial strain and, therefore, lateral expansions and contractions (Poisson's effect) arise from the axial stresses associated with lateral inertia (Graff, 1975). The additional friction term (after Paikowsky and Whitman, 1990) was included under the assumption that the soil is stationary (having no inertia effects), and the action of the friction forces does not violate any of the previous assumptions.

The so-called "wave equation methods" are based on a numerical solution of the onedimensional wave equation. The numerical solution utilizes mathematical models for the pile and the pile-soil system. When the one-dimensional wave equation numerical solution is used for pre-driving analysis, the driving system is also modeled.

In 1960, Smith developed a numerical model to simulate the dynamic behavior of the hammer-pile-soil system during driving. This model is represented by a series of discrete masses and springs used for solving the one-dimensional wave equation (see figure 2). The soil resistance is modeled via a spring, slider, and dashpot, which represent the static and dynamic soil resistances (see figure 3). The elasto-plastic soil model is employed for the static soil resistance in Smith's solution. The distance traveled by the pile toe during the elastic deformation of the soil is represented by the soil quake (Smith, 1960). As the elastic limit of the soil is reached (represented by the slider in sequence with the spring), plastic deformation takes place. The plastic deformation, or irreversible compression of the soil, is denoted by the permanent set of the soil (see figure 3).

According to this model, point A represents the ground resistance buildup to the ultimate resistance, R_u . Plastic failure occurs as the ground resistance has reached its maximum and the adjacent pile segment displaces, plastically, to point B. Unloading the soil at point B produces an elastic rebound, equal to the quake, to point C. The





Figure 2. Smith's model simulating the hammer-pile-soil system for use with the one-dimensional wave equation (Smith, 1960).



Figure 3. Soil-pile model (left) and the corresponding elastoplastic soil resistance-displacement relationship (after Smith, 1960).

permanent set is, therefore, equal to the distance OC, which, in turn, is equal to distance AB (Smith, 1960). The static soil resistance-displacement relationship, as presented by Smith (1960), is modeled by a spring (K_s) and a slider, where W represents the mass of the pile element.

The dynamic component of the soil's resistance is assumed to be viscous (soil-type related) and is, therefore, velocity-dependent. This dynamic resistance is modeled by a dashpot (J) parallel to the spring (see figure 3). The resisting soil force (R_{max}) developed under each hammer blow is a combination of the static and dynamic soil resistances:

$$R_{\max} = R_s + R_d \tag{5}$$

where

 R_{max} = total resistance

 $R_s = static resistance$ $R_d = dynamic resistance.$

The wave equation formulation is used in two general ways: pre-driving analysis and post-driving analysis.

3.3.2 Pre-Driving Analysis

The so-called "wave equation analysis" utilizes the one-dimensional wave equation to predict dynamic pile behavior before construction and models the pile-soil system and the driving system (i.e., the hammer, cushion, and capblock), as suggested by Smith (1960). This computerized solution is used for the evaluation of the penetration resistance (i.e., blow count) and the driving stresses in the modeled pile under given conditions. The static capacity is then determined by relating the computed static capacity-penetration resistance relationship for a certain energy rating to observed dynamic resistances during driving. Such analyses enable engineers to determine a suitable pile-site-equipment combination.

3.3.3 Post-Driving Analysis - CAPWAP/TEPWAP

Post-driving analyses utilize the measured force signal (calculated from strain readings) and the measured velocity signal (integrated from acceleration readings) obtained near the pile top during driving. These analyses model the pile-soil system as shown in figure 4 with the element denoted as number 3 representing the point of measurement. The velocity signal is used as a boundary condition at that point while varying the parameters describing the soil resistance in order to match the calculated and measured force signals. These parameters include the side and tip quake, side and tip damping, the pile shaft resistance, and the pile tip resistance. Additional parameters may be used to describe soil resistance and rebound ratio for unloading different from that of loading. The process is described in the form of a flow chart in figure 5. The subscripts msd. and cal. denote measured and calculated values, respectively. Iterations are performed by changing the soil-model variables for each pile element in contact with the soil until the best match between the force signals is obtained. The results of these analyses are assumed to represent the actual distribution of the ultimate static capacity of the pile.

This procedure was first suggested by Goble, Likins, and Rausche (1970), utilizing the computer program CAPWAP. Similar analyses were developed by others (see Paikowsky, 1982 and Paikowsky and Whitman, 1989) utilizing the program code TEPWAP.

3.3.4 Wave Equation Analysis - Discussion

Post-driving analyses utilize the measured force and velocity waves, hence, the energy delivered to the pile in these models is exact. The models can consider the "damping" at



Figure 4. Notations used for model of pile and soil in TEPWAP analysis (Paikowsky, 1982).



Figure 5. Flow chart describing the analysis process using TEPWAP (Paikowsky, 1982).

each depth by utilizing different damping parameters for each of the discrete units and, therefore, account, to some degree, for different energy losses in the surrounding soils and the various pile-type effects. Such analyses may result in a force distribution along the pile that differs from the actual one, but by keeping the energy balanced, the calculated total resistance may be accurate (a case study of large instrumented piles that showed such results was presented by Paikowsky, 1982). A method that presents a simplified solution for the wave propagation phenomenon (i.e., the Case method, see section 3.5), with the attempt to correlate the energy losses to the soil type at the tip, does not capture the actual phenomena and does not necessarily keep a balance of energy. The resulting factor (J_c) is difficult (if not impossible) to correlate to the soil type at the pile tip. A simple field method that predicts pile capacities in "real-time" remains attractive, however, because of its ability to monitor pile capacity during driving.

3.4 FIELD ANALYSIS AND THE PILE-DRIVING ANALYZER

Capacity evaluation in the field is attractive because of the potential to increase quality control and to improve construction efficiency of deep foundation systems. The procedure of monitoring pile driving by dynamic measurements is well established. Early large-scale studies (e.g., Michigan State Highway Commission, 1965; Texas Highway Department, 1973; and Ohio Department of Transportation, 1975; see also Highway Research Record, 1967 and Goble *et al.*, 1970) led to the development of commercial systems that enable complete and relatively easy acquisition of dynamic measurements and analysis during driving. These dynamic measurements include acceleration and strain readings recorded at the pile top under each hammer blow. The most popular acquisition and analysis system in the United States is the pile-driving analyzer (PDA) (see Pile Dynamics Inc., 1990).

The PDA calculates a number of different physical quantities, including force (from strain readings), velocity and displacement (from acceleration readings), maximum delivered energy (to the pile top), and tension and compression stresses. These results are used to predict the pile capacity, as well as to examine the hammer performance, stresses in the pile, and pile integrity. The PDA predicts pile capacities in the field by utilizing a simplified evaluation method, known as the Case method.

3.5 THE CASE METHOD

3.5.1 General

The Case method (see Goble *et al.*, 1970 and Rausche *et al.*, 1975), is a simple field procedure used by the PDA to estimate pile capacities. Analysis by the Case method is based on the assumptions of a uniform elastic pile, ideal plastic soil behavior, and a simplified wave propagation formulation. Employed are force and velocity measurements taken at the pile top and a correlation between the soil at the pile tip to a damping parameter.

3.5.2 The Case Method Equation

The Case method calculates the total soil resistance (RTL) active during pile-driving, using the following equation:

$$RTL = \frac{[F(TI) + F(TI + \frac{2L}{C})]}{2} + [v(TI) - v(TI + \frac{2L}{C})] + \frac{MC}{2L}$$
(6)

where

F(T1)	=	measured force at the time T1
F(T1+2L/C)	=	measured force at the time T1 plus 2L/C
v(T1)	=	measured velocity at the time T1
v(T1+2L/C)	=	measured velocity at the time T1 plus $2L/C$
L, M	=	length and mass of the pile, respectively
С	=	speed of wave propagation in the pile.

Different variations of the Case method have been developed taking T1 as the time of impact or modified to include a time delay constant allowing higher RTL values to be obtained. The time T1 is defined, in equation form, as:

$$TI = TP + \delta \tag{7}$$

time of the impact peak TP where δ time delay. =

The time delay is required in soils capable of large deformations before achieving full resistance (see figure 6). A time delay is also used in situations where the hammer impact is uneven (PDA Manual, 1990).

The total resistance calculated is a combination of the static resistance (S) which is displacement-dependent, and the dynamic resistance (D) which is velocity-dependent. Therefore, the total resistance (Goble et al., 1975) is:

$$RTL = S + D \tag{8}$$

Several factors that influence the pile-soil system must be considered when the total predicted resistance is evaluated. These factors include the damping coefficient, timedependent soil strength changes, and refusal driving when the soil's resistance is not fully mobilized under a single hammer blow.

3.5.3 Case Damping Coefficient

The dynamic resistance D is considered to be viscous in nature, hence, a function of the velocity at the pile toe (V_{toe}) and a damping constant (J) where:

$$D = J * V_{toe} \tag{9}$$

By applying the wave propagation theory, the pile toe velocity can be calculated as a



Figure 6. Force and velocity traces showing two impact peaks indicative of driving in soils capable of large deformations.

function of the velocity at the pile top (Goble et al., 1975):

$$V_{top} = 2 V_{top} - \frac{L}{MC} RTL$$
(10)

where

pile length

Μ pile mass =

=

L

C R wave speed of the pile material =

= total resistance V_{top}

velocity at pile top. =

 V_{top} is taken as the pile top velocity at the time T1.

According to Goble *et al.* (1975), remolding effects cause the majority of the damping resistance to be concentrated near the pile tip. Consequently, the damping constant is determined according to the soil type at the pile tip. In most cases, the damping constant (J) is proportional to the pile properties (EA/C) and, therefore, is represented by a dimensionless coefficient (J_c) using the following equation:

$$J = J_c \frac{EA}{C}$$
(11)

where

 J_c = dimensionless Case damping coefficient E = elastic modulus of the pile material A = pile cross-sectional area C = wave speed of the pile material.

The recommended values for J_c have changed since the initial estimates made by Goble *et al.* (1975) as a result of improvements to the PDA and continued research in this area. In 1975, Goble *et al.* (1975) published recommended J_c values for various soil types. These recommendations have been revised in PDA Manual-Model GCPC (1990). Both sets of recommended J_c values are given in the following table:

Soil Type at Pile Tip	Goble et al., 1975	PDA Manual, 1990
clean sand	0.05	0.10 to 0.15
silty sand	0.15	0.15 to 0.25
sandy silt	0.2	-
silt	0.3	0.25 to 0.40
silty clay / clayey silt	0.55	-
silty clay	-	0.40 to 0.70
clay	1.1	0.70 to 1.00

Table 2. Recommended J_c values according to the soil type at the pile tip.

It is suggested that J_c values less than 0.10 are unlikely. Large J_c values result in more conservative capacity predictions, and the range of $J_c = 0.5$ to 1.0 can cause large capacity differences (PDA Manual, 1990). J_c can be back-calculated from static load test results and applied to other piles nearby, provided they are driven in similar soil strata. Negative damping coefficients are physically meaningless and are set to zero should they occur. If load testing to failure is not conducted at a particular site, subsurface investigation of the underlying soil strata must be carried out to provide the necessary information needed to estimate J_c .

3.5.4 Case Method Variations

Several variations of the Case method have evolved for the analysis of different driving situations and soil types. The variations are similar in that they all begin with the initial total resistance prediction (RTL) of equation (6). Five distinct methods are used to employ the predicted RTL: Damping Factor Method, the Maximum Resistance Method, the Minimum Resistance Method, the Unloading Method, and the Automatic Method. A brief review of each of these methods follow (for details, see the 1990 PDA Manual).

(a) The Damping Factor Method, RSP

The Damping Factor Method uses the velocity at the toe of the pile (V_{toe}) of equation (10), which may be rewritten as:

$$V_{toe} = V_{top} + \frac{[F(TI) - RTL]}{\frac{MC}{L}}$$
(12)

and the Case damping constant (J_c) is nondimensionalized by the pile impedance (MC/L), to determine the static capacity (RSP) (PDA Manual, 1990). The equation, which was discussed in the last section, utilizes damping constants empirically derived from static load tests where:

$$RSP = RTL - J \frac{MC}{L} * V_{toe}$$
(13)

This expression is the standard Case method equation used for normal driving conditions.

(b) The Maximum Resistance Method, RMX

The Maximum Resistance Method uses the RSP equation with 2L/C as a fixed quantity. The time T1 used in the RSP equation is varied between the impact time (TP) and TP + 30 ms to find the corresponding maximum RSP value, denoted as RMX (see figure 6).

Originally (Goble *et al.*, 1967), it was proposed to choose the time T1 as the time when the pile top velocity becomes zero (referred to (e), the automatic method, RAU). Time delay methods were then developed (Goble *et al.*, 1975). The most familiar one is T1=TP, the time of maximum velocity. This was then modified to T1=TP + δ , where δ is a time-delay constant required to enable full resistance to be developed. The maximum resistance method (RMX) is a variation of this approach, where T1 will result in the maximum static resistance (R_s). This T1 is not necessarily the same one that will produce the maximum total resistance RTL. RMX can be used in cases of large soil quakes or short rise times where the full resistance is not mobilized by the time the stress wave reaches the pile toe. This method is advantageous for large displacement piles with substantial end-bearing. The RMX resistance may not, however, develop until unacceptably large displacements occur. Caution should be taken when using RMX in silts and clays with high damping factors because over-predictions may result.

(c) The Minimum Resistance Method, RMN

Tension cracks, splices, and changes in cross-sectional area may vary the wave speed along a single pile. To compensate for these changes, the Minimum Resistance Method uses the first or second peak as the impact time (T1) in the capacity equation. The tip reflection time, T2 or (T1+2L/C), is varied through the 2L/C "window," which is centered around T2 and is ± 20 percent of 2L/C. The minimum capacity (RMN) is determined using the tip reflection time. This method can be used with confidence if the blow count is less than 40 blows per foot (131 blows per meter) (PDA Manual, 1990).

(d) The Unloading Method, RSU

For long piles with high frictional resistance, the measured velocity can become negative before a reflection from the tip is observed at time T2. Under such conditions, the upper portion of the pile experiences decreasing displacement or rebounding. This results in an unloading of the upper soil layers resistance and the computed capacities are under-predicted. The Unloading Method compensates for this by calculating the total friction in the upper unloading layers from the force velocity difference. This friction is then divided by two thus yielding the correction. The unloading resistance (RSU) then is:

$$RSU = RTL + K - J[F(tI) + v(TI) * \frac{MC}{L} - RTL - K]$$
(14)

where

the unloading correction coefficient.

The correction coefficient is calculated from:

=

$$K = \frac{[F(T3) - V(T3)(\frac{MC}{L}) - F(TP) + V(TP)(\frac{MC}{L})]}{2}$$
(15)

where T3 = 2TP + 2l/C - T0 and T0 is the time of zero velocity (before 2L/C) (PDA Manual, 1990).

(e) The Automatic Method, RAU

Κ

The Automatic Method computes the capacity (RAU) for the first time where the computed pile toe velocity (V_{toe}) is zero. This method, originally proposed by Goble *et al.* (1967), does not select a damping coefficient because damping must be zero when V_{toe} is zero; therefore, the resistance at this time is completely static. This method

provides an exact solution for the end-bearing for piles with no skin friction and is recommended for use on piles with very little frictional resistance. Another variation of this method attempts to convert any skin friction into end-bearing resistance. This is proposed for piles having moderate skin friction, but are unaffected by J (PDA Manual, 1990).

3.5.5 Evaluation

(a) Critical Discussion

Two fundamental questions should be addressed regarding the Case Method approach:

- What is the time (T1) that should be used to calculate the total resistance (RTL)?
- What is the meaning and reliability of the Case damping factor?

Based on the various methods described in section 3.5.4, the Case method produces a range of results according to the way in which it is employed. The "right" way and the "correct" T1 are questionable, and depend on the driving system and soil and pile conditions.

The Case damping coefficient J_c is based on viscous damping in a dimensionless form. Thus, the dynamic resistance is correlated to the calculated velocity at the tip of the pile, and J_c is assumed to be related to the soil type at the pile's tip. To find the J_c to be used for different soils, the damping coefficient was calculated to fit failure loads obtained from static load tests. These damping coefficients were calculated for a range of ± 20 percent of the load test results, resulting in ranges of the J_c coefficient that were then ascribed to each soil type (Goble *et al.*, 1975).

The correlation between J_c and soil type is questionable and may or may not be feasible. The following section evaluates the use of J_c and demonstrates that the pile's dynamic resistance is influenced by several additional factors that cannot be appropriately considered through the use of the Case damping factor. A detailed examination of the J_c parameter is presented in section 8.2.1.

(b) Review of Existing Experience

The Case method has been the subject of different comparison studies attempting to evaluate it's reliability. When static load testing is conducted on a pile, the corresponding Case damping coefficient can be obtained through back-calculating. This coefficient can then be compared to typical J_c values recommended to be used with the given soil conditions. Such information enables the determination of the reliability of the Case method for individual testing sites. Comparisons between the Case method and CAPWAP analysis results (in place of static load testing) have also been conducted (see, for example, Thompson and Goble (1988) or Riker and Fellenius (1988)).

The Trow Company (1978) examined 226 piles and 40 static load tests at 21 different sites. Their report concluded that the Case Method was shown to be in closer agreement with static load tests than dynamic formulas. For end-bearing piles, the range of the applicable damping factor is narrow, and the use of J_c values between 0 and 0.3 led to predictions within \pm 25 percent of the load test results (excluding piles in till at one site). However, for friction piles, the choice of damping values was critical for the correct prediction of the capacity, and the tested pile's capacity was about twice the predicted one.

Four full-scale static load tests were conducted offshore and analyzed by Paikowsky (1979-1982). Open-ended pipe piles (48 in and 60 in [1219 mm and 1524 mm] diameters) were dynamically monitored during driving in a predominantly calcareous sand soil profile. The Case damping coefficient values (J_c) for capacity predictions in the range of ± 20 percent from the load test results are presented in figure 7. In order to be consistent with the data analysis of chapter 6, the J_c values of figure 7 are based on the same data used for CAPWAP and TEPWAP analyses that were somehow different from the one observed in the field. These values varied between $J_c = 0.06$ to 0.37, fitting the load tests, and in different ranges for each of the individual cases (e.g., $0.18 \leq J_c \leq 0.52$ for T-2/A and -0.20 $\leq J_c \leq 0.31$ for T-2/B).

Despite the Case method being used in only one of its forms, a significant scatter exists in the "recommended" damping coefficient field values that are considered more accurate than values from a general data set (see table 2).

A pile-testing study that began in 1980 was conducted in Milwaukee, WL to establish foundation design criteria, such as the most suitable pile type and driving depth (see Riker and Fellenius, 1988). This project was undertaken because of the extensive pile installation program required for the construction of a wastewater plant (3,000 to 4,000 driven piles). The test piles consisted of steel H-piles and closed-ended pipe piles, with varying thicknesses, and were founded in glacial soil deposits. Approximately 40 piles were monitored during initial driving and/or during restriking, using a pile driving analyzer (PDA). All of these piles were analyzed using CAPWAP, and from these results, a J, value was back-calculated for each pile. This analysis allowed engineers to correlate J_c values for the remaining piles at the site, provided they are founded in similar soils. Similarly, the Case method was performed on each test pile and capacity predictions were obtained using the calibrated J_c factors determined from the CAPWAP analyses. The results of this comparison show that when using pile-site-calibrated J_{c} factors for thick-walled steel pipe piles, the Case method predictions were within 20 percent of the CAPWAP results. Riker and Fellenius concluded that in light of the consistency of the J_c values at this site, the reliability of the Case method for rapid field predictions was demonstrated. They also cautioned, however, that additional CAPWAP analyses are necessary if other pile types are to be used at this site.

A comparison study between static load tests to failure and the Case method was carried



Figure 7. Case damping (J_c) values for capacity prediction of offshore piles in the range of ±20 percent from load test results (after Paikowsky, 1982).

out in Europe by Bustamante and Weber in 1983 (Bustamante and Weber, 1988). This study consisted of dynamically monitoring six different shaped-steel H-piles using a PDA' and load testing them to failure. The piles were tested at two different sites, and the general soil profiles consisted of sandy and clayey soils, respectively. The study results indicated that the predictions made by the Case method and CAPWAP were in agreement with capacities determined by static load testing. However, the Case damping coefficients for the sandy site required calibration from CAPWAP results or static load test results.

Thompson and Goble (1988) tested 25 piles at 9 different sites across the eastern regions of Canada and the United States. All of the piles were founded in granular soils and were dynamically monitored using the PDA. CAPWAP analyses had been performed at the beginning of restriking (BOR). The results confirmed that the Case damping constants required to match CAPWAP capacities were high compared to recommended values. These high damping constants varied from 0.24 to 0.70 in the same soil on the same site and from 0.24 to 0.85 for all nine sites. These values are in sharp contrast to the $J_c = 0.05$ that was recommended to be used in sand by Goble *et al.* (1975) and $J_c =$ 0.10 to 0.15 recommended by the PDA Manual (1990). Higher damping constants than expected will result in capacity over-predictions by the Case method. Thompson and Goble pointed out that their wide-range data set eliminated the possibility of treating these results as consequence of localized geographic or geologic conditions, and suggested that since they could not find an explanation for these high values, every project involving piles driven in sand should be calibrated for the correct J_c value.

Paikowsky and Chernauskas (1992) examined nine piles that were monitored during driving at the end of driving and/or at the beginning of restrike and were driven into soils ranging from sandy-silt to rock and till. Their study included static load tests to failure, whereby the failure loads were then employed to back-calculate Case damping factors. The results indicated that there is no specific correlation between the damping coefficient and the soil type. Thompson and Goble (1988) further concluded that it may be necessary at some projects to incorporate CAPWAP analyses with every pile to confirm the predictions by the Case method.

3.5.6 Capacity Predictions

The static resistance of the pile is predicted by subtracting the dynamic resistance from the total resistance (equation 8). As the static resistance may be time-dependent, it is often necessary to restrike piles and conduct dynamic analyses sometime after the end of initial driving (EOD). Setup may cause the static capacity to increase, while relaxation may cause the static capacity to decrease. Setup most often occurs in cohesive soils due to either (1) dissipation of excess pore pressure in the vicinity of the pile after driving or (2) thixotropy (an increase in strength with time without changing the water content) and a variety of reasons not always well-understood that may be referred to as "aging" (Schmertmann, 1991).

Soil relaxation most often occurs when piles are driven into dense fine sand or silts, shearing the soil beyond its peak resistance to residual strength. This results in smaller long-term frictional resistance. Although relaxation occurs less frequently than setup, its determination may be crucial. Restriking can lead to a more economical foundation system in the event of setup, and can prevent major structural problems in the event of

relaxation (Likins et al., 1990).

When driving reaches refusal (e.g., a set of 0.1 in [2.5 mm] or less, most often regarded as 12 blows per inch [0.47 blows per millimeter]), the Case method may under-predict the static capacity of the pile. This is consistent with the concept that the driving operation must induce failure in the pile-soil system. If the pile experiences a small permanent set, or none at all, then the soil resistance is not fully mobilized (which indicates that the pile-soil system is mostly within the elastic range). Under such conditions, the predicted static capacity relates to the mobilized value only, often resulting in an under-prediction (PDA Manual, 1990).

3.5.7 Summary

The dynamic analysis of pile driving is based on the one-dimensional wave equation that describes the stress propagation through a slender elastic body. An additional term that accounts for the external forces acting on that body is added to the equation in order to consider the soil resistance. Traditionally, this resistance is considered to consist of static and dynamic components, as previously described. Practically, however, the dynamic component (even though represented by viscous damping) accounts for other energy losses, such as radiation, soil inertia, true damping, and more (Paikowsky and Whitman, 1989). These factors are determined by the pile shape, the acceleration at the pile toe, and the surrounding soil and, hence, cannot be correlated only to the soil type at the pile tip, as suggested by the Case method. The wave equation type of solutions (including CAPWAP) can consider the damping at each depth of pile penetration and, therefore, account for the different surrounding soils and pile type. The Case method simplified solution is not capable of this damping consideration. The correlation of the energy losses to the pile tip velocity and the soil type at the tip oversimplifies the complex phenomena; the resulting damping factor is difficult to correlate, leading to unreliable predictions. The accuracy of the Case method as a means of analyzing driven piles in the field will be further examined in chapter 8, based on the analysis of data set PD/LT in appendix A.

CHAPTER 4 - THE ENERGY APPROACH

4.1 BACKGROUND

While the static soil resistance is represented relatively adequately by the elasto-plastic soil model (see figure 3), the viscous damping accounts practically for various energy losses such as radiation, soil inertia (at the pile tip in particular), true damping, and viscosity in cohesive soils. As such, the model parameters (i.e., damping coefficients) cannot be calibrated on the basis of soil type alone. If such a calibration was possible, there would be no need to use different damping coefficients for the same soil next to the toe or the skin.

This observation has three major implications:

- 1. The success of the soil model in correctly representing the physical phenomena next to the pile is really controlled by its ability to account for the energy losses (in particular, those due to dynamic actions).
- 2. Calibration of the soil model parameters cannot be done on the basis of soil type alone. The calibration requires consideration of the combination of the pile and soil types (mainly small vs. large displacement piles), driving resistance, and, in addition, awareness of the installation details during construction (e.g., the use of jetting or preaugering).
- 3. A byproduct of 1 and 2 can explain why one method of analysis fails while the other succeeds (e.g., the Case method and CAPWAP).

The prediction of static capacity from pile driving, either by dynamic equations or by the one-dimensional wave equation, requires a balance of energy (i.e., the total energy that is transferred to the pile through the driving system is equal to the work done by the resisting forces during penetration).

Even though most of the theoretical and semi-empirical dynamic formulas were based on the energy principle, their reliability is very low, for the following reasons (see section 3.2 for discussion):

• Their analysis of Newtonian impact between the ram and the cushion/capblock system is theoretically invalid and, therefore, it led

to incorrect predictions of the amount of energy transferred to the pile.

The elastic soil-pile rebound (quake) was estimated or calculated based on a static approach.

Analyses such as CAPWAP, on the other hand, utilize dynamic measurements and, therefore, the transferred energy is known. With the appropriate pile and soil modeling, the number of unknowns is limited and the different energy losses can be accounted for indirectly through dynamic resisting forces based on viscous dampers, as previously discussed.

4.2 UNDERLYING CONCEPT

٠

The concept of the "Energy Approach," in which basic energy relations are used in conjunction with dynamic measurements, was presented by Paikowsky (1982). Limited additional studies were carried out by Paikowsky (1984), McDonnell (1991), and Paikowsky and Chernauskas (1992). The underlying concept of this approach is the energy balance that is developed between the total energy delivered to the pile and the work done by the pile/soil system. The required "real-time" prediction in the field calls for a simplified solution and, therefore, does not consider the propagation process, while distinguishing between:

- Energy loss from elastic soil/pile deformations.
- Work done by the static resistance on plastic soil deformations.
- Energy loss due to various combined factors associated with the pile penetration (i.e., damping, radiation, inertia, etc.).

4.3 THE ENERGY EQUATION

The energy delivered to the pile is:

$$E_{n} = \int V(t)F(t)dt \tag{16}$$

where

V(t) = velocity signal at the pile top for the analyzed blow F(t) = force signal at the pile top for the analyzed blow.

The velocity signals are obtained by measurements of acceleration, $a_{cc}(t)$, where:

$$V(t) = \int a_{cc}(t)dt \tag{17}$$

The force signals are obtained by processing the measurements of strain, $\epsilon(t)$, whereby:

$$F(t) = \epsilon(t) E A \tag{18}$$

where E = modulus of elasticity of the pile material A = cross-sectional area of the pile.

These measurements and calculations are immediately processed by the data acquisition system after each hammer blow.

The force/displacement relations of the pile/soil system are assumed to be elasto-plastic, which is consistent with the basic dynamic equations and static resistance of soil models in the wave equation analyses.

The total work done by such a system (elastic and plastic), therefore, will be (referring to figure 1):

$$W = R_{\mu}(S + \frac{Q}{2}) \tag{19}$$

where

R.,

0

 yield resistance
 quake denoting the combined elastic deformation of the pile and soil

S = set denoting the plastic deformation.

The quake is determined by finding the maximum displacement reduced by the plastic deformation (permanent set) under each hammer blow, such that:

$$Q = D_{\max} - S \tag{20}$$

where D_{max} = maximum value of $\int V(t) dt$.

The permanent set can theoretically be determined by $D_{fin} = final value of \int V(t)dt$. However, the displacement is the second integration of the measured acceleration. Any offset in the acceleration measurement (e.g., due to DC voltage in the accelerometers) will have a relatively small effect on D_{max} , but a much greater effect on D_{fin} (for further discussion, see experimental work by Bernardes, 1989). It is more practical to use the field blow count, such that S = Set = 1/BPI (blows per inch) (see figure 8).

The maximum resistance under the above assumptions is obtained from $E_n = W$, and becomes the proposed Energy Approach (uncorrected):

$$R_{\mu} = \frac{E_n}{Set + \frac{(D_{max} - Set)}{2}}$$
(21)

This resistance can be taken as the maximum possible resistance and can be correlated to the predicted static capacity (P_u) by a correlation factor, such that:

$$P_u = K_{sp} * R_u \tag{22}$$

where $K_{sp} =$

"static pile" correlation factor accounting for all dynamic energy losses.

The K_{sp} factor is correlated to pile type (small vs. large displacement), soil type (mainly granular vs. cohesive), and driving resistance.

4.4 ENERGY LOSSES AND SOIL INERTIA

4.4.1 General Considerations

Soil inertia is a major factor contributing to the energy loss during driving. As such, a substantial portion of the dynamic resistance should be a function of two parameters:

- Mass/volume of the displaced soil that is a function of the pile geometry, namely, small vs. large displacement piles.
- Acceleration of the displaced soil, especially at the tip that conveniently can be examined as a function of the driving resistance.

4.4.2 Soil Displacement

The volume of the displaced soil is identical to the volume of the penetrating pile, excluding the cases in which pile plugging takes place (Paikowsky and Whitman, 1990). The piles, therefore, can be classified as small (e.g., H and open pipe) and large (e.g., closed pipe and concrete) piles. Additional classification of open-pipe piles can take place according to a tip-area ratio similar to that used for soil samplers (Paikowsky *et al.*, 1989).

As most of the soil displacement takes place at the tip area, the classification of piles can



Figure 8. The proposed way of obtaining the combined quake, Q (soil and pile). [Not to scale.]

be better served by looking at the ratio between the piles embedded surface area and the area of the pile tip:

$$A_R = \frac{A_{skin}}{A_{uip}}$$
(23)

where

 A_R = pile area ratio A_{skin} = pile's surface area in contact with soil A_{tip} = area of the pile tip.

According to this ratio, a pile that is traditionally referred to as a "large displacement" pile can behave like a small displacement pile if it is driven deep enough. Because the frictional resistance of a pile increases as the pile skin area in contact with soil increases, the effect of the soil mobilized at the tip decreases. As the pile's embedded surface area and the skin friction increases, the energy losses resulting from the mobilization of the soil mass at the pile tip will decrease relative to the energy losses along the side of the pile. For example, the area ratio for cylindrical (closed-end) piles is:

$$A_R = \frac{2\Pi R * D}{\Pi R^2} = \frac{2D}{R}$$
(24)

in which D = penetration depth R = pile radius.

For the same pile diameter, this area ratio increases linearly with depth, e.g., a 14-in (356-mm) diameter pile will have an area ratio of 69 at the depth of 20 ft (6.1 m) and an area ratio of 360 at the depth of 105 ft (32 m). It is clear that the effect of soil inertia at the tip in the second case will be substantially smaller than that in the first case and the pile may be classified as a "small displacement pile." A quantitative boundary between "small" and "large" displacement piles on the basis of the area ratio is presented in section 8.5.

4.4.3 Soil Acceleration

The energy loss through the work performed by the inertia forces at the displacement of the soil mass at the tip is directly related to the acceleration of this mass. The direct evaluation of these accelerations are beyond the scope of the present research. The indirect evaluation of these accelerations can be performed through the driving resistance, which is the measure of the pile's displacement under each hammer blow. With low driving resistance there is high velocity (i.e., free-end analogy) and high acceleration at the pile tip, hence, high inertia of the tip soil mass. This results in a soil inertia "force" that, when multiplied by the pile displacement at the tip, produces a large loss of energy. In the case of high driving resistance (hard driving), there is little, if any, mobilization of the tip soil mass and the acceleration at the tip is very low. Therefore, the corresponding energy loss is small.

4.4.4 Expected Performance

In summation, according to the above hypothesis, the largest loss of "unknown" energy occurs when large displacement piles experience easy driving (large tip displacement). The smallest loss of "unknown" energy occurs with small displacement piles driven under high blow counts (hard driving).

Considering the preceding criteria, the Energy Approach should theoretically produce two distinct trends:

- In the case of high "unknown" energy losses, i.e., in easy driving of piles with small area ratios, the Energy Approach predictions should yield a tendency of over-prediction. Hence, R_u is expected to be higher than the actual resistance as the large energy losses were not considered. As a result, K_{sp} is expected to be smaller than unity ($K_{sp} < 1.0$).
- In the case of small "unknown" energy losses, i.e., hard driving of piles with large area ratios, the Energy Approach predictions should yield a tendency of under-prediction. Hence, R_u is expected to be smaller than the actual resistance as there are only small energy losses and the full capacity may not have been developed. As a result, K_{sp} is expected to be higher than unity ($K_{sp} > 1.0$).

-

CHAPTER 5 - DATA BASE BUILDUP

5.1 GENERAL

In order to examine the dynamic analyses and calibrate the Energy Approach method, extensive case study data was assembled. The information was divided into two major categories describing two data sets: set PD/LT and set PD. Data set PD/LT contains data for piles on which dynamic measurements, office analyses (CAPWAP or TEPWAP), and a static load test to failure have been conducted. Data set PD contains data for piles that were monitored by dynamic measurements during driving, followed by office analyses and occasionally a static proof test (not to failure). Section 1.3 outlines the source and/or reference of the obtained data. The following chapter describes the procedures used for analyzing the case studies comprising the data sets.

5.2 DATA SET PD/LT

The piles of data set PD/LT were analyzed in two stages: a static load test analysis followed by a dynamic measurements analysis. The static load test analysis was intended to produce a representative static resistance (denoted by R_s) for each pile, using several load test interpretation methods. The dynamic measurements analyses involved several different methods, including the application of computer programs specifically developed for the analysis of dynamic measurements taken during driving.

5.2.1 Static Load Test Analysis

A universal criterion capable of establishing the ultimate capacity of a pile is essential in improving the accuracy of static load test interpretations. Various ultimate load criteria have been proposed and used by researchers and design organizations (see, for example, Vesic, 1977 and Fellenius, 1989). Significant disagreements remain among these methods as they are based on different principles and produce different values under varying pile types and sizes, load test procedures, and surrounding soils.

Vesic (1972) pointed out that interpreting a pile's ultimate load based solely on a visual examination of its load-settlement curve (i.e., shape of the curve) may be misleading and can result in different pile capacities depending on the scale used to plot the curve. Figures 9 and 10 demonstrate this point by presenting the same load-settlement relations using two different scales. Figure 9 shows a load-settlement curve indicating a pile capacity of approximately 140 kips (623 kN) whereas the curve in figure 10 suggests that the pile's displacement at 140 kips (623 kN) may still be based on the elastic compression of the pile and that the pile capacity is approximately 170 kips (756 kN).

39 **Preceding page blank**



Figure 9. Load-settlement curve of pile-case 95 with the elastic compression line inclined at 20 degrees.



Figure 10. Load-settlement curve of pile-case 95 with a scale that does not consider the elastic compression of the pile (following Vesic, 1977).

One solution to this problem is to implement a common scale, based on the pile's elastic deformation. When plotting load-settlement curves, the elastic deformation of a fixed end, frictionless pile is expressed as:

$$\delta = \frac{PL}{EA} \tag{25}$$

where

 $\delta = calculated elastic deformation of the pile$ P = applied loadL = pile lengthE = elastic modulus of the pile materialA = cross-sectional area of the pile.

The elastic compression line obtained by equation 25 is based on the assumption that all of the load applied to the pile top is transferred to the pile toe. To implement a scale proportional to all load settlement curves, the elastic compression line should be inclined at an angle of about 20 degrees to the load axis (see figure 11).

In order to facilitate this scale, all of the load-settlement curves in set PD/LT were digitized using the program DIGITIZE, developed at University of Massachusetts-Lowell by Chernauskas and Paikowsky. These curves were then replotted, using the graphics software GRAPHER, to produce curves that were proportional to each pile's elastic compression line inclined at 20 degrees.

After replotting, each load-settlement curve was analyzed using five different failure load interpretation procedures: Davisson's Criteria, the Shape of Curve method, Limited Total Settlement methods ($\Delta = 1$ in [25.4 mm] and $\Delta = 0.1B$), and DeBeer's method.

(a) **Davisson's Criteria** (Davisson, 1972), or offset limit, defines the failure load of a pile as the load corresponding to the settlement that exceeds the elastic compression of the pile (δ) by an offset (X) equal to 0.15 in (3.8 mm) plus a factor equal to the diameter of the pile divided by 120. The offset is simply:

$$X = 0.15 + \frac{B}{120}$$
(26)

where B = diameter of the pile in inches.

The Davisson's Criteria line is parallel to the elastic compression line and predicts the failure load at its intersection with the load-settlement curve. Figure 11 illustrates the use of Davisson's failure criteria for load-settlement relations of pile-case 50, yielding a capacity of 817 kips (3634 kN).

(b) The Shape-of-Curve Method is a failure load approximation that usually yields a



Figure 11. Load-settlement curve for pile-case 50 with the elastic compression line inclined at approximately 20 degrees.

range of values over which the pile is considered at or near failure. The boundaries of this range can be determined by examining the minimum curvature in the loadsettlement curve through lines drawn tangent to the load-settlement curve (similar to the method proposed by Butler and Hoy (1977)). The failure range is relatively easy to define for load-settlement curves that exhibit general failure or plunging failure (rapid settlement with slightly increased loads) (see figure 11 for example). Piles that experience local failure, or non-plunging failure, are difficult to analyze using the shapeof-curve method because of the uniform changes in slope of lines drawn tangent to the curve. Figure 11 illustrates the use of the shape-of-curve procedure, yielding an estimated capacity range of between 685 kips and 825 kips (3047 kN and 3670 kN) with a representative average of 755 kips (3358 kN).

(c) The Limited Total Settlement Methods, $\Delta = 1$ in (25.4 mm) and $\Delta = 0.1B$ (Terzaghi, 1942), define the failure load as the load corresponding to settlements of 1 in and 0.1B, respectively, where B is the diameter of the pile. These methods are not applicable in many cases. For example, the elastic compression for a very long steel pile often exceeds 1 in (25.4 mm) and/or 0.1B without inducing any plastic deformation in the soil. Figure 11 shows as an example, a load-settlement curve for pile-case 50, a 24-in (610-mm) square concrete pile that experiences a plunging failure well before a displacement of 1 in (25.4 mm). Also, it is obvious that a settlement of 0.1B, or 2.4 in (61 mm) in this case, does not represent the failure load of this pile and, therefore, is not applicable.

(d) DeBeer's log-log Method (DeBeer, 1970) defines the failure load as the load corresponding to the intersection of two distinct slopes created by the load-settlement data plotted using logarithmic scales. Figure 12 illustrates the use of DeBeer's criteria for the load-settlement curve of pile-case 50, leading to an estimated capacity of 748 kips (3327 kN). The two slopes are especially visible for piles that experience plunging failures, yet when using DeBeer's method piles that undergo local failures, the result may be a range of values. As mentioned earlier, each load-settlement curve was digitized from the standard linear plots that they were presented on and the data was stored. This data was later plotted in logarithmic scales to utilize DeBeer's method.

(e) The Representative Static Capacity: The capacity results for each method were reviewed independently, based on the load-settlement curves for each pile. After considering the pile type, soil type, size of each pile, and the load test procedure, unrealistic results were eliminated, and the acceptable values were averaged, yielding a final static capacity (R_s). For example, for pile-case 50, presented in figures 11 and 12, the considered criteria were: Davisson's = 817 kips (3634 kN), shape of curve = 685-825 kips (3047-3670 kN), 1.0-in settlement = 887 kips (3945 kN), 0.1B settlement = NA, and DeBeer's = 748 kips (3327 kN). Excluding the 0.1B settlement method, which is not applicable, and 1.0-in (25.4 mm) settlement, which is clearly beyond the failure, the average of all the criteria led to a final static resistance assessment of $R_s = 773$ kips (3438 kN).

5.2.2 Dynamic Measurements Analysis

The analyses performed on piles in data set PD/LT employed office analysis (i.e., CAPWAP or TEPWAP) as well as several computer programs developed to process and manage force and velocity signals, including DIGITIZE, PDAP, INTEGRATE, and FILECHNG.

The dynamic analyses were performed in different ways depending on the completeness of each pile case. In all cases, the pile geometry (i.e., type, material, length of



Figure 12. Load-settlement data plotted on a logarithmic graph for pile-case 50 to determine the failure load according to DeBeer's method.

penetration, the soil at the pile's tip and side, and the blow count) was known before any type of analysis was initiated. The individual cases were divided into three distinct groups:

(a) Group 1 — pile cases with complete CAPWAP summaries, including
$$E_{max}$$
, D_{max} , F1, and V1.

(b) <u>Group 2</u> –	pile cases with incomplete CAPWAP summaries,			
	such as those missing E_{max} , D_{max} , F1, and/or V1.			

(c) <u>Group 3</u> — pile cases that were analyzed using TEPWAP.

(a) GROUP 1 - Complete CAPWAP Analyses

Pile group 1 contains the complete cases available in data set PD/LT. The most common adjustment necessary for the pile cases in this group was a ratio correction between the force at impact (F1) and the velocity at impact (V1). Theoretically, the force and velocity multiplied by the pile impedance are identical under a passing disturbance, as long as no other external forces act. The ratio between these values is:

$$\frac{VI\left(\frac{EA}{C}\right)}{FI}$$
(27)

where	Ε	=	modulus of elasticity of the pile material
	Α	=	cross-sectional area of the pile
	С	=	wave speed of the pile

and should be equal to unity. An acceptable ratio was considered to be 1.0 ± 0.1 . Beyond this ratio, a linear multiplier was applied to either or both parameters (force, velocity, or both) and to their byproducts, e.g., displacement and energy. The ratio between force and velocity may also be influenced by the precompression of a diesel hammer and hammer misalignment.

Precompression in a diesel hammer occurs as the air-fuel mixture is compressed by the ram just prior to combustion. This results in a force that is applied to the pile top. However, as the force is applied relatively slowly and before the actual impact between the ram and the pile top, there is not a corresponding velocity wave. This scenario results in a discrepancy between the impact force (F1) and the impact velocity (V1(EA/C)), as shown in figure 13. The force and velocity traces of pile-case 1, driven with a Delmag 30 diesel hammer, are shown in figure 13. The observed relations indicate the need for a force reduction (Δ_{total}), which is equal to the difference between Δpk and Δps . Prior to a correction, the ratio (V1(EA/C)/F1) for pile-case 1 was 0.874. The factor (Δ_{total}) represents the number of units by which the force must be reduced in order to produce an acceptable ratio according to equation 28. The magnitude of Δ_{total} and the reduction of F1 are performed as follows:

$$\Delta_{total} = \Delta_{pk} - \Delta_{ps} = 2 \text{ units} = \frac{2 \text{ units}}{38.5 \text{ units}} \times 250 \text{ kips} = 13 \text{ kips} (58 \text{ kN}) \quad (28)$$



Figure 13. Force and velocity (V^*EA/C) traces of pile-case 1, a steel HP12x74 that needed a force correction (not to scale).

$$FI = FI_{uncorrected} - \Delta_{total} = 335.4 \text{ kips} - 13 \text{ kips} = FI_{corrected} = 322.4 \text{ kips} (1434 \text{ kN})$$
(29)

The corrected F1 yields a new V1(EA/C)/F1 ratio, an adjusted E_{max} , and a corresponding uncorrected Energy Approach prediction (R_u) as follows:

$$\frac{VI(\frac{EA}{C})}{Fl_{corrected}} = 0.909$$
(30)

$$E_{\text{max}} = 18 \ \text{kip} - \text{ft} \times \frac{322.4 \ \text{kips}}{335.4 \ \text{kips}} = 17.3 \ \text{kip} - \text{ft} = R_{\mu} = 362 \ \text{kips} \ (1610 \ \text{kN}) \ (31)$$

The procedure for correcting F1 is also performed in a similar manner for adjusting V1 and the corresponding D_{max} , where $D_{max} = \int V(t) dt$. This is sometimes necessary when

either there is a significant hammer-pile misalignment that creates disturbance in the force and velocity measurements or there is a discrepancy in the measurement itself. The correction procedure for decreasing V1(EA/C) also uses the factor Δ_{total} as determined by the discrepancy in the F1 and V1(EA/C) measurements where Δ_{total} is converted to units of force. Similarly, V1(EA/C) is decreased by:

$$VI\left(\frac{EA}{C}\right)_{corrected} = VI\left(\frac{EA}{C}\right)_{uncorrected} - \Delta_{total}$$
(32)

producing a corrected ratio:

$$\frac{(VI\frac{EA}{C})_{corrected}}{FI}$$
(33)

and an adjusted D_{max}:

$$Dmax_{corrected} = \int VI_{corrected} dt$$
(34)

The corresponding uncorrected Energy Approach prediction is calculated using the adjusted D_{max} as follows:

$$Ru = \frac{Emax}{Set + \frac{Dmax_{corrected}}{2}}$$
(35)

(see chapter 4 for Energy Approach details). It should be noted that it is sometimes necessary to correct both the force and the velocity measurements given the proper circumstances. In general, very few pile-cases required correction, the majority of which needed very small adjustments. These corrections usually had an insignificant effect on the obtained J_c and Ru values.

After the static load test analysis and the dynamic analysis were completed, the Case damping coefficient (J_c) was back-calculated using equation 6 as outlined by Goble *et al.* (1980).

(b) GROUP 2 - Incomplete CAPWAP Analyses

The pile cases categorized in group 2 include piles from data set PD/LT that were analyzed via CAPWAP. Difficulties associated with retrieving and accumulating complete pile data cause pile cases to require more analysis in order to produce missing information essential for the study. Typical information missing from pile cases included E_{max} (the maximum energy delivered to the pile top) and D_{max} (the maximum displacement of the pile top). A typical pile case in group 2 includes a static load test plot, subsurface site information, blow count records, and CAPWAP predictions at EOD, BOR, and/or EOR, excluding the CAPWAP summary tables. The CAPWAP summary tables include pile characteristics, Case method predictions and crucial dynamic measurements (V_{max} , V_{fin} , V1*Z, F1, F_{max} , D_{max} , D_{fin} , E_{max} , and E_{fin}). In order to determine these missing dynamic parameters, a program was developed at UMASS-Lowell called INTEGRATE (written by L. Chernauskas). This program was specifically developed to calculate the uncorrected Energy Approach and the Case method similar to a more extensive and versatile program called PDAP (Pile Driving Analysis Program), which was developed by Paikowsky (1984). The program PDAP uses recorded field data from the PDA, enables it's manipulation and correction, and produces an Energy Approach prediction and a range of Case method predictions based on all the different variations for different J_c values.

INTEGRATE processes digitized force and velocity (V*EA/C) traces (see figure 14 for example) and, using the pile parameters as given by the user, produces the dynamic measurements listed above. INTEGRATE also calculates the uncorrected Energy Approach prediction and back-calculates the Case damping coefficient (J_c) using the following relationship:

$$J_{c} = \frac{RTL - FINAL Rs}{VI * \frac{EA}{C} - FI - RTL}$$
(36)



Figure 14. Digitized force and velocity multiplied by the impedance (EA/C) traces for pile-case 192 used for input into INTEGRATE.
UMASS-LOWELL GEOTECHNICAL ENGINEERING DYNAMIC PILE TESTING

SUMMARY OF INPUT PARAMETERS

FILE	33P1BOR
PILE LOCATION.	SITE 33
DATE OF ANALYSIS	2-10-92
PILE DESIGNATION	33P1-BOR
	HP12x74
HAMMER TYPE	B-400
NOMINAL ENERGY OF HAMMER (ft-kips)	46
PENETRATION DEPTH (tt)	114.4
2L/C (msecs)	14.39
TIME INTERVAL (msecs)	.1
PILE IMPEDANCE - EA/C (kip/sec/ft)	38.9
FINAL BLOW COUNT (bl/in)	16
T2 (offset from T1) (msecs)	14. 39

SUMMARY OF OUTPUT PARAMETERS

DMAX	0 787
DEIN	0 164
HAMMER EFFICIENCY (%)	69 14
EMAX (kip-ft)	31.80
EFIN (kip-ff)	25.55
VMAX (H/sec)	15 78
VFIN (ft/sec)	0.420
FMAX (kins)	637.38
EFIN (kips)	42.02
	-0.017
F1 (kins)	637.38
F2 (kins)	192 10
V1 (ft/sec)	15.78
V2 (ft/sac)	3.59
(V17EA/C)/E1	0.963

PILE CAPACITY (kips)

DAVISSON'S CRITERIA	800
SHAPE OF CURVE	800
Δ = .1B	598
Δ = 1 inch	52 2
DEBEERS LOG METHOD	800
FINAL R _s	800
CASE RTL	792
CAPWAP	715
ENERGY APPROACH RU (uncorrected)	89 8

Figure 15. INTEGRATE output of pile-case 192 showing the back-calculated Case J_c value and the Energy Approach prediction.

(Goble et al., 1980). The static load test results are denoted FINAL R_s, and must be supplied by the user. An example of the results of an INTEGRATE analysis of the force and velocity traces shown in figure 14 for pile-case 192 (33P1BOR) is shown in figure 15. After reviewing the force and velocity (EA/C) traces for a given pile case and the $(V1^{*}EA/C)/F1$ ratio, calculated by INTEGRATE, any necessary corrections and corresponding adjustments to E_{max} and D_{max} can be made, as outlined in section 5.5.2(a); and the uncorrected Energy Approach calculations can be performed.

(c) GROUP 3 - TEPWAP Analyses

Several pile cases in data set PD/LT were lacking the CAPWAP office analysis and, therefore, required wave match analysis to be performed. These pile cases were categorized in group 3 and all of them were analyzed using a computer program called TEPWAP. TEPWAP (Paikowsky, 1982; Paikowsky and Whitman, 1990; and Chernauskas, 1993) utilizes a procedure somewhat similar to the CAPWAP analysis described by Goble *et al.* (1970). This program allows the input of the measured velocity at the pile top as a function of time, solving for a set of parameters describing the soil resistance (dynamic and static) along the pile (see section 3.3.3). Adjustments of the matches are made until the calculated force at the top matches that measured. A good agreement between CAPWAP and TEPWAP analyses was presented by Paikowsky (1982) and further confirmed by Chernauskas (1993).

The pile cases in group 3 were initially analyzed in the same manner as those in group 2, whereby their force and velocity traces were digitized with respect to time using the program DIGITIZE and processed using INTEGRATE. After these steps were successfully completed, three data files were created for each case: an input file, an identification file, and a pile/soil file. An input file for TEPWAP is created using the program DIGPWAPE that processes digitized force and velocity traces and prepares them in the same manner as the PDA. Figures 16 and 17 show the identification file and the pile/soil file for pile-case 191, respectively. These files, along with the digitized force and velocity traces (see figure 18 for example), are necessary for TEPWAP analyses. Iterations are performed, where the user is required to adjust the soil properties (i.e., side and tip damping and quake, and side and tip resistance) until an acceptable force at the top (obtained from the above procedure) to the measured force at the top of pile-case 191.

This particular pile case appears to be exhibiting pile plugging near the tip as indicated by the sudden observed force "jump" near 2L/C and again near 4L/C. Pile plugging is most commonly associated with open-pipe piles or H-piles. It usually refers to the phenomenon that occurs when soil enters the open-pipe pile during driving until the inner-soil cylinder develops sufficient resistance to prevent further soil intrusion (see Paikowsky *et al.*, 1989; Paikowsky and Whitman, 1990). The development of friction along the web of an H-pile can also develop enough resistance to prevent soil intrusion, causing the H-pile to become "plugged." When an H-pile becomes plugged, it then

Figure 16. Example of the pile identification information of pile-case 191 used as input for the TEPWAP analysis.

UNIVERSITY OF MASSACHUSETTS - LOWELL GEOTECHNICAL ENGINEERING TEPWAP ANALYSIS

SOIL AND PILE PROPERTIES ALONG PILE ELEMENTS

3

element no.	dist from gauges	area	weight	stiffn	regist	sum of resist	damp	quake	quake rebnd ratio	upwrd resist ratio
	(fī)	(sq.in)	(ibs.)	(k/in)	(kips)	(kips)	(s/ft)	(in.)	(%)	(%)
					_					
з	5.3	21.8	390.1	10364	0.0	439.0	.000	û. 000	0.0	0.0
4	10.5	21.8	390.1	10384	5.0	434.0	.020	0.300	100.0	-50.0
5	15.8	21.8	390.1	10364	5.0	429.0	.020	0.300	100.0	-50.0
6	21.0	21.8	390.1	10 364	5.0	424.0	020	0.300	100.0	-50.0
7	28.3	21.8	390.1	10384	5.0	419.0	.020	0.300	100.0	-50.0
8	31.8	21.8	390.1	10364	5.0	414.0	.020	0.300	100.0	-50.0
9	38.8	21.8	390.1	10364	0.0	414.0	.010	0.300	100.0	-50.0
10	42.1	21.8	390.1	10364	0.0	414.0	.010	0.300	100.0	-50.0
11	47.3	21.8	390.1	10364	50	414.0	010	0.300	100.0	-50.0
12	52.0	21.8	390.1	10364	0.0	414.0	010	0.300	100.0	-50.0
13	57.9	21.8	390.1	10364	0.0	414.0	.010	0.300	100.0	-50.0
14	63.1	21.8	390.1	10364	5.0	409.0	.010	0.300	100.0	-50.0
15	68.4	21.8	390.1	10364	5.0	404.0	.010	0.300	100.0	-50.0
16	73.8	21.8	390.1	10364	8.0	396.0	.010	0.300	100.0	-50.0
17	78.9	21.8	390.1	10364	8.0	388.0	.010	0.300	100.0	-50.0
18	84.2	21.6	390.1	10364	8.0	380.0	.010	0.300	100.0	-50.0
19	88.4	21.8	390.1	10364	8.0	372.0	.010	0.300	100.0	-50.0
20	94.7	21.8	390.1	10364	8.0	364.0	.010	0.300	100.0	-50.0
21	99.9	21.8	390.1	10384	8.0	356.0	.010	0.300	100.0	-50.0
22	105.2	21.8	390.1	10364	8.0	348.0	.01 0	0.300	100.0	-50.0
23	110.5	21.5	390. t	10364	8.0	340.0	.010	0.300	100.0	-50.0
24	115.7	21.8	390.1	10364	80.0	260 0	.090	0.150	100.0	-50.0
25	121.0	21.8	390.1	٥	100.0	160.0	.080	0.150	100.0	-50.0
tip					160.0	00	.080	0.150	100.0	

Figure 17. Example of the soil and pile properties used along the pile elements of pile-case 191 as input for the TEPWAP analysis.



Figure 18. Measured force and velocity multiplied by the impedance (EA/C) traces of pile-case 191 used by the TEPWAP analysis.



Figure 19. Comparison between measured force near the top of pile-case 191 and the calculated force from TEPWAP analysis.

UNIVERSITY O GEOTE T	F MASSA CHNICAL EPWAP A	ACHUSEI . ENGINE ANALYSIS	TS - LOWELL ERING S
****************	******	********	****************
SUMN	ARY OF F	INAL RES	ULTS
END RESISTANCE SIDE FRICTION TOTAL CAPACITY	- -	160.0 k 279.0 k 439.0 k	ips ips ips
PERCENT IN FRICTIO	N -	63.6%	
TOP QUAKE SET AT TIP		0.79 ind 0.32 ind	thes thes
THE SET CORRESPO		Blows Pe	r Inch.
MAXIMUM CALCULAT MAXIMUM MEASURE	ED ENER	GY - Y -	32.362 kip*ft 32.674 kip*ft
ENERGY DIFFERENC	E (CALC -	MEAS) -	-0.312 kip*ft
MAXIMUM COMPRES IN ELEMENT #3 ON I	SION FOR	ICE IS - 1 32	636.49 kips
MAXIMUM TENSION I IN ELEMENT #19 ON	FORCE IS	- N 181	-139.18 kips
THE MAXIMUM STRE	SS IS -		29.20 kips/sq in.

Figure 20. Summary of the final results from TEPWAP analysis performed on pile-case 191.

assumes the penetration characteristics of a large displacement pile (i.e., with a closed rectangular tip). Pile plugging is shown to have the following marked effects: significant contribution to the capacity of piles driven in sand; delay in capacity gain with time for piles driven in clay; and changes in the behavior of piles during installation, causing it to differ from that described by the models commonly used to predict and analyze pile driving (Paikowsky and Whitman, 1990). Further investigation into pile-case 191 shows that the H-pile is embedded over 114 ft (35 m) into silty sand. These conditions are ideal for pile plugging to occur and, therefore, plugging can be attributed to the force match disagreement at 2L/C and again at 4L/C by TEPWAP as shown in figure 14.

The final summary of results from TEPWAP analyses are produced for each case (see figure 20 for example). These summaries allow the user to investigate the compressive and tensile stresses developed in the pile during driving (e.g., concrete piles) as well as the side and tip resistance and the measured and calculated energy delivered to the pile.

All of the pile cases that were analyzed using TEPWAP are footnoted in the data set tables in chapter 6. The Case damping coefficients for these cases were calculated as part of the INTEGRATE output as previously stated.

5.3 DATA SET PD

Data set PD contains information related to 403 piles; the vast majority were sorted from information related to 428 piles provided by Pile Dynamics, Inc. of Cleveland, Ohio, as part of their support of the Energy Approach method research. Large portions of the PD data set analysis were performed by McDonnell (1991). The data set contains the following information:

- Pile identification, which also refers to the time of measurement, e.g., end of driving (EOD) or beginning of restrike (BOR).
- Soil type on the side and at the tip of the piles.
- Pile type, geometry, material, and modulus of elasticity.
- Hammer type and blow count.
- Resistance obtained by CAPWAP analysis.
- All parameters pertinent to the CAPWAP analysis, e.g., damping factors and quake values.
- Maximum energy, force, velocity, and displacement of the analyzed blow.

Resistance obtained from different Case method evaluations.

The data set is subgrouped according to pile and soil types, as shown in table 3.

Pile Type/Soil Type	Sand and Silt	Clay and Till	Rock	N/A*	Total		
Small Displacement	26	21	29	-	76		
Large Displacement	92	50	78	22	242		
Miscellaneous#	40	21	19	5	85		
Total	158	92	126	27	403		
 Soil type not available. Miscellaneous piles include timber, monotube, pipe with H beams, etc. 							

Table 3. Subgrouping of the piles in data set PD (indicating the number of piles in each group).

The large size of data set PD provides an excellent basis for the examination of any possible parameter relations. The complete summarized data set is presented in table 24. Correlations between the Energy Approach vs. CAPWAP predictions for the various pile/soil combinations shown in table 3 are presented in chapter 9. The total number of correlations is 15 (see table 3 for number of cases in each category).

CHAPTER 6 - DATA SET PD/LT

6.1 GENERAL

This chapter summarizes the pile cases in data set PD/LT. Four tables are used to group the information into four categories (see appendix A). The groups are as follows: site and pile information (table 20), pile driving and dynamic measurements (table 21), parameters of dynamic analyses (table 22), and pile capacity based on static load test results and dynamic analyses (table 23). The following sections discuss the breakdown of these tables and provide:

- Details of where the information was gathered for each column.
- Methods used to produce the information.
- Definitions of any symbolism used.

For a list of references and contributors to data set PD/LT, see section 1.3.

6.2 SITE AND PILE INFORMATION - TABLE 20

(a) Columns 1-4

The first four columns of table 20 list the case number for each case in data set PD/LT (208 total), the pile-case number, the site reference number, and the site location, respectively. A number is assigned to each pile-case in column 1 for all four tables to provide easier transition from one table to another. The next column lists the pile-case number that corresponds to the pile number as labeled in individual site plans and reports. Included in the pile-case numbers are extensions that designate the time of driving when measurements were taken (e.g., EOD=end of driving, BOR=beginning of restrike, EOR=end of restrike, DD=during driving, and BORL=BOR after load test). A reference number is assigned to each pile-case depending on which project the particular pile was driven and the location column lists the general area where the driving site is located (e.g., county, state, province, or country).

(b) Columns 5-8

The next four columns of table 20 provide pile information, including the pile geometry and the depth to which the pile was driven at the time of analysis. Column 5 briefly lists the pile type according to its material and its cross-sectional dimensions. For example, the notation HP, CEP or CP, and OEP represent a steel H-pile, a steel closed-end pipe pile, and a steel open-end pipe pile, respectively, whereas PSC, VC, and RC represent a pre-stressed concrete pile, voided concrete pile, and simply reinforced-concrete pile. Any timber piles listed refer to those that were treated prior to driving. Following the pile type notations are the dimensions of the pile. For the closed-end and open-end pipe piles, the wall thickness dimensions can be back-calculated from the piles cross-sectional area listed in column 6. Typically, the pile length below gauges and its penetration depth, shown in columns 7 and 8, were taken from the field driving records (when available) as reported by the field engineer. Many times, the length below gauges is reported as a general value for several piles at one site (e.g., length below gauges = pile length - 3 ft [0.91 m]). There are some cases in which there is no indication as to exact lengths and, instead of assuming the field conditions, the length below gauges according to the CAPWAP results is used.

(c) Columns 9 and 10

The soil type at the side and tip of each pile-case are listed in the final two columns of table 20. This information is obtained from subsurface investigation reports and boring logs and it is considered essential to all pile-cases in data set PD/LT. The soil descriptions listed under side and tip are generalized according to the basic nature of the soil. For example, a pile that is reported to have a sandy silt with traces of clay is listed as sandy-silt. Also, soil types listed in the following manner, cl-sa-silt, for instance, refer to a clayey sandy silt with the most predominant soil listed at the end of the classification. Several abbreviations are used to condense the soil descriptions, these include: sa = sand, si = silt, cl = clay, ti = till, gr = gravel, d. = dense, l. = loose, clcr = calcareous, and carb = carbonious.

6.3 PILE DRIVING AND DYNAMIC MEASUREMENTS - TABLE 21

The pile driving and dynamic measurements information of each pile-case are summarized in table 21.

(a) Columns 1 and 2

In accordance with table 20, the first and second columns in this table list the case number and the pile-case number of each pile-case.

(b) Columns 3-5

The following three columns provide relevant hammer information for each case, such as the hammer type, the rated hammer energy, and the maximum energy delivered to the pile top. The letter abbreviations used denote the manufacturers name, for instance: B=Bermingham, D=Delmag, K=KC=Kobelco, Con=CN=Conmaco, LB=Link Belt, ICE=International Construction Equipment, KB=Kobe, Vul=Vulcan,M=MH=Mitsubishi, and DE=MKT. The abbreviations are followed by the model size (i.e., B-400 refers to a Bermingham 400 diesel hammer). The rated hammer energies are shown according to the manufacturers recommendations. The energy delivered refers to the maximum delivered energy, which is based on the dynamic measurements and was usually determined from office analyses (i.e., CAPWAP/TEPWAP). It should be noted that often some discrepancy exists between the measured energy in the field as calculated by the PDA to the one reported by CAPWAP. This may be a result of several reasons:

- "Correction" of the waves for better proportionality before carrying out the office analysis.
- Older PDA models require the storage of data in an analog form on magnetic tapes. The data retrieval in those cases always contains some error.
 - Field analysis may provide an average value while the office analysis refers to one particular blow. For reasons of consistency, whenever possible, the delivered energy value refers to the one reported by CAPWAP as the maximum energy (E_{max}).

(c) Column 6

The blow count (reported in blows per inch, BPI) is listed in the sixth column of table 21. Several times the blow count records were only in blows per foot and it was, therefore, necessary to convert these values to blows per inch. This conveniently allows the pile set to be derived in units of inches (set = 1/blows per inch). An asterisk follows each blow count that was converted from blows per foot to blows per inch.

(d) Columns 7-10

Following the blow count is the pile impedance, velocity at impact (V_{imp}) , force at impact (F_{imp}) , the ratio [(V1*EA/C)/F1], and the maximum pile displacement (D_{max}) . As mentioned in chapter 5, the pile impedance is used to examine the ratio between the velocity and the force waves. The impedance is calculated using:

$$\frac{EA}{C} \tag{37}$$

where E = modulus of elasticity of the pile material at the point of measurement

- A = cross-sectional area of the pile at the point of measurement
- C = wave speed of the pile.

The impedance is reported in kips per foot per second. The velocity at impact, V_{imp} (ft/s); the force at impact, F_{imp} (kips); and the maximum displacement of the pile, D_{max} (in), are obtained from dynamic measurements (see chapter 4). These values were typically taken from CAPWAP summaries and/or INTEGRATE results. The ratios between velocity and force [(V1*EA/C)/F1], reported in table 21, were those corrected when necessary, as discussed in chapter 5.

6.4 PARAMETERS OF DYNAMIC ANALYSES - TABLE 22

The parameters associated with dynamic analyses (i.e., quake and damping) are listed in table 22.

(a) Columns 1 and 2

The first two columns of table 22 list the case numbers and pile-case reference numbers consistent with tables 20 and 21.

(b) Column 3

The Case damping coefficient (J_c) reported in column 3 was back-calculated using the static load test results (R_s) as the "predicted capacity" for each particular pile and the "standard form" of the Case method utilizing equation 36.

(c) Columns 4 and 5

Columns 4 and 5 present the pile impedance and the calculated 2L/C, respectively. The magnitude 2L/C is the time that it takes for a wave to reach the pile tip and reflect back to the pile top. This term is reported in milliseconds; L represents the pile length below gauges (feet) and C represents the wave speed of the pile material (feet per second).

(d) Columns 6-9

The last four columns list the tip and side quake and the tip and side damping, respectively. These values are used as input into CAPWAP or TEPWAP analyses and were obtained from their summaries. Those values that were used in TEPWAP analyses are denoted with an asterisk. The quake values are reported in inches and the damping is reported in units of seconds/feet.

6.5 PILE CAPACITY: STATIC TESTS AND DYNAMIC ANALYSES - TABLE 23

The static load test results for each pile in data set PD/LT were analyzed using five different failure load interpretation procedures: Davisson's Criteria, Shape-of-Curve method, the Limited Total Settlement methods ($\Delta = 1$ inch and $\Delta = 0.1B$), and DeBeer's method. These procedures are discussed in detail in chapter 5.

(a) Columns 1-3

The case number, pile-case reference number, and load test type are listed in the first three columns. The load test types have been abbreviated: S=standard, Q=quick, SM=slow maintained, LLT=Louisiana load test, FQ=Florida modified quick, and CRP=constant rate of penetration.

(b) Columns 4-8

Following the load test type column are the five load test interpretation methods used (all results are given in kips). The abbreviation NA refers to methods that were not applicable.

(c) Column 9

The static resistance (R_s) represents the average of the resistances given by the five methods (see chapter 5 for discussion).

(d) Columns 10-12

The last three columns report the capacity predictions from CAPWAP or TEPWAP, the Energy Approach, and the Energy Approach correction factor (K_{sp}) . The predictions based on TEPWAP analyses are denoted with an asterisk. The CAPWAP/TEPWAP and Energy Approach predictions are reported in kips.

CHAPTER 7 - DATA SET PD

7.1 PILE/SOIL AND DYNAMIC MEASUREMENTS OF DATA SET PD - TABLE 24

The information of data set PD was provided by Pile Dynamics, Inc. of Cleveland, Ohio, as part of their support of the Energy Approach method research. Table 24 in appendix B summarizes the information describing the pile geometry, skin and toe soil, and dynamic measurements of the piles comprising data set PD (403 in all). Initially, this data set consisted of 428 pile-cases, however, 25 cases were removed because they were either duplicates or they were missing information. Table 24 categorizes the PD pile-cases according to pile type and soil type. A summary of these categories is presented in table 3. The correlations between the Energy Approach and CAPWAP are presented in chapter 9.

(a) Columns 1 and 2

The first two columns in table 24 list the reference number and the pile name according to designations made by Pile Dynamics, Inc.

(b) Columns 3 and 4

The side and toe soil are abbreviated in a similar manner to table 20, however, there are several additional soil types included: alluv=alluvial, clayston=claystone, coopermar=coopermarl, limestn=limestone, sastone=sandstone, overburd=overburden, dolom=dolomite, cobbl=cobbles, til=till, tilall=alluvial till, and sigr=silty gravel.

(c) Columns 5-9

The pile type and geometry are given in column 5 and are abbreviated in a similar fashion to table 20. The length below gauges, cross-sectional area, and modulus of elasticity are listed in columns 6, 7, and 8, respectively, and their units are as shown. Hammer type is listed in column 9 and abbreviations are consistent with those in table 20. Additional abbreviations include RAY=Raymond and IHC=IHC Hydrohammer.

(d) Columns 10-14

The dynamic measurements are reported in columns 10 through 13 and are listed as follows: FMX=maximum force at the pile top (kips), EMX=maximum energy delivered to the pile (kip-ft), VMX=maximum velocity at the pile top (ft/s), and DMX=the maximum displacement of the pile top (in). Column 14 contains the blow count for each pile-case reported in blows per inch.

(e) Columns 15 and 16

The last two columns list the CAPWAP predictions (in kips) and the corresponding Energy Approach predictions (in kips) for each pile-case, respectively.

63 Preceding page blank

7.2 SIDE/TIP QUAKE AND DAMPING PARAMETERS OF DATA SET PD - TABLE 25

Table 25 in appendix B summarizes the quake and damping parameters used for both the side and tip of each PD pile-case. The first five columns are identical to table 24, however, the pile-cases are listed in ascending order according to the reference numbers in column 1.

(a) Columns 6 and 7

The quake parameters used for the side and tip soil are listed in columns 6 and 7, respectively. These values are reported in inches.

(b) Columns 8 and 9

The last two columns of table 25 list the damping parameters used for the side and tip soil of each pile-case (reported in seconds per foot).

CHAPTER 8 - ANALYSIS OF DATA SET PD/LT

8.1 OVERVIEW

8.1.1 Purpose

The aim of this chapter is to present the analysis of the pile-cases in data set PD/LT in two forms by using:

- Graphical correlations, e.g., between static load test results and dynamic predictions (i.e., CAPWAP/TEPWAP and the Energy Approach), considering different factors such as pile and soil type, time of driving, and driving resistance.
- Statistical analyses in combination with the graphical correlations in order to establish conclusions and recommendations.

8.1.2 Outline

Three different types of correlations were investigated for the pile cases of data set PD/LT. The three categories and their rationales are presented below.

(a) Damping Parameters-Soil Type Correlations

One of the basic concepts presented in this research is that the different damping parameters fulfill the need for absorbing energy rather than truly representing either the soil or the physical phenomena it is subjected to. As such, correlations were built between the different damping parameters (Smith side and tip and the Case damping) and soil type, in order to examine the existence or nonexistence of such relations. The correlations of this category are presented in section 8.2.

(b) Prediction Methods-Load Test Capacity

Three dynamic analysis methods are examined throughout this research: (1) the office analyses (CAPWAP/TEPWAP), the field analysis (the Case method), and the proposed Energy Approach. Correlations were built between the predictions of CAPWAP/TEPWAP analysis and the Energy Approach analysis to the actual capacity based on the load test results. No correlations were built between the Case method predictions and the load test results, due to the following reasons:

- The method has a variety of shapes in which it can be implemented (see section 3.5), hence, no "unique" value would be valid.
- Previous studies (see section 3.5.5) suggested limited accuracy.

• The method is based on the notion of an existing correlation between the J_c damping parameter and the soil type at the tip. This was proven not to exist in the correlations described in group (a) above (see section 8.1.2).

In order to examine the influence of different factors (e.g., pile shape, driving resistance, time of driving, soil type, etc.), these correlations were built from the most generic cases (e.g., CAPWAP vs. load test results for all piles) to the private cases (e.g., CAPWAP predictions vs. load test results for small displacement piles in sand).

The correlations of this category are presented in sections 8.3.2 and 8.3.3 for different pile and soil types and in section 8.3.4 for different driving times. Their statistical analyses and interpretations are presented in section 8.4.

(c) Office Method/Field Method Predictions

Data set PD/LT contains information that is difficult to obtain. In general, very few load tests are carried out and, of those, only a small portion are carried out to failure. A considerably smaller portion is monitored dynamically during driving. As such, a strong correlation between the dynamic methods themselves may prove beneficial where load test data is not available. Correlations between the different predictions can therefore be compared to those obtained for data set PD for which static resistance is not available. These correlations are presented as part of sections 8.3 through 8.6 and are compared to those of data set PD in chapter 9.

8.2 DAMPING PARAMETERS AND SOIL TYPE GRAPHICAL CORRELATIONS

As previously discussed in chapter 4, viscous damping accounts practically for different energy losses, including radiation, soil inertia, and viscosity in cohesive soils. The damping parameters and their calibrations based on soil type have therefore been questioned.

8.2.1 Case Method Damping

The Case damping coefficient (J_c) is based on viscous damping in a dimensionless form, and is assumed to be related to the soil type at the pile's tip. Figure 21 presents the back-calculated Case damping coefficients for data set PD/LT vs. soil type at the pile tip. J_c was calibrated using the static capacity R_s and the "standard" Case method, as outlined by Goble *et al.* (1975) (see equations 6 and 36). It is shown that for the 208 pile-cases reported, no specific correlation exists between the soil type and the damping coefficient. Moreover, the obtained negative damping coefficients have no physical meaning and should be reviewed only for the purpose of illustrating the limitations of the J_c coefficient.

8.2.2 Smith Damping

Figures 22 and 23 compare the Smith damping coefficients (side and tip) used by CAPWAP/TEPWAP to the soil type at the side and tip of the pile, respectively. Corrections were not made to the office analysis, hence, the capacities obtained by the presented analysis reflect the predicted capacity and not the actual static capacity. A large variation of the damping parameter values can be observed for each soil type. Consequently, no specific correlation was made between the damping coefficients and soil type. These relations are further examined for the pile-cases of data set PD in chapter 9 and for all combined cases (581) in chapter 10.

8.3 DYNAMIC PREDICTION-STATIC CAPACITY GRAPHICAL CORRELATIONS

8.3.1 Correlations Breakdown

The graphical relationships between CAPWAP/TEPWAP and the Energy Approach to the static load test results were produced for all PD/LT pile-cases. These relationships are shown in the form of scatter plots (scattergrams). These plots were necessary as no statistical analysis can provide the actual observed information. The scatter plots were further divided into subgroups based on:

- Pile type (i.e., large and small displacement), presented in section 8.3.2.
- Soil type at the pile tip, presented in section 8.3.3.
- Time of driving (i.e., EOD = end of driving, BOR = beginning of restrike), presented in section 8.3.4.

The scatter plots for each different subgroup are shown in a consistent order: (1) static load test vs. CAPWAP/TEPWAP predictions, (2) static load test vs. Energy Approach predictions, and (3) CAPWAP/TEPWAP predictions vs. Energy Approach predictions. A flow chart illustrating the breakdown of all cases is presented as table 4. Each correlation graph includes a first-order best-fit line through zero (shown as the solid line), the corresponding coefficient of determination (r^2) , and a set of dashed lines representing the ratio between the actual capacity over the predicted one to allow the assessment of over- and under-predictions. For example, points falling on a dashed line labeled 0.80 designates an over-prediction, where the actual static capacity is 80 percent of the predicted capacity. It should be noted that this ratio is a direct multiplier, hence, the ratio represents the value that when multiplied by the prediction will give the "correct" capacity. This is the inverse to the ratio of the predicted over measured capacity used, for example, by Olson and Dennis (1989) or Briaud *et al.* (1988). The



Table 4. Breakdown of all PD/LT categories.

breakdown of the best-fit line (using linear regression) for all cases is presented in tables 5, 6, and 7.

8.3.2 Pile Type Correlations

(a) All Piles

The following graphs compare static load test results, CAPWAP/TEPWAP, and the Energy Approach, based on the pile type and the soil type at the pile tip.

Figures 24, 25, and 26 present all PD/LT pile-cases in all types of soil. As indicated earlier, all relationships are shown in the following sequence:

- (1) CAPWAP/TEPWAP vs. Static Capacity (figure 24).
- (2) Energy Approach vs. Static Capacity (figure 25).
- (3) CAPWAP/TEPWAP vs. Energy Approach (figure 26).

The information in figure 24 indicates that a large scatter exists when comparing the predicted capacity of the office analyses to the actual static capacity. The predicted capacity ranges from over-predictions of about 0.6 (predicted over actual ≈ 1.7) to a maximum under-prediction of 4.4, with most cases falling within the under-prediction ratio of 2.5 (predicted over actual ≈ 0.4). Overall, the tendency is of under-prediction, with the best-fit line (forced through zero) indicating a ratio of 1.265.

Figure 25 also exhibits a scatter when comparing the Energy Approach predictions to the actual load test results. The predictions range from under-predictions of 1.67 (predicted over actual ≈ 0.6) to over-predictions of 0.45, with most cases falling within the over-prediction ratio of 0.50 (predicted over actual ≈ 2.0). The best-fit line indicates an overall over-prediction with a ratio of 0.839.

It is important to note that the range of under-prediction to over-prediction of the office analyses is about twice that of the Energy Approach. The maximum over-prediction of CAPWAP is 0.57 and the under-prediction is 4.41, compared to the Energy Approach method that ranges between 0.45 and 1.74. These numbers indicate a range of under- to over-prediction of 7.74 for the office analyses, compared to 3.88 for the Energy Approach. This important observation becomes clearer when scattergrams are built as the relationships between the ratio of the actual capacity over the predicted capacity (the slopes in figures 24 and 25) versus the predicted capacity. These relationships are presented in figures 27 and 28 (for the office method and the Energy Approach method, respectively) and clearly demonstrate the large scatter in the prediction ratios in the office methods when compared to that of the Energy Approach. The linear best-fit lines of the data in figures 27 and 28 are:

 $\begin{array}{l} K_{sw} = 1.4867 - 0.00024 R_{u} \\ K_{sp} = 1.0259 - 0.00013 Q_{u} \end{array}$

Ksw = Static Load Test Results / CAPWAP or TEPWAP predictions								
Pile- Linear Regression								
Case	Number		Best Fit	Forced through	gh Zero			
Group		x-coefficient	y-intercept	r squared	x-coefficient	r squared		
ААА	206	1.127	97.3	0.707	1.265	0.692		
AAS	141	1.128	112.2	0.767	1.272	0.749		
AAC	51	1.057	140.7	0.413	1 <u>.31</u> 9	0.383		
AAR	14	0.937	-14.0	0.581	0.908	0.580		
AEA	97	1.065	151.7	0.779	1.248	0.740		
ABA	109	1.344	-36.2	0.616	1.284	0.614		
LAA	162	1.315	32.2	0.555	1.372	0.554		
LAS	118	1.360	3.6	0.595	1.366	0.595		
LAC	43	1.164	118.3	0.411	1.391	0.393		
LEA	68	1.450	37.2	0.530	1.529	0.528		
LBA	94	1.385	-48.3	0.598	1.307	0.596		
SAA	44	1.074	38.8	0.934	1.108	0.932		
SAS	23	1.048	138.4	0.968	1.142	0.952		
SAC	8	0.854	104.8	0.688	1.021	0.653		
SAR	13	0.980	-35.4	0.37 8	0.908	0.376		
SEA	29	1.073	52.2	0.936	1.113	0.933		
SBA	15	0.922	76.3	0.812	1.069	0.785		

Table 5. Linear-regression analysis of K_{sw} for selected PD/LT pile-cases.

Pile-case legend:

XXX

 first letter denotes pile type: A = all piles, L=large displacement, and S=small displacement.

- second letter denotes time of measurement: A=anytime E=end of driving, and B=beginning of restrike.

 third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

Ksp = Static Load Test Results / Energy Approach predictions								
Pile- Linear Regression								
Case	Number		Best Fit		Forced throu	gh Zero		
Group		x-coefficient	y-intercept	r squared	x-coefficient	r squared		
AAA	208	0.736	111.5	0.723	0.839	0.703		
AAS	141	0.721	130.3	0.725	0.831	0.700		
AAC	53	0.789	74.2	0.675	0.872	0.666		
AAR	14	0.864	-18.3	0.745	0.830	0.744		
AEA	98	0.791	126.6	0.830	0.900	0.804		
ABA	110	0.677	111.8	0.597	0.786	0.578		
LAA	164	0.668	160.8	0.579	0.832	0.534		
LAS	118	0.634	184.2	0.548	0.816	0.489		
LAC	45	0.784	88.4	0.669	0.882	0.656		
LEA	69	0.787	145.2	0.648	0.966	0.603		
LBA	95	0.669	119	0.569	0.780	0.550		
SAA	44	0.816	58.2	0.920	0.856	0.916		
SAS	23	0.795	129.3	0.930	0.863	0.916		
SAC	8	0.767	28.1	0.713	0.804	0.711		
SAR	13	0.935	-57.5	0.628	0.829	0.619		
SEA	29	0,809	73.2	0.922	0.851	0.917		
SBA	15	0.914	-7. <u>1</u>	0.737	0.902	0.737		

Table 6. Linear-regression analysis of K_{sp} for selected PD/LT pile-cases.

Pile-case legend:

 first letter denotes pile type: A = all piles, L=large displacement, and S=small displacement.

- second letter denotes time of measurement: A=anytime E=end of driving, and B=beginning of restrike.

- third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

Kew = CAPWAP or TEPWAP predictions / Energy Approach predictions								
Pile- Linear Regression								
Case	Number		Best Fit	Forced through	gh Zero			
Group		x-coefficient	y-intercept	r squared	x-coefficient	r squared		
AAA	206	0.573	73.1	0.782	0.641	0.766		
AAS	141	0.592	54.3	0.810	0.637	0.802		
AAC	51	0.479	121.8	0.607	0.624	0.539		
AAR	14	0.720	96.8	0.783	0.901	0.730		
AEA	97	0.675	26.1	0.861	0.698	0.859		
АВА	109	0.424	173.6	0.693	0.593	0.553		
LAA	162	0.420	163.9	0.701	0.589	0.554		
LAS	118	0.415	172.1	0.732	0.586	0.571		
LAC	43	0.423	150.3	0.567	0.600	0.446		
LEA	68	0.406	158.7	0.611	0.612	0.407		
LBA	94	_0.411	181	0.670	0.581	0.549		
SAA	44	_0.742	32.1	0,939	0.764	0.937		
SAS	23	0.751	-1.4	0.942	0.750	0.942		
SAC	8	0.876	-75.3	0.986	0.779	0.971		
SAR	13	0.552	189.4	0.557	0.900	0.329		
SEA	29	0.736	35.4	0.940	0.757	0.939		
SBA	15	0.971	-80.2	0.871	0.834	0.851		

Table 7. Linear-regression analysis of K_{ew} for selected PD/LT pile-cases.

Pile-case legend:

XXX

 first letter denotes pile type: A = all piles, L=large displacement, and S=small displacement.

- second letter denotes time of measurement: A=anytime E=end of driving, and B=beginning of restrike.

- third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

in which R_u and Q_u are the predicted capacities (in kips) by the office method and the Energy Approach method, respectively. These best-fit linear-regression lines indicate that:

- Predictions are not a function of the load, although both equations indicate a reduction in the ratios $(K_{sw} \text{ and } K_{sp})$ with the increase of the load. This increase is very small for both prediction methods.
- Average K_{sp} -value with load is 1.03, where the average K_{sw} with load is 1.49.

A few additional general observations can be made regarding the trends shown in figures 24, 25, 27, and 28:

- No specific correlations seem to exist between accuracy in prediction and soil type.
- Small displacement piles seem to have significantly less scatter than the one observed for the large displacement piles.

The relationship between the predicted capacities of the office methods and the Energy Approach is shown in figure 26. The information demonstrates consistent correlation within the range of 1.00 to 0.40. The best-fit line forced through zero is the ratio of 0.641, which means that the Energy Approach predictions are about 1.56 times those of the office predictions. This ratio is close to: (1) the ratio between the mean prediction ratio of the office methods (1.367) and the Energy Approach (0.925), which leads to 1.48; and (2) the best-fit ratio of the office methods (1.265) and the Energy Approach (0.839), which leads to 1.51.

In observing figure 26, it can also be noted that the scatter of the small displacement piles is much smaller than that of the large displacement piles. Moreover, the ratio of best fit for the small displacement is 0.764 (see figure 34) with a mean value of 0.796 (see table 5). This observation has a special meaning as it indicates that in the cases where small soil inertia takes place, both methods are in much better agreement, with the Energy Approach prediction only about 1.3 times that of the office methods.

(b) Large Displacement Piles

The relationships pertaining to large displacement piles for all soil types are shown in figures 29 through 31.

The relationship between the office analyses and the actual static capacity for large displacement piles shows significant over-predictions. The best-fit line yields an increase from 1.265 — obtained for all piles in figure 24 — to 1.370. The prediction ratios range from 4.41 to 0.57 with most cases in the range of 2.50 to 0.80 (actual over-prediction).

This emphasizes the outlined notion that energy loss takes place mainly due to soil inertia. Hence, the signal matching techniques using viscous damping models can not correctly represent the actual mechanism and, as a result, under-predict the actual pile capacity. It can also be mentioned in this context that the best-fit line ratio for all small displacement piles (to be presented in figure 32) is 1.108.

The information in figure 30 yields similar results to those in figure 25, with a considerable scatter between Energy Approach predictions and actual load test results. No significant changes can be seen from figure 25 through 28 as the best-fit ratio for large displacement piles is consistent at 0.832.

Figure 31 indicates a high correlation between the methods, similar to that presented in figure 26 with the CAPWAP/TEPWAP over Energy Approach ratio ranging from about 1.00 to 0.40. The best-fit line through zero produced a prediction ratio of 0.589 (CAPWAP/TEPWAP over Energy Approach).

(c) Small Displacement Piles

The relationships pertaining to small displacement piles for all soil types are shown in figures 32 through 34.

The correlation between the office method predictions and the actual static capacity for small displacement piles is shown in figure 32. The general trend indicates a relatively good agreement with a best-fit line forced through zero producing a ratio of 1.108, much closer to the desired ratio of 1.0 than that for the large displacement piles. The scatter is substantially smaller than that for the large displacement piles. The prediction ratios range from an over-prediction of about 2.5 to an under-prediction of 0.6 (actual over-prediction) with the majority of data falling between 1.25 and 0.60. While no clear trend can be seen on the basis of soil type, the capacity of all piles driven in rock seem to be over-predicted by the office method. This distinction seems to be more related to driving resistance as all of these cases present high driving resistance of over 10 BPI.

The Energy Approach predictions vs. the actual load test results for small displacement piles are presented in figure 33. The presented relationship indicates a small scatter with ratios ranging from about 1.67 to 0.60, with the majority of the data falling between 1.10 and 0.60. The best-fit line through zero yields a ratio of 0.856, which is only slightly higher than the ratios in figures 25 and 30. It should be noted that the scatter of both methods, the Energy Approach predictions in figure 33 and the office methods in figure 32, are very small when compared to that observed for the large displacement piles, as indicated by the coefficients of determination.

The information in figure 34 indicates that a distinct trend has developed between the two methods of analysis. The data shows very small scatter with ratios ranging between 1.00 and 0.60, with the best-fit ratio through zero equal to 0.764.

(d) Intermediate Conclusions

.

•

See tables 5 through 7 for statistical data. Different correlations have been investigated on the basis of pile type.

A relatively large scatter appears in the predictions of both dynamic methods, the office methods and the Energy Approach, for all cases (AAA). While the office methods under-predict on the average ($K_{sw} = 1.265$), the Energy Approach over-predicts ($K_{sp} = 0.839$). Both scatters are reflected through:

(1) Relatively low coefficient of determination for the best-fit line through zero ($r^2=0.692$ and 0.703 for the office methods and the Energy Approach, respectively).

(2) High intercept for unforced best-fit lines (y-intercept = 97.3 kips and 111.5 kips [432.8 kN and 496 kN] for the office and Energy methods, respectively).

- Much better correlations and a smaller scatter appear for both methods when predicting the capacity of small displacement piles compared to large displacement piles. For the office methods, the best-fit ratios and coefficients of determination are 1.372, $r^2=0.554$ and 1.108, $r^2=0.932$ for large and small displacement piles, respectively. For the Energy Approach, the best-fit ratios and coefficients of determination are 0.832, $r^2=0.534$ and 0.856, $r^2=0.916$ for large and small displacement piles, respectively.
 - As a result of the above, both methods seem to correlate very well to each other in all cases. A similar ratio is produced for the relationship between the predictions of CAPWAP/TEPWAP and the Energy Approach, regardless of pile type. This ratio varies between 0.641 for all piles to 0.589 for large displacement piles and 0.764 for small displacement piles. The coefficient of determination for the best-fit line through zero, however, is the highest for the small displacement piles ($r^2=0.937$), compared to 0.554 for the large displacement piles. This may imply that both methods encounter the same difficulties under the same conditions in spite of the fact that the Energy Approach does not consider any dynamic resistance while the office methods consider dynamic resistance through viscous damping.
 - No clear trends in the predictions appear on the basis of soil type at the tip, whereby predictions in all types of soil exist throughout, without any particular order. This conclusion is observational only and requires a quantitative evaluation that is presented in the following section.

8.3.3 Pile-Soil Type Correlations

The PD/LT pile-cases were subgrouped according to the different tip-soil types in an effort to investigate possible trends developing according to end-bearing soils. The correlations follow the sequence outlined in section 8.3.2 for three tip-soil conditions: sand and silt, clay and till, and rock.

(a) Sand and Silt

Figure 35 shows the correlation between the office method predictions and the actual static capacity for a PD/LT pile-case in sand and silt. The results remain consistent with figures 24 and 27 as they continue to under-predict. The best fit through zero shows an under-prediction ratio of 1.272. The scatter is, however, smaller for predictions in sand and silt, with the ratio ranging from 2.5 to 0.80 (load test over-prediction) and the coefficient of determination is 0.749.

The correlation between the Energy Approach predictions and the actual static capacity in sand and silt is shown in figure 36. The scatter is consistent with that of figure 25, with a best fit ratio very similar at 0.831. The ratio range is unchanged and it is difficult to see any different trends based on the sand and silt subgroup. It is noticeable, however, that all predictions pertaining to small displacement piles are contained within a narrow range approximately between 0.80 and 1.60.

The information in figure 27 indicates a good agreement between the office analysis predictions and the Energy Approach predictions for piles driven in sand and silt. The best-fit line forced through zero yields a ratio of 0.639, which is very consistent with the correlations of figures 26 and 31. The range of ratios remains between 1.00 and 0.40, with the majority of the points falling between 1.00 and 0.60. These results suggest that the sand and silt end-bearing soil has little effect on the overall trend of the prediction ratios.

(b) Clay and Till

The relationships in figure 38 between the office analysis predictions and the actual load test results for clay and till result in a similar best-fit line to the one obtained for the relationships in sand and silt. Considering the difference in the number of data points, however, it seems that the 51 cases of piles in clay and till are scattered much more relative to the 139 cases of piles in sand and silt. As a result, the coefficient of determination of the cases in sand and silt is much higher than that of clay and till (0.749 compared to 0.383 for the best-fit line through zero). The best-fit line yielded a ratio of 1.319, which far exceeds the best-fit ratio of figure 24 in which the small displacement piles were included. It is also interesting to note that the best-fit line resulted in a ratio of 1.057 with an intercept of 141 kips (627 kN). Time effects have not been considered in figure 38 and the data represents all states of EOD and BOR. Time effects will be addressed in section 8.3.4.

The relationships between the Energy Approach predictions and the actual static capacity is shown in figure 39. This information indicates a similar scatter (see figures 25, 30, 33, and 36) among the predictions, with the best-fit line remaining nearly unchanged at 0.872 (actual over-prediction). Based on figures 36 and 39, it appears that soil type, alone, has little effect on the overall performance of the Energy Approach.

Figure 40 demonstrates the consistency that has been evident in figures 26, 31, 34, and 37. The correlation between the office analysis predictions and the Energy Approach remain within a range of 1.00 and 0.40, with a distinct trend developing around the 0.80 line. The best-fit ratio (forced through zero) is 0.624 and a comparison with the sand and silt best-fit ratio (figure 37) shows a similar value. The coefficient of determination for the clay and till cases is substantially lower, however, and is approximately 0.539, compared to 0.802 for the best-fit line through zero for the pile cases in sand and silt. This shows that although, on the average, the ratio is unchanged, the agreement between the methods is more scattered for piles in clay.

(c) Rock

The correlation between the office analysis predictions and the actual static capacity for piles end-bearing on rock showed a considerably better prediction ratio with a considerable scatter. The best-fit line in figure 41 yielded an under-prediction ratio of 0.908 with all points falling almost exclusively in the range of 1.25 and 0.60 (actual over-prediction), yielding a poor coefficient of determination of 0.580. These results may be attributable to three reasons: (1) all the piles driven into rock are small displacement piles (except for one), (2) the driving resistance in the majority of cases (13 out of 14) ranges between 10 and 44 blows per inch (0.394 and 1.73 blows per mm), and (3) the presented subset contains only 14 pile-cases.

The information in figure 42 for the correlation between the Energy Approach predictions and the actual static capacity in rock produced very good results, showing excellent agreement that is consistently within a range between 1.00 and 0.60. The best-fit line shows a ratio of 0.823 with a coefficient of determination of 0.744, which is consistent with the other correlations between the Energy Approach and the actual static capacity previously mentioned.

Figure 43 indicates a very good correlation between the office analysis predictions and the Energy Approach predictions. The best-fit line forced through zero yields a ratio of 0.901 and all data points fall within ± 20 percent of the best-fit line.

(d) Intermediate Conclusions

See tables 5, 6, and 7 for statistical data.

Different correlations have been investigated on the basis of soil-type conditions at the tip.

- The office analysis relationships seem to be less scattered for the predictions of piles in sand compared to those in clay. Both best-fit line coefficients indicate a similar ratio for both soil types, 1.272 and 1.319 for sand and clay, respectively. Their coefficients of determination differ substantially however, $r^2=0.749$ and 0.383 for sand and clay, respectively. The "free" trend best-fit line for both cases show an intercept of 112 kips and 141 kips (498 kN and 627 kN) for sand and clay, respectively. The coefficients of determination of these lines are similar, however, to those for the lines forced through zero.
- The relationships of the Energy Approach analyses seem to be consistent for both clay and sand pile-cases. The best-fit ratios through zero and coefficients of determination are 0.831, $r^2 = 0.700$ and 0.872, $r^2 = 0.666$ for sand and clay, respectively.
- The relationships between the predicted capacity of piles in rock and the static capacity is different for both methods. The Energy Approach shows consistency in the best-fit coefficient and the coefficient of determination when compared to the sand and clay cases. The office analyses present a much better best-fit line with a relatively high scatter. The presented relationships for rock have been discussed separately and represent a separate case due to the small number of piles and the fact that all of them are small displacement piles driven in a high driving resistance.
- Less scatter appeared for the small displacement piles under all categories of soil types. This is in agreement with the previous section's conclusion that examined the pile-type case.
- A consistent ratio appears between the predictions of both methods and pile types. Higher scatter exists for the predictions in clay compared to sand ($r^2=0.539$ in clay vs. $r^2=0.802$ in sand).

8.3.4 Correlations of Pile and Soil Type for Different Driving Time

Further relationships were developed to examine any trends that may take place as a direct result of the time during driving for which the predictions were made. The subgrouping includes pile type (large displacement and small displacement) and time of driving (EOD = end of driving and BOR = beginning of restrike).

Two comments made in regard to these comparisons are:

• The EOD condition is of great importance as ideally we would like to accurately find the pile capacity at the end of driving state, which also enables us to control driving according to our real-time predictions.

The BOR records consist of different driving times after the initial EOD. These records were lumped together as one group. As such, the actual setup time and stage in which the driving took place was not considered. For the cases that were examined independently, consistent improvements were observed with each elapse of time since EOD.

(a) All Piles - EOD

Figure 44 presents the relationship between the office analysis predictions and the actual static capacity for all PD/LT piles in all types of soil at the end of driving (AEA). The results show a scatter with the prediction ratio ranging from 4.41 to 0.57, consistent with the best-fit lines produced in figure 24 for predictions at anytime during driving (AAA). The best-fit line forced through zero produced an under-prediction ratio of 1.248, or about +25 percent of the actual static capacity. The coefficient of determination improves somewhat from $r^2 = 0.692$ for all cases to 0.740 for the EOD conditions. Moreover, it seems that the under-prediction can be mostly attributed to the large displacement piles, whereas the predictions for the small displacement piles seem to concentrate within a zone of lower and more accurate load test over-prediction ratios. With regard to the best-fit line of the relationships in figure 44, it should be noted that the presented best-fit line is the one forced through zero (origin of axis). In most other cases, the slope of the forced best-fit line does not differ much from that of the unforced minimum square best-fit line. For the data presented in figure 44 the situation is different. The unforced best-fit line has a slope of 1.065 (see table 6) with a y-intercept of 152 kips (676 kN). This, again, implies some consistent under-prediction for the office methods in analyzing the EOD records.

The information in figure 45 indicates a prediction range from 1.67 to 0.60 for the correlation between the Energy Approach and the actual static capacity at EOD (AEA). The general scatter is substantially smaller than that of the office methods in figure 44, with a coefficient of determination of $r^2=0.804$. The best-fit prediction ratio increases substantially from the correlation for all cases (figure 25, ratio of 0.839) to 0.901 (prediction over actual ≈ 1.11).

It is important to note that for all the cases where substantial under-predictions took place in the office analyses, reasonable predictions were achieved by the Energy Approach. Observing figure 44, it can be seen that in many cases, the predictions exceed the line denoted by 1.67 (load test 67 percent higher than the prediction) up to a ratio of 4.4. All these cases are within the 1.67 line of the Energy Approach. Although not easily explained, in many cases in which improvement in prediction of the office method was observed with time, more accurate predictions were obtained by the Energy Approach at the EOD.

Figures 46 and 47 present the same data as that presented in figures 44 and 45, in the form of scattergrams of the actual over-prediction ratio versus the predicted capacity.

The aforementioned observations are enhanced by the data presentations of figures 46 and 47, emphasizing the relatively good predictions of the Energy Approach.

Figure 48 presents the correlation between the predictions of the office methods to the Energy Approach. As in previous similar correlations, the scatter between the methods is much smaller than that between the individual methods and the actual static capacity. It is interesting to note that the majority of the small displacement piles concentrate in a narrow band approximately between 0.7 and 1.0. This means that both methods produce very similar results for small displacement piles. From figures 44 and 45, it can also be concluded that both methods produce relatively accurate predictions for the small displacement piles.

(b) All Piles - BOR

The correlation between the actual static capacity and the office analysis predictions based on measurements at the beginning of restrike (ABA) is shown in figure 49. The range of predictions is between 2.5 and 0.80, with a best-fit prediction ratio of 1.284. The majority of the predictions reside within the 1.28 to 0.80 range with a general scatter higher ($r^2=0.614$) than that observed in figure 44, where predictions were based on end-of-driving measurements ($r^2=0.740$).

Figure 50 indicates that a much greater scatter exists for Energy Approach predictions at the beginning of restrike than for predictions made at the end of driving (see figure 45). The tendency is to over-predict more for restrikes with the prediction ratios ranging from 1.25 to 0.40. Consequently, the best-fit prediction ratio (0.786) is lower than that of figure 45 and the coefficient of determination is $r^2=0.578$, compared to 0.804 for EOD conditions.

The results presented in figure 50 are in sharp contrast to those shown in figure 45. While the Energy Approach provided much better predictions for the EOD condition compared to the office methods, it resulted in a larger scatter at the BOR state. In many cases, where improvement was observed with additional restrikes with time for the office methods, no such improvement (or, in many cases, worse predictions) were obtained by the Energy Approach.

Figure 51 exhibits a substantial scatter when compared to figure 48 for EOD predictions. The correlation between the two prediction methods is, however, considerably better than that observed in figures 49 and 50. The scatter exists mainly between 1.00 and 0.40 (CAPWAP/TEPWAP over Energy Approach) with a ratio of 0.593 for the slope of the best fit through zero (CAPWAP/TEPWAP over Energy Approach).

(c) Large Displacement Piles - EOD

Figure 52 shows the correlation of the office analysis predictions and the actual static capacity for large displacement at EOD (LEA). There is a significant scatter ($r^2=0.528$) ranging between 4.41 and 0.74, with most data between 2.50 and 1.00. The best-fit line

prediction ratio is 1.529. The data in figure 52 indicates the difficulties in analyzing records of large displacement piles and the shortcoming of the office methods for the EOD state.

The information in figure 53 indicates relatively good agreement of the Energy Approach and the actual static capacity for large displacement piles at EOD. Although the scatter ranges from 1.67 to 0.60 and the coefficient of determination, $r^2 = 0.603$, the majority of points lie within ± 20 percent of the actual static capacity, whereby the best-fit line yields a prediction ratio of 0.966. The relative accuracy of the Energy Approach for those cases is surprising and not yet well understood.

The relationship of the prediction methods large displacement piles at EOD is shown in figure 54. In general, the tendency appears to be within the 1.00 and 0.60 range, with a best-fit ratio of 0.612. This ratio meets the substantial under-prediction of the office methods and the relatively high accuracy of the Energy Approach.

(d) Large Displacement Piles - BOR

Figure 55 presents the correlation of the office analysis predictions and the actual static capacity for large displacement piles at BOR (LBA). The correlation demonstrates improved accuracy of the office analyses for BOR compared to the results obtained in figure 52 for EOD. In general, most of the data points fall between 2.0 to 0.80, with the best-fit line as a ratio of 1.307 and a coefficient of determination, $r^2 = 0.596$. Figure 55 shows improved predictions for large displacement piles relative to the EOD state, but poor predictions relative to those obtained for small displacement piles.

The information in figure 56 indicates a significant scatter for the correlation of the Energy Approach and the actual static capacity for large displacement piles at BOR. This is in contrast to the results obtained in figure 53 for the predictions of large displacement piles at EOD. The prediction ratios range from 1.53 to 0.40, with a best-fit ratio of 0.780 and a coefficient of determination, $r^2 = 0.550$.

The correlation shown in figure 57, between the prediction methods at BOR, remains consistent with previous findings. The ratios range from 1.00 to 0.40, with very few predictions outside of this cange.

(e) Small Displacement Piles - EOD

Figures 58, 59, and 60 present the relationships between the load test results and the office predictions, load test results and the Energy Approach predictions, and the relationships between the prediction methods for small displacement piles at EOD (SEA). Based on previous observations: (1) predictions for small displacement piles (see figures 32, 33, and 34) were much better than those for large displacement piles, and (2) predictions for end of driving (see figures 44, 45, and 46) were better than those at the beginning of restrike, especially for the Energy Approach. Therefore, the combined criteria resulted with very good relationships, as expected.

Figure 58 shows that the office method best-fit line is $K_{sw} = 1.113$ and $r^2 = 0.933$. The relationships have the second best coefficient of determination of all combination cases examined in table 5. The other similarly high correlations and accuracy were obtained for all small displacement piles (SAA) and their subgroup (SAS).

Figure 59 indicates a similar trend for the Energy Approach, yielding a best-fit line with a K_{sp} -ratio of 0.851 and $r^2 = 0.917$. These results are similar to those of all small displacement piles (SAA) and those in sand (SAS).

Figure 60 reflects the outcome of figures 58 and 59, with a best-fit correlation of $K_{rw} = 0.757$ and $r^2 = 0.939$.

(f) Small Displacement Piles - BOR

Figures 61 and 62 present the relationships between the predictions of the dynamic methods and the load test results for a small subgroup (12 cases in the figures and 15 cases in the statistical analysis) of small displacement piles at the beginning of restrike in all soils.

The obtained coefficients are $K_{sw} = 1.069$, $r^2 = 0.785$, and $K_{sp} = 0.962$, $r^2 = 0.737$, which indicate the following:

- The predictions for the BOR state are more scattered than for the EOD state, even for small displacement piles only.
- Out of the entire BOR group, the predictions for the small displacement piles are much better than those for the large displacement piles.

Figure 63 presents the relationships between the two prediction methods for 12 small displacement PD/LT piles in all types of soil at BOR (SBA), indicating a good correlation between them.

(g) Intermediate Conclusions

See tables 5, 6, and 7 for statistical data. Different correlations have been investigated based on the time of driving.

• Based on the data of figures 44 (46), 45 (47), 49, and 50, it is evident that both dynamic methods perform better for the end of driving (EOD) condition than for beginning of restrike (BOR). This is especially true for the Energy Approach method, which shows excellent predictions for all cases of EOD condition (AEA). The conclusions regarding the office method are different. On one hand, there is an improvement for EOD when compared to the overall cases (AAA); on the other hand, the BOR cases, as shown in figure 44, do not reflect correctly the accuracy of the method. As mentioned earlier, a closer look at the time of driving showed consistent improvement of the office methods with time. The data of figure 49 may, therefore, not correctly represent the accuracy of the method, which may improve when examined, for example, for only the last BOR of each case.

Based on the data of figures 52, 53, 55, and 56, it is clear that the capacity predictions for large displacement piles are problematic for both dynamic analyses, CAPWAP/TEPWAP, and the Energy Approach. CAPWAP/TEPWAP seem to produce, however, similar results at BOR than at EOD (see figures 44 and 49) subjected to the aforementioned comments. The Energy Approach, on the other hand, produces more accurate results at the end of driving than at the beginning of restrike (see figures 45 and 50). The relationships between the prediction methods, CAPWAP/TEPWAP, and the Energy Approach, show strong correlation between the methods regardless of the time of driving (see figures 43 and 48).

The above conclusion becomes more clear when comparing the performance of the large displacement piles and small displacement piles for the same driving time. For example, the office method, when comparing AEA (figure 44) to LEA (figure 52) and SEA (figure 58), clearly shows that the predictions for large displacement piles at EOD is very poor compared to that of the small displacement piles at EOD. The same conclusion holds true for beginning of restrike, demonstrating again the importance of the pile type. Similar conclusions are obtained by checking the Energy Approach method for AEA (figure 45), LEA (figure 53), and SEA (figure 59).

8.4 STATISTICAL ANALYSIS OF DATA SET PD/LT

٠

A statistical analysis of the correlations of data set PD/LT was performed in order to quantify the accuracy of both the office analysis and the Energy Approach predictions as well as the correlation between them. The statistical analysis was performed in three stages:

(I) Determination of the first-order best-fit lines (forced through zero and yintercept) by linear regression, in combination with the sample coefficient of determination (r^2) to measure the accuracy of the best fit (note that the coefficient of determination is a square of the coefficient of correlation (r)).

- (II) Examination of the fitness of the data to known probability distribution functions (PDF).
- (III) Determination of the mean and the standard deviation of the individual ratios (e.g., load test to Energy Approach) as a measure of variability.

8.4.1 Linear-Regression Analysis

The results of the linear regression analysis performed on selected subgroups of table 4 and the presented graphical relationships of section 8.3 are summarized in tables 5, 6, and 7.

The tables summarize the different subgroups for the ratios between: (1) the static resistance to the office method predictions (K_{sw}) in table 5, (2) the static resistance to the Energy Approach predictions (K_{sp}) in table 6, and (3) the relationship between the predictions of the office methods and those of the Energy Approach in table 7.

The first two columns of each of the tables list the pile-case subgroup and the total number of cases considered in that group. The number of cases shown in the tables may be equal or greater than the numbers shown in the figures for the same correlations. This occurs when some of the data points exceed the dimensions of the plots. Linear regression was preformed for each group to determine: (1) the best-fit line ratio, (2) the best-fit line ratio forced through zero, and (3) the coefficient of determination for each analysis. The results are listed in columns 3, 4, and 5 for the best-fit line forced through zero, in each table. For example, the best-fit line forced through zero for the K_{sp} coefficients calculated for the subgroup AEA (all piles, at EOD, for all soils) was found to have a slope of 0.900 with a coefficient of determination, $r^2 = 0.804$. The sample coefficient of determination (r^2) for each subgroup was determined to measure the representativeness (accuracy) of the best fit and the best fit through zero.

The coefficient of determination (r^2) represents the proportion of the sum of squares of deviations of the y-values about their mean, and it is a measure of the contribution of "x" in prediction "y". By definition, a scatter at higher x-values will influence this coefficient more than a scatter close to the origin of axes. The coefficient of determination varies between 0 and 1; the first indicating no correlation or contribution and the last $(r^2=1)$ is a perfect match where all the points fall on the best-fit, least-squares line. For example, $r^2=0.6$ means that 60 percent of the sum of squares of deviations of the observed y-values about their mean is attributed to the linear relations between y and x. (actual vs. predicted). In other words, 60 percent of the variability in y is explained by the regression equation. According to Ryan (1989), a meaningful correlation is obtained with $r^2 \ge 0.80$, which coincides with $p \le 0.0011$; p is the probability of obtaining an F-value as or larger than the calculated value. This value of $r^2=0.8$ may be rigorous relative to
correlations in geotechnical engineering. The results, therefore, may be reviewed in the following ranges (Veneziano, 1993):

$r^2 \ge 0.80$	good correlation
$0.60 \le r^2 < 0.80$	moderate correlation
$r^2 < 0.60$	poor correlation.

Table 5 presents the results of the K_{sw} analysis and the best correlation of all subgroups was found to be for all small displacement piles in all soils (SAA) and in sand and silt (SAS). Reasonable correlation was found for all piles based on the end of driving records, especially for the small displacement piles. Poor correlations were found for all piles at BOR in all soils (ABA) and for all large displacement piles (LAA), both at EOD (LEA) and at BOR (LBA).

The coefficients of table 6 indicate that the Energy Approach presents slightly better correlation overall for all cases (AAA) than that of the office methods, where both methods show moderate correlation according to the above coefficient of determination standards. The Energy Approach method shows very high accuracy for small displacement piles in all soils at all times (SAA), mostly due to its excellent performance in sand and silt (SAS). The Energy Approach prediction shows excellent correlation for all piles at EOD in all soils (AEA), producing a best fit through zero sample coefficient of distribution of 0.804, mostly again due to the high accuracy for the small displacement piles (SEA). Low accuracy was also determined for all piles at BOR in all soils (ABA), similar to that of the office methods.

Table 7 enables the examination of under what conditions both methods predict similarly or differently, indicating that, in general, the correlation between the methods is stronger than the correlation between the individual methods and the actual capacity with especially strong correlations in the cases where both methods predict well, namely small displacement piles and end of driving.

8.4.2 Actual Distributions of the K Coefficients and their Probabilistic Models

The K coefficients are defined as follows:

$$K_{sw} = \frac{Actual static capacity}{CAPWAP/TEPWAP prediction}$$
(38)

$$K_{sp} = \frac{Actual static capacity}{Energy Approach prediction}$$
(39)

$$K_{ev} = \frac{CAPWAP/TEPWAP prediction}{Energy Approach prediction}$$
(40)

These ratios are equivalent to the ratios marked and denoted by the straight lines on the scatter plots of sections 8.3.2, 8.3.3, and 8.3.4.

The distributions of the individual K coefficients for all PD/LT pile-cases are presented in figures 64, 66, and 68 in the form of histograms. The cumulative frequency distribution of K_{sw} and K_{sp} are presented in figures 65 and 67, respectively. The histograms were plotted for K coefficients ranging from 0.0 to >3.0 in 0.1 intervals and include all the available information. The left y-axis shows the total number of K coefficient occurrences, whereas the right y-axis shows the frequency (normalized number of occurrences).

The common parameters most often used to evaluate prediction methods are the mean and standard deviation of the normal distribution. The normal distribution best represents occurrences ranging from $-\infty$ to $+\infty$ with the highest probability at the mean. The ratio between the actual capacity to the predicted one (or its inverse) is limited between 0 to $+\infty$ and, hence, its distribution is not symmetrical. Even though, in many cases where the data is "normally" distributed, the normal distribution will represent it in a reasonable fashion (e.g., see figure 68). In many other cases, the normal distribution is incapable of correctly reflecting the accuracy (represented by the mean) and the precision (represented by the standard deviation) of the predicting method. A better probability distribution function for cases ranging from 0 to $+\infty$ is the log-normal distribution. A simple transformation can be performed from the mean and standard deviation of the normal distribution to the log-normal distribution parameters (see, for example, Benjamin and Cornell, 1970), which allows plotting of the log-normal distribution. Both distributions, normal and the corresponding (transformed) log-normal distributions, were plotted for the ratios between actual capacity to the predictions of the office methods (K_{sw} , figure 64) and the actual capacity to the Energy Approach predictions (K_{so} , figure 65). In any case, the actual data must be reviewed as scatter graphs (section 8.3.2, 8.3.3, and 8.3.4) and histograms before the establishment of any conclusions.

The information presented in figure 64 for the K_{sw} coefficients (actual capacity over CAPWAP/TEPWAP predictions) indicates a concentration of cases (about 50 percent of all cases) between 0.9 and 1.3, with a significant scatter of the other 50 percent of the cases across a wide range of K values from 0.57 to 4.41. A normal distribution curve was added to the actual data based on the analysis results presented in table 8. The actual data seems to differ from the normal distribution and explains the relatively large standard deviations of the K_{sw} . The "transformed" log-normal distribution seems to better represent the actual data, but yet, falls short of representing it accurately. An

attempt to improve the log-normal distribution representation of the actual data was carried out by decreasing the standard deviation parameter ($\ln \sigma_x$, note: not that of the standard deviation). The results are shown in the form of a log-normal distribution and plotted using dashed lines that seem to represent the peak and over-prediction side better, but do not seem to represent the under-prediction side as well. Figure 65 presents the cumulative frequency distribution of the K_{sw} ratio. Due to the large range of values, a gradual increase in the cumulative frequency distribution takes place for values of K_{sw} greater than 1.3.

Figure 66 shows the distribution of the K_{sp} values for all PD/LT pile-cases and the data fits reasonably well with the normal distribution description. The "transformed" lognormal distribution seems to fit the data even better, allowing good representation of the data skewness. About 75 percent of the cases fall in the range between 0.6 and 1.2, with the mean at 0.925. An attempt to improve the log-normal distribution was carried out by decreasing the standard deviation $(\ln \sigma_x)$. The results, again, are better only for part of the data, showing better agreement with the peak and the underestimation, and worse representation for the overestimated capacities. Figure 67 presents the cumulative frequency distribution of the K_{sp} ratio. A moderate increase exists for about 50 cases between 0.4 and 0.7, followed by a sharp increase of about 150 cases between 0.7 and 1.2. The distribution ends with a moderate slope of about 10 cases, up to about 1.7.

The distribution of the K_{ew} coefficients is presented in figure 68. The results of this distribution indicate excellent correlation between the office analysis predictions and the Energy Approach predictions (CAPWAP/TEPWAP over Energy Approach), represented well by the normal distribution.

8.4.3 Mean and Standard Deviation Analysis

Table 8 presents the statistical analysis for all PD/LT correlations listed in table 4 (see chapter 6 and tables 20 through 23 in appendix A for details). The first column of table 8 lists the pile-case group according to the abbreviation system shown in table 4. The table reports the normal distribution mean and standard deviation for each subgroup in a similar sequence mentioned in section 8.3.2. The subcolumns for each K coefficient list the total number of pile-cases analyzed, the mean value determined, and the corresponding standard deviation. It should be emphasized that even for cases in which better representation is given through the log-normal distribution, the mean and the standard deviation remain a powerful tool for the evaluation of the accuracy, through the mean, and for the precision, through the standard deviation.

8.5 INTERPRETATION OF THE CONTROLLING PARAMETERS

8.5.1 Overview

Section 4.4.4 outlined the expected performance of the dynamic analyses based on the hypothesis that the majority of the energy is lost through soil inertia. This hypothesis was partially confirmed by the results presented in the previous sections. A closer examination of the controlling parameters and their influence on the accuracy of the dynamic analyses follows.

8.5.2 Dynamic Predictions - Pile Area Ratio Graphical Correlations

To investigate a possible relationship between the office analysis, the Energy Approach predictions, and pile geometry, K_{sw} and K_{sp} values were correlated with the pile area ratio (see sections 4.4.2 and 4.4.3).

Figures 69 and 70 present the correlation between the K_{sw} -values and the pile area ratio (A_R) . The data are presented using two scales (linear and logarithmic) to allow the assessment of the many cases for which A_R varies between approximately 90 to 300, which create a "spot" when presented in a linear scale. For pile area ratios less than 350, a significant scatter can be observed with K_{sw} -values exceeding 2.0. In general, K_{sw} -values closer to unity appear as A_R increases. Some scatter appears, however, at very large A_R ratios that may indicate the influence of additional parameters on the K_{sw} -values (e.g., driving resistance).

Figures 71 and 72 present the correlation between K_{sp} and the pile area ratio. Significantly smaller scatter appears in the K_{sp} -values compared to that of the K_{sw} -values. The general trend is similar to that of figure 69 — most scatter appears within a zone in which the pile area ratio is smaller than 350.

The pile area ratio seems to enable the quantification of the definition of large displacement and small displacement piles. The information from figures 69 through 72 suggest that considerable consistency is developed for pile area ratios > 350. From these correlations, it was concluded that large displacement piles can be defined as those with pile area ratios < 350 and small displacement piles defined as those with pile area ratios > 350. The following section examines the relationship between the prediction of the dynamic analyses and the driving resistance, as the complementary factor to the pile type in controlling the soil's inertia (see section 4.4 for background).

8.5.3 Dynamic Predictions - Driving Resistance Graphical Correlations

Figure 73 presents the ratio between the load test results over the office method predictions (K_{sw}) vs. blow count at the time of measurement for all PD/LT pile-cases

Pile-		Ksw			Ksp			Kew	
Case	No.	Mean	Standard	No.	Mean	Standard	No.	Mean	Standard
Group			Deviation			Deviation			Deviation
AAA	206	1.367	0.5334	208	0.925	0.2932	206	0.712	0.1815
AAS	141	1.385	0.4758	141	0.942	0.3127	141	0.702	0.1771
AAC	51	1.443	0.6760	53	0.906	0.2689	51	0.681	0.1736
AAR	14	0.906	0.1922	14	0.827	0.1402	14	0.925	0.1073
AEA	97	1.478	0.6167	98	1.023	0.3073	97	0.743	0.1844
AES	58	1.534	0.5310	58	1.089	0.3244	58	0.742	0.1766
AEC	28	1.598	0.7576	29	0.971	0.2742	28	0.672	0.1729
AER	11	0.877	0.1957	11	0.810	0.1510	11	0.936	0.1132
ABA	109	1.268	0.4257	110	0.838	0.2509	109	0.684	0.1748
ABS	83	1.282	0.4049	83	0.840	0.2605	83	0.674	0.1731
ABC	23	1.254	0.5160	24	0.827	0.2354	23	0.692	0.1777
ABR	e	1.010	0.1667	3	0.888	0.0816	C.	0.887	0.0890

Table 8. Statistical analysis of K coefficients for all PD/LT pile-cases.

 first letter denotes pile type: A=all piles, L=large displacement, S=small displacement. : XXX Pile-case legend:

second letter denotes time of measurements: A=anytime, E=EOD, B=BOR.
 third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

Pile-		Ksw			Ksp			Kew	
Case Group	No.	Mean	Standard Deviation	No.	Mean	Standard Deviation	No.	Mean	Standard Deviation
LAA	162	1.399	0.5250	164	0.925	0.3056	162	0.689	0.1756
LAS	118	1.366	0.4448	118	0.927	0.3246	118	0.693.	0.1714
LAC	43	1.501	0.6969	45	0.925	0.2559	43	0.670	0.1842
LAR	1	•	•	1	•	•	+	•	1
LEA	68	1.574	0.6177	69	1.069	0.3192	68	0.718	0.1830
LES	44	1.535	0.5142	44	1.108	0.3427	44	0.750	0.1773
LEC	24	1.646	0.7802	25	0.975	0.2577	24	0.661	0.1827
LER		I	1	1	•	1	•	•	1
LBA	94	1.272	0.4041	95	0.827	0.2554	94	0.668	0.1678
LBS	74	1.265	0.3657	74	0.819	0.2613	74	0.659	0.1595
LBC	19	1.317	0.5399	20	0.854	0.2433	19	0.683	0.1904
LBR	1	•		1	•		1	1	,

Table 8. Statistical analysis of K coefficients for all PD/LT pile-cases (continued).

: XXX Pile-case legend:

first letter denotes pile type: A=all piles, L=large displacement, S=small displacement.
second letter denotes time of measurements: A=anytime, E=EOD, B=BOR.
third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

Pile-		Ksw			Ksp			Kew	
Case Group	No.	Mean	Standard Deviation	No.	Mean	Standard Deviation	No.	Mean	Standard Deviation
SAA	44	1.250	0.5542	14	0.926	0.2445	44	0.796	0.1798
SAS	23	1.486	0.6127	23	1.021	0.2313	23	0.746	0.2028
SAC	8	1.132	0.4679	8	0.820	0.3172	8	0.738	0.0854
SAR	13	0.906	0.2001	13	0.822	0.1448	13	0.921	0.1104
SEA	29	1.252	0.5616	29	0.935	0.2616	29	0.802	0.1772

Table 8. Statistical analysis of K coefficients for all PD/LT pile-cases (continued).

 first letter denotes pile type: A=all piles, L=large displacement, S=small displacement. :XXX Pile-case legend:

 second letter denotes time of measurements: A=anytime, E=EOD, B=BOR. third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

0.1784

0.715

14

0.2606

1.029

14

0.6013

1.530

7

SES

0.0797

0.738

4

0.4120

0.949

4

0.6067

1.309

4

SEC

0.1132

0.936

7

0.1510

0.810

7

0.1957

0.807

=

SER

0.1904

0.784

15

0.2150

0.908

15

0.5591

1.247

15

SBA

0.2392

0.793

თ

0.1912

1.009

ŋ

0.6604

1.418

ດ

SBS

0.1033

0.737

4

0.1436

0.691

4

0.2442

0.955

4

SBC

0.0461

0.839

2

0.1154

0.888

3

0.1960

1.064

2

SBR

(AAA). As indicated in chapter 6, the blow count per inch was often calculated based on records of blows per foot.

There is considerable scatter for all driving resistances (especially at the two extremes, namely, very low blow count (less than 10 BPI) and very high blows) at refusal (no set). It also can be noted that the predictions for the small displacement piles present, on average, much better performance than that of the large displacement piles, including the area of low driving resistance.

Figure 74 presents the ratio between the load test results to the Energy Approach predictions (K_{sp}) , in the same format as that of figure 73. Considerably less scatter appears in the figure compared to that of figure 73. A large range of K_{sp} (from over-prediction of $K_{sp} \approx 0.4$ to under-prediction of about $K_{sp} \approx 1.7$) appears at the range of small resistance to driving of about 0 to 10 BPI. A few additional observations can be made in relationship to figure 74:

- In the majority of the cases, the Energy Approach over-predicts, however, there is improvement with the increase of driving resistance.
- Most of the significant under-predictions exist in the low-resistance zone.
- Very good performance appears at very high driving resistance, when actually no displacement takes place under each blow.

8.5.4 Dynamic Predictions - Driving Resistance and Time of Driving Graphical Correlations

Additional subdivision of the dynamic analyses prediction ratios vs. driving resistance was conducted based on the time of driving, namely, end of driving (EOD) and beginning of restrike (BOR).

(a) All Piles at EOD

Figure 75 presents the correlation between driving resistance and K_{sw} coefficient for all piles at EOD (AEA) and the results indicate a scatter similar to the results shown in figure 73. A major scatter remains at low driving resistances when the full resistance of the soil is mobilized.

The correlation between the K_{sp} coefficients for all piles in all soil types at EOD (AEA) and driving resistance is presented in figure 76, and the results are similar to those of figure 60. This is consistent with the findings of sections 8.3 and 8.4 for piles at the end of driving, with the emphasis on under-prediction cases at the low-resistance zone. Most of the under-prediction cases observed in figure 74 for the low blow count seem to be a result of the EOD cases as shown in figure 76. These cases are confined, however,

mostly within a zone of blow count between 0 to 6 BPI, which may, as a result, be defined as "easy driving."

(b) All Piles at BOR

Figure 77 represents the relationship of driving resistance and K_{sw} -values for all piles at BOR (ABA). A scatter among the predictions for driving resistances ranging from 0 to 25 blows per inch (0.98 blows per mm) is observed. The scatter appears, however, to be substantially smaller than that observed for EOD in figure 75, with lower underpredictions.

In both cases (EOD and BOR), the office analysis predictions produced a scatter. At the BOR, however, a large concentration of cases appear around the K_{sw} =1 and the K_{sw} -values are lower cases than those observed in figure 75.

Figure 78 presents the relationship of driving resistance and K_{sp} -values for all piles at BOR (ABA). A major scatter, mostly to the over-prediction side, appears in figure 78. When comparing figures 76 and 78 to figure 74, it appears that:

- The Energy Approach tends to over-predict at the low driving resistance for BOR cases and under-predict at the EOD cases. It should be emphasized that both under-prediction and over-prediction at the lowresistance zone appears in both EOD and BOR. The extreme overpredictions, however, exist only at the BOR and the extreme underpredictions exist only at the EOD.
- On the average, the performance of the Energy Approach at EOD is better than that at BOR, especially for piles with driving resistances greater than 6 BPI (0.24 blows per mm).

8.5.5 Dynamic Predictions - Driving Resistance and Pile-Type Graphical Correlations

Section 8.5.2 examined the relationship between the dynamic predictions and the pile area ratio and concluded that small displacement piles can be referred to as piles with $A_R > 350$. Section 8.5.3 examined the relationship between the dynamic predictions and the driving resistance and determined the effect of the driving resistance on the accuracy of the predictions of both dynamic analyses.

The subdivision of the dynamic predictions vs. driving resistance to small and large displacement, based on the pile area ratio definition, is presented in this section.

(a) Small Displacement Piles

The relationship between K_{sw} and the driving resistance for small displacement piles with $A_R > 350$ is presented in figure 79. The office analysis, in general, appears to perform better for small displacement piles than for the large displacement piles. When

comparing the data in figure 79 to that of 69 and 70, a relatively good agreement exists between the predicted and observed capacity with the exception of very low and very high driving resistances. This agreement suggests that the relatively high underpredictions of the office methods for the small displacement piles are associated with either very low driving resistances (which result in high inertia of the soil mass) of less than 6 BPI (0.24 blows per mm), or a very high driving resistance (which results in a lack of full-capacity mobilization). In relationship to figure 79 and the following figures, it should be clear that the criteria for distinguishing between small and large displacement piles is the area ratio of A_R =350. As such, the open symbols in those figures refer to piles that, by observation, would be considered as large displacement piles (e.g., square concrete pile), however, their area ratio of A_R >350 would categorize them as small displacement piles as explained in section 4.4 and concluded in section 8.5.2.

Figure 80 presents the relationship between driving resistance and K_{sp} for piles with $A_R > 350$. Excellent agreement exists between the predictions of the Energy Approach and the observed static capacity for all driving resistances. Two major conclusions can be made regarding the data in figure 80:

- The influence of the pile type on the performance of the dynamic methods is evident. The mean K_{sp} for figure 80 is 0.938 with a standard deviation of 0.239, which indicates an excellent performance.
- The highest scatter and over- and under-predictions occur at the lower resistance zone of less than 6 BPI (0.24 blows per mm).

In reference to figures 79 and 80, it should be noted that with the new definition of small/large displacement piles based on the pile area ratio of 350, the piles that were previously considered as large displacement (i.e., open symbols) fit well into the general trend of the small displacement piles.

(b) Large Displacement Piles

Figures 81 and 82 present the relationships between K_{sw} and K_{sp} to the driving resistance for large displacement (pile area ratio <350), respectively. The data in figure 81 indicates that substantial scatter appears in the predictions of the office method. Overprediction takes place especially for the low blow count (figure 79) and under-predictions appear to exist for all driving resistances. When compared to predictions of the small displacement piles (figure 79), the existing scatter and under-prediction seem to be much more significant.

Figure 82 indicates that much larger scatter and inaccuracy in prediction exists for the large displacement piles when compared to the small displacement piles (figure 80). The inaccuracy is, however, highly related to the driving resistance with a decrease in scatter (mainly due to the decrease in the under-prediction) and an increase in accuracy with

the increase in the driving resistance. The predictions above approximately 10 BPI (0.39 blows per mm) seem to be much better than those below that resistance.

8.5.6 The Effect of the Combined Major Controlling Parameters on the Accuracy of the Dynamic Predictions

(a) Breakdown of Combinations

The previous correlations that were presented throughout chapter 8 indicated the following factors as the major controlling parameters:

- Pile type, according to the pile area ratio, distinguishing between large displacement piles with $A_R < 350$ and small displacement piles with $A_R > 350$.
- Time of driving, distinguishing between end of driving (EOD) records to analyses on records obtained at some time later at the beginning of restrike (BOR).
- Driving resistance, distinguishing between easy driving of less than 6 BPI (0.24 blows per mm) to intermediate driving resistance between 6 and 12 BPI (0.24 and 0.47 blows per mm) with high driving resistance above 12 BPI (0.47 blows per mm).
- Type of soil, distinguishing between predictions of piles predominately in clay vs. those driven in granular materials.

Different combinations of these factors are presented in the following sections with a summary of their statistical data presented in table 9.

(b) Combinations of Pile Type and Driving Resistance

The previously mentioned criteria for pile type and driving resistance assisted in establishing the following combinations:

- (1) Small displacement piles with easy driving; $A_R > 350$ and blow count < 6 BPI (0.24 blows per mm) shown in figures 83 and 84 for K_{sw} and K_{sp} , respectively.
- (2) Small displacement piles with hard driving; $A_R > 350$ and blow count > 6 BPI (0.24 blows per mm) shown in figures 85 and 86 for K_{sw} and K_{sp} , respectively.
- (3) Large displacement piles with easy driving; $A_R < 350$ and blow count < 6 BPI (0.24 blows per mm) shown in figures 87 and 88 for K_{sw} and K_{sp} , respectively.

Pile	Driving	Time	-	Ksw			Ksp	
Area	Resistance	of	No.	Mean	Standard	No.	Mean	Standard
Ratio		Driving			Deviation			Deviation
<350	ail piles	anytime	144	1.427	0.543	146	0.920	0.317
<350	0-6 BPI	anytime	64	1.374	0.512	64	0.962	0.347
<350	> 6 BPI	anytime	80	1.469	0.567	82	0.887	0.288
<350	all piles	EOD		1.643	0.654	57	1.068	0.345
<350	0-6 BPI	EOD	36	1.545	0.569	36	1.102	0.349
<350	> 6 BPI	EOD	20	1.820	0.769	_21	1.026	0.340
<350	all piles	BOR	88	1.290	0.407	89	0.825	0.257
<350	0-6 BPI	BOR	28	1.155	0.319	28	0.783	0.254
<350	> 6 BPI	BOR	60	1.352	0.430	61	0.844	0.258
>350	all piles	anytime	57	1.247	0.502	57	0.938	0.239
>350	0-6 BPI	anytime	16	1.542	0.595	16	1.031	0.259
>350	> 6 BPI	anytime	41	1.133	0.414	41	0.902	0.224
>350	all piles	EOD	39	1.151	0.408	39	0.902	0.240
>350	0-6 BPI	EOD	12	1.476	0.492	12	1.021	0.291
>350	> 6 BPI	EOD	27	1.161	0.473	27	0.928	0.214
>350	all piles	BOR	18	1.225	0.530	18	0.897	0.240
>350	0-6 BPI	BOR	4	1.740	0,901	4	1.062	0.154
>350	> 6 BPI	BOR	14	1.078	0.274	14	0.850	0.243

Table 9. Statistical analysis of the area ratio, resistance, and time of driving combination.

1 BPI = 0.039 blows per mm

Pile-case legend:

<350 - pile area ratio definition of large displacement piles.

>350 - pile area ratio definition of small displacement piles.

0-6 BPI - low driving resistance, resulting in full mobilization of the soil resistance.

> 6 BPI - intermediate (6 to 12 BPI) and high driving resistance of more than 12 BPI. (4) Large displacement piles with hard driving; $A_R < 350$ and blow count > 6 BPI (0.24 blows per mm) shown in figures 89 and 90 for K_{sw} and K_{sp} , respectively.

The above four combinations clearly suggest (with the limitation of the small number of pile-cases for some combinations):

- Small displacement piles with high driving resistance present very good prediction conditions for the dynamic methods.
- These conditions are followed by the predictions for small displacement piles with easy driving resistance (especially for the Energy Approach).
- Less favorable conditions result from the predictions of large displacement piles with high resistance (especially for the office methods).
- The worst conditions are presented for the large displacement piles with easy driving where both dynamic methods predict poorly with a high scatter.

(c) Combinations of Pile Type, Driving Resistance, and Time of Driving

The above combinations were further investigated, incorporating the time of driving into the above criteria as follows:

- (I) Small displacement piles with easy driving at the end of driving; $A_R > 350$ and blow count < 6 BPI (0.24 blows per mm) shown in figures 91 and 92 for K_{sw} and K_{sp} , respectively.
- (II) Small displacement piles with hard driving at the end of driving; $A_R > 350$ and blow count > 6 BPI (0.24 blows per mm) shown in figures 93 and 94 for K_{sw} and K_{so} , respectively.
- (III) Large displacement piles with easy driving at the end of driving; A_R < 350 and blow count < 6 BPI (0.24 blows per mm) shown in figures 95 and 96 for K_{sw} and K_{sp}, respectively.
- (IV) Large displacement piles with hard driving at the end of driving; $A_R < 350$ and blow count > 6 BPI (0.24 blows per mm) shown in figures 97 and 98 for K_{sw} and K_{sp} , respectively.
- (V) Small displacement piles at the beginning of restrike for all driving resistances; $A_R > 350$ shown in figures 99 and 100 for K_{sw} and K_{sp} , respectively.

(VI) Large displacement piles at the beginning of restrike for all driving resistances; $A_R < 350$ shown in figures 101 and 102 for K_{sw} and K_{sp} , respectively.

These combinations again suggest the following trends (with the limitation of the small number of pile-cases for some combinations):

- Small displacement piles at the end of driving (EOD) with high driving resistance present very good prediction conditions for the office methods and even better conditions for the predictions of the Energy Approach.
- The office methods present a considerable scatter for large displacement piles at the end of driving, especially in the cases with high driving resistance. The Energy Approach presents very good predictions under these conditions as can be observed for the 20 pile-cases shown in figure 98.
- The prediction conditions for small displacement piles at EOD with low driving resistance presented difficulties for the office methods, while good predictions were obtained by the Energy Approach. Again, this conclusion may be affected by the limited number of pile-cases for this combination.
- The least favorable prediction conditions for the end of driving state for the office methods occur for large displacement piles with high driving resistance.
- The predictions conditions for small displacement piles at BOR with all driving resistances yield good results for the Energy Approach and a moderate scatter for the office predictions. This conclusion should once again be subjected to the limited number of pile-cases for this combination.
- Small variation was observed between easy and hard driving resistances that may have been the result of the scatter produced in both methods.



Figure 21. Tip soil conditions vs. calculated case damping coefficient (J_c) based on static load test results for 208 PD/LT pile-cases.



Figure 22. Side soil conditions vs. Smith side damping coefficient based on CAPWAP/TEPWAP results.



Figure 23. Tip soil conditions vs. Smith tip damping coefficient based on CAPWAP/TEPWAP results.



Figure 24. Static load test results vs. CAPWAP or TEPWAP predictions for 204 PD/LT pile-cases in all types of soil (AAA).



Figure 25. Static load test results vs. Energy Approach predictions for 202 PD/LT pile-cases in all types of soil (AAA).



Figure 26. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 201 PD/LT pile-cases in all types of soil (AAA).



Figure 27. K_{sw} vs. CAPWAP/TEPWAP predictions for 206 PD/LT pile-cases in all types of soil (AAA).



Figure 28. K_{sp} vs. Energy Approach predictions for 208 PD/LT pile-cases in all types of soil (AAA).



Figure 29. Static load test results vs. CAPWAP or TEPWAP predictions for 162 large displacement PD/LT pile-cases in all types of soil (LAA).



Figure 30. Static load test results vs. Energy Approach predictions for 163 large displacement PD/LT pile-cases in all types of soil (LAA).



Figure 31. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 161 large displacement PD/LT pile-cases in all types of soil (LAA).



Figure 32. Static load test results vs. CAPWAP or TEPWAP predictions for 42 small displacement PD/LT pile-cases in all types of soil (SAA).



Figure 33. Static load test results vs. Energy Approach predictions for 40 small displacement PD/LT pile-cases in all types of soil (SAA).



1 kip = 4.448 kN

Figure 34. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 38 small displacement PD/LT pile-cases in all types of soil (SAA).



Figure 35. Static load test results vs. CAPWAP or TEPWAP predictions for 139 PD/LT pile-cases in sand and silt (AAS).



Figure 36. Static load test results vs. Energy Approach predictions for 136 PD/LT pile-cases in sand and silt (AAS).



Figure 37. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 136 PD/LT pile-cases in sand and silt (AAS).



Figure 38. Static load test results vs. CAPWAP or TEPWAP predictions for 51 PD/LT pile-cases in clay and till (AAC).



Figure 39. Static load test results vs. Energy Approach predictions for 53 PD/LT pile-cases in clay and till (AAC).



Figure 40. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 51 PD/LT pile-cases in clay and till (AAC).



Figure 41. Static load test results vs. CAPWAP or TEPWAP predictions for 14 PD/LT pile-cases in rock (AAR).



Figure 42. Static load test results vs. Energy Approach predictions for 14 PD/LT pile-cases in rock (AAR).


Figure 43. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 14 PD/LT pile-cases in rock (AAR).



Figure 44. Static load test results vs. CAPWAP or TEPWAP predictions for 96 PD/LT pile-cases in all types of soil at EOD (AEA).



Figure 45. Static load test results vs. Energy Approach predictions for 94 PD/LT pile-cases in all types of soil at EOD (AEA).



Figure 46. K_{sw} vs. CAPWAP/TEPWAP predictions for 97 PD/LT pile-cases at EOD in all types of soil (AEA).



Figure 47. K_{sp} vs. Energy Approach predictions for 98 PD/LT pile-cases at EOD in all types of soil (AEA).



Figure 48. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 94 PD/LT pile-cases in all types of soil at EOD (AEA).



Figure 49. Static load test results vs. CAPWAP or TEPWAP predictions for 108 PD/LT pile-cases in all types of soil at BOR (ABA).



Figure 50. Static load test results vs. Energy Approach predictions for 108 PD/LT pile-cases in all types of soil at BOR (ABA).



Figure 51. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 108 PD/LT pile-cases in all types of soil at BOR (ABA).



Figure 52. Static load test results vs. CAPWAP or TEPWAP predictions for 68 large displacement PD/LT pile-cases in all types of soil at EOD (LEA).



Figure 53. Static load test results vs. Energy Approach predictions for 69 large displacement PD/LT pile-cases in all types of soil at EOD (LEA).



Figure 54. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 68 large displacement PD/LT pile-cases in all types of soil at EOD (LEA).



Figure 55. Static load test results vs. CAPWAP or TEPWAP predictions for 94 large displacement PD/LT pile-cases in all types of soil at BOR (LBA).



Figure 56. Static load test results vs. Energy Approach predictions for 94 large displacement PD/LT pile-cases in all types of soil at BOR (LBA).



Figure 57. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 93 large displacement PD/LT pile-cases in all types of soil at BOR (LBA).



Figure 58. Static load test results vs. CAPWAP or TEPWAP predictions for 22 small displacement PD/LT pile-cases in all types of soil at EOD (SEA).



Figure 59. Static load test results vs. Energy Approach predictions for 20 small displacement PD/LT pile-cases in all types of soil at EOD (SEA).



Figure 60. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 20 small displacement PD/LT pile-cases in all types of soil at EOD (SEA).



Figure 61. Static load test results vs. CAPWAP or TEPWAP predictions for 12 small displacement PD/LT pile-cases in all types of soil at BOR (SBA).



Figure 62. Static load test results vs. Energy Approach predictions for 12 small displacement PD/LT pile-cases in all types of soil at BOR (SBA).



Figure 63. CAPWAP or TEPWAP predictions vs. Energy Approach predictions for 12 small displacement PD/LT pile-cases in all types of soil at BOR (SBA).



Ksw histogram and frequency distribution for 206 PD/LT pile-cases (AAA)

Figure 64. Histogram and frequency distributions of K_{sw} for 206 PD/LT pile-cases in all types of soil (AAA).



Figure 65. Cumulative frequency distribution of K_{rw} for 206 PD/LT pile-cases in all types of soil (AAA).



Ksp histogram and frequency distribution for 208 PD/LT pile-cases (AAA)

Figure 66. Histogram and frequency distributions of K_{sp} for 208 PD/LT pile-cases in all types of soil (AAA).



Figure 67. Cumulative frequency distribution of K_{sp} for 208 PD/LT pile-cases in all types of soil (AAA).



Kew histogram and frequency distribution for 206 PD/LT pile-cases (AAA)

Figure 68. Histogram and frequency distribution of K_{ew} for 206 PD/LT pile-cases in all types of soil (AAA).



Figure 69. K_{sw} vs. the pile area ratio (A_R) for 201 PD/LT pile-cases in all types of soil.



Figure 70. K_{sw} vs. the pile area ratio (A_R) for 201 PD/LT pile-cases in all types of soil (logarithmic scale).



Figure 71. K_{sp} vs. the pile area ratio (A_R) for 203 PD/LT pile-cases in all types of soil.



Figure 72. K_{sp} vs. the pile area ratio (A_R) for 203 PD/LT pile-cases in all types of soil (logarithmic scale).



Figure 73. K_{sw} vs. blow count (BPI) for 206 PD/LT pile-cases in all types of soil (AAA).



Figure 74. K_{sp} vs. blow count (BPI) for 208 PD/LT pile-cases in all types of soil (AAA).



Figure 75. K_{sw} vs. blow count (BPI) for 95 PD/LT pile-cases in all types of soil at EOD (AEA).



Figure 76. K_{sp} vs. blow count (BPI) for 96 PD/LT pile-cases in all types of soil at EOD (AEA).



Figure 77. K_{sw} vs. blow count (BPI) for 109 PD/LT pile-cases in all types of soil at BOR (ABA).



Figure 78. K_{sp} vs. blow count (BPI) for 110 PD/LT pile-cases in all types of soil at BOR (ABA).


Figure 79. K_{sw} vs. blow count (BPI) for 57 PD/LT pile-cases with pile area ratios >350 in all types of soil.



Figure 80. K_{sp} vs. blow count (BPI) for 57 PD/LT pile-cases with pile area ratios >350 in all types of soil.



Figure 81. K_{sw} vs. blow count (BPI) for 144 PD/LT pile-cases with pile area ratios <350 in all types of soil.



Figure 82. K_{sp} vs. blow count (BPI) for 146 PD/LT pile-cases with pile area ratios <350 in all types of soil.



Figure 83. K_{sw} vs. blow count (BPI) for 16 PD/LT pile-cases with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil.



Figure 84. K_{sp} vs. blow count (BPI) for 16 PD/LT pile-cases with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil.



Figure 85. K_{sw} vs. blow count (BPI) for 41 PD/LT pile-cases with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil.



Figure 86. K_{sp} vs. blow count (BPI) for 41 PD/LT pile-cases with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil.



Figure 87. K_{sw} vs. blow count (BPI) for 64 PD/LT pile-cases with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil.



Figure 88. K_{sp} vs. blow count (BPI) for 64 PD/LT pile-cases with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm) in all types of soil.



Figure 89. K_{sw} vs. blow count (BPI) for 80 PD/LT pile-cases with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil.



Figure 90. K_{sp} vs. blow count (BPI) for 82 PD/LT pile-cases with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm) in all types of soil.



Figure 91. K_{sw} vs. blow count (BPI) for 12 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm).



Figure 92. K_{sp} vs. blow count (BPI) for 12 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts <6 BPI (0.24 blows/mm).



Figure 93. K_{sw} vs. blow count (BPI) for 27 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm).



Figure 94. K_{sp} vs. blow count (BPI) for 27 PD/LT pile-cases at EOD with pile area ratios >350 and blow counts >6 BPI (0.24 blows/mm).



Figure 95. K_{sw} vs. blow count (BPI) for 36 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm).



Figure 96. K_{sp} vs. blow count (BPI) for 36 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts <6 BPI (0.24 blows/mm).



Figure 97. K_{sw} vs. blow count (BPI) for 20 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm).



Figure 98. K_{sp} vs. blow count (BPI) for 21 PD/LT pile-cases at EOD with pile area ratios <350 and blow counts >6 BPI (0.24 blows/mm).



Figure 99. K_{sw} vs. blow count (BPI) for 18 PD/LT pilecases at BOR with pile area ratios >350 and all blow counts.



Figure 100. K_{sp} vs. blow count (BPI) for 18 PD/LT pilecases at BOR with pile area ratios >350 and all blow counts.



Figure 101. K_{sw} vs. blow count (BPI) for 88 PD/LT pilecases at BOR with pile area ratios < 350 and all blow counts.



Figure 102. K_{sp} vs. blow count (BPI) for 89 PD/LT pilecases at BOR with pile area ratios <350 and all blow counts.

CHAPTER 9 - ANALYSIS OF DATA SET PD

9.1 INTRODUCTION

9.1.1 Purpose

This chapter presents the graphical and statistical analysis of the pile-cases of data set PD. Graphical relationships in the form of scattergrams considering pile type and soil type are presented. A statistical analysis was performed in combination with the graphical relationships in an effort to correlate the results of chapter 8 with pile-cases that were not load tested to failure.

9.1.2 Overview

Two different types of correlations were examined for the pile-cases of data set PD. These can be summarized as follows:

(a) Damping Parameters - Soil-Type Correlations

Smith damping parameters (side and tip) obtained from CAPWAP results were correlated to the soil type at the side and tip of the pile, respectively. These graphical relationships are presented in section 9.2.

(b) Office Method - Field Method Predictions

The relationship between the office analysis predictions and the Energy Approach predictions of data set PD were obtained. These relationships can be compared to the correlations of data set PD/LT that were presented in the form of the coefficient K_{ew} , the ratio of CAPWAP or TEPWAP predictions over the Energy Approach predictions. Strong correlations between the two prediction methods may prove beneficial where load test data is not available. This approach can be especially useful since piles are dynamically monitored far more often than they are load tested to failure; hence, large data sets can be accumulated. The subgrouping of these correlations is consistent with table 3.

9.2 SMITH DAMPING PARAMETERS AND SOIL-TYPE CORRELATIONS

Figure 103 presents the relationship between Smith side damping parameters and the soil conditions along the pile shaft for 378 pile-cases analyzed by CAPWAP. The parameters shown are those obtained directly from the CAPWAP analyses performed on the pile-cases of data set PD. No corrections were performed on these parameters. A

substantial scatter exists in figure 103 with no clear correlation between the damping and the soil type at the pile shaft.

The information in figure 104, presenting the relationship between Smith tip damping parameters and tip soil conditions, indicates that no specific correlation can be made. These results are similar to those obtained in figures 22 and 23 for data set PD/LT.

9.3 CAPWAP AND THE ENERGY APPROACH CORRELATIONS

The following graphs compare CAPWAP and Energy Approach predictions based on pile type and soil type at the pile tip. The pile-type subgrouping includes large and small displacement piles, as well as miscellaneous piles (see table 3). The indicated slopes of the lines are identical to the parameter K_{ew} , which is the ratio of CAPWAP predictions to the Energy Approach predictions.

9.3.1 All Piles - All Soils

The relationship between the predicted capacities of CAPWAP and the Energy Approach for 398 PD pile-cases is shown in figure 105. The information indicates a good agreement between the two types of analyses, with the majority of data points in the ratio range of 1.00 to 0.60. The best-fit line through zero yields a ratio of $K_{ew} =$ 0.695 (CAPWAP over Energy Approach) with a coefficient of determination $r^2 = 0.699$. It can be seen that the small displacement piles (solid symbols) are concentrated in a narrow band, indicating a very good correlation of the two prediction methods for these pile-cases.

9.3.2 Large Displacement Piles

The following graphs compare the CAPWAP results to that of the Energy Approach predictions for large displacement piles.

(a) All Cases

Figure 106 presents all 238 large displacement pile-cases in all types of soil. The data points range from approximately 1.10 to 0.20. Overall, good agreement is presented with the best-fit line through zero at 0.676 (CAPWAP over Energy Approach) and a general trend between 1.00 and 0.60. The coefficient of determination is $r^2 = 0.650$.

(b) Sand and Silt

The information in figure 107 indicates excellent correlation between CAPWAP predictions and Energy Approach predictions for cases of large displacement piles in sand and silt. Most of the data points lie within the range of 1.00 to 0.60. The best-fit line is at 0.669 with $r^2 = 0.812$.

(c) Clay and Till

Figure 108 shows the correlation of CAPWAP and Energy Approach predictions for 50 large displacement pile-cases in clay and till. The majority of data points fall on or near the 0.80 line, with other cases reaching 0.40 and slightly below. The best-fit line through zero is at $K_{rw} = 0.600$ with $r^2 = 0.404$.

(d) Rock

The relationship between CAPWAP and the Energy Approach for 78 cases of large displacement piles found in rock is shown in figure 109. The information indicated a general scatter with the majority of data points falling between 1.00 and 0.60. The best-fit line through zero yielded $K_{ew} = 0.652$ with $r^2 = 0.572$.

(e) Unknown Soil Type

The correlation of the prediction methods for the 22 cases of large displacement piles in unknown soil types is presented in figure 110. It can be seen that regardless of soil type, good agreement is generally observed in these cases between the CAPWAP and Energy Approach predictions. The obtained best-fit line through zero is $K_{ew} = 0.844$ with $r^2 = 0.589$.

(f) Intermediate Conclusions

٠

- Generally good agreement exists between the predictions of CAPWAP to those of the Energy Approach for large displacement piles. The obtained relationship for all large displacement piles, at all times and in all types of soil (242 cases), is similar to that obtained for the corresponding cases in data set PD/LT.
- The breakdown of the piles to the different soil types shows that the highest correlation between the methods exists for piles driven in sand and silt. The worst correlation is obtained for piles driven in clay and till.
 - No subdivision was made regarding the time of driving. The analysis of data set PD is, therefore, equivalent to all time of driving cases in data set PD/LT.

9.3.3 Small Displacement Piles

The correlations of the two dynamic analysis predictions for small displacement piles in all soil types are presented in figures 111 through 114.

(a) All Cases

The information in figure 111 indicates an outstanding correlation between the CAPWAP and Energy Approach predictions for 76 small displacement pile-cases in all

soil types. The data points are almost exclusively within the range of 1.00 and 0.60 with the best-fit line at $K_{rw} = 0.800$ and $r^2 = 0.826$.

(b) Sand and Silt

The correlation for 26 small displacement pile-cases in sand and silt is shown in figure 112. A very well-defined relationship is observed with all data points within the range of 1.00 and 0.60. The best-fit line forced through zero is shown with a ratio of $K_{ew} = 0.807$ and $r^2 = 0.922$.

(c) Clay and Till

Figure 113 presents the correlation between CAPWAP and Energy Approach predictions for 21 small displacement pile-cases in clay and till. These data points are also indicating a relatively good correlation with the best-fit line at $K_{ew} = 0.723$ and $r^2 = 0.736$. The data points are within the range of 1.00 and 0.60, however, there is a larger scatter than that observed in sand and silt. This is also indicated by the reduction in the value of the calculated coefficient of determination.

(d) Rock

Figure 114 presents the comparison of the dynamic prediction methods for 29 small displacement pile-cases on rock. The relationship yields similar results to those of figures 111 through 113, with an excellent correlation between the prediction methods. The best-fit ratio is equal to $K_{ew} = 0.838$ with $r^2 = 0.797$.

(e) Intermediate Conclusions

- A better agreement with better correlation was found between the office method and the Energy Approach for small displacement piles when compared to large displacement piles. As both data sets contained a large number of cases (242 large displacement and 76 small displacement pilecases for all soil types at all driving times), the findings reflect the importance of pile type in the accuracy of the predictions.
- The predictions for small displacement piles in sand were found to match and correlate better than those in clay. In both cases, a better fit was found when compared to the large displacement pile-cases with the respective soil type. These results indicate that the soil type is secondary to the pile type as factors shaping the prediction results.

9.3.4 Miscellaneous Piles

The relationships of CAPWAP and Energy Approach predictions for miscellaneous piles for different soil conditions are presented in figures 115 through 119.

(a) All Cases

The correlation between the two prediction methods is presented in figure 115 for 85 miscellaneous pile-cases in all soil types. It is shown that there is excellent agreement between these predictions, with the majority of data points falling in the range of 1.00 to 0.60. The best-fit ratio is $K_{ew} = 0.763$ (CAPWAP over Energy Approach) with $r^2 = 0.873$.

(b) Sand and Silt

Figure 116 shows similar agreement between the two prediction methods for 40 miscellaneous pile-cases in sand and silt. The best-fit ratio equals 0.787 and most of the data points are within ± 20 percent of the 0.80 line with $r^2 = 0.857$.

(c) Clay and Till

The correlation of the predictions for 21 miscellaneous pile-cases in clay and till are shown in figure 117. Very good correlations are obtained with a best-fit ratio of $K_{cw} = 0.735$. The majority of data points lie within the 1.00 to 0.60 range, with a coefficient of determination of $r^2 = 0.783$.

(d) Rock

The information of figure 118 indicates a very good agreement between the CAPWAP and Energy Approach predictions for 19 pile-cases of piles found in rock. The best-fit ratio is 0.742 with $r^2 = 0.899$.

(e) Unknown Soil Type

Figure 119 presents the correlation results for five cases of miscellaneous piles driven in unknown soils. Good correlation is obtained from this small and non-specific data set.

9.4 STATISTICAL ANALYSIS OF DATA SET PD

In order to quantify the correlations obtained from the graphical relationships of section 9.3, a statistical analysis was performed as follows:

- (1) Determination of the first-order best-fit lines (forced through zero and y-intercept) by linear regression along with the sample coefficient of determination (r^2) to measure the quality of the best-fit line.
- (2) Determination of the mean and standard deviation of the K_{ew} ratio (CAPWAP over Energy Approach) as a measure of the accuracy (through the mean) and precision (through the standard deviation) of the calculated ratio distribution.

Kew = CAPWAP predictions / Energy Approach predictions							
Pile-	Linear Regression						
Case	Number		Best Fit		Forced through Zero		
Group		X-coefficient	y-intercept	r-squared	X-coefficient	r-squared	
AA	403	0.603	75.4	0.723	0.695	0.699	
LA	242	0.593	77.0	0.667	0.676	0.650	
LS	92	0.580	81.8	0.841	0.669	0.812	
LC	50	0.407	142.4	0.570	0.600	0.404	
LR	78	0.563	89.7	0.591	0.652	0.572	
LN	22	0.814	31.5	0.590	0.844	0.589	
SA	76	0.751	22.7	0.830	0.800	0.826	
SS	26	0.774	13.0	0.924	0.807	0.922	
sc	21	0.651	32.2	0.747	0.723	0.736	
SR	29	0.728	56.3	0.817	0.838	0.797	
МА	85	0.678	51.8	0.892	0.763	0.873	
MS	40	0.705	42.5	0.871	0.787	0.857	
мс	21	0.640	46.6	0.807	0.735	0.783	
MR	19	0.652	76.8	0.924	0.742	0.899	
MN	5	0.816	61.3	0.933	0.955	0.898	

Table 10. Linear regression analysis of K_{ew} for PD pile-cases.

Pile-case legend:

XX

first letter denotes pile type: A = all piles,
L=large displacement, S=small displacement,
and M=miscellaneous piles.

- second letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, R=rock, and N=not available.

9.4.1 Linear Regression Analysis

The results of the linear regression analysis performed on the subgroups of data set PD are presented in table 10. The first two columns of table 10 report the pile-case subgroups and the total number of pile-cases included in the analysis, respectively. This analysis is similar to that which was performed in section 8.4.1 for data set PD/LT. The results of the best-fit linear regression performed for each subgroup are listed in columns 3, 4, and 5. Column 3 shows the first-order best-fit ratio, and the corresponding intercept is presented in column 4. Column 5 shows the sample coefficient of determination (r^2) for each subgroup. The coefficients for the best-fit ratio forced through zero are listed in columns 6 and 7. Column 6 presents the first-order best-fit sample coefficient of determination.

Table 10 indicates a relatively consistent best-fit ratio (forced through zero) for all pile types and soil types. It can be seen that pile type is the controlling factor in the resulting best-fit ratio as very little change is seen for large or small displacement piles in different soils. For example, considering small displacement piles, the most extreme best-fit ratios range from 0.723 for piles found in clay and till to 0.838 for piles found in rock. Similarly, for large displacement piles in sand, clay, and rock, the best-fit ratios range from 0.676 to 0.600. Excellent coefficients of determination are reported for all small displacement piles ($r^2 = 0.826$) and, in particular, in sand ($r^2=0.922$). This is in comparison to all large displacement piles ($r^2 = 0.600$) that improve in sand only to $r^2 =$ 0.812, compared to clay with $r^2 = 0.404$.

9.4.2 Mean and Standard Deviation Analysis

Table 11 presents the results of the statistical analysis, evaluating the mean and standard deviation of the PD subgroups outlined in table 3. The first two columns are consistent with table 10 and they report the pile-case subgroup and the total number of pile-cases included in each analysis, respectively. The mean of all cases was found to be 0.774 with mean values obtained for the subgroups in the range of 0.701 to 0.863 (with the exception of miscellaneous piles in unknown soil types, which represent a subgroup of only five piles). Overall, the values obtained are very consistent with very good standard deviations compared to those obtained for the relationships between the predictions and the actual capacity. This suggests that the prediction methods may be similar in their analysis and, based on the mean values, it appears that soil type has a lesser effect on the correlation between CAPWAP and the Energy Approach predictions.

9.5 SUMMARY AND CONCLUSIONS

Data set PD contains information that allows capacity predictions to be conducted on 403 pile-cases, based on dynamic measurements. As no comparison can be made to the actual static resistance, the results serve two purposes:

Pile-	Kew = CAP	WAP/ Energy	P/ Energy Approach		
Case	Number	Mean	Standard		
Group			Deviation		
AA	403	0.774	0.2099		
LA	242	0.742	0.2359		
LS	92	0.754	0.1753		
LC	50	0.701	0.2093		
LR	78	0.722	0.2202		
LN	22	0.860	0.4528		
SA	76	0.813	0.1255		
SS	26	0.815	0.0862		
sc	21	0.741	0.1396		
SR	29	0.863	0.1229		
MA	85	0.827	0.1734		
MS	40	0.861	0.1410		
мс	21	0.806	0.2297		
MR	19	0.821	0.1544		
MN	5	1.022	0.1233		

Table 11. Statistical analysis of K_{ew} for PD pile-cases.

Pile-case legend:

XX

first letter denotes pile type: A=all piles,
L=large displacement, S=small displacement,
and M=miscellaneous piles.

- second letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, R=rock, and N=not available.

- They can be compared to the pile-cases of data set PD/LT to allow assessment of trends found in that data set.
- They indirectly serve as an excellent indicator for the controlling parameters through the conditions in which the different prediction methods are close to each other or different from each other.

The following conclusions are based on the scattergrams presented in figures 103 through 119:

- 1. No correlations seem to exist between soil type and damping parameters for either Smith damping at the pile side or the pile tip.
- 2. General comparisons between the best-fit linear regression of the different subgroups in data set PD/LT (tables 5 through 7) and in data set PD (table 10) indicate a reasonably good agreement between the two independent data sets. Some of the major parameters are summarized in table 12 below.

Pile- Case Group	K _{ew} Coefficient						
	PD/LT			PD			
	number	x-coefficient	r²	number	x-coefficient	٢²	
AAA	206	0.641	0.766	403	0.695	0.699	
LAA	162	0.589	0.554	242	0.676	0.650	
LAS	118	0.571	0.586	92	0.669	0.812	
LAC	43	0.446	0.600	50	0.600	0.404	
SAA	44	0.764	0.937	76	0.800	0.826	
SAS	23	0.750	0.942	26	0.807	0.922	
SAC	8	0.779	0.971	21	0.723	0.736	

Table 12. Linear regression summary of selected PD/LT and PD subgroups.

For many of these cases, r^2 can serve as a good indicator of the agreement as mentioned above. The assigned x-coefficient refers to the slope of the best-fit line forced through zero. A more realistic comparison may be obtained through the slope of the natural best-fit line.

- 3. General comparisons between the parameters of the normal distribution of the different subgroups in data set PD/LT (table 8) indicate a reasonably good agreement between the two independent data sets. Some of the major parameters are summarized in table 13.
- 4. Based on the data, it seems that both methods predict fairly similarly in the case of small displacement piles and, in particular, in sand. The small displacement piles present higher mean values, higher x-coefficients, higher coefficients of determination, and smaller standard deviation ratios. This conclusion verifies the fact that when small soil inertia and soil damping exist, both methods give similar results.
- 5. The cases related to the large displacement piles exhibit lower xcoefficients, lower coefficients of determination, and lower mean values, while having higher standard deviation values. This indicates that, in the case of large displacement piles, the dynamic methods differ from each other as the damping modeling has an active role in the CAPWAP analysis of these cases.
- 6. The consistent pattern of better agreement in the predictions for piles in sand compared to those in clay indicates the relative importance of the soil type. This, however, is secondary to the importance of pile type.

Pile	Kew Coefficient					
Case Group	PD/LT			PD		
- -	number	mean	standard deviatio n	number	mean	standard deviatio n
AAA	206	0.712	0.182	403	0.774	0.210
LAA	162	0.689	0.176	242	0.742	0.236
LAS	118	0.693	0.171	92	0.754	0.175
LAC	43	0.670	0.184	50	0.701	0.209
SAA	44	0.796	0.180	76	0.813	0.126
SAS	23	0.746	0.203	26	0.815	0.086
SAC	8	0.738	0.085	21	0.741	0.140

Table 13. Statistical analysis summary of selected PD/LT and PD subgroups.



Figure 103. Side soil conditions vs. Smith side damping based on CAPWAP results for 372 PD pile-cases.



Figure 104. Tip soil conditions vs. Smith tip damping based on CAPWAP results for 377 PD pile-cases.


Figure 105. CAPWAP predictions vs. Energy Approach predictions for 398 PD pile-cases in all types of soil.



Figure 106. CAPWAP predictions vs. Energy Approach predictions for 238 large displacement PD pile-cases in all types of soil.



Figure 107. CAPWAP predictions vs. Energy Approach predictions for 89 large displacement PD pile-cases in sand and silt.



Figure 108. CAPWAP predictions vs. Energy Approach predictions for 50 large displacement PD pile-cases in clay and till.



Figure 109. CAPWAP predictions vs. Energy Approach predictions for 76 large displacement PD pile-cases in rock.



Figure 110. CAPWAP predictions vs. Energy Approach predictions for 22 large displacement PD pile-cases in unknown soil types.



Figure 111. CAPWAP predictions vs. Energy Approach predictions for 76 small displacement PD pile-cases in all types of soil.



Figure 112. CAPWAP predictions vs. Energy Approach predictions for 26 small displacement PD pile-cases in sand and silt.



1 kip = 4.448 kN

Figure 113. CAPWAP predictions vs. Energy Approach predictions for 21 small displacement PD pile-cases in clay and till.



Figure 114. CAPWAP predictions vs. Energy Approach predictions for 29 small displacement PD pile-cases in rock.



Figure 115. CAPWAP predictions vs. Energy Approach predictions for 85 miscellaneous PD pile-cases in all types of soil.



Figure 116. CAPWAP predictions vs. Energy Approach predictions for 40 miscellaneous PD pile-cases in sand and silt.



Figure 117. CAPWAP predictions vs. Energy Approach predictions for 21 miscellaneous PD pile-cases in clay and till.



Figure 118. CAPWAP predictions vs. Energy Approach predictions for 19 miscellaneous PD pile-cases in rock.



Figure 119. CAPWAP predictions vs. Energy Approach predictions for five miscellaneous PD pile-cases in unknown soil types.

CHAPTER 10 - SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

10.1 SUMMARY

Two methods are currently employed for the analysis of dynamic measurements obtained during pile driving. Both methods are based on the solution of the one-dimensional wave equation for the stress wave traveling through the pile following the hammer's impact. The first method, an office analysis, utilizes a numerical solution of a mathematical model for the pile-soil system under measured boundary conditions (e.g., the computer codes CAPWAP or TEPWAP). The other method, a field analysis known as the Case Method, which is based on a simplified closed-form solution and empirical correlations, provides an instantaneous evaluation of the pile capacity following each hammer blow.

Substantial experience suggests the existence of major limitations in the field method. In addition, no large-scale evaluation has been carried out for the office methods since their development.

A simplified method, based on the energy balance between the total energy delivered to the pile and the work done by the pile/soil systems, is proposed as an alternative field method. This method, entitled the Energy Approach, assumes elasto-plastic load displacement pile-soil relations. Calculated transferred energy and maximum pile displacement from the measured data together with the field blow count are used as input parameters for the Energy Approach. The method does not consider the propagation process and is aimed at providing a real-time pile capacity prediction in the field. The Energy Approach simplified analysis considers the energy loss from elastic soil/pile deformations and the work done by the static resistance due to plastic soil deformation.

The stress-wave-based solutions represent the external forces acting on the penetrating pile as a stationary soil resistance. Traditionally, this resistance consists of static and dynamic components. The static component is usually considered to be elasto-plastic and the dynamic component is represented by viscous damping.

It was presented and argued (in this research) that this type of formulation does not correctly represent the physical phenomena associated with pile driving. The dynamic resistance component needs to stand for phenomena such as soil inertia, wave radiation, and true damping. These factors are determined by the pile shape, penetration depth, acceleration at the pile toe, and the surrounding soil and, hence, cannot be correlated through viscous damping parameters to soil type alone. The energy loss due to various combined factors associated with the pile penetration, such as damping radiation and inertia, are not considered directly by the Energy Approach. As such, the method serves as an excellent indicator for pointing out the physical phenomena that should account for dynamic energy losses during driving.

Two large data sets were gathered at the University of Massachusetts at Lowell. One, PD/LT, contains 208 dynamic measurement cases on 120 piles monitored during driving, followed by a static load test to failure. The data were obtained from various sources and reflect varying combinations of soil-pile-driving systems. The other, PD, contains data on 403 piles monitored during driving and was provided by Pile Dynamics, Inc. of Cleveland, Ohio. All cases were examined and analyzed.

Data set PD/LT was analyzed for the static resistance, dynamic measurements, office analysis predictions, Case damping coefficient, and the Energy Approach predictions. Data set PD was analyzed for CAPWAP analysis and the Energy Approach predictions.

The results of this study invalidate the concept of a unique recommended correlation between the viscous damping parameters and soil type in both wave-based analyses. It is shown that energy losses should be attributed more to soil inertia rather than soil damping. As such, energy losses are mostly pile-shape-dependent, in addition to the soil type and driving resistance influences.

A pile-shape parameter denoted as area ratio (A_R) was introduced as a quantitative measurement for the pile shape. The area ratio allows one to distinguish between large and small displacement piles on the basis of their soil mobilization at the pile tip relative to their skin-friction contact area.

The accuracy of the dynamic methods, when compared to the actual static capacity, and the relations between the predictions themselves, provided insight into the controlling mechanisms and the preferable conditions for these methods. It was found that best results are obtained for small displacement piles (with area ratio $A_R > 350$). The worst analysis conditions are for large displacement piles in clay under low driving resistance (<6 BPI [0.24 blows per mm]).

The Energy Approach method was found to provide excellent evaluations of pile capacity under all conditions. The method is, therefore, proposed to be used in the field for instantaneous capacity determination. The predictions of this method were found, on the average, to provide more accurate evaluations than the sophisticated office methods, especially for records obtained at the end of initial driving. The Energy Approach is, therefore, also proposed to be used as an independent tool to evaluate the office methods.

10.2 CONCLUSIONS

The research investigated four general correlations:

- (1) Damping parameters vs. soil type.
- (2) Load test results vs. office methods (CAPWAP/TEPWAP) predictions using the parameter K_{sw} = load test capacity/office method prediction.
- (3) Load test results vs. Energy Approach predictions using the parameter K_{sp} = load test capacity/Energy Approach prediction.
- (4) CAPWAP/TEPWAP vs. Energy Approach using the parameter K_{ew} = office method prediction/Energy Approach prediction that is also equivalent to K_{ew} = K_{sp}/K_{sw}

The conclusions based on the graphical and statistical analyses presented in the preceding chapters are summarized as follows:

1. Viscous damping does not truly represent the physical phenomena through which energy is lost and, hence, cannot be viewed as intrinsic to soil type.

Figure 21 presents the relationship between the back-calculated case-damping parameter, J_c (which was required to provide the actual measured static resistance), to the soil type at the tip. No correlation can be observed in this figure. Moreover, in many cases, the obtained damping parameters are negative, which has no physical meaning.

Figures 120 and 121 present the relationship between soil conditions and Smith side and tip damping for all PD/LT and PD pile-cases combined (581 cases combining figures 22 and 104, 23 and 105, respectively). Figures 120 and 121 present the damping parameters that were used in the analyses in order to obtain the best signal match between the calculated and measured signals. No correlation was found between the damping parameter used in these analyses and soil type.

2. The capacity predictions for small displacement piles resulted in higher accuracy and substantially lower scatter for both dynamic methods when compared to the predictions and the scatter obtained for large displacement piles. (See, for example, figures 29 and 30 compared to figures 32 and 33).



Figure 120. Side soil conditions vs. Smith side damping based on CAPWAP/TEPWAP results for 581 pile-cases.



Figure 121. Tip soil conditions vs. Smith tip damping based on CAPWAP/TEPWAP results for 581 pile-cases.

Small and large displacement piles can be defined according to area ratio $(A_R>350$ for small displacement piles and $A_R<350$ for large displacement), as presented and discussed in sections 4.4 and 8.5.4 (see figures 70 and 72).

- 3. The above conclusion is reinforced by the excellent correlations that were obtained between the prediction methods for the small displacement pile cases (see tables 12 and 16). These observations show that energy is lost mainly due to soil inertia as a result of the mobilization of the soil mass at the pile tip. The correlations of section 8.3 (see tables 5 and 8) indicate that soil type has very little effect on the accuracy of the Energy Approach predictions. As such, correlations were examined based on pile type, driving resistance, and time of driving (see section 8.5).
- 4. Correlations between driving resistance and dynamic predictions do not lead to definitive conclusions (see table 9). Figures 73 through 102 and reanalysis of the prediction coefficients on the basis of blow counts between 0 to 10 BPI (0.39 blows per mm) and over 10 BPI indicate the following trends:
 - Small displacement piles with high driving resistance will result in a small loss of energy due to soil inertia and, therefore, more accurate predictions, as the actual pile resistance is similar to the maximum resistance during driving. The results of both methods of analysis performed well for that category. For example, the mean and standard deviation for 25 small displacement piles ($A_R > 350$) driven in the range of 0 to 10 BPI (0.39 blows per mm) is $K_{sw} = 1.360$, $\sigma_x = 0.5581$ and $K_{sp} = 0.939$, $\sigma_x = 0.2788$, compared to the 32 pile-cases driven under resistances higher than 10 BPI (0.39 blows per mm) that resulted in $K_{sw} = 1.159$, $\sigma_x = 0.4422$ and $K_{sp} = 0.929$, $\sigma_x = 0.2185$.
 - Large displacement piles with low driving resistance will result in a large loss of energy due to soil inertia and less accurate predictions, as the actual pile resistance is the difference between the maximum pile resistance during driving and the large energy loss. (For this category, the Energy Approach predicts well for EOD and over-predicts for BOR while the office methods seem to under-predict for EOD and improve with time.) For example, the mean and standard deviation for 101 large displacement pile cases ($A_R < 350$) driven at the range of 0 to 10 BPI (0.39 blows per mm) is $K_{sw} = 1.353$, $\sigma_x = 0.4879$ and $K_{sp} = 0.906$, $\sigma_x = 0.3257$, compared to the 43 pile-cases driven in resistances higher than 10 BPI (0.39 blows per mm), which resulted in $K_{sw} = 1.601$, $\sigma_x = 0.6279$ and $K_{sp} = 0.951$, $\sigma_x = 0.2961$.

5. The End of Driving (EOD) condition is of special interest as it represents the ability of the methods to predict the capacity during driving and to evaluate for the most common state. The predictions for EOD were examined, in particular, in figures 44 (and 46), 45 (and 47), 75, 76, and tables 5, 6, 8, and 9. The data clearly indicate very good predictions and correlations of the Energy Approach under all categories with better performance for small displacement piles. For example, 97 piles at EOD resulted in $K_{sw} = 1.478$, $\sigma_x = 0.6167$ and $K_{sp} = 1.023$, $\sigma_x = 0.3073$. These numbers improved for the subgroup of 29 small displacement piles showing $K_{sw} = 1.252$, $\sigma_x = 0.5616$, and $K_{sp} = 0.935$, $\sigma_x = 0.2616$. The large mean and standard deviation ratios for the K_{sw} -coefficient suggest limitations of the office analysis methods for all piles at the end of driving, but, in particular, for large displacement piles.

10.3 RECOMMENDATIONS

10.3.1 General

The recommendations are comprised of three parts. One part (sections 10.3.2, 10.3.3, and 10.3.4) describes the major prediction parameters and their statistical evaluation for the different pile-cases. The statistical evaluation is shown in the form of:

- Determination of the first-order best-fit line forced through zero (xcoefficient) and the measure of its accuracy through the coefficient of determination (r^2). Section 8.4.1 reviewed these parameters, mainly indicating that good correlation exists for $r^2 \ge 0.8$ and that $0.6 \le r^2 < 0.8$ indicates a moderate correlation only.
- Mean and standard deviation of the normal distribution. The mean represents the accuracy of the prediction (the ability to predict the measured ultimate static capacity) and the precision of the method refers to the scatter, which is represented by the standard deviation (the smaller the scatter, the lower the standard deviation).

In examining a certain pile-case category, it is advised to check both the x-coefficient and the mean as measures of the prediction accuracy and check the coefficient of determination and the standard deviation as measures of the scatter. It is also advised to look at the actual data presented in the scattergram associated with the particular case.

The second part of the recommendations refers to a discussion regarding the factors of safety that are associated with the predictions of the office methods and the Energy Approach. The third part lists several recommendations for the implementation of the methods and potential future improvements.

10.3.2 The Performance of the Office Methods (CAPWAP/TEPWAP)

Table 14 summarizes the major numerical parameters obtained through the analysis of data set PD/LT, concerning the performance of the office analyses. Only the pile-cases that contained a significant number of cases and/or could indicate an important influence were included. Table 14 indicates the following:

- For all piles at any time of driving in all soils, the office method underpredicts the actual static capacity by about 30 percent with a relatively large scatter. The scatter is mostly due to low accuracy in the prediction of cases involving large displacement piles (see LAA compared to SAA) and driving in clay (see AAC compared to AAS). It must be emphasized that a separate observation (not presented in this study) shows a clear improvement of the office method predictions with time. The accuracy of the method when analyzing records close to the time of load testing is, therefore, not evident in the data.
- The major single parameter controlling the accuracy of the method is the pile type. The accuracy of the method and its scatter reduces substantially for small displacement piles at any time of driving in all soils. It is further evident with the accuracy of the small displacement piles at the end of driving for which the office method presented excellent results with a mean and x-coefficient close to 1 and $r^2 = 0.95$.

10.3.3 The Performance of the Energy Approach

Table 15 summarizes the major numerical parameters obtained through the analysis of data set PD/LT concerning the performance of the Energy Approach. Only the pile groups that contained a significant number of pile-cases and/or could indicate an important influence were included. The Energy Approach was proposed as the field method and, hence, the performance at the end-of-driving condition is emphasized. Table 15 indicates the following:

• For all piles at any time of driving in all soils, the Energy Approach overpredicts the actual static capacity by about 8 percent with a noticeable scatter that is, however, significantly smaller than that of the office method. As in the prediction of the office methods, the scatter is mostly due to lower accuracy in the prediction of cases involving large displacement piles (see LAA compared to SAA) and driving in clay (see AAC compared to AAS).

- Good correlation exists for predictions related to end of driving and small displacement piles. This is evident through the high coefficients of determination (r²) and small standard deviations for these cases.
- The mean prediction ratio for all cases at the end of driving (AEA) is 1.0. Higher accuracy is obtained for small displacement piles (SEA) compared to large displacement piles (LEA).

10.3.4 The Correlation Between the Office Methods and the Energy Approach

Tables 16 and 17 summarize the correlations obtained between the two methods under the different pile cases. Table 16 has a similar format to that of tables 14 and 15 and is based on the PD/LT data set. The parameters in table 16 referring to the end-of-driving conditions present excellent correlations between the methods, except for predicting large displacement piles for which each of the methods encountered its own difficulties.

Table 17 is based on the data combined in both data sets (PD and PD/LT) and, hence, refers to 609 pile-cases. The low correlations were again obtained for large displacement piles (LAA), especially when driven in clay (LAC).

The obtained relationships of tables 16 and 17 can perform as excellent guidelines when comparing the results of the office methods to that of the Energy Approach.

10.3.5 Factors of Safety and Risk Analysis

(a) General

Factor of safety in the current common use is the factor that we apply to our prediction in order to come up with an allowed capacity for which we would feel freedom from meaningful risk.

Risk is defined (see, for example, Briaud and Tucker, 1988) as the probability (P) that the predicted ultimate capacity (Q_p) divided by the factor of safety (F.S.) exceeds the measured ultimate load (Q_m) :

$$R = P[(\frac{Q_p}{F.S.}) > Q_m]$$
(41)

The calculated K-values (K_{sw} and K_{sp}) as presented throughout this research study are the ratio of $K = Q_m/Q_p$, using the above notation. The risk can therefore be rewritten in the following format:

$$R = P[K \cdot F.S. < 1] \tag{42}$$

where $K = K_{sw}$ or K_{sp} .

As the construction cost is directly related to the factor of safety, we are interested in several forms of that factor:

- What is the minimum factor of safety that will allow us absolute safety?
- What is the risk associated with any factor of safety?
- What is the actual factor of safety when considering the inaccuracy of the prediction method?

These aspects are discussed in the following section.

(b) Absolute Safety Based on Data Set PD/LT

The data sets were searched for the worst over-prediction ratio. The absolute factor of safety was defined as the one that should have been used in this case in order to make certain that the allowed capacity would not exceed the ultimate capacity. The results of the analysis based on this approach are summarized in table 18 in the following manner:

Columns 1 through 3 detail the method of analysis, pile-case category, and number of cases related to that category.

Column 4 indicates the minimum K-factor $(K_{sw} \text{ or } K_{sp})$ in the related data set. The K_{min} -value is associated with the maximum over-prediction ratio.

Column 5 is the inverse ratio of K_{min} , indicating the absolute factor of safety that would have been needed in this case in order to guarantee that the allowed capacity would not exceed the ultimate static capacity.

Column 6 takes into consideration the average built-in risk or safety that exists in each of the methods. The office methods under-predict on the average, such that the mean K_{sw} for the AAA category is 1.367. Using, in addition, a factor of safety of 1.75 means that the actual mean factor of safety is 1.367 x 1.75, which results in 2.40. The Energy Approach is over-predicting on the average. The mean K_{sp} for the AAA category is 0.925, which means that when employing a factor of safety of 2.44, the actual mean factor of safety is 0.925 x 2.44 = 2.26.

Column 6 indicates that although the Energy Approach requires somewhat higher factors of safety in order to cover the worst over-prediction case, the actual factor of safety that is used when considering the accuracy of the method is smaller than that of the office

Pile-	Number	Ar	- WSM	= Load Test/CA	PWAP or	TEPWAP
Case	of	Area	Linear Best Fit	through Zero	Nom	al Distribution
Group	Cases	Ratio	x-coefficient	r-squared	mean	standard deviation
AAA	206	all	1.265	0.692	1.367	0.5334
AAS	141	all	1.272	0.749	1.385	0.4758
AAC	51	all	1.319	0.383	1.443	0.676
SAA	60	>350	1.108 (44)	0.932	1.237	0.4916
LAA	144	<350	1.372 (102)	0.554	1.427	0.5433
AEA	26	all	1.248	0.740	1.478	0.6167
SEA	41	>350	1.113 (29)	0.933	1.250	0.4838
LEA	56	<350	1.529 (68)	0.528	1.643	0.6544

Table 14. Linear regression and statistical analysis of Ksw for selected PD/LT pile-cases.

 First letter denotes pile type: A=all piles, L=large displacement, and S=small displacement. Pile-case legend: XXX

second letter denotes time of measurement: A=anytime,

E=end of driving, and B=beginning of restrike. - third letter denotes soil type: A=all soils, S=sand and

silt, C=clay and till, and R=rock.

Pile-	Number	Ar	Ks	sp = Load Test	(Energy Ap	proach
Case	of	Area	Linear Best Fit	through Zero	Non	nal Distribution
Group	Cases	Ratio	x-coefficient	r-squared	mean	standard deviation
AAA	206	all	0.839	0.703	0.925	0.2932
AAS	141	all	0.831	0.700	0.942	0.3127
AAC	53	ail	0.872	0.666	906.0	0.2689
SAA	41	>350	0.856 ⁽⁴⁴⁾	0.416	0.960	0.2347
LAA	144	<350	0.832 (164)	0.534	0.920	0.3165
AEA	86	all	0.900	0.804	1.023	0.3073
SEA	41	>350	0.851 (29)	0.917	0.960	0.2347
LEA	57	<350	0.966 (69)	0.603	1.068	0.3453

Table 15. Linear regression and statistical analysis of Ksp for selected PD/LT pile-cases.

Pile-case legend: XXX

first letter denotes pile type: A=all piles, L=large displacement, and S=small displacement.

second letter denotes time of measurement: A=anytime, E=end of driving, and B=beginning of restrike.
 third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

220

									_
y Approach	Distribution	standard deviation	0.203	0.215	0.175	0.198	0.147	0.155	0.126
PWAP/Energy	Normal	mean	0.753	0.721	0.720	0.687	0.807	0.783	0.740
APWAP or TE	through Zero	r-squared	0.727	0.609	0.690	0.447	0.937	0.950	0.914
Kew = C	Linear Best Fit	x-coefficient	0.670	0.638	0.617	0.600	0.772	0.754	0.752
Number	of	Cases	609	404	210	93	120	49	29
Pile-	Case	Group	AAA	LAA	LAS	LAC	SAA	SAS	SAC

and PD pile-cases.
ΛLT
PD
for selected
3
X
of
sis
al y
an
R
tic
ıtis
Sta
Ъ
I B
ion
ess
ŭ
2
Linear
16.
<u>e</u>
Tab

ter denotes pile type: A=all piles, L=larg	ement, and S=small displacement.
- first let	displac
XXX	

Pile-case legend:

second letter denotes time of measurement: A=anytime E=end of driving, and B=beginning of restrike.
 third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

	_				
rgy Approach	mal Distribution	standard deviation	0.179	0.147	0.184
PWAP/Enei	LION	mean	0.743	0.813	0.695
APWAP or TE	through Zero	r-squared	0.861	0.947	0.398
Kew = C	Linear Best Fit	x-coefficient	0.699	0.762	0.598
Ar	Area	Ratio	all	>350	<350
Number	of	Cases	35	39	56
Pile-	Case	Group	AEA	SEA	LEA

Table 17. Linear regression and statistical analysis of Kew for selected PD/LT pile-cases.

Pile-case legend: XXX - first letter denotes pile type: A=all piles, L=large displacement, and S=small displacement. - second letter denotes time of measurement: A=anytime,

E=end of driving, and B=beginning of restrike. - third letter denotes soil type: A=all soils, S=sand and silt, C=clay and till, and R=rock.

methods. This situation is especially clear for the end-of-driving cases where a factor of safety of 2.0 actually means an average factor of safety of 2.0 for all cases.

Column 7 examines the maximum factor of safety that will be employed for the worst under-prediction ratio, using this approach. Since the maximum under-prediction ratio for the office method is $K_{sw_{max}} = 4.42$, the maximum actual factor of safety that will result from using an F.S. of 1.75 is 1.75 x 4.42 = 7.74.

The small scatter for the Energy Approach is again demonstrated for all the end-ofdriving predictions where the use of a factor of safety of 2.0 will result in a maximum conservative factor of safety of only 4.24.

Method of Analysis	Pile- Case Group	No. of Cases	K _{min}	Factor of Safety (F.S.)	F.S. x mean K	F.S. x K _{max}
CAPWAP/TEPWAP	AAA	206	0.57	1.75	2.40	7.74
CAPWAP/TEPWAP	AEA	97	0.57	1.75	2.59	7.74
Energy Approach	AAA	208	0.41	2.44	2.26	5.28
Energy Approach	AEA	98	0.51	1.96	2.01	4.24

Table 18. Absolute factor of safety based on data set PD/LT.

(c) Factor of Safety and the Associated Risk Based on the Actual Data

The PD/LT data set was used to prepare the relationships between the applied factor of safety and its associated risk as defined earlier. The procedure was described by Briaud and Tucker (1988) and contains the following steps:

- 1. Select an arbitrary F.S. (factor of safety).
- 2. Calculate the risk of failure as the ratio between the number of piles in the data set for which $Q_p/Q_m > F.S.$ over the total number of piles in that data base.
- 3. Repeat steps 1 and 2 for different F.S. values.
- 4. Plot the obtained relations between the applied factor of safety and the associated risk.

This analysis was carried out for three pile group cases (AAA, AEA, and SEA) for each of the two prediction methods. Figures 122, 124, and 126 are related to the office

method predictions and figures 123, 125, and 127 are related to the Energy Approach predictions. An accurate prediction occurs when the predicted value is equal to the failure value and, hence, associated with a risk of 100 percent for a factor of safety of 1. A smaller risk with F.S. = 1 reflects on under-prediction. For example, according to figure 122, 77.7 percent of the piles (AAA cases) will be safe using CAPWAP and F.S. = 1 as the method under-predicts in most cases. In order to include the bias of the prediction method itself, the relationships between the applied factor of safety and the mean over-prediction ratio (or the mean actual factor of safety) were added in each chart. For example, figure 122 indicates that using a factor of safety of 1.2 for all cases of the office method, will result in a risk of 5.8 percent. This factor of safety, however, is actually equivalent to a factor of safety of 1.64 when considering the mean of $K_{sw} = 1.367$ for the AAA pile-case group.

Table 19 summarizes numerically, based on figures 122 through 127, a few representative factors of safety and their associated risks. The numerical values show the accuracy and reliability of the Energy Approach, especially for the end-of-driving analysis. The use of a factor of safety of 1.6, for example, will be associated with an actual F.S. of 1.6 and a risk of 2.1 percent for the Energy Approach, while the same factor of safety means an actual F.S. of 2.3 and risk of 1.1 percent for the office method.

(d) Factor of Safety and the Associated Risk Based on the Probability Distribution Function

The risk associated with the factor of safety can also be evaluated based on the probabilistic models. The models associated with the distribution of the predictions were presented in section 8.4.

The use of these evaluations can be done in the following way:

$$R = P[K \times F.S. < 1] = P[K < \frac{1}{F.S.}]$$
(43)

using x = 1/F.S.

for a normal distribution

$$P[K < x] = P[U \le (\frac{x - m_x}{\sigma_x})] = F_u(\frac{x - m_x}{\sigma_x}) = F_u(u)$$
(44)

for a log-normal distribution

$$P[K < x] = P[\ln K \le \ln x] = F_{\mu}(\frac{\ln(x/m_x)}{\sigma_{\ln x}})$$
(45)

where F_u is obtained directly from the standard tables of the normal distribution function.

For example, using the log-normal distribution for K_{sp} for AAA pile-cases (see figure 66):

$$m_x = 0.8818, \sigma_{inx} = 0.3094$$

F.S.	1/F.S.	U	Fu	$\mathbf{R} = \mathbf{P}[\mathbf{K} < 1/\mathbf{F}.\mathbf{S}.]$	R (table 19)
1.0	1	0.4066	0.6591	65.9%	67.8%
1.6	0.625	-1.2517	1-0.8944	10.6%	14.4%
1.8	0.556	-1.4932	1-0.93189	6.8%	7.7%
2.0	0.500	-1.8337	1-0.96712	3.3%	4.8%
2.5	0.400	-2,5549	1-0.996	0.4%	0%

These numbers fit very well with the risk presented in figure 123 and table 19.

10.3.6 Recommendations for Implementation

I. The simplicity of the Energy Approach formulation together with its high accuracy at the end of driving makes it an ideal method of analysis to be used in the field and as a check for the office methods.

The following factors of safety are recommended to be used with the Energy Approach predictions:

- F.S. = 2.50 for all piles in all cases (AAA, mean $K_{sp} = 0.93$).
- F.S. = 2.00 for all end-of-driving cases (AEA, mean $K_{sp} = 1.00$).
- F.S. = 2.00 for all small displacement piles ($A_R > 350$) in all cases (SAA, mean $K_{sp} = 0.94$).

The following recommendations are made for improving the use of the Energy Approach method:

1. Implement the Energy Approach as part of the Pile-Driving Analyzer (PDA) routine analysis. This will ensure more accurate field predictions that may be further enhanced by using K_{sp} correction factors based on the pile group cases as shown in the previous section.

- 2. The limited accuracy of the displacements obtained from acceleration measurements brought the use of the blow count for set evaluation. It was found that, in many cases, blows are counted along a distance of a foot rather than an inch, even during the final penetration. Whenever records of blows per foot were replaced by measurements along 1 in (25.4 mm), a significant improvement was obtained for the estimated set and, as a result, in the accuracy of the Energy Approach predictions. It is proposed to remediate the problem by measuring blows along 1 in (25.4 mm) of penetration together with the following recommendations.
- 3. The Energy Approach can be improved by analysis based on average blows per inch, average E_{MAX} , average D_{MAX} , and average capacity for the last inch during driving rather than for a single blow.
- II. Both methods of analysis, the simple Energy Approach and the stress-wave-based formulation, require the recording of the pile displacement with time. Currently, we either double integrate the acceleration measurements to obtain displacement with time, or integrate once, using the velocity in the numerical solution of the wave equation. Direct and accurate displacement measurements can be currently obtained, instead of using accelerometers. Such measurements based, for example, on laser devices will enhance substantially the accuracy of all dynamic methods.

Prediction method	CAPWAP/TEPWAP			Energy Approach		
Pile Case	AAA	AEA	SEA	AAA	AEA	SEA
No. of Cases	206	95	39	208	96	39
F.S.	1.00	1.00	1.00	1.00	1.00	1.00
Actual F.S.*	1.37	1.44	1.15	0.92	1.00	0.90
Risk (%)	22.3	16.8	41.0	67.8	60.4	41.0
F.S.	1.20	1.20	1.20	1.20	1.20	1.20
Actual F.S.*	1.64	1.73	1.38	1.11	1.20	1.08
Risk (%)	5.8	7.4	25.6	42.3	31.3	25.6
F.S.	1.40	1.40	1.40	1.40	1.40	1.40
Actual F.S.	1.91	2.02	1.61	1.30	1.40	1.26
Risk (%)	2.4	3.2	5.1	22.6	9.4	5.1
F <i>.</i> S.	1.60	1.60	1.60	1.60	1.60	1.60
Actual F.S.*	2.19	2.31	1.84	1.48	1.60	1.44
Risk (%)	1.0	1.1	2.6	14.4	2.1	2.6
F.S.	1.80	1.80	1.80	1.80	1.80	1.80
Actual F.S.*	2.46	2.59	2.07	1.67	1.80	1.62
Risk (%)	0	0	0	7.7	1.0	0
F.S.	2.00	2.00	2.00	2.00	2.00	2.00
Actual F.S.*	2.74	2.88	2.30	1.84	2.00	1.80
Risk (%)	0	0	0	4.8	0	0
F.S.	2.5	2.5	2.5	2.5	2.5	2.5
Actual F.S.*	3.43	3.60	2.88	2.30	2.50	2.25
Risk (%)	0	0	0	0	0	0

Table 19. Factor of safety and associated risk.

Actual F.S. = F.S. x mean OPR

mean OPR = mean Over-Prediction Ratio



Figure 122. Risk analysis of CAPWAP/TEPWAP predictions for 206 PD/LT pile-cases in all types of soil.


Figure 123. Risk analysis of Energy Approach predictions for 208 PD/LT pile-cases in all types of soil.



Figure 124. Risk analysis of CAPWAP/TEPWAP predictions for 95 PD/LT pile-cases in all types of soil at EOD.



Figure 125. Risk analysis of Energy Approach predictions for 96 PD/LT pile-cases in all types of soil at EOD.



Figure 126. Risk analysis of CAPWAP/TEPWAP predictions for 39 small displacement ($A_R > 350$) PD/LT pile-cases in all types of soil at EOD.



Figure 127. Risk analysis of Energy Approach predictions for 39 small displacement ($A_R > 350$) PD/LT pile-cases in all types of soil at EOD.

APPENDIX A - DATA SET PD/LT

Table 20. Site and pile information for PD/LT.

No.	Pile-Case Number	Refer. No.	Location	Pile Type	Pile	Length Beiow	Penetr Depth	Soli	Туре
				.,,,	(in ²)	Gauges (ft)	(11)	Side	Tip
1	FN1-EOD	i-48 0	Omaha NE	HP10x42	12.40	72.0	72_0	sitty clay	10
2	FN1-BOR1	⊢48 0	Omaha NE	HP10x42	12.40	72.0	72.1	silfy clay	ยม
3	FN1-80R2	H480	Omana NE	HP10x42	12.40	72.0	73.0	silty clay	UH
4	FN2-EOD	1480	Omaha NE	PSC12'sq	144.00	62.0	65.0	silty clay	till
5	FN2-BOR	H480	Omaha NE	PSC12*sq	144.00	62.0	65.0	sility clay	tiu
6	FN3-EOD	H480	Omaha NE	PSC14'sq	196.00	62.0	56.0	silty clay	ยม
7	FN3-BOR	H480	Omaha NE	PSC14'sq	196.00	62.0	56 .0	slitty clary	111
8	FN4-EOD	1-480	Omaha NE	CEP12.75	19.20	68 .0	66.0	slity clay	ซม
9	FN4-BOR	1480	Omaha NE	CEP12.75*	19.20	66.0	66.0	silty clay	មា
10	FIA-EOD	Sille 1	kowa	HP14x89	26.10	117.5	114.1	clayey sand	send
11	FA-BOR	Sille 1	kowa	HP1 4x89	26.10	117.5	114.1	clayey sand	send
12	FIB-EOD	Site 1	lowa	CEP 14*	21,20	97.5	94.1	clayey sand	sand
13	FIB-BOR	Site 1	lowa	CEP 14*	21.20	97.5	94.1	clayey sand	sand
14	F01-EOD	Cim S-1	Oidahoma	CEP 28*	67.70	60.3	60.2	silty sand	silty sand
15	FO1-BOR	Cim S-1	Oklahoma	CEP 26*	67.70	60.3	60.2	silty sand	silty sand
16	FO2-EOD	Cim S-1	Oidahoma	PSC24'oct	470.90	61.5	63.0	silty sand	silty sand
17	FO2-BOR	Cim S-1	Oklahoma	PSC24'oct	470.90	61.5	63.1	sitty sand	sitty sand
18	FO3-EOD	Cim S-2	Okiahoma	HP14x117	34.40	110.0	63.7	sa-si-clay	clayey sand
19	FO4-EOD	Cim S-2	Oklahoma	RC24'sq	576.00	60.3	45.0	sa-si-clay	clayey sand
20	FO4-BOR	Cim S-2	Oklahoma	RC24'sq	576.00	60.3	55.8	sa-si-clay	clayey sand
21	FOR1-EOD	Aisea	Oregon	PSC20"sq	393.00	131.0	125.5	sand & silt	siltstone
22	FOR1-BOR	Alsea	Oregon	PSC20*sq	393.00	131.0	125.6	sand & silt	alitstone
23	FM5-E00	Site A	Maine	CEP 18"	27.50	117.3	99.0	clay & sand	sand
24	FM5-BOR	Site A	Maine	CEP 18"	27.50	101.0	99,1	clay & sand	sand
25	FM17-EOD	Site B	Maine	CEP 18*	27.50	77.8	71.1	<u>uu</u>	ยแ
26	FM17-BOR	Site B	Maine	CEP 18"	27.50	77.8	71.3	111	
27	FM23-EOD	Site B	Maine	CEP 18"	27.50	56.8	50.7	till	38
28	FM23-BOR	Site B	Maine	CEP 18"	27.50	56.8	50.8	tin	210
29	FC1-EOD	Crook	Colorado	CEP12.75*	9.82	33.5	33,5	sand	sand
30	FC1-BOR	Crook	Colorado	CEP12.75*	9.82	33.5	33.9	sand	sand
31	FC2-EOD	Crook	Colorado	CEP12.75*	8.82	27.5	26.5	sand	sand

235 Preceding page blank

No.	Pilo-Case Number	Refer.	Location	Plie Type	Pile Area	Length	Penetr	Soil	Туре
	1.00000000000			.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(in ²)	Gauges (ft)	(1)	Side	Тір
32	FC2-BOR	Crook	Colorado	CEP12.75*	9.82	27.5	26.9	sand	sand
33	FMI1-EOD	RL 115	Missouri	CEP 14*	16.10	83,0	83.0	sand-gravel	sand
34	FMI1-BOR	RL 115	Missouri	CEP 14"	16.10	63.0	83.1	sanci-gravel	sand
36	FMI2-EOD	RL 115	Missouri	CEP 14*	16.10	61.5	61.0	sand-gravel	sand
36	FM12-BOR	RL 115	Missouri	CEP 14"	16.10	61.5	61.0	sand-gravel	sand
37	FWA-EOD	3 rd lake	Washingtn	CEP 48"	111.3	152.0	24.8	till-gravel	till
38	FWA-BOR	3 rd lake	Washingtn	CEP 48"	111.3	152.0	24.9	tili-gravel	
39	FWB-EOD	3 nd lake	Washingth	CEP 48"	111.3	140.0	109.0	till-gnavel	tu
40	FWB-BOR	3 rd lake	Washingtn	CEP 48"	111.3	140.0	109.3	tiit-gravel	till
41	FA1-EOD	H1 65	Alabama	PSC 18'sq	324.00	63,0	64.0	silty sand	sitty sand
42	FA1-BOR1	F1 65	Alabama	PSC 18'sq	324.00	63.0	64.5	silty sand	sitty sand
43	FA1-BOR2	F165	Alabama	PSC 18'sq	324.00	63.0	64.8	sitty sand	silty sand
44	FA2-E00	I-1 65	Alabama	PSC 18*sq	324.00	73.0	75.0	slifty sand	aity sand
45	FA2-BOR1	⊦1 65	Alabama	PSC 18'sq	324.00	73.0	75.3	sitty sand	sitty sand
48	FA2-BOR2	⊦165	Alebama	PSC 18'sq	324.00	73.0	75.5	silty sand	sitty sand
47	FA3-EOD	⊦1 66	Alabama	PSC 24'sq	489.00	63.0	64.0	sility sand	slity sand
48	FA3-BOR1	⊦165	Alabama	PSC 24*sq	489.00	63.0	64.1	slity sand	silty sand
49	FA3-BOF2	⊦165	Alabama	PSC 24'sq	489.00	63.0	64.5	silty sand	sitty sand
50	FA4-EOD	⊦165	Alabama	PSC 24'sq	489.00	73.0	75.0	silly sand	sitty sand
51	FA4-BOR1	⊦165	Alabama	PSC 24'sq	489.00	73.0	75.1	silty sand	sitty sand
52	FA4-BOR2	⊦1 65	Alabama	PSC 24'sq	489.00	73.0	75.2	sitty sand	ality sand
53	FA5-EOD	⊦165	Alabama	PSC 36'sq	898.00	70.0	73.0	silty send	silty sand
54	FA5-BOR	⊦165	Alabama	PSC 36*sq	898.00	70.0	73,1	silty sand	sitty sand
55	FV15-EOD	WBJ	Vermont	HP14x73	21.40	92.0	75.0	silt-d.sand	sand gravel
56	FV15-BOR	WRJ	Vermont	HP14x73	21.40	\$2.0	75.8	sill-d.sand	sand gravei
57	FV10-EOD	WBJ	Vermont	HP14x73	21.40	92.0	90.0	slit-d.sand	sand gravel
58	FV10-BOR	WRJ	Vermont	HP14x73	21.40	92.0	\$0.4	slit-d.sand	sand gravel
59	FMN2-EOD	RL 18	Minnesola	HP14x73	21.4	97.0	96.0	sa-si-clay	fat clay
60	FMN2-BOR	R£ 18	Minnesota	HP14x73	21.4	97.0	96.1	sa-si-clay	fat clay
61	FP5-EOD	Tioga	Penn.	Monotube	7.00	34.5	23.6	sandy grvi	sandy grvt
62	FP5-BOR	Tioga	Penn.	Monotube	7.00	34.5	23.8	sandy grvl	sandy grvl

Table 20. Site and pile information for PD/LT (continued).

No.	Pile-Case Number	Refer.	Location	Plie Type	Plie	Length	Penetr	Soil 1	уре
				.,,,,	(In ²)	Gauges (ft)	(ft)	Side	Tip
63	FKG-EOD	RI.27	Kentucky	PSC14*8Q	196.00	72.0	34.7	soft clay	dense
64	FKG-BOR	RL27	Kentucky	PSC14*sq	196.00	72.0	34.7	soft clay	dense
65	FL3-EOD	RL415	Louisiana	PSC24*sq	463.00	100.0	84.3	sitty clay	sitty sand
- 86	FL3-BOR1	Pt.415	Louisiana	PSC24*sq	463.00	100.0	84.3	sifty clay	silly sand
67	FL3-BOR2	FL415	Louisiana	PSC24*sq	463.00	100.0	84.3	sility clay	silty sand
68	CA1-EOD	Site C-L	O.S. Ont	CEP 9.6"	15.42	172.0	154.3	si-sa-ciay	31-60-1 111
69	CA1-BOR	Site C-L	Q.S. Ont	CEP 9.6"	15.42	172.0	154.3	si-sa-ciay	si-se-till
70	CA2-BOR	Site C-L	O.S. Ont	CEP 9.6*	15.42	112.5	110.1	si-sa-ciary	si-sa-ciay
71	CAS-BOR1	Sile A	N.Y. Ont	CEP11.73*	11.98	67,0	83.2	fill-sand	sand
72	CAS-BOR2	Site A	N.Y. Ont	CEP11.73*	11.98	67.0	65.6	fill-sand	sand
73	CA3/8-BOR	Marina	Bar. Ont	CEP10.24*	8.74	73.8	64.4	sand-sit	siit
74	CA24-BOR	Sate D	Tor. Ont	CEP12.75	14.54	38.6	38.6	sand	sand
75	CA8-BORI	Site E	Ham. Ont	CEP12.75*	14.54	60.2	54.0	sa-si-tili	siit-tiit
76	CA8-BOR2	Site E	Ham. Ont	CEP12.75	14.54	60.2	54.0	88-5i-till	s#t-t#l
77	CA8-EOR	Sitte E	Ham. Ont	CEP12.75	13.55	60,2	54.0	88-5i-titi	8117-CN
78	WC3-EOD	White	Florida	PSC24"sq	576.00	48.4	27.3	lad.sand	dense
79	WC3-BOR1	White	Florida	PSC24*sq	576.00	48.4	27.5	isd.sandi	dense
80	WC3-BOR2	White	Fiorida	PSC24*8q	578.00	37.5	27.5	iad.aand	dense
81	WC8-EOD	White	Florida	PSC24*sq	576.00	39.5	28.3	isd.sand	dense
82	WC8-BOR1	White	Florida	PSC24"sq	576.00	39.5	28.5	isd.sandi	dense
63	WC8-BOR2	White	Florida	PSC24*sq	576.00	28.0	27.5	isd.sand	dense
84	WB9-BOR	West	Florida	PSC30"sq	645.50	130.0	128.5	clayey sand	clayey
85	WB15-BOR	West	Florida	PSC30*sq	645.50	105.0	103.6	sand	silt-clay
- 86	T1/A-EOD	offshore	taraei	OEP 60"	212.00	138.5	52.8	cicr sand	sand
87	T1/AALT	offshore	Israel	OEP 60"	212.00	173.9	53.8	cicr sand	sand
88	T1/B-EOD	offshore	Israel	OEP 60"	212.00	216.2	101.7	cicr sand	sand
89	T2/A-EOD	offshore	israel	OEP 48"	111.33	117.1	52.5	cicr sand	8800
90	12/B-EOD	offshore	Israel	OEP 48"	111.33	260.5	182.1	cicr sand	sand
91	35-1-BOR	C.N.R.	Toronto	HP12x74	21.80	60.1	48.5	cl-sa-sit	slity sand
92	35-4-80R	C.N.R.	Toronia	CEP12.75	9.80	52.2	48.2	ci-sa-siti	silty sand
83	35-5-BOR	C.N.R.	Toronio	HP12x74	21.80	100.2	90.5	ci-sa-silit	silty sand

Table 20. Site and pile information for PD/LT (continued).

N 0.	Plie-Case	Refer.	Location	Pile	Plie	Length	Penetr	Soil	Туре
		~~		.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(in ²)	Gauges (ft)	(ft)	Side	Tip
94	35-8-BOR	C.N.R.	Toronio	CEP12.75'	9.80	105.4	90.0	ci-sa-silt	silty sand
96	35-7-BOR	C.N.R.	Toronto	T.Timber	157.00	44.4	41.6	ci-sa-silt	silty sand
96	35-10-BOR	C.N.R.	Toronto	PSC 12'sq	144.00	50.0	48.0	ci-sa-sili	ailty sand
97	E2-BOA	DFI	Raleigh	PSC 12'sq	144.00	43.5	44.5	ci-sa-silt	ci-sa-sit
98	635-BOR	Mahonig	Penn.	HP12:53	15.50	68 .8	66.0	sand-sill	elit
99	LB21-BOR	Site A	NĂ	PSC 20*sq	400.00	36.0	36.0	siit-sand	silt-sand
100	LB20-BOR	Site B	Ň	PSC 20"sq	400.00	51.0	55.0	sand	sand
101	LC3-BOR	Site C	NĂ	PSC 20*sq	400.00	115.0	86.0	ci-sa-sifi	ci-sa-silt
102	LIN16-BOR	Site D	NĂ	PSC 20"sq	400.00	155.0	94.0	CI-88-91#	ci-88-sitt
103	LE37-BOR	Site E	NA	PSC 10"sq	100.00	60.0	50.0	ci-sa-sit	limestone
104	LE64-BOR	Site F	NA	PSC 10*90	100.00	60.0	58.0	ci-sa-sili	88-ci-8itt
105	ST1-EOD	Sile H	Florida	PSC 18'sq	324.00	66.0	44.0	-	carb sand
108	ST2-EOD	Site P	Florida	PSC 18*sq	324.00	62.0	40.0	•	carb sand
107	ST9-BOR	H664	Virginia	PSC 54'sq	770.00	131.0	109.0		silt-clay
108	ST46-EOD	Castletn	New York	CEP 10"	5.80	40.0	38.0	sitt-sand	siit-sand
109	GZA3-EOD	CMc	Prov. RI	CEP13.38	20.30	143.0	125.5	silt-sand	gr-sa-sili
110	GZA5-EOD	Civic	Prov. Al	CEP 9.75*	15.50	138.0	93.8	slit-sand	tiil-shale
111	GZ46-E00	Civic	Prov. RI	CEP 9.75*	15.50	171.0	156.0	silt-sand	gr-sa-sill
112	GZB8C-EOD	Chric	Prov. RI	CEP 10"	18.40	116.0	89.5	siit-sand	sift
113	GZBP2-EOD	Смс	Prov. RI	CEP13.38"	20.30	143.7	106.0	slit-sand	gr-sa-sik
114	GZE8-EOD	CMc	Prov. RI	CEP13.38*	20.30	97.0	92.3	silt-sand	5i-88-1iii
115	GZZ5-EOD	Deer is,	Boston MA	CEP 14*	21.20	87.0	87.0	tili⊢clay	till
116	GZO5-EOD	Deer Is.	Boston MA	CEP 14*	21.20	87.0	54.0	tili-clay	UN
117	GZCC5-EOD	Deer is.	Boston MA	CEP 14"	21.20	117.0	80.0	till-clay	tiir
118	GZL2-EOD	Deer is.	Boston MA	CEP 14	21.20	117.0	83.0	tili-ciay	1UI
119	GZP: 4-EOD	Deer la.	Boston MA	CEP 14*	21.20	105.0	60.5	tili-clay	till
120	GZP11-EOD	Deer is.	Boston MA	CEP 14"	21.20	105.0	5 6.5	til-clay	ti#
121	GZP12-EOD	Deer is.	Boston MA	CEP 14*	21.20	115.5	69.0	tili-clay	415
122	GZB22-EOD	NWS	Colt Neck	OEP 36"	54.00	138.0	118.0	sand-ciay	silt-ciay
123	GZW1-EOR	Water	Vermont	CP12.75	14.60	126.0	99.5	silty sand	sand
124	A54-EOD	HICC	Australia	RC10.6*sq	117.22	67.9	67.6	slity clay	clay

Table 20. Site and pile information for PD/LT (continued).

No.	Pile-Case Number	Refer.	Location	Plie Type	Pile Area	Length Balow	Penetr	Soll Type	
				.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(in ²)	Gauges (ft)	(11)	Side	тър
125	A54-BOR	HICC	Australia	RC10.8*sq	117.22	67.9	67.8	slity clay	clay
128	A147-EOD	HICC	Australia	RC10.8*sq	117.22	67.9	67.6	ality clay	ciay
127	A147-BOR	HICC	Australia	RC10.8"sq	117.22	67.9	67.6	silty clay	clay
128	GF19-EOD	Site 1	Pgh. PA	HP10x42	12.30	58.5	49.5	grvl-snd-sit	shale
129	GF110-EOD	Site 1	Pgh. PA	HP12x74	21.70	57.0	49.7	grvf-and-alt	shale
130	GF222-EOD	Site 2	Pgh. PA	HP12x74	21.70	67.0	61.1	grvi-snd-slt	shale
131	GF224-EOD	Site 2	Pgh. PA	Monotube	9.70	53.0	29.6	grvi-and-sit	grvi-and-sit
132	GF312-EOD	Site 3	Pgh. PA	HP12x74	21.70	33.0	28.2	and-givi-shi	shale
133	GF313-EOD	Sile 3	Pgh. PA	HP10x57	16.70	35.0	31.5	and-grvf-shi	claysione
134	GF412-EOD	Site 4	Pgh. PA	HP12x74	21.70	48.5	33.6	grvl-snd-sil	claystone
135	GF413-EOD	Site 4	Pgh. PA	HP10x57	16.70	34.2	34.6	grvl-and-sit	claystone
136	GF414-EOD	Ste 4	Pgh, PA	HP10x57	16.70	47.5	34.7	grvi-and-sit	clayatone
137	GF415-EOD	Site 4	Pgh. PA	HP12x74	21.70	47.5	34.1	grvi-and-sit	claystone
138	EF62-EOD	Ottawa	Canada	CP 9.625*	15.54	-	62.3	si-se- ci	till
139	EF167-BOR	Otlawa	Canada	CP 9.625"	15.54	•	68.9	8 -58- Cl	tui
140	A3-EOD1	Apalach	Florida	VC 24"sq	462.90	94.0	63.4	clayey sand	84nd
141	A3-BORI	Apalach	Rorida	VC 24"8q	462.90	94.0	63.4	clayey sand	sand
142	A3-EOD2	Apalach	Florida	VC 24"sq	462.90	94.0	90.3	ciayey sand	sand
143	A3-BOR2	Apalach	Porida	VC 24"sq	462.90	94.0	90.4	clayey sand	sand
144	A3-80R3	Apelach	Florida	VC 24"8q	462,90	89.3	90.6	clayey sand	clayey sand
145	A14-DD1	Apalach	Florida	VC 24"sq	462.90	107.0	45.0	sandy clay	sand
148	A14-002	Apalach	Florida	VC 24"sq	462.90	107.0	47.0	sandy clay	sand
147	A14-BOR1	Apalach	Florida	VC 24*8q	462.90	107.0	58.5	clayey sand	sand
148	A14-BOR2	Apalach	Florida	VC 24"sq	462.90	75.0	58,8	clayey sand	cand
149	A25-EOD	Apalach	Florida	VC 24"sq	462,90	106.0	55.1	clayey sand	uand
150	A25-BOR1	Apalach	Florida	VC 24"sq	462.90	106.0	55.2	clayey sand	Send
151	A25-B0R2	Apalach	Florida	VC 24*8q	462.90	59.3	55.4	clayey sand	Sand
152	A25-BOR3	Apalach	Fiorida	VC 24"8q	462.90	59.3	55.5	clayey sand	send
153	A18-EOD	Apalach	Florida	PSC18'sq	324.00	65.0	60.6	sandy clay	sand
154	A15-BOR1	Apalach	Florida	PSC18'sq	324.00	65.0	60.6	sandy clay	sand
155	A18-BOR2	Apalach	Rorida	PSC18'sq	324.00	62.2	61.0	sandy clay	sand

Table 20. Site and pile information for PD/LT (continued).

No.	Plie-Case Number	Refer.	Location	Pile Type	Plie	Length Below	Penetr	Soil	Туре
				.,,	(in ²)	Gauges (11)	(11)	Side	Тір
158	A41-EOD	Apalach	Florida	VC 24"sq	462.90	91.0	52.0	clay	sand
157	A41-BOR1	Apalach	Fiorida	VC 24"sq	462.90	91.0	52.0	clay	sand
158	A41-BOR2	Apalach	Florida	VC 24"sq	462.90	81.5	52.8	clay	sand
159	A101-EOD	Apalach	Florida	VC 24*sq	462.90	88.0	61.8	clay	clayey sand
160	A101-BOR1	Apalach	Florida	VC 24"sq	462.90	88.0	61.8	clay	clayey sand
161	A101-BOR2	Apalach	Fiorida	VC 24*sq	462.90	71.5	62.1	clay	clayey sand
162	A133-EOD	Apalach	Florida	VC 24°sq	462.90	130.0	103.9	ciayey sand	sandy clay
163	A133-BOR	Apalach	Florida	VC 24*sq	462.90	115.7	104.9	clayey sand	sandy clay
164	A145-EOD	Apalach	Fiorida	VC 24"sq	462.90	132.0	102.9	clayey sand	sand
185	A145-BOR1	Apalach	Florida	VC 24 °s q	462.90	132.0	102.9	clayey sand	sand
166	A145-BOR2	Apalach	Fiorida	VC 24*sq	462.90	115.1	103.0	ciayey sand	sand
167	CB3-BOR	Choctw	Fiorida	PSC24*sq	576.00	77.9	77.0	clayey sand	sand
168	CB3-BORL	Choctw	Florida	PSC24*sq	576.00	79.9	77.8	clayey sand	sand
169	CB5-BOR	Choctw	Fiorida	VC 30*sq	645.53	87.0	53.1	clayey sand	sand
170	CB5-BORL	Choctw	Florida	VC 30*sq	845.53	61.1	54.0	clayey sand	sandy clay
171	CB11-BORL	Choctw	Fiorida	VC 30"sq	645.53	97.6	85.7	clayey sand	clayey sand
172	CB11-EORL	Choctw	Florida	VC 30"sq	645.53	97.6	85.8	clayey sand	clayey sand
173	CB17-BOR1	Choctw	Fiorida	VC 30*sq	645.53	97.0	77.7	clayey sand	clayey sand
174	CB17-BOR2	Choctw	Florida	VC 30*sq	645.53	97.0	77.8	clayey sand	clayey sand
175	CB17-BORL	Choctw	Florida	VC 30*sq	645.53	90.0	77.9	clayey sand	clayey sand
176	CB17-DRL	Choctw	Fiorida	VC 30"sq	645.53	90.0	7 8 .2	ciayay sand	clayey sand
177	CB23-BOR	Choctw	Fiorida	VC 30"sq	645.53	96.0	80.3	clayey sand	sand
178	CB23-BORL	Choctw	Florida	VC 30*sq	645.53	96.0	82.7	clayey sand	sand
179	CB29-BORL	Choctw	Fiorida	VC 30"sq	645.53	95 .1	84 .5	clayey sand	clayey sand
180	CB29-EORL	Choctw	Florida	VC 30*sq	645.53	95 .1	84.5	ciayey sand	clayey sand
181	CB35-BOR1	Choctw	Fiorida	VC 30*sq	645.53	97 .1	78.5	clayey sand	clayey sand
182	CB35-BOR2	Choctw	Florida	VC 30*sq	645.53	97 .1	78.9	clayey sand	clayey sand
183	CB35-BORL	Choctw	Florida	VC 30°sq	645.53	89.1	79.1	clayey sand	clayey sand
184	CB41-EOR	Choctw	Florida	VC 30*sq	645.53	102.3	64.7	sandy clay	sandy clay
185	CB41-BOR	Choctw	Florida	VC 30"sq	645.53	101.3	64.7	sandy clay	sandy clay
186	CB41-BORL	Choctw	Florida	VC 30*sq	645.53	79.0	65.4	sandy ciay	sandy clay

Table 20. Site and pile information for PD/LT (continued).

NO.	Pile-Case Number	Refer. No.	Location	Pile Type	Pile Area	Length Below	Penetr Deoth	Soll	Туре
				-1F-	(in ²)	Gauges (ft)	(11)	Side	Тір
187	CB26-EOD	Choctw	Florida	PSC24"8q	578.00	80.1	62.5	clayey sand	sand
188	CB26-BOR	Choctw	Fiorida	PSC24"3q	576.00	80.1	62.6	clayey sand	sand
189	C826-EOR	Choctw	Florida	PSC24"8q	576.00	80.1	64.8	clayey sand	sandy clay
190	CB26-BOR2	Choctw	Fiorida	PSC24"sq	576.00	65.0	65.0	sandy clay	sandy clay
191	33P1-EOD	Site P	Ontarlo	HP 12x74	21.60	120.9	114.4	ci-sa-silt	sitty send
192	33P1-BOR	Site P	Ontario	HP 12x74	21.80	120.9	114.4	ci-sa-si#	silly send
193	33P1-EOR	Site P	Ontario	HP 12x74	21.80	120.9	114.4	ci-sa-siit	silty sand
194	33P2-EOD	Sile P	Ontario	CP 12.75	9.80	149.6	107.2	ci-sa-silt	slity sand
195	33P2-80R	Site P	Oniario	CP 12.75*	9.60	111.0	107.2	ci-sa-sili	sifty sand
196	33P2-EOR	Site P	Ontario	CP 12.75'	9.80	111.0	107.2	ci-sa-sitt	sility sand
197	33P4-EOD	Sile P	Ontario	PSC 12*sq	144.00	55.0	54.2	ci-sa-sili	ci-sill-till
198	33P5-EOD	Site P	Ontario	≠14 Timber	144.9	43.0	28.4	ci-sa-silt	ci-siit-tii
199	TRD22-EOD	Site R	Ontario	HP 12x74	21.80	22.5	20.1	sand	tii
200	TRO22-BOR	Stie R	Ontario	HP 12x74	21.80	22.5	20.1	sand	till
201	TRE22-EOD	Sale R	Ontario	HP 12x74	21.80	30.0	25.7	sand	rock
202	TRE22-BOR	Sile R	Ontarlo	HP 12x74	21.80	30.0	25.7	sand	rock
203	TRP5X-EOD	Sille R	Ontario	HP 12x53	15.60	25.0	25.2	sand	rock
204	TRP5X-BOR	Sille R	Ontario	HP 12x53	15.60	25.0	25.2	sand	rock
205	TR131-BOR	Site R	Ontario	CP 7.063*	7.90	26.8	NA	sand	rock
206	TRAH-EOR	Sile S	Brunswick	HP 12x89	26.50	138.0	126.0	clayey siti	sandy gravel
207	TRBH-BOR	Site S	Brunswick	HP 12x89	26.50	114.3	102.1	clayey silt	sandy gravel
208	TR8P-EOR	Site S	Brunswick	CP 12.75'	12.40	110.0	104.0	clayey sill	sandy gravel

Table 20. Site and pile information for PD/LT (continued).

1 in = 25.4 mm $1 \text{ in}^2 = 645.2 \text{ mm}^2$ 1 ft = 0.305 m

NG.	Pile-Case Number	Hammer Type	Rated Hammer	Delivered Energy	Blow Count	impedence EA/C	Vimp	F _{imp}	<u>VEA/C</u> F	Dmax
			Energy (Mp-fi)	(kip-ft)	(8PI)	(ldps/ft/s)	(ft/s)	(kips)		(in)
1	FN1-EOD	D-30	54.2	17.30	2.83	22.13	13.24	322.4	0.909	.793
2	FN1-BOR1	0-30	54.2	18.42	8.00	22.13	13.24	315.2	0.930	.813
3	FN1-BOP2	D-30	54.2	20.15	15.00	22.13	13.04	308.9	0.934	.837
4	FN2-EOD	D-30	54.2	12.70	3.50	60.49	7.35	462.0	0.963	.444
5	FN2-BOR	D-30	54.2	12.35	5.00	60.49	8.36	508.9	0.994	.409
6	FN3-EOD	D-30	54.2	9.90	9.17	85.89	6.14	558.0	0.943	.386
7	FN3-BOR	D-30	54.2	16.20	6.00	85.89	7.88	691,1	0.979	.459
8	FN4-EOD	D-30	54.2	15.55	2.50	34.26	12.97	478.5	0.929	.531
G.	FN4-BOR	D-30	54.2	17.40	5.00	34.26	13.23	475.8	0.953	.517
10	FIA-EOD	K-25	51.5	23.38	3,33	46.60	14.90	667.8	1.039	.686
11	FIA-BOR	K-25	51.5	18.88	1.83	46.60	15.00	657.6	1.062	.558
12	FIB-EOD	K-25	51.5	25.40	5 .83	37,80	15.10	555.3	1.027	.689
13	FIB-BOR	K-25	51.5	22.47	2.50	37.80	15.30	548.2	1.054	.675
14	FO1-EOD	DE110	93 .5	18.06	5.67	120.80	5.40	788.7	0.827	.413
15	FO1-BOR	DE110	93.5	37.47	5.00	120.80	8.90	1223.6	0.873	.541
16	FO2-EOD	DE110	93.5	18.26	5.08	199.10	3.90	805.9	0.963	.453
17	FO2-BOR	DE110	93.5	31.37	12.00	199.10	5.70	1114.0	1.018	.450
18	FO3-EOD	DE110	93.5	16.40	18.67	61.40	8.60	489.5	0.828	.625
19	FO4-EOD	DE110	93.5	9.81	11.67	214.60	2.50	575.3	0.933	.269
20	FO4-BOR	DE110	93.5	22.73	1.00	214.60	4.70	1011.7	0.997	.362
21	FOR1-EOD	D-46-23	105	30.11	9.17	159.00	6.10	946.0	1.025	.838
22	FOR1-BOR	D-48-23	105	23.77	77.33	159.00	5.80	918.1	1.004	.462
23	FM5-EOD	K-45	92.6	27.00	1.29	49.08	10.73	550.5	0.957	1.033
24	FM5-BOR1	K-45	92.8	40.20	3.00	49.08	13.30	658.2	0.992	.981
25	FM17-EOD	K-45	92.8	39.50	1.42	49.08	11.14	590.8	0.926	1.105
28	FM17-BOR	K-45	92.6	36.50	3.00	49.08	13.17	697.7	0.926	.788
27	FM23-E0D	K-45	82.6	33.30	1.33	49.08	11.37	559.0	0.998	1.190
28	FM23-BOR	K-45	92.8	31.00	2.00	49.08	10.57	508.1	1.021	1.247
29	FC1-EOD	KC-25	51.5	15.47	3.50	17.54	13.40	272.8	0.862	.799
30	FC1-BOR	KC-25	51.5	16.19	3.80*	17.54	13.40	276.2	0.851	.806
31	FC2-EOD	KC-25	51.5	18.07	3.67*	17.54	14.90	290.0	0.901	.606

Table 21. Pile driving and dynamic measurements for PD/LT.

* - Denotes blow count (BPI) based on blows per foot.

No.	Pile-Case Number	Hammer Type	Rated Hammer	Delivered Energy	Biow Count	Impedence EA/C	V _{imp}	Fimp	<u>YEA/C</u> F	Dmax
			Energy (kip-ft)	(idp-fi)	(BPI)	(kips/ft/s)	(fl/s)	(idps)		(in)
33	FC2-BOR	KC-25	51.5	13.66	4.00*	17.54	13.10	268.1	0.857	.630
33	FMI1-EOD	ICE-640	40.0	11.00	3.00	28.73	8.10	250.3	0.919	.738
34	FMI1-BOR	ICE-840	40.0	12.00	3.00	28.73	8.80	304.5	0.830	.632
35	FMI2-EOD	ICE-640	40.0	11. 66	1.42*	28.73	7.10	230.1	0.686	.861
36	FM12-BOR	ICE-640	40.0	13.58	3.00	28.73	9.00	285.7	0.905	.834
37	FWA-EOD	Con300	90.0	44.90	47.00	198.62	9.80	1925.	1.011	.920
38	FWA-BOR	Con300	90.0	33.30	7.00	198.62	8.80	1708.	1.023	.550
39	FWB-EOD	Con300	90.0	47.20	30.00	198.62	8.50	1715.	0.990	.630
40	FWB-BOR	Con300	90.0	39.30	15.00	198.62	7.60	1516.	0.996	.670
41	FA1-EOD	K-45	92.8	17.53	1.50*	145.72	4.37	628.7	1.013	.727
42	FA1-BOR1	K-45	92.8	9.19	7.00	145.72	3.60	547.9	0.957	.335
43	FA1-BOR2	K-45	92.8	21.84	7.00	145.72	7.30	1074.	0.990	.481
44	FA2-EOD	K-45	92.8	21.22	3.50*	145.72	3.98	639.0	0.908	.611
45	FA2-BOR1	K-45	92.8	22.67	7.00	140.52	6.90	1024.	0.946	.430
46	FA2-BOP2	K-45	92.8	20.80	5.00	145.72	6.70	1025.	0.952	.357
47	FA3-EOD	K-45	92.8	22.79	2.83	221.53	3.31	729.1	1.006	.646
48	FA3-BOR1	K-45	92.8	15.22	6.00	219.99	3.41	782.4	0.959	.324
49	FA3-BOR2	K-45	92.8	18.31	5.00	221.53	5.30	1199.	0.979	.274
50	FA4-EOD	K-45	92.8	19.08	8.42*	221.53	3.5 6	782.4	1.007	.437
51	FA4-BOR1	K-45	92.8	16.82	8.00	219.93	5.19	1108.	1.030	.255
52	FA4-BOR2	K-45	92.8	20.42	18.00	219.93	6.76	1468.	1.013	.283
53	FA5-EOD	D-62-22	153.2	37.08	7.67*	403.88	5.10	2106.	0.978	.446
54	FAS-BOR	D-62-22	153.2	45.51	5.00	403.88	7.40	3016.	0.991	.288
56	FV15-E0D	MKT-358	22.0	10.00	4.17 ⁴	38.19	10.40	403.2	0.965	.475
56	FV15-BOR	MKT-358	22.0	12.23	9.00	38.19	14.20	522.8	1.037	.643
57	FV10-EOD	MKT-358	22.0	10. 98	2.87*	38.19	10.70	417.8	0.978	.490
58	FV10-BOR	MKT-35B	22.0	13.98	2.00	38.19	16.30	609.1	1.022	.675
59	FMN2-EOD	ICE-90S	90.0	28.29	1.83*	38.19	15.00	627.6	0.913	.879
60	FMN2-BOR	ICE-90S	90.0	29.14	25.00	38.19	16.50	675.6	0.932	.802
61	FP5-EOD	D-12	22.0	7.58	5.42	12.49	13.80	177.4	0.972	.677
82	FP5-BOR	D-12	22.0	7.55	13.00	12.49	14.90	192.5	0.967	.570

Table 21. Pile driving and dynamic measurements for PD/LT (continued).

' - Denotes blow count (BPI) based on blows per foot.

NO.	Plie-Case Number	Hammer Type	Rated Hammer	Delivered Energy	Blow Count	Impedence EA/C	V _{imp}	Fimp	<u>VEA/Ç</u> F	Dmax
			Energy (kip-fl)	(kip-ft)	(8PI)	(ldps/ft/s)	(TT/S)	(idps)		(in)
63	FKG-EOD	1.8-520	31.0	8,31	23.25	80.23	4.28	353.5	0.971	.466
64	FKG-BOR	LB-520	31.0	7.76	18.00	80.23	4.55	373.3	0.978	.407
65	FL3-EOD	Vul-020	60.0	14.60	1.67 [*]	203.74	3.28	678.3	0.985	.757
- 66	FL3-BOR1	Vu i-02 0	60.0	17.03	4.00	203.74	3.70	811.4	0.929	.433
67	FL3-BOR2	Vul-020	60.0	14.43	11.00	203.74	3.70	795.4	0.948	.297
68	CA1-EOD	B-400	48.0	20.32	21.33	27.87	15.75	432.2	1. 016	1.051
69	CA1-BOR	B-400	46.0	18.96	40.00	27.87	14.44	427.9	0.941	1.025
70	CA2-BOR	B-400	48.0	16.74	14.00	27.87	15.09	424.1	0.992	.863
71	CAS-BOR1	35kdrop	38.7min	30.48	25.00	21.38	15.09	341.8	0.944	1.312
72	CA5-BOR2	49kdrop	54.2min	31.44	11.00	21.38	13.45	307.3	0.936	1.298
73	CA3/8-BOR	ICE 40S	40.0	19.03	4.23	15.60	15.42	275.6	0.873	1.001
74	CA24-BOR	D-12	24.0	8.83	50.00	13.44	14,11	215.7	0.879	.500
75	CAB-BOR1	D-30-13	66.0	41.23	10.00	25.93	17.33	494.4	0.909	1.185
78	CA6-BOR2	D-30-13	66.0	42.68	6.67	25.93	17.91	502.7	0.924	1.230
77	CA6-EOR	D-30-13	66.0	37.60	8.00	25.93	16.83	\$17.7	1.045	1.156
78	WC3-EOD	Delmag	106.0	17.50	9.33	268.75	4.23	1122.6	1.013	.452
79	WC3-BOR1	Delmag	105.0	16.90	9.33	268.75	4.25	1179.6	0.968	.412
80	WC3-BOR2	Delmag	105.0	17.89	6.67	268.75	3.47	1042.4	0.895	.403
81	WC8-EOD	Deimag	105.0	17.60	5.00	265.88	4.47	1191.0	0.998	.508
82	WC8-BOR1	Delmag	105.0	18.24	8.00	265.88	4.50	1224.8	0.977	.489
63	WC8-BOR2	Delmag	105.0	26.28	6.67	265.88	5.04	1330.1	1.007	.667
84	WB9-BOR	Con300	90.0	39.87	6.67	271.52	6.42	1748.2	0.997	.383
85	WB15-BOR	Con300	90.0	34.70	5.00	269.50	5.84	1533.7	1.025	.375
86	T1/A-EOD	D-55	125.0	44.99	7.37	378.57	8.01	2967	1.022	.260
87	T1/A-ALT	D-55	125.0	151.50	2.29	378.57	12.50	4423	1.070	.830
88	T1/B-EOD	M-2500	NA	172.73	2.03	378.57	12.70	4787	1.004	.870
89	T2/A-EOD	D-55	125.0	\$ 0.62	4.83	198.75	9.40	1891	0.968	.570
90	T2/B-EOD	M-2500	NA	168.68	5.08	198.75	13.60	2814	0.961	1.169
91	35-1-BOR	B-400	46.0	13.10	1.83.	38.93	10.20	423.0	0.939	.600
92	35-4-BOR	B-400	46.0	23.20	5.68*	17.50	17.94	377.0	0.833	1.010
93	36-5-BOR	B-400	46,0	17.70	10.31	38.93	14.50	584.0	0.967	.590

Table 21. Pile driving and dynamic measurements for PD/LT (continued).

*- Denotes blow count (BPI) based on blows per foot.

NO.	Pile-Case Number	Hammer Type	Raled Hammer	Delivered Energy	Biow Count	Impedence EA/C	V _{kmp}	Fimp	<u>VEA/C</u> F	D _{maax}
			(kip-ft)	(kip-fi)	(BPI)	(kips/ft/s)	(ft/s)	(kips)	_	(ln)
94	36-6-BOR	B-400	46.0	26.10	16.67°	17.50	18,60	372.0	0.875	1.130
95	35-7-BOR	8-225	29.0	9.90	2.53	18.97	10.70	217.0	0.935	.910
96	35-10-BOR	B-400	48.0	11.10	6.00	60.56	9.10	522.0	1.055	.460
97	E2-BOR	Conm65	26.5	15.01	10.00	64.20	7.96	527.7	0.966	.392
96	63S-80R	ICE-640	40.0	12.13	4.50	30.34	10.85	327.1	1.006	.597
99	LB21-BOR	VUL-510	50.0	13.47	4.00*	169.40	4.60	821.7	0.948	.373
100	LB20-BOR	VUL-510	50.0	14.77	8.00	161.80	5.90	966.0	0.988	.311
101	LC3-BOR	D-48-23	107.0	39.40	7.00	181.90	8.30	1437.0	0.935	.666
102	LIN16-BOR	D-48-23	107.0	28.00	10.00*	161.90	5.90	1087.0	0.879	.579
103	LE37-BOR	VUL-01	15.0	5.40	10.00	38.80	5.74	226.3	0.977	.434
104	LE64-BOR	VUL-01	15.0	6.90	5.50	38.80	6.10	247.2	0.957	.422
105	ST1-EOD	D-36-13	84.0	33.13	2.42*	123.00	8.30	1035.4	0.986	.848
106	ST2-EOD	D-36-13	84.0	33.03	3.42	132.20	8.33	861.1	0.973	.899
107	ST9-BOR	CN5300	150.0	45.70	7.86	339.10	6.10	2076.0	0.996	.442
108	ST46-EOD	VUL-1	15.0	5.50	2.67*	10.35	10.00	102.5	1.010	.790
109	GZA3-EOD	ICE-840	40.0	16.12	20.00	36.20	10.80	362.1	1.079	.864
110	GZA5-EOD	ICE-840	40.0	17.36	6.00	27.80	10.20	300.9	0.942	1.062
111	GZA6-EOD	ICE-840	40.0	13.40	15.00	27.70	8.10	219.8	1.021	1.076
112	GZBBC-EOD	ICE-640	40.C	17.67	20.00	37.80	8.90	362.2	0.929	.886
113	GZBP2-EOD	ICE-640	40.0	9.57	20.00	36.20	6.80	258.7	0.952	.710
114	GZB6-EOD	ICE-640	40.0	15.91	11.00	27.70	10.90	344.5	0.878	.834
115	GZZS-EOD	ICE1070	72.6	28.73	4.20	37.80	13.30	533.8	0.942	1.000
118	GZOS-EOD	ICE1070	72.6	23.71	4.20	37.80	14.20	568.1	0.945	.878
117	G2CC5-EOD	ICE1070	72.8	34.05	5.40	37.80	14.60	590.6	0.934	1.180
118	GZL2-EOD	ICE1070	72.6	25.81	9.00	37.80	12.90	500.6	0.974	.984
119	GZP14-EOD	ICE1070	72.6	25.68	5.00	37.60	11.40	502.3	0.858	.882
120	GZP11-EOD	ICE1070	72.8	16.13	5.30	37.80	11,20	471.5	0.898	.782
121	GZP12-EOD	ICE1070	72.6	34.64	12.60	37.80	12.70	499.5	0.961	1.155
122	GZB22-EOD	MH728	135.0	55.17	8.50	111.00	11.60	1326.1	0.971	.859
123	GZW1-EOR	K-25	47.0	12.79	12.00	26.10	12.21	339.3	0.939	.776
124	A54-EOD	Banut-6	34.72	21.05	3.63	50.82	8.73	407.0	1.090	.862

Table 21. Pile driving and dynamic measurements for PD/LT (continued).

* - Denotes blow count (BPI) based on blows per foot.

NO.	Pile-Case Number	Hammer Type	Rated Hammer	Delivered Energy	Blow Count	impedence EA/C	Vimp	Fimp	<u>VEA/C</u> F	0 _{mex}
			(kdp-ft)	(kip-ft)	(BPI)	(kips/fi/s)	(fi/s)	(kips)		(in)
125	A54-BOR	Banut	34.72	25.67	18.14	50.82	10.40	491.0	1.077	.827
126	A147-EOD	Banut	34.72	19.69	1.95*	48.44	8.14	404.2	0.976	.860
127	A147-BOR	Banut	34.72	25.30	6.68*	47.20	9.14	437.3	0.987	.780
128	GF19-EOD	LB-520	NA	9.40	20.00	29.80	10.55	342.8	0.918	.470
129	GF110-EOD	LB-520	NA	10.18	4 4 .00	36,73	10.43	449.0	0.900	.380
130	GF222-E00	ICE-640	NA	16.60	20.00	38.73	12.59	503.0	0.969	.590
131	GF224-EOD	ICE-640	NA	21.00	5.00	17,31	15.70	258.9	1.050	.900
132	GF312-EOD	LB-520	NĂ	6.86	18.00	38.73	9.16	396.1	0.899	.285
133	GF313-EOD	LB-520	NA	10.05	20.00	29.80	10.81	352.1	0.915	.403
134	GF412-EO0	LB-520	NA	8.49	39.00	38.73	9.61	408.8	0.91 1	.35 9
135	GF413-EOD	LB-520	NA	9.07	39.00	29.80	10.74	360.4	0.888	.418
136	GF414-EOD	ICE-640	NA	16.47	48.00	29.80	11.32	372.0	0.907	.607
137	GF415-EOD	ICE-640	NA	12.25	28.00	38.73	10.39	442.8	0.909	.455
138	EF62-EOD	D30-32	52.0	27.29	6.10	27.95	17.01	532.8	0.886	.866
139	EF167-BOR	030-32	52.0	25.81	6.10	27.99	15.29	479.8	0.892	.827
140	A3-EOD1	Vul-020	60.0	18.74	6.00	209.66	3.52	743.8	0.992	.548
141	A3-BOR1	Vul-020	60.0	17.36	7.00	209.66	2.87	622.5	0.967	.488
142	A3-EOD2	Vul-020	60.0	18.85	3.42	209.66	3.40	788.8	0.904	.538
143	A3-BOR2	Vul-020	60.0	16.87	4.00	209.66	3.09	870.8	0.966	.412
144	A3-BOR3	Vul-020	60.0	21.93	30.00	209.66	3.63	753.4	1.010	.337
145	A14-DD1	Con-300	90.0	29.81	8.75	291.07	3.52	1028.3	0.996	.614
146	A14-DD2	Con-300	90.0	30.91	10.83	291.07	4.33	1216.8	1.034	.597
147	A14-BOR1	Con-300	©0.0	40.87	3.00	291.07	6.26	1679.8	1.085	.544
148	A14-BOR2	Con-300	90.0	22.63	20.00	291.07	3.16	962.9	0.955	.318
149	A25-E00	Vul-020	60.0	22.52	4.00	207.40	3.61	726.3	1.031	.735
150	A25-BOR1	Vul-020	60.0	19.06	8.00	207.40	3.12	651.2	0.994	.563
151	A25-B0R2	Vul-020	60.0	22.20	20.00	207.40	3.82	767.2	1.033	.499
152	A25-BOR3	Vul-020	60.0	22.13	20.00	207.40	3.78	753.8	1.040	.521
153	A16-EOD	Vul-010	32.5	11.52	3.17	150.55	3.98	571.0	1.049	.598
154	A16-BOR1	Vul-010	32.5	10.78	6.00	150.55	3.60	534.0	1.015	.457
155	A18-BORZ	Vul-010	32.5	9.04	7.67	150.55	3,19	457.0	1.051	.299

Table 21. Pile driving and dynamic measurements for PD/LT (continued).

'- Denotes blow count (BPI) based on blows per foot.

No.	Plie-Case Number	Hammer Type	Raied Hammer	Delivered Energy	Blow Count	impedence EA/C	V _{imp}	Fimp	VEA/Ç F	D _{max}
			Energy (kip-ft)	(Idp-fi)	(8PI)	(kips/ft/s)	(1/1)	(kips)		(Լո)
150	A41-EOD	Vul-002	80.0	21.94	4.16*	205.97	3.74	788.6	0.977	.604
157	A41-BOR1	Vul-020	60.0	25.41	5.00	205.97	4.31	828.1	1.072	.853
158	A41-BOR2	Vul-020	60.0	21.62	8.00	205.97	4.22	865.8	1.004	.497
159	A101-EOD	Vul-020	60.0	20.95	2.91	208.34	3.68	704.4	1.088	.718
160	A101-BOR1	Vui-020	60.0	21.20	6.00	208.34	3.89	744.3	1.089	.538
181	A101-BOR2	Vui-020	60.0	14.74	24.00	208.34	3.08	643.9	0.997	.360
162	A133-EOD	Vui-020	60.0	18.04	5.25	212.62	4.29	832.0	1.096	.654
163	A133-BOR	Vui-020	80.0	15.42	60.00	212.62	3.47	756.1	0.976	.354
164	A145-EOD	Vul-020	80.0	18.67	5.25	212.71	3.72	771.0	1.026	.626
165	A145-BOR1	Vul-020	60.0	17.50	13.00	212.71	3.01	652.0	0.962	.487
166	A145-BOR2	Vul-020	60.0	16.52	48.00	212.71	3.59	748.6	1.020	.411
167	CB3-BOR	Vul-020	60.0	16.55	10.00	255.08	3.08	766.8	1.025	.306
168	CB3-BOPL	Vu⊢020	60.0	15.85	10.00	255.08	3.37	808.4	1.063	.281
169	CB5-BOA	ICE200S	100.0	15.34	12.00	291.95	3.31	897.6	1.077	.282
170	CB5-BORL	ICE200S	100.0	24.97	16.00	291.95	4.25	1289.2	0.962	.451
171	CB11-BORL	ICE200S	100.0	28.36	18.00	318.10	5.10	1634.8	0.992	.322
172	CB11-EORL	ICE200S	100.0	29.12	16.00	318.10	5.07	1630.5	0.989	.320
173	CB17-BOR1	ICE200S	100.0	29.19	16.00	297.16	4.95	1483.6	0.991	.332
174	C817-BOR2	ICE200S	100.0	36.58	15.33	297.16	6.15	1822.5	1.003	.416
175	CB17-BORL	ICE200S	100.0	20.50	36.00	297.16	4.21	1256.9	0.995	.300
176	CB17-DRL	ICE200S	100.0	26.85	16.50	297.16	4.64	151 8.1	0.908	.332
177	CB23-BOR	ICE200S	100.0	14.07	8.00	309.73	2.64	842.0	0.971	.266
178	CB23-BORL	ICE200S	100.0	22.88	12.00	309.73	4.45	1407.5	0.979	.337
179	C829-BORL	ICE200S	100.0	8.89	26.00	288.28	211	639.3	0.961	.211
180	CB29-EORL	ICE200S	100.0	16.92	20.00	288.28	3.51	1018.4	0.994	.330
181	CB36-BOR1	ICE200S	100.0	31.33	8.73 [°]	293.99	5.14	1394.6	1.064	.637
182	CB35-BOR2	ICE200S	100.0	22.70	20.00	293.99	4.78	1340.4	1.048	.333
183	CB35-BORL	ICE200\$	100.0	19.60	13.00	293.99	4.19	1232.9	0.999	.288
184	CB41-EOR	ICE200S	100.0	32.18	15.17	302.20	5.03	1503.8	1.011	.558
185	CB41-BOR	ICE200S	100.0	27.09	24.00	302.20	5.23	1555.1	1.016	.489
196	CB41-BORL	ICE200S	100.0	21.50	8.70	302.20	4.62	1450.5	0.963	.329

Table 21. Pile driving and dynamic measurements for PD/LT (continued).

' - Denotes blow count (BPI) based on blows per foot.

NO.	Pile-Case Number	Hammer Type	Raied Hammer Foerov	Delivered Energy	Blow Count	Impedence EA/C		Fimp	<u>VEA/C</u> F	D _{max}
			(kip-ft)	(kip-ft)	(BPI)	(kips/ft/s)	(ît/s)	(klps)		(in)
187	C826-EOD	Vul-020	60.0	15.53	4.75	261.84	2.87	754.2	0.996	.461
188	CB26-BOR	Vul-020	60.0	22.67	5.45	261.84	3.62	947,4	1.001	.507
189	CB26-EOR	Vul-020	60.0	25.40	10.00	261.84	3.83	1034.1	0.970	.537
190	CB26-BOR2	Vul-020	60.0	20.93	12.00	261.84	3.50	937.9	0.977	.368
191	33P1-EOD	B-400	46.0	32.67	12.00	38.90	15.38	615.4	0.972	1.110
192	33P1-BOR	B-400	46.0	31.60	16.00	38.90	15.78	637.4	0.963	.787
193	33PI-EOR	B-400	46.0	32.50	no set	38.90	16.60	656.0	0.984	.845
194	33P2-EOD	B-400	46.0	32.84	39.00	17.54	18.44	281.3	1.025	1.859
195	33P2-BOR	B-400	46.0	30.97	76.00	17.54	16.66	317.0	0.922	1.418
198	33P2-EOR	8-400	46.0	31.24	no sel	17.54	17.28	337.6	0.885	1.374
197	33P4-EOD	6-400	46.0	24.47	5.00	65.66	10.92	789.3	0.964	.685
198	33P5-EOD	8-225	29.0	8,41	10.67	24.56	8.75	240.9	0.892	.527
199	TRD22-EOD	D-12	22.5	9.78	30.00	38.70	10.50	394.7	1.005	.391
200	TR022-BOR	D-12	22.5	7.83	20.00	38.70	8.62	410.3	0.886	.323
201	TRE22-EOD	D-22	40.0	15.19	22.00	38.70	13.33	502.5	1.003	.461
202	TRE22-BOR	D-22	40.0	15.18	10.00	38.70	14.28	60 1.7	0.896	.415
203	TRP5X-EOD	D-12	22.5	9.17	38.00	27.80	12.99	376.1	0.960	.409
204	TRP5X-BOR	D-12	22.5	9.70	25.00	27.80	12.13	361.5	0.933	.435
205	TR131-BOR	D-12	22.5	7.10	4.00	14.10	11.10	158.0	0.991	.759
206	TRAH-EOR	8-225	29.0	9.50	no set	46.70	10.30	489.0	0.990	.404
207	TRBH-BOR	B-225	29.0	12.50	2.50	46.80	11.10	532.0	0.974	.727
208	TR8P-EOR	B-225	29.0	8.60	4.87	21.60	12.90	306.0	0.918	.468

Table 21. Pile driving and dynamic measurements for PD/LT (continued).

* - Denote blow count (BPI) based on blows per foot.

1 kip-ft = 1.36 kN-m 1 BPI = 0.039 blows per mm 1 kip/ft/s = 14.59 kN/m/s 1 ft/s = 0.305 m/s 1 kip = 4.448 kN 1 in = 25.4 mm

Nio.	Ple-Case	Case	EA/C	2L/C	Пр	Side	Пρ	Side
	NUMDer	Jc	(kips/s/ft)	(ms)	(in)	(in)	Uamping (s/ft)	Damping (s/ft)
1	FN1-EOD	0.884	22.13	8.57	.200	.100	.070	.170
2	FN1-BOR1	1.564	22.13	8.57	.100	.100	.400	.130
3	FN1-BOR2	2.217	22.13	8.57	.100	.100	.580	.110
4	FN2-EOD	0.356	60.49	9.65	.200	.220	.050	.270
5	FN2-BOR	0.637	60.49	9.55	.080	.150	.100	.330
6	FN3-EOD	0.066	85.89	9.55	.120	.060	.290	.800
7	FN3-BOR	0.283	85.89	9.55	.210	.070	.340	.310
8	FN4-EOD	0.377	34.26	7.85	.150	.120	.050	.150
9	FN4-BOR	0.703	34.26	7.85	.100	.110	.050	.180
10	FIA-EOD	-0.454	46.60	13.98	.300	.170	.479	.049
11	FIA-BOR	-0.453	46.60	13.98	.050	.100	.597	.065
12	FIB-EOD	0.147	37.80	11.59	.200	.150	.066	.059
13	FIB-BOR	-0.025	37.80	11.59	.150	.100	.129	.068
14	F01-EOD	0.133	120.80	7.17	.280	.100	.139	.136
15	FO1-BOR	0.720	120.60	7.17	.260	.100	.092	.092
18	FO2-EOD	-0.150	199.10	9.41	.230	.100	.049	.185
17	FO2-BOR	0.285	199.10	9.41	.250	.100	.039	.166
18	FO3-EOD	-0.826	61.40	13.08	.050	.080	.675	.082
19	F04-E00	-1.655	214.60	10.48	.130	.100	.115	.127
20	FO4-BOR	-0.570	214.60	10.48	.200	.120	.244	.193
21	FOR1-EOD	-0.578	159.00	20.96	.380	.250	.060	.179
22	FOR1-BOR	-0.503	159.00	20.96	.220	.220	.246	.185
23	FM5-EOD	0.022	49.08	13.96	.320	.100	.041	.086
24	FM5-BOR	0.677	49.08	12.02	.390	.100	.097	.074
25	FM17-EOD	0.438	49.06	9.26	.530	.090	.077	.076
26	FM17-BOR	0.877	49.08	9.26	.200	.100	.050	.142
27	FM23-EOD	0.259	49.08	6.75	.400	.080	041	.454
28	FM23-BOR	0.146	49.08	6.75	1.000	.210	.045	.090
29	FC1-EOD	-0.226	17.54	3.98	.300	.157	.036	.036
30	FC1-BOR	-0.278	17.54	3.96	.330	.140	.030	.032
31	FC2-EOD	-0.229	17.54	3.27	.330	.148	.041	.026

Table 22. Parameters of dynamic analysis for PD/LT.

NO.	Pile-Case	Case	EA/C	2L/C	Пр	Side	Тір	Side
			(ktps/s/ft)	(ms)	(in)	(in)	(s/ft)	(s/ft)
32	FC2-BOR	-0.393	17.54	3.14	.330	.150	.029	.024
33	FMI1-EOD	-0.010	28.73	9.87	.100	.100	.051	.047
34	FMI1-BOR	0.113	26.73	9.87	.150	.100	.096	.017
36	FMI2-EOD	0.196	26.73	7.32	.140	.100	.030	.060
36	FMI2-BOR	0.348	26.73	7.32	.150	.100	.055	.030
37	FWA-EOD	0.125	198.62	18.08	.500	.260	.095	.303
38	FWA-BOR	0.213	198.82	16.08	.301	.251	.189	.137
39	FWB-EOD	0.272	198.62	16.66	-	-		-
40	FWB-BOR	0.141	198.62	16.66	-	•	-	-
41	FA1-EOD	0.043	145.72	9.40	.100	.100	.123	.234
42	FA1-BORT	0.241	145.72	9.07	.200	.060	.368	.367
43	FA1-BOR2	0.373	145.72	9.07	.250	.100	.363	.322
44	FA2-EOD	-0.106	145.72	10.51	.420	.100	.098	.215
45	FA2-BOR1	0.207	140.52	10.90	.250	.100	.282	.205
48	FA2-BOR2	0.577	145.72	10.51	.170	.130	.323	.327
47	FA3-EOD	-0.199	221.53	9,16	.350	.100	.153	.329
48	FA3-8031	0.085	219.99	9.06	.200	.070	.513	.398
49	FA3-80,72	0.313	221.53	9.00	.205	.080	.309	.396
50	FA4-ECD	-0.204	221.53	10.43	.250	.100	.153	.297
51	FAA-BORT	0.331	219.93	10.51	.120	.080	.334	.390
52	FA4-BOR2	0.567	219.93	10.51	.150	.100	.282	.356
53	FA5-EOD	0.133	403.88	10.07	.330	.120	.395	.302
54	FA5-BCR	0.414	403.88	10.45	.240	.070	.398	.395
55	FV15-E00	0.085	38,19	10.95	.300	.100	_1 40	.102
56	FV15-BOR	0.009	38.19	10.95	.300	.100	.424	.089
57	FV10-EOD	0.478	38,19	10.95	.300	.100	.377	.202
58	FV10-BOR	0.143	38.19	10.95	.340	.125	.276	.164
59	FMN2-EOD	-0.223	38.19	11.19	.500	.160	.079	.104
60	FMN2-BOR	0.025	38.19	11.19	.150	.152	.085	.114
81	FP5-EOD	0.183	12.49	4.10	.200	.040	.037	.0191
82	FP5-BOR	0.350	12.49	4.10	.190	.045	.030	.068

Table 22. Parameters of dynamic analysis for PD/LT (continued).

NO.	Pie-Case	Case	EA/C	2L/C	Tip	Side	Tip	Side
		"c	(kips/s/ft)	(ms)	(in)	(in)	(\$/ft)	(s/ft)
63	FKG-EOD	-0.286	80.23	11.39	.250	.090	.104	.256
64	FKG-BOR	-0.1 89	80.23	11.39	.140	.060	.113	.282
65	FL3-EOD	-0.119	203.74	15.04	.400	.100	.246	.269
66	FL3-BORI	0.149	203.74	15.04	.250	.150	.378	.392
67	FL3-BOR2	0.394	203.74	15.04	.250	.124	.523	.507
68	CA1-EOD	-0.120	27.87	20.47	.140	.140	.089	.083
66	CA1-BOR	-0.108	27.87	20.47	.130	.130	.089	.075
70	CA2-BOR	0.556	27.87	13.39	.100	.100	.097	.105
71	CAS-BORI	-0.354	21.58	7.97	.362	.100	.012	.088
72	CAS-BOR2	-0.408	21.58	7.97	.327	.217	.035	.024
73	CA3/8-BOR	0.765	15.60	8.78	.374	.276	.096	.118
74	CA24-BOR	0.295	13.44	4.59	.177	.118	,113	.077
76	CA6-BOR1	0.040	25.93	7.16	.354	.276	.048	.050
76	CA6-BOR2	-0.006	25.93	7.18	.394	.258	.047	.052
77	CA6-EOR	-0.347	25.93	7.16	.335	.256	.037	.062
78	WC3-EOD	-0.063	268.75	6.80	.400	.100	.168	.066
79	WC3-BOR1	-0.056	268.75	6.80	.350	.130	.038	.196
80	WC3-BOR2	-0.042	268.75	5.36	.320	.080	.087	.137
81	WC8-EOD	-0.038	265.88	5.54	.420	.100	.118	.143
82	WC8-BORI	-0.044	265.88	5.54	.471	.080	.127	.212
83	WC8-BOP2	-0.005	265.88	3.93	.610	.100	.048	.311
64	WB9-BOR	0.455	271.52	20.00	.260	.050	.433	.251
85	WB15-BOR	0.457	269.50	16.28	.225	.060	.242	.486
- 36	T1/A-EOD	0.265	378.57	16.49	.150	.050	.070	.115
B7	T1/A-ALT	0.257	378.57	20.70	.200	.100	.157	.079
88	T1/B-EOD	0.153	378.57	25.74	.060	.060	.021	.047
89	T2/A-EOD	0.346	196.75	13.94	.150	.040	.235	.087
90	T2/B-EOD	0.057	198.75	31.01	.070	.070	.154	.033
91	35-1-BOR	0.068	38.93	7.15	.250	.100	.114	.043
92	35-4-BOR	0.315	17.50	6.21	.300	.100	.024	.033
-933	35-5-BOR	0.382	38.93	11.93	.040	.040	.042	.063

Table 22. Parameters of dynamic analysis for PD/LT (continued).

No.	Pile-Case	Case	EA/C	2L/C	Tip	Side	Тір	Side
		"c	(kips/s/ft)	(ms)	(in)	(in)	(s/fl)	(s/ft)
94	35-6-BOR	0.266	17.50	12.55	.100	.060	.001	.093
95	35-7-BOR	0.007	18.97	8.88	.200	.100	.040	.047
96	35-10-BOR	-0.182	80.56	7.69	.250	.040	.019	083
97	E2-BOR	0.589	64.20	6.31	.260	.100	.120	.175
98	63S-BOR	0.755	30.34	8.19	.280	.100	.027	.264
96	LB21-BOR	0.129	169.40	5.50	.310	.100	.128	.169
100	LB20-BOR	0.164	161.80	7.91	.230	.120	.211	.211
101	LC3-BOR	0.389	161.90	20.31	.350	.250	.192	.137
102	LIN16-BOR	0.787	161.10	24.50	.220	.120	.293	.337
103	LE37-BOR	0.082	38.80	10.00	.140	.080	.181	.650
104	LE64-BOR	0.218	38.80	10.00	.105	.070	.148	.132
105	ST1-EOD	0.242	123.0	11.52	.300	.080	.054	.061
106	ST2-EOD	-0.102	132.3	9.96	.600	.080	.017	.020
107	ST9-BOR	0.500	339.1	24.19	.220	.100	.322	.148
108	ST46-EOD	0.010	10.35	4.52	.400	.150	.033	.042
109	GZA3-EOD	-0.240	36.20	17.00	.330	.150	.053	.050
110	GZA5-EOD	0.000	27.80	16.50	.320	.150	.030	.050
111	GZA6-EOD	-0.180	27.70	20.37	.250	.125	.118	.053
112	GZBBCEOD	-0.234	37.80	13.79	.058	.050	.091	.075
113	GZBP2EOO	0.361	36.20	17.08	.040	.050	.051	.129
114	GZB6-EOD	0.174	27.70	11 .56	.240	.120	.061	.064
115	GZZS-EOD	0.281	37.80	10.34	.450	.350	.171	.238
116	GZO5-EOD	0.481	37.80	10.34	.580	.100	.059	.810
117	GZCC5EOD	0.471	37.80	13.91	.430	.220	.029	.117
118	GZL2-EOD	-0.038	37.80	13.91	.530	.320	.137	.244
119	GZP14-EOD	0.457	37.80	12.48	.450	.100	.077	.102
120	GZP11-EOD	0.268	37.80	12.48	.100	.100	.063	.178
121	GZP12-EOD	0.247	37.80	13.73	.110	.170	.038	.186
122	GZB22-EOD	0.812	111.0	18.71	.065	.065	.207	.126
123	GZW1-EOR	0.094	26.10	15.02	.170	.100	.118	.142
124	A54-EOD	-0.765	50.82	12.07	.138	.099	.199	.101

Table 22. Parameters of dynamic analysis for PD/LT (continued).

No.	Plie-Case	Case	EA/C	2L/C	Тір	Side	Tip	Side
	num09r	Jc	(kips/s/ft)	(ms)	(in)	(in)	(s/ft)	(s/fl)
125	A54-BOR	-0.149	50.82	12.07	.100	.343	.088	.109
126	A147-EOD	-0.599	48.44	11.18	.669	.100	.068	.112
127	A147-BOR	0.310	47.20	10.89	.219	.100	.075	.100
128	GF19-EOD	0.280	29.80	9.45	.110	.100	.035	.062
129	GF110-EOD	-0.117	36.73	6.79	.160	.110	.034	.117
130	GF222-EOD	0.221	38.73	7.97	.140	.130	.065	.078
131	GF224-EOD	0.008	17.31	6.30	.080	.030	.046	.023
132	GF312-EOD	0.830	38.73	3.93	.120	.080	.115	.057
133	GF313-EOD	1.124	29.80	4.16	.150	.080	,133	.043
134	GF412-EOD	1.355	38.73	5.78	.1 20	.120	.058	.026
136	GF413-EOD	1.056	29.60	4.07	.100	.120	.064	.029
136	GF414-EOD	1.390	29.80	5.85	.120	.110	.043	.012
137	GF415-EOD	0.622	38.73	5.88	.130	.100	.058	.027
138	EF82-EOD	0.093	27.95	-	<u> </u>	-	· ·	
139	EF167-BOR	0.835	27.99	. <u> </u>				-
140	A3-EOD1	-0.392	209.68	13.43	.330	.120	.110	.230
141	A3-BOR1	-0.714	209.66	13.43	.270	100	.130	.160
142	A3-EOD2	-0.329	209.68	13.43	.250	.150	.160	.180
143	A3-80R2	-0.464	209.66	13.43	.020	.080	.150	.260
144	A3-BOR3	0.209	209.66	12.78	.170	.100	.260	.220
145	A14-DD1	-0.130	291.07	15.35	.390	.100	.130	.280
146	A14-D02	-0.009	291.07	15.35	.370	.140	.110	.280
147	A14-BORI	0.213	291.07	15.35	.100	.120	.220	.220
148	A14-BOR2	0.402	291.07	10.76	.200	.150	.120	.230
149	A25-E00	-0.267	207.40	15.31	.350	.120	.060	.120
150	A25-BOR1	-0.198	207.40	15.31	.320	.100	.100	.110
151	A25-BOR2	0.010	207.40	15.31	.380	.270	.310	.100
152	A25-BOR3	-0.008	207.40	15.31	.380	.250	.260	.190
153	A16-EOD	0.064	150.55	9.05	.230	.100	.150	.100
154	A16-BOR1	0.103	150.55	9.05	.330	.100	.160	100
155	A16-BOR2	0.332	150.55	8.66	.240	.080	.650	.160

Table 22. Parameters of dynamic analysis for PD/LT (continued).

NO.	Pile-Case	Case	EA/C	2L/C	Тφ	Side	Τιρ	Side
	Number	J _e	(kips/s/fi)	(ms)	Quake (in)	Guake (in)	Damping (s/ft)	Damping (s/ft)
156	A41-E00	-0.027	205.97	13.24	.290	.080	.150	.080
157	A41-BOR1	0.034	205.97	13.24	.370	.090	.140	.090
158	A41-BOR2	0.090	205.97	8.95	.350	.100	.130	.100
159	A101-EOD	-0.375	208.34	12.65	.400	.120	.040	.310
160	A101-BOR1	-0.142	208.34	12.65	.120	.080	.120	.160
161	A101-BOR2	0.273	208.34	10.26	.100	.090	.200	.210
162	A133-EOD	-0.195	212.62	18.31	.360	.180	.260	.210
163	A133-BOR	0.217	212.62	16.30	.130	.130	.210	.190
164	A145-EOD	-0.454	212.71	18.59	.190	.090	.150	.240
165	A145-BOR1	-0.338	212.71	18.59	.170	.170	.170	.270
166	A145-BOR2	-0.019	212.71	16.21	.1 60	.140	.210	.210
167	CB3-BOR	0.647	255.08	11.38	.190	.100	.563	.317
168	CE3-BORL	0.520	255.08	11.67	,190	.110	.527	.379
169	C85-80R	-0.395	291.95	12.45	.140	.100	.314	.929
170	CES-BORL	-0.188	291.95	8.74	.300	.100	.227	.405
171	CB11-BORL	0.363	318.10	12.81	.140	.180	1.335	.249
172	CB11-EORL	0.259	318.10	12.81	.120	.170	.560	.538
173	C817-BOR1	0.029	297.18	1 3.63	.130	.210	.413	.258
174	CB17-BOR2	0.128	297.16	13.63	.250	.160	.318	.277
175	CB17-BORL	-0.092	297.18	12.65	.220	.030	.350	.125
176	CB17-DRL	0.024	297.18	12.65	.250	.010	.328	.031
177	CB23-BOR	0.312	309.73	12.95	.140	.130	.658	.284
178	C823-BORL	0.405	309.73	12.95	.050	.170	1.674	.535
179	CB29-BORL	-0.213	288.28	13.78	.090	.100	.707	.483
180	C829-EORL	0.017	268.28	13.78	.200	.100	.129	.812
181	CB35-BOR1	-0.311	293.99	13.79	.240	.100	.113	.095
182	CB35-BOR2	-0.167	293.99	13.79	.180	.100	.113	.433
183	CB35-BORL	-0.029	293.99	12.66	.090	.170	.700	.226
184	CB41-EOR	-0.161	302.20	14,14	.260	.100	.141	.209
185	CB41-BOR	0,118	302.20	14.00	.260	.110	.127	.198
186	CB41-BORL	-0.117	302.20	10.92	.140	.130	.351	.392

Table 22. Parameters of dynamic analysis for PD/LT (continued).

.

No.	Plie-Case	Case	EA/C	2L/C	Тір	Side	Tip	Side
		Jc	(kdps/s/ft)	(ms)	(in)	(in)	(9/ft)	(s/fi)
187	CB26-EOD	-0.408	261.84	11.40	.210	.120	.078	.105
188	CB26-BOR	-0.137	261.84	11.40	.270	.110	.099	.059
189	CB26-EOR	-0.096	261.84	11.40	.330	.090	.056	.056
190	CB26-80F2	-0.164	261.84	9.25	.230	.100	.176	.736
191	33P1-EOD	-0.141	38.90	14.39	.150*	.300*	.080	.010*
192	33P1-BOR	-0.017	38.90	14.39	.060	.040	.030	.038
193	33P1-EOR	-0.240	38.90	14.39	.100	.100	.012	.028
194	33P2-EOD	-0.182	17.54	17.81	.400*	.200*	.150*	.020*
195	33P2-BOR	-0.125	17.54	13.21	.300	.300	.048	.046
198	33P2-EOR	-0.096	17.54	13.21	.300	.300	.010	.033
197	33P4-EOD	0.152	67.30	10.45	.100*	.025	.100*	.050*
198	33P5-EOD	0.509	21.21	6.94	.090*	.100*	.040*	.040*
199	TRD22-EOD	0.371	38.70	2.68	.150	.100	.015	.224
200	TRD22-BOR	0.157	38.70	2.68	.160	.100	.106	.216
201	TRE22-EOD	0.681	38.70	3.53	.100*	.100*	.100*	.100 [*]
202	TRE22-BOR	0.411	38.70	3.53	.250	.100	.018	.135
203	TRP5X-EOD	0.418	27.80	2.94	.150	.100	.020	.111
204	TRP5X-BOR	-0.059	27.80	2.94	.150	.100	.013	.128
205	PR131-BOR	0.162	14.10	3.15	.300	.300	.034	.235
206	TRAH-EOR	-0.310	46.70	16.24	.200	.100	.025	.753
207	TRIBH-BOR	0.138	46.80	13,40	.050	.050	1.040	.206
206	TRBP-EOR	-0.730	21.80	12.94	.025*	.100*	.200*	.100*

Table 22. Parameters of dynamic analysis for PD/LT (continued).

1 kip/s/ft = 14.6 kN/s/m 1 in = 25.4 mm 1 s/ft = 3.281 s/m

No	Plie-Case Number	Lond Teni	Deviason's Criteria	Shape of	∆ =1"	∆= 0.1B	DeBeer	Static Resist	CAPWAP TEPWAP	Energy Appr.	K _{ap}
		iype	(Idpa)	(kips)	(kips)	(kipe)	(kips)	r∿_ (k⊴ps)	(kips)	rر (kips)	д, Я,
1	FN1-EOD	Q	304	300	304	304	300	300	230	362	0.829
2	FN1-BOR1	٩	304	300	304	304	_300	300	375	484	0.620
3	FN1-BOR2	Q	304	300	304	304	_ 300	300	431	535	0.561
4	FN2-EOD	Q	358	354	362	366	356	354	228	418	0.847
5	FN2-BOR	Q	358	354	362	368	358	354	305	487	0.727
6	FN3-EOD	٩	378	370	382	393	366	374	179	480	0.779
7	FN3-BOR	a	378	370	382		368	374	297	621	0.602
8	FN4-EOD	٩	284	280	288	292	282	280	244	401	0.698
9	FN4-BOR	Q	284	260	268	292	282	280	266	582	0.481
10	FIA-EOD	٩	928	934	772	910	920	\$30	367	569	1.634
11	FIA-BOR	٥	928	934	772	910	920	930	731	689	1.349
12	FIB-EOD	Q	650	480-640	650	NA	648	650	511	708	0.918
13	FIB-BOR	a	650	480-640	650	NA	648	650	521	696	0.934
14	FO1-EOD	a	598	500-560	672	NA	544	567	496	718	0.811
15	FO1-BOR	Q	598	500-560	672	NA	544	557	700	1169	0.476
16	FO2-EOD	Q	760	750	780	800	754	750	530	646	1.161
17	FO2-BOR	Q	760	750	750	800	754	750	731	1168	0.648
18	FC3-EOD	<u>a</u>	778	700-850	816	862	820	820	566	584	1.404
19	FO4-EOD	<u> </u>	1700	1400	1716	1800	1664	1650	658	763	2.163
20	FO4-BOR	0	1700	1400	1716	1800	1664	1650	767	1269	1.300
21	FOR1-EOD	Q	1360	1350	1168	1600	1400	1380	559	839	1.651
22	FOR1-BOR	a	1360	1350	1168	1600	1400	1380	729	1207	1.143
23	FM5-EOD	Q	440	360-440	526	NA	461	420	346	357	1.178
24	FM5-BOR	a	440	360-440	526	NA	461	420	499	734	0.572
25	FM17-EOD	0	408	375-440	541	NA	430	447	424	524	0.853
26	FM17-80R	a	408	375-440	541	NA	430	447	526	781	0.572
27	FM23-EOD	٩	342	290-330	378	NA	356	340	323	412	0.825
28	FM23-BOR	a	342	290-330	378	NA	356	340	340	426	0.798
29	FC1-EOD	Q	316	320-380	370	372	358	340	270	342	0.994
30	FC1-BOR	Q	316	320-360	370	372	358	340	265	363	0.937
31	FC2-EOD	Q	368	350-400	442	NA	336	376	375	402	0.935

Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT.

	Die Case	beol	Deviseon's	Shape	A = 1	A=0.1P	Delleer	Static	CADWAR	Energy	
	Number	Test	Criteria	of	2-1	4-9.18		Resist	TEPWAP	Appr.	φ α^
;		туре	(kips)	Cuive (kips)	(kips)	(kips)	(kips)	R, (Kips)	(kips)	R. (kips)	₿ . R,
32	FC2-BOR	٥	368	350-400	442	NA	336	376	340	353	1.065
33	FMI1-EOD	Q	330	288-317	333	NA	320	310	285	246	1.260
34	FMI1-BOR	0	330	288-317	333	NA	320	310	319	306	1.013
35	FMI2-EOD	0	209	160	NA	NA	126	160	184	179	0.894
36	FMI2-BOR	a	209	160	NA	NA	126	160	217	279	0.573
37	FWA-EOD	SM	1300	1300	1300	NA	1150	1300	295	1145	1.135
38	FWA-BOR	SM	1300	1300	1300	NA	1150	1300	652	1154	1.127
39	FWB-EOD	SM	1000	1200	1000	NA	1497	1225	plug	1708	0.717
40	FWB-BOR	SM	1000	1200	1000	NA	1497	1225	plug	1280	0.957
41	FA1-EOD	S	370	325-350	419	NA	334	345	205	302	1,142
42	FA1-BOR1	S	370	325-350	419	NA	334	345	257	462	0.747
43	FA1-BOR2	s	370	325-350	419	NA	334	345	382	840	0.411
44	FA2-EOD	S	550	480-550	568	NA	541	535	428	568	0.942
45	FA2-BOR1	3	550	480-550	588	NA	541	535	489	950	0.563
48	FA2-BOR2	s	550	480-550	588	NA	541	535	599	896	0.597
47	FA3-EOD	s	625	500-640	679	NA	648	.614	340	547	1.122
48	FA3-BORI	5	625	500-640	679	NA	648	614	307	744	0.825
49	FA3-BOR2	3	625	500-640	679	NA	648	614	587	826	0.743
50	FA4-EOD	S	817	685-825	887	NA	748	773	446	772	1.001
51	FAA-BORT	S	817	685-825	887	NA	748	773	604	1062	0.728
52	FA4-BOR2	S	817	685-825	687	NA	748	773	852	1448	0.534
53	FA5-EOD	s	1140	1050	1186	NA	939	1074	662	1543	0.696
54	FA5-BOR	S	1140	1050	1158	NA	939	1074	945	2238	0.480
55	FV15-EOD	٩	315	300-350	372	440	246	315	194	336	0.938
56	FV15-BOR	٩	315	300-350	372	440	246	315	198	369	0.810
57	FV10-EOD	a	345	230-300	400	484	240	313	159	305	1.026
58	FV10-BOR	٩	345	230-300	400	484	240	313	179	285	1.098
59	FMN2EOD	٩	765	720-740	722	752	724	740	342	476	1.555
60	FMN2BOR	٩	785	720-740	722	752	724	740	652	831	0.890
61	FP5-EOD	۵	243	220-235	NA	NA	211	227	210	211	1.076
62	FPS-BOR	٩	243	220-235	NA	NA	211	227	239	280	0.611

Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT (continued).

f				<u>, </u>		_					
No	Plie-Case Number	Lond Test Type	Davisson's Criteria	Shape of Curve	∆ =1"	∆=0.18	DeBeer	Static Resist R_	CAPWAP TEPWAP	Energy Appr. R.	K _{ap} R
			(kips)	(kips)	(kips)	(kips)	(Kips)	(kips)	(kips)	(kips)	R
63	FKG-EOD	a	368	480-520	_530	NA	475	465	288	392	1.188
64	FKG-BOR	٩	366	480-520	530	NA	475	465_	295	403	1.154
65	FL3-EOD	ЦТ	400	400	NA	NA	400	400	136	258	1.550
66	FL3-BORI	шŤ	400	400	NA	NA	400	400	272	598	0.669
67	FL3-BOR2	щτ	400	400	NA	NA	400	400	350	893	0.448
68	CA1-EOD	s	540	500-560	390	390	530	533	410	444	1.200
69	CA1-BOR	s	540	500-560	390	390	530	533	500	433	1.231
70	CA2-BOR	s	366	320-400	370	370	355	380	342	430	0.884
71	CAS-BORI	S	468	460-500	500	NA	460	480	409	540	0.888
72	CAS-BOR2	S	468	460-500	500	NA	460	480	489	543	0.883
73	CA38-BOR	a	189	200-230	271	271	227	230	241	369	0.623
74	CA24-BOR	S	242	220-260	NA	NA	248	243	207	403	0.595
75	CA6-BOR1	s	660	620-660	590	650	640	660	610	782	0.844
76	CA6-BOR2	S	660	620-660	590	650	640	660	584	742	0.889
Π	CA8-EOR	S	680	620-660	590	650	640	660	558	704	0.938
78	WC3-EOD	FQ	610	550-850	NA	NA	620	ē10	509	751	0.812
79	WC3BOR1	FQ	610	550-850	NA	NA	620	610	506	781	0.781
80	WC380P2	FQ	610	550-850	NA	NA	620	810	536	777	0.785
81	WC6EOD	FQ	453	445-545	NA	NA	537	495	450	597	0.829
82	WC8BOR1	FQ	453	445-545	NA	NA	537	495	480	713	0.694
83	WC6BOR2	FQ	453	445-545	NA	NA	537	495	443	772	0.641
84	W89-80R	FQ	900	830-680	925	NA	855	884	941	1789	0.500
85	WB15BOR	FQ	820	740-790	833	NA	767	766	805	1448	0.529
86	T1/A-EOD	SM	1984	1984	1984	NA	1808	1984	1775	2729	0.726
87	T1/A-ALT	SM	1984	1984	1954	NA	1808	1984	1800	2870	0.690
- 88	T1/B-EOO	SM	2868	2425	NA	NA	1852	2648	2368	3042	0.871
89	T2/A-EOD	SM	1345	1323	NA	NA	1654	1470	1252	1872	0.785
90	T2/B-EOD	SM	3285	>2204	NA	NA	NA	3080	2778	2964	1.040
9 1	35-1-BOR	s	322	320-350	354	366	318	325	260	274	1.164
92	35-4-BOR	s	3330	300-330	334	342	314	320	360	466	0.684
83	35-5-BOR	S	612	580-620	600	608	600	600	650	616	0.971

Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT (continued).

* - Davisson's reduced for creep.

* - Extrapolated. (Load taken to 2200 kips.)

No	Plie-Case Number	Load Test	Davisson's Criteria	Shape of	∆ = 1*	∆=0 .1B	DeBeer	Static Resist	CAPWAP TEPWAP	Energy Appr.	Kap
		Туре	(kips)	Curve (tips)	(kips)	(kips)	(kips)	R _s (kips)	(kips)	R. (kips)	<u>B</u> , R,
94	35-6-BOR	s	800	500-550	530	548	526	530	580	526	1.007
8	35-7-BOR	s	122	120-170	152	146	144	142	139	183	0.776
96	35-10-BOR	S	402	370-420	432	444	378	400	334	425	0.941
97	E2-BOR	٩	415	375-395	NA	NA	369	390	420	732	0.533
98	63S-BOR	CRP	264	250-272	292	NA	258	268	279	3220	0.838
8	LB21-BOR	S	380	360	ş	484	260	360	361	519	0.694
100	L820-80R	S	580	480	NĂ	NĂ	480	530	474	813	0.652
101	LC3-BOR	S	620	600	680	z	560	640	812	1169	0.547
102	LIN16-BOR	5	600	600	600	NA	600	600	562	985	0.609
103	LE37-BOA	S	250	240	270	NA	230	250	197	241	1.037
104	LE64-BOR	s	270	240	NA	NA	220	260	232	274	0.949
105	ST1-EOD	S	344	280-320	NA	NA	300	344	505	630	0.546
108	ST2-EOD	s	510	540	NA	NA	500	540	616	665	0.812
107	ST9-BOR	S	920	720-640	920	NA	800	900	807	1927	0.487
108	ST48-EOD	S	NA	104	NA	NA	104	104	82	113	0.920
109	GZA3-EOD	Q	440	500	460	520	480	480	365	424	1.132
110	GZA5-EOD	٩	256	160-210	320	314	270	296	293	339	0.873
111	GZA8-EOD	a	188	350	316	306	350	328	275	281	1.160
112	GZBBCEOD	Q	440	500-560	500	590	560	530	413	453	1.169
113	GZBP2EOD	٩	280	340	340	324	290	320	317	302	1.059
114	GZB6-EOD	Q	380	420	456	530	360	390	341	413	0.944
115	GZZS-EOD	Q	464	420-470	540	NA	410	440	214	557	0.790
118	GZOS-EOD	a	480	440-480	800	NA	490	486	205	511	0.951
117	GZCCSEOD	٩	450	480-520	520	750	NÅ	490	492	599	0.818
118	GZ12-EOD	٩	640	600-660	690	760	530	660	267	566	1.167
119	GZP14-EOD	٩	390	360-400	440	500	480	420	305	570	0.737
120	GZP11-EOO	Q	250	340-420	380	440	430	386	239	399	0.967
121	GZP12-EOD	٥	500	600	630	NA	NA	560	520	674	0.831
122	GZB22-EOD	a	1120	1120	1040	NA	840	1060	1109	1357	0.781
123	GZW1-EOR	٥	360	335-400	404	416	353	380	250	357	1.064
124	A54-E00	CRP	652	630-652	618	638	639	638	383	464	1.437

 Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT (continued).

20	Plie-Case Number	Load Tesi	Davisson's Criteria	Shape of	<u>∆</u> =1*	▲ =0.1B	DeBeer	Static Resist	CAPWAP TEPWAP	Energy Appr.	Kap
		Туре	(kips)	Curve (kips)	(kips)	(kips)	(kips)	R (kips)	(kips)	R _u (kips)	R,
125	A54-BOR	CRP	652	630-652	618	638	639	638	611	696	0.914
126	A147-E00	CRP	558	547	565	580	540	552	259	339	1.628
127	A147-BOR	CRP	558	547	555	560	540	552	564	653	0.845
128	GF10-EOD	Q	330	400-460	380	360	325	397	398	434	0.915
129	GF110-EOD	Q	500	500-800	560	560	450	550	457	605	0.909
130	GF222-EOD	Q	580	540-600	590	590	540	570	512	623	0.916
131	GF224-EOD	Q	ź	450-470	NĂ	NA	465	463	419	458	1.011
132	GF312-EOD	Q	340	300-310	NĂ	NĂ	280	310	405	483	0.642
133	GF313-EOD	Q	334	320-330	NA	NA	334	330	448	532	0.620
134	GF412-EOD	Q	240	240-280	294	294	200	272	455	530	0.513
135	GF413-EOD	Q	300	280-320	350	350	270	300	428	491	0.611
136	GF414-EOD	٩	360	360-420	420	420	320	390	524	630	0.619
137	GF415-EOD	Q	460	460-520	540	540	440	500	561	599	0.835
138	EF62-EOD	Q	502	440-510	466	456	480	477	522	636	0.750
139	EF167-BOR	Q	271	267	279	277	267	272	479	625	0.438
140	A3-EOD1	FQ	958	850-940	960	NA	958	939	472	629	1.493
141	A3-BOR1	FQ	958	850-940	960	NA	958	939	538	660	1.423
142	A3-E002	FQ	958	850-940	960	NA	958	939	368	545	1.723
143	A3-BOR2	FQ	968	850-940	960	NA	958	939	462	612	1.534
144	A3-BOR3	FQ	958	850-940	960	NA	958	939	825	1421	0.661
145	A14-DD1	FQ	N	860-945	NA	NA	908	905	684	982	0.922
148	A14-DD2	FQ	NA	860-945	NA	NA	908	905	741	1076	0.B41
147	A14-BORT	FQ	NA	860-945	NA	NA	908	905	604	1118	0.809
148	A14-80P2	FQ	NA	660-945	NA	NA	908	905	962	1476	0.613
149	A25-E00	FQ	715	750-840	840	NA	845	800	459	549	1.457
150	A25-BOR1	FQ	715	750-840	840	NA	845	800	555	685	1.203
151	A25-80R2	FQ	715	750-840	840	NA	845	800	452	970	0.825
152	A25-BOR3	FQ	715	750-840	840	NA	845	800	442	930	0.860
153	A15-EOD	FQ	315	275-315	350	NA	272	308	224	303	1.017
154	A16-BORI	FQ	315	275-315	350	NA	272	308	282	415	0.742
155	A16-BOR2	FQ	315	275-315	350	NA	272	308	298	505	0.610

 Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT (continued).

No	Plie-Case Number	Lond Teal	Devisson's Criteria	Shape	∆ =1"	∆=0.1B	DeBøer	Static Resist	CAPWAP TEPWAP	Energy Appr.	Кър
		Туре	(kips)	Curve (kips)	(kips)	(kips)	(kips)	R, (kips)	(Nips)	R _u (kips)	<u>R</u> ,
156	M1-EOD	FQ	524	500-525	540	NA	536	530	431	624	0.849
157	M1-BORI	FQ	524	500-525	540	Š	536	530	503	715	0.741
158	A41-BOR2	FQ	524	500-525	540	NĂ	536	530	565	834	0.835
159	A101-EOD	FQ	812	800-840	NA	NA	800	810	517	474	1.709
160	A101-BOR1	FQ	812	800-840	NĂ	N	800	810	669	722	1.122
181	A101-BOR2	FQ	812	800-840	NĂ	NĂ	600	810	803	881	0.919
162	A133-EOD	FQ	808	780-860	810	NA	868	826	311	513	1.610
163	A133-BOR	FQ	806	780-860	810	NA	868	826	780	998	0.828
164	A145-EOD	FQ	976	860-950	975	NA	913	940	353	549	1.712
165	A145-BOR1	FQ	976	860-950	975	NA	913	940	841	745	1.262
166	A145-BOP2	FQ	976	860-950	975	NA	913	940	761	918	1.024
167	CB3-BOR	FQ	500	488-500	470	NA	472	484	564	978	0.495
188	CB3-BORL	FQ	500	488-500	470	NA	472	484	502	998	0.485
169	CB5-BOR	FQ	1250	1240	1325	NA	1170	1246	566	1006	1.236
170	CB5-BORL	FQ	1250	1240	1325	NA	1170	1246	584	1167	1.068
171	CB11-BORL	FQ	1435	1370	1430	NA	1364	.1400	814	1803	0.776
172	CB11-EORL	FQ	1435	1370	1430	NA	1364	1400	639	1827	0.768
173	CB17-BOR1	FQ	1515	1400	1500	NA	1400	1453	820	1776	0.818
174	CB17-BOR2	FQ	1515	1400	1500	NA	1400	1453	749	1824	0.797
175	CB17-BORL	FQ	1515	1400	1500	NA	1400	1453	683	1501	0.968
1 76	CB17-DRL	FQ	1515	1400	1500	NA	1400	1453	845	1641	0.885
177	CB23-BOR	FQ	643	640-810	732	NA	758	702	619	864	0.813
178	CB23-BORL	FQ	643	540-810	732	NA	758	702	444	1306	0.538
179	CB29-BORL	FQ	917	870-960	960	NA	910	926	776	856	1.083
180	CB29-EORL	FQ	917	870-960	960	NA	910	926	449	1069	0.868
181	CB36-BOR1	FQ	1463	1400	1490	NA	1400	1437	812	1001	1.438
182	C835-BOP2	FQ	1463	1400	1490	NA	1400	1437	949	1422	1.011
183	CB35-BORL	FQ	1463	1400	1490	NA	1400	1437	909	1269	1.115
184	CB41-EOR	FQ	1410	1380	1435	NA	1357	1396	857	1238	1.128
185	CB41-BOR	FQ	1410	1380	1435	NA	1357	1396	850	1225	1.140
186	C641-BORL	FQ	1410	1360	1435	NA	1357	1396	485	1162	1.201

 Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT (continued).

₩0	Plie-Czae Number	Loadi Test Type	Deviason's Criteria (kips)	Shape of Curve (kips)	∆ = 1* (kips)	∆ = 0.1B (kdps)	DeBeer (kips)	Static Resist R _s (kips)	CAPWAP TEPWAP (kips)	Energy Appr. R _u (kips)	Kap Ra Ru
187	C826-EOO	FQ	960	850-950	1000	NA	1000	965	488	555	1.739
188	C826-80R	FQ	960	850-950	1000	NA	1000	965	619	788	1.225
189	CE25-EOR	FQ	960	850-950	1000	NA	1000	965	716	957	1.008
190	CB25-BOP2	FQ	960	850-950	1000	NA	1000	965	563	1113	0.867
191	33P1-EOD	5	>800	800	520	600	800	800	439	657	1.218
192	33P1-80R	5	>800	800	520	600	800	800	715	896	0.891
193	33P1-EOR	S	>800	800	520	600	800	800	650	923	0.867
194	33P2-EOD	S	490	450-500	450	490	460	490	290 [°]	418	1.172
195	33P2-BOR	S	490	450-500	450	490	460	490	355	520	0.942
196	33P2-EOR	S	490	450-500	450	490	460	490	401	546	0.897
197	33P4-EOD	S	468	350-500	558	592	470	500	400 [°]	625	0.800
196	33P5-EOD	8	164	160-200	244	264	200	200	143*	248	0.806
199	TR022-EOD	S	354	350	NA	Ň	356	350	432	553	0.633
200	TRD22-BOR	S	354	350	NA	NA	356	350	294	504	0.694
201	TRE22-EOD	S	558	570	NA	NĂ	570	570	575 [*]	720	0.792
202	TRE22-BOR	S	556	570	NA	NA	570	570	818	707	0.806
203	TRP5X-EOD	S	410	500-550	510	560	400	475	484	506	0.939
204	TRP5X-BOR	9	410	500-550	510	560	400	475	395	490	0.969
205	TR131-BOR	S	140	160-200	210	200	200	150	166	169	0.888
206	TRAHEOR	S	730	650-700	600	650	640	650	218	564	1.152
207	TRBH-BOR	S	325	275-300	337	352	304	300	100	266	1.128
206	TRBP-EOR	S	340	>300	340	340	325	330	248*	306	1.078

 Table 23. Pile capacity based on static load test and dynamic analysis for PD/LT (continued).

1 kip = 4.448 kN 1 in = 25.4 mm

	_	I	1						A	7	Ľ	N	U	X	В	•	D	A	1 A	1 2	sE	1	P	D														
ENERGY	APPROACH	KIPS	1528	1 4	743	167	495	199	478	644	111	475	212	1178	1709	1211	714	489	714	553	858	404	204	231	688	702	308	466	400	629	530	366 3	690	454	60 E	475	B 03	4 04
CAPWAP	Rut	KIPS	616	333	337	123	365	183	179	297	101	226	226	643	655	655	507	373	507	416	646	185	68	8	598	329	237	282	469	346	390	348	615	39 3	277	425	720	308
	BLOWS/	INCH	5 00	2.00	1.50	3.42	5.00	7.00	9.17	7.00	1.83	5 00	14.42	11.00	14.00	5.00	13 50	15.00	13 50	2.25	26 00	5.00	2 00	5.00	28.00	166.67	44 67	25.33	2.67	52.33	5 75	2.06	0 0'6	100.00	833.33	2.67	14.00	5.58
	DMX	z	0.453	0.824	0.632	0.848	0.409	0.438	0.386	0.459	0.587	0.444	0.232	0.257	0.453	0 374	0.501	0.513	0.501	1 326	0.590	0.463	0.862	0.409	0.618	0.150	0.120	0 175	0.594	0 145	0.693	0.814	1110	0 317	0.233	0 747	0 469	1.112
	XMV	FT/S	10.10	8.47	4.64	11.70	8.60	10.30	6.10	8.10	4.00	7.40	3.10	3.80	6.60	7.60	6 .03	B 20	0 .03	13 41	7.20	7.90	8.20	4.10	5.00	3.30	1.90	2.90	4.30	3.60	5.50	4.50	6.20	5.90	3.20	5.70	4 70	14.93
	EMX	K-FT	41.61	24.33	40.19	7.93	12.56	4.81	99.6	16.16	5.22	12.74	2.66	17.08	37.34	28.96	17 09	11.83	17 09	40.79	22.47	11.16	11.59	5.86	18.74	4.56	1.83	4.16	16.13	4.30	19.16	19.72	23.62	6.18	3.02	22.20	18 08	21.71
	FINX	KIPS	1581.0	1156.3	1684.1	161.0	541.0	184.0	558.0	717.0	223.0	476.0	192.0	1049.0	1922 0	1279.0	854 2	480.3	854 2	761.1	6360	472.0	426.0	286.0	542.0	435.0	248.0	424.0	551.0	451.0	664.0	0.795	482.0	385.0	219.0	460.0	604.0	330.3
	HANNER		D46-32	D 46 13	C5300	D12	020	D12	030	030	MKT DE33	D30	MKT DE33	D46-32	D46-32	D46-32	K35	KCE 640	K35	K35	VUL 512	VUL 06	VUL 06	7UL 06	VUL 512	CON 65	CON 65	CON 65	CON 160	CON 65	CON 160	CON 160	CON 160	VUL 01	VUL 01	CON 160	CON 160	VUL 508
	E MOD	KSI	6097.2	5651.5	5122.0	30000.0	5452.0	30000.0	5934.0	5934.0	6190.0	5452.0	6190.0	6500.0	6609.2	6790.0	30000.0	4644 0	30000.0	30000	6120.0	5120 0	5120.0	5120.0	6120.0	5057.0	5057.0	5220.0	5177.5	5220.0	5037.6	4658.0	4658.0	6116.0	6116.0	4658.0	4658 0	0.0000E
	AREA	ţN2	324.0	324.0	0.008	7.0	144.0	7.0	196.0	196.0	96.5	144.0	3 6.5	576.0	576.0	324 0	548	144.0	548	312	196.0	144.0	144.0	144.0	196.0	256.0	256.0	256.0	324 0	256.0	324.0	196.0	196.0	144.0	144.0	196.0	324.0	12.1
	LEN.	F	67.0	32.5	63.0	37.0	62.0	36.0	62.0	62.0	20.0	62.0	730	0.77	76 0	710	603	668	603	910	070	58.0	58.0	58.0	97.0	35.0	35.0	35.0	84.0	35.0	84.0	0.77	0.77	53.0	53.0	0 / /	84.0	98.5
	PILE	түре	PSC 18	PSC18	PSC30	CEP12x0.18	PSC 12	CEP12x0.18	PSC 14	PSC 14	PSC	PSC 12	PSC	PSC 24	PSC 24	PSC 18	CEPIPE 24	PSC 12	CEPIPE24	CEPIPE 1	PSC 14	PSC 12	PSC 12	PSC 12	PSC 14	PSC 16	PSC 16	PSC 16	PSC 18	PSC 16	PSC 18	PSC 14	PSC 14	PSC 12	PSC 12	PSC 14	PSC 18	CEPIPE 1
~	ΤOE	SOIL	ō	ರ	ರ	ರ	ರ	ರ	ರ	ರ	ರ	ರ	ರ	ರ	ರ	ರ	CL SA	ಜ ರ	с s	CL TL	CLSA	CLSA	CLSA	CLSA	CLSA	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	CLSI	TIL ALL
PILES IN CLAY	SKIN	SOIL	CLSA	ರ	ರ	ALLUV	CLSI	ALLUV	CLSI	CLSI	SASI	CLSI	SASI	CLSA	CLSA	CLSA	CL SA	SUCL	CL SI	CL TIL	SA	CLSA	CLSA	CLSA	SA	SACL	SACL	SACL	CLSI	SACL	CLSI	CLSI	CLSI	SASI	SASI	CLSI	CLSI	ALLUV
DISPLACEMENT	PILE	NAME	ROR TP1	83E 2 RE	S4PC N20	DD TP2	BOR T2	BOR TP3	EOD T3	BOR T3	EOD TP1799	EOD T2	RES TP1799	BOR TP2	BOR TP5	BOR TP4	EOD PN7	BOR L-8	BOR PN7	rer og	BOR TP1235	BOR PNH20	EOD PNH20	BOR PNF2	BOR TP1259	BOR PN111	EOD PN111	EOD PN110	DD TP15	BOR PN110	DD TP15	DD TP16	DO TP16	EOD TP6	BORTP6	DD TP16	RES TP15	EOD 258
LARGE	REF.	NO	362	321	319	20	155	69	162	156	328	161	329	363	366	365	90 1	235	410	252	181	276	277	275	192	224	226	225	ŝ	223	57	2 3	61	1	10	99	62	258

Table 24. Pile/soil and dynamic measurements of data set PD.

APPENDIX B - DATA SET PD

			Tab	le 24. Pile/so	il and o	dynami	c measur	ements of e	lata set	PD (co	ntinuco	Ĵ.			
														CAPWAP	ENERGY
REF.	PILE	SKIN	TOE	PILE	LEN.	AREA	EMOD	HAMMER	FMX	EMX	XMV	DMX	BLOWS/	Ruft	APPROACH
<u>Ö</u>	NAME	SOIL	SOIL	түре	Ŀ	IN2	KSI		KIPS	K-FT	F1/S	z	INCH	KIPS	KIPS
5		ζ	Ē		145.0	0.501	1366.0		0 77 0	30.53	0f F	1 070	1 000	Υ.	634
		; ; ;			2.2	2.06	0.0000			3 3	20.21	P 100 0	8.2	3	3
ន្ល	X PINAS	CLSA		CEP 14X0.5		717	0.0000			3 1	N-01	0.33		8	
8	EOD PN126	ರ	TILL	PSC 14	70.0	196.0	4000.0	VUL 140C	520.0	19.73	6.70	0.649	11.00	230	640
8	RES PN50	ರ	זנו	PSC 14	137.0	196.0	4974.0	ICE 640	324.0	9.58	4.20	0.486	40.00	99C	450
51	EOD B611	ช	TILL	PSC 16	145.0	256.0	4355.0	ICE 660	635.0	26.13	6.40	0.783	16.00	571	742
356	X PNB5	CLSA	TIL	CEP 14x0.5	87.0	21.2	30000.0	D16-32	590.0	25.18	15.50	0.810	3.75	178	561
358	BOR PN126	ರ	TILL	PSC 14	70.0	196.0	4000.0	VUL 140C	434.0	11.84	5.80	0.430	21.00	442	595
8	RES PN20	ರ	TILL	PSC 14	137.0	196.0	4974.0	ICE 640	352.0	10.21	4.60	0.512	80.00	428	467
361	EOD PN177	ช	TILL	PSC 14	70.0	196.0	3920.0	VUL 140C	476.0	19.22	6.10	0.665	12.00	489	616
3	RES PN9	ರ	TILL	PSC 14	128.0	196.0	4974.0	ICE 640	407.0	15.23	5.50	0.668	25.00	6 4	516
359	BOR PN177	บ	TILL	PSC 14	72.0	196.0	4000.0	VUL 140C	413.0	11.51	5.60	0.446	15.00	370	539
ß	EOD PN10	ರ	TILL	PSC 14	130.0	196.0	4974.0	ICE 640	339.0	11.29	4.50	0.670	12.00	328	360
357	X PNG3	CLSA	TILL	CEP 14x0.5	88.0	21.2	30000.0	D16-32	637.0	29.32	16.80	0.865	2:00	260	661
ጽ	EOD PN1	ರ	TILL	CEP 10.75	0 .0C	5.9	30000.0	CON 65	213.0	9.57	13.20	0.849	5.33	198	222
LARG	E DISPLACEMENT F	PILES SOIL TY	PE UNKNO	NM											
378				PSC 12	42 0	144.0	6230.0		583 5	12.52	8.35	0.403	10.42	483	601
259	BOR 174	ALLUV	ALLUV	CÉPIPE 1	975	12.1	30000	VUL 508	305.6	16.93	14.56	0.928	7 00	257	380
333	X PN29K	¥	M	CEP 20x0.5	77.0	30.6	30000.0	MKT DE70	897.0	24.07	14.80	0.513	16.00	794	1004
332	X PN25BK	AN	M	CEP 20x0.5	67.0	30.6	30000.0	MIKT DE 70	749.0	18.84	13.90	0.525	17.00	674	774
9	EOD PN7E3	¥	AN	PSC 30	125.0	0.006	5554.1	CON 300	1652.0	29.81	4 20	0 371	8.00	1090	1442
7	EOD PN210	AA	M	CEP 12.75	80.0	14.6	30000	VUL SOC	224.0	5.93	8.56	0.485	7.50	181	230
334	X PN30K	AN	¥	CEP 20x0.5	70.0	30.6	30000.0	MKT DE70	815.0	21.67	13.40	0 537	40 00	925	925
R	EOR PN30	A	A	PSC 30	63.0	900.0	5344.2	CON 300	1746.0	34.10	4.50	0.474	6.00	974	1277
133	EOD PN7	NA	M	PSC NU	165.0	0.006	4976.0	CON 300	1477.0	27.96	4.20	0070	17.50	1497	1468
ŝ	RES PN7-61	MA	MA	PSC 18 NU	54.0	324.0	5540.0	CON 5125	1377.0	34.51	9.20	0.606	5.33	490	1044
278	BOR K521C2	¥	M	PSC 14	88.0	196.0	4720.0	VUL 506	530.0	13.58	6.80	0.425	23.33	582	697
279	BOR M27A1	¥	M	PSC 14	93.0	196.0	5122.0	VUL 506	517.0	15.11	7 80	0.483	29.50	565	702
22	EOD PN1	¥	AN	PSC 24 NU	0.66	576.0	5055.0		1210.0	23.54	5.40	0.487	3.17	200	704
280	BOR M29C3	AN	AA	PSC 14	0.86	196.0	4759.4	VUL 506	395.0	11.21	4.40	0.419	0.0	481	547
ğ	RES PN2	AN	M	PSC 18	48 0	324 0	5540.0	CON 5125	753.0	17.12	4.20	0.501	000	285	547
132	EOD PN6	¥	MA	PSC	134.0	0.006	5057.0	CON 300	1463.0	34 77	3.90	0.490	0.0	1211	1368
66 2	BOR 20W6	¥	M	PSC24	92 0	576.0	5688.5	CON E5-2	929.8	21.65	4.19	0.374	00.0	836	266
8	BOR PNPE28	A	M	PSC 30	125.0	0.002	5440.5	CON 300	1757.0	33.05	4.30	0.352	00.0	1507	550
ଞ୍ଚ	EOD PN25	AN	NA	PSC 30 NU	162.0	0.006	5300.0	CON 300	1159.0	18.38	2.60	0.299	00.0	1261	1341
281	BOR M29C3	ž	M	PSC 14	98.0	196.0	4768.6	VUL 506	592.0	17 91	7.20	0.539	13.67	490	702
37	DD PN25	¥	A	PSC 30 NU	162.0	0006	5290.0	CON 300	1586.0	26.76	4.20	0.332	13.33	1290	1578

264
														CAPWAP	ENERGY
REF.	PILE	SKIN	TOE	PILE	LEN.	AREA	EMOD	HAMMER	FMX	EMX	XWX	XWO	BLOWS/	Ruft	APPROACH
g	NAME	soll	soll	TYPE		Ĩ	KSI		KIPS	K-FT	FT/S	z	INCH	KIPS	KIPS
9	EOD PN165	ş	¥	CEP 12.75	0.77	14.6	30000.0	VUL 50C	261.0	6.73	10.06	0.504	71.7	185	251
			2												
			4												
21	BOR TN	AGDITE	AGDITE	PSC 16	38.0	256.0	4280.3	ICE 660	587.0	12.79	5.50	0.371	300.00	578	820
19	RES TN12	AGDITE	AGDITE	PSC 16	65.0	256.0	4278.0	ICE 660	439.0	10.77	4.60	0.403	633.33	532	639
20	BOR CT	AGDITE	AGDITE	PSC 16	65.0	256.0	4278.0	ICE 660	493.0	11.41	4.80	0.384	404.00	715	709
327	X PN213E2	SASI	COOPMAR	PSC 18	38.0	324.0	5500.0	D46-13	908.0	23.51	6.70	0.916	1.08	291	307
125	BOR ET2	CLSI	COOPMAR	PSC 24	91.0	478.0	5302.7	D46-23	1048.0	22.81	5.20	0.524	8.00	337	844
3	BOR PN225	CLSI	COOPWAR	PSC 18 NU	113.0	294.0	30000.0	ICE 1070	694.0	18.93	5.40	0.436	6.00	290	754
8	BOR PN121R9	ល	COOPMAR	PSC 24	76.0	478.0	6060.0	D46-23	1123.0	23.07	5.10	0.643	4.00	224	620
288	BOR PN234A3	SASI	COOPMAR	PSC 18	42.0	324.0	5542.6	D46-13	554.0	11.92	3.40	0.538	2.17	257	286
65	BOR PN120R9	N	COOPMAR	PSC 24	82.0	478.0	0.0903	D46-23	07.0	16.19	4.20	0.477	6.00	257	604
129	BOR ET3	CLSI	COOPMAR	PSC	91.0	477.0	5469.7	D46-23	1194.0	27.69	5.70	0.392	8.33	651	1298
324	X PN201E2	SASI	COOPMAR	PSC 18	35.0	324.0	5500.0	D46-13	846.0	22.40	6.20	1 105	1.17	224	274
208	BOR21B12	COOPMAR	COOPMAR	OCT24	76.0	477.0	5898.5	D46-23	1225.3	27.75	5.93	0 539	11.08	264	1058
131	RES PN120	CLSI	COOPMAR	PSC	82.0	477.0	5 898 5	D46-23	935.0	15.82	4.10	0.393	8.00	236	553
2 04	PN8B43BO	COOPMAR	COOPMAR	OCT.18	117.0	268.4	5898.5	ICE 1070	624.6	18.07	<u>6.06</u>	0 641	2.33	304	405
199	C-41 BOR	COOPMAR	COOPMAR	PSC 18	103.0	324.0	5469 7	CON 125E	732.1	16.85	44,4	0.414	24.00	621	887
326	X PN209E3	SASI	COOPMAR	PSC 18	36.0	324.0	5500.0	D46-13	932.0	22.89	6 70	0 810	1.58	346	381
385	BOR B43P	COOPMAR	COOPMAR	OCT.18	117 0	268 0	5898 5	ICE 1070	852.1	23.91	6.60	0 477	8.00	605	953
325	X PN205E3	SASI	COOPMAR	PSC 18	36.0	324.0	5500.0	D46-13	806.0	19.03	5.90	0.749	1.75	339	346
106	BOR PN123	S	COOPMAR	PSC 24	73.0	478.0	5:0095	D46-23	1054.0	23.68	5.70	0.764	1.33	318	375
63	RES PN119R24	S	COOPMAR	PSC 24	82.0	478.0	6062.0	D16-23	1214.0	22.48	5 40	0 449	3.33	310	720
206	BOR 14B1	COOPMAR	COOPMAR	OCT24	82.0	477.0	5896 0	D46-23	1119.9	24.89	5.55	0.465	5.00	263	898
5	RES PN122B23	S	COOPMAR	PSC 24	73.0	478.0	6062.0	D16-23	1874.0	36.04	8.20	0.675	3.33	312	887
205	PN25BOR	COOPMAR	COOPMAR	OCT.24	82 0	477.0	5898 5	DEL46-23	1136.0	27.14	5.42	0.376	18.17	422	1511
107	BOR PN123R7	ß	COOPMAR	PSC 24	710	478.0	6062.0	D46-23	955.0	20.20	5.20	0.702	1 25	295	323
254	B1P26BOR	SA	LIMESTN	PSC18"SQ	82.0	324.0	4660 6	DEL D36	963.5	20.58	7.12	0.560	4.00	372	610
425	F20.5 MI	S	LIMESTN	PSC24	71.0	576.0	5224.4	D4623	1353.5	21.84	5.39	0 371	22.00	925	1259
386	BOR B12P	SASI	LIMESTN	PSC18	0.77	324.0	5057.0	DEL D36	664.1	12.49	4.27	0.349	65.58	921	832
167	BOR PN898B	SA	LIMESTN	PSC 14	97.0	496.0	4896.5	030	497.0	12.38	5.80	0.447	8.00	458	519
407	B.2P.16B	SA CL	LIMESTN	PSC24	0.67	576.0	5611.1	D46-32	1640.4	28.45	6.41	0.527	00 [.] E	293	794
253	BORPN290	SASI	LIMESTN	PSC14	48.1	196.0	5158.2	LINK520	420.4	6.26	4.23	0.296	6.67	324	336
120	BOR PN115	CLSA	LIMESTN	PSC 12	52.0	196.0	6074.6	ICE 520	550.0	10.39	5.70	0.335	21.00	658	652
144	BOR PNSE25	SA	LIMESTN	PSC 30	104.0	0 006	5469 7	CON 300	1333.0	24.69	0 C C	0 307	5.00	782	1169
398	BOR-E6/4	ರ	LIMESTN	PSC16"SQ	580	256 0	4660 6	90A	632.7	17.63	6.14	0.422	5.50	585	101
143	BOR PN5E22	SA	LIMESTN	PSC 30	120.0	0.006	5469.7	CON 300	1362.0	27 15	3 60	0.333	20.00	915	1701

														CADMAD	ENEDGY
REF.	PILE	SKIN	TOE	PILE	LEN.	AREA	E MOD	HANANER	FMX	EMX	XWV	XWC	BLOWS/	Rult	APPROACH
0N	NAME	SOIL	SOIL	түре	F	IN2	KSI		KIPS	K-FT	FT/S	Z	INCH	KIPS	KIPS
142	BOR PN5E19	SA	LIMESTN	PSC 30	116.0	0.006	5469.7	CON 300	1487.0	30.13	3.50	0.322	30.00	913	2035
256	EODP.26	SASTONE	LIMESTN	PSC18"SQ	82.0	324.0	4722.7	DEL D36	1093.4	33.26	8.11	0.817	3.00	587	694
422	PN B EOD	SA	LIMESTN	PSC24	0.17	576.0	5469.7	CON 300	1318.1	31.57	5.19	0.473	13.00	1396	1378
285	EOD TP3	CLSA	LIMESTN	PSC 10 NU	58.0	100.0	5120.0	ICE 520	375.0	14.98	8.60	0.768	5.50	256	379
284	BOR PNA4W	CLSA	LIMESTN	PSC 14	32.0	196.0	5122.0	ICE 520	360.0	5.18	9 .0	0.241	56.67	436	481
255	B11P50BO	SA	LIMESTN	PSC18"SQ	77.0	324.0	4660.6	DEL D36	841.3	21.75	6.61	0.617	12.00	619	745
393	EOD-IS-3	SA SI	LIMESTN	PSC10	58.0	100.0	4658.0	7011	248.7	5.32	6.09	0.397	8.33	242	247
257	B13P48BO	SA	LIMESTN	PSC18"SQ	0.77	324.0	4660.6	DEL D36	841.3	21.75	6.61	0.617	12.00	619	745
415	EOD PN1	SA	LIMESTN	PSC18	52.4	324.0	4976.4	D16-32	626.1	11.36	4.73	0.369	7.00	454	533
387	EORPN20	SA	LIMESTN	PSC1875Q	75.0	324.0	4660.6	DEL D36	419.9	6.47	3.16	0 284	5.67	330	337
119	BOR PN11	CLSA	LIMESTN	PSC 12	53.0	144.0	6343.5	ICE 520	466.0	11.10	6.20	0.430	18.00	455	549
388	EODPN23	SA	LIMESTN	PSC18-SQ	75.0	324.0	4660.6	DEL D36	419.9	6.47	3.16	0 284	5.67	330	337
196	X TP3	CLSA	LIMESTN	CEP	74.0	9.6	30000.0	LB 440	247.0	7 75	9.50	0.733	20.00	250	238
236	BOR 36	CLAY	LIMESTN	PSC	52.0	196 0	6199 4	MKT DE 4	556.6	6.82	6.17	0.221	20 00	372	603
207	PN9B BOR	MARK	MARK	OCT 24	75.0	477.0	5898.5	DEL 46-2	1329.7	29 60	5 91	0.464	12.50	258	1306
411	BOR 13B6	OVERBURD	MARL	OCT 24	1150	26 0	30000	D46-23	1378.9	31.86	6.19	0 485	35.00	749	1469
417	BOR2/59	MARL	MARL	OCT 18	1150	261 0	5898 5	ICE 1070	847.2	24 17	6 70	0.494	5.50	526	858
418	BOR 4/58	MARL	MARL	OCT 18	1150	261 0	5 898 5	ICE 1070	882.4	26.51	6 84	0.534	5 50	513	688
412	BOR 5B 5	MARL	MARL	OCT 18	116.0	268 0	5898 5	ICE 1070	819.3	2129	6 65	0 458	8.00	508	877
169	EOD PNB4	TILL	ROCK	CEP 18	0 69	275	30000 0	ICE 1070	592 0	19.35	10.90	0 690	6.00	450	542
282	16OS 00	CLSA	ROCK	PSC 10 NU	470	100 0	4950 0	VUL 50C	211.0	4.63	5.20	0.353	90 OC	213	288
283	EOD AB345	CLSA	ROCK	PSC 10	47.0	100.0	4430 0	VUL NO.1	166 0	371	4 40	0 352	8.00	180	187
126	EOD TP1	SASI	ROCK	PSC 12	57.0	144.0	51383	CON 65E5	548.0	15.24	06.6	0.496	10,00	401	614
168	EOD PNB1	TILL	ROCK	CEP 18	69.0	27.5	30000.0	ICE 1070	570.0	17.32	10 30	0 647	4.50	484	478
127	EOD TP2	SASI	ROCK	PSC 12	57.0	144.0	5057.0	CON 65E5	601.0	17.16	10 30	0.580	15 00	386	637
139	BOR PN7N6	SA	SASTONE	PSC 30	100.0	0.006	5469.7	CON 300	1181.0	2135	2 90	0.300	15.00	1477	1397
312	BOR TP8	SA	SASTONE	PSC 14	77.0	196.0	4720.0	ICE 640	765 0	19.72	6.20	0.488	14.00	507	846
138	BOR PN7E	SA	SASTONE	PSC 30	1160	0.006	5288.2	CON 300	1319.0	28.41	3.50	0 374	15 00	1200	1547
306	BOR TP1	SA	SASTONE	PSC 14	77.0	196.0	4720 0	ICE 640	517.0	10.98	5.70	0.352	36.00	544	694
311	BOR TP7	SA	SASTONE	PSC 14	0.77	196.0	4720.0	ICE 640	467.0	11.22	5.20	0 378	10.00	407	563
307	BOR TP10	SA	SASTONE	PSC 14	77.0	196.0	4720.0	ICE 640	661.0	14.21	6.40	0.359	37.50	604	884
60 2	BOR TP5	SA	SASTONE	PSC 14	77.0	196.0	4720.0	ICE 640	316.0	7.22	3.80	0 371	11.50	316	378
141	EOD PNSE18	SA	SASTONE	PSC 30 NU	155.0	0.006	6243.1	CON 300	2231.0	41.64	5.20	0.423	25.00	1492	2158
316	EOD TP8	SA	SASTONE	PSC 14	0.77	196.0	4720.0	KCE 640	509.0	13.57	5.50	0.521	6.00	354	474
146	EOD PN4	SA	SASTONE	PSC 30	134.0	0.006	5385.8	CON 300	1211.0	28.71	3.10	0.432	8.33	1106	1248
308	BOR TP3	SA	SASTONE	PSC 18	0.77	324.0	4767.2	ICE 640	608.0	8.76	4.40	0.243	28 50	476	756
145	BOR PN5	SA	SASTONE	PSC 30	105.0	0.006	5329.8	CON 300	1868.0	35.56	5.10	0.437	5.00	761	1340
5	EOD PN6E25	SA	SASTONE	PSC 30 NU	155.0	0.006	6163.6	CON 300	1512.0	28.53	3.60	0.433	55.05	1364	1479

(continued).
PD
set
data
Jo
c measurements
dynami
and
le/soil
. Pil
Table 24.

														CAPWAP	ENERGY
REF S	PILE	SKIN	₽₹	PILE	N. L	AREA	E MOD	HAMMER	FMX	EMX	VMX S	XWO 4	BLOWS/	Rult	APPROACH
2		201	SOL			7NI	2				2	z			S IN
313	BOR TP9	SA	SASTONE	PSC 14	0.77	196.0	4720.0	ICE 640	614.0	14.46	7.40	0.393	39.00	515	829
314	EOD TP1	SA	SASTONE	PSC 14	0.77	196.0	4720.0	ICE 640	494.0	13.56	5.50	0.705	2.50	234	295
315	EOD TP3	SA	SASTONE	PSC 18	17.0	324.0	4720.0	ICE 640	665.0	10.92	4.70	0.382	13.00	357	571
123	EOD PN22SE	SA	SASTONE	PSC 30	132.0	0.009	5500.0	CON 300	1127.D	26.85	2.80	0.398	8.33	1167	1244
310	BOR TP6	SA	SASTONE	PSC 18	0.77	324.0	4814.4	ICE 640	740.0	13.42	5.50	0.362	0 0.6	320	681
74	EOD PNE17	ß	SHALE	CEP 11x0.4	41.0	12.2	30000.0	LB 640	531.0	23.70	15.40	0.913	20:00	200	610
			!												
LARGI	E DISPLACEMENT	PILES IN SA	g												
212	EOD PN316	CLSI	SA	PSC 12	52.0	144 0	5875.4	CON 65	391.0	10 12	6 30	0 483	7.58	470	395
213	EOD PN332	CLSI	SA	PSC 12	62.0	144.0	5652 1	CON 65	411.0	11.65	6 80	0.712	2.42	203	248
197	BOR PN14	SASI	SA	PSC 14	92.0	196 0	5496.7	VUL 010	628.0	16 64	6.70	7 60.0	5.00	690	699
81	BOR TP11	SASI	SA	PSC 36	47.0	487.0	6720.0	CON 300E	1287.0	31.80	5 50	0.464	10.00	9/6	1353
215	BOR P5	SA CL	SA	PSC 14"	42.5	196 0	5469.7	KOBE 25	495.8	10 07	5.77	0.317	15.00	413	631
209	DD PN69	CLSI	SA	PSC 12	72.0	144.0	6053 5	CON 65	366.0	9.01	5 80	0.512	10 50	400	356
198	BOR PN24	SASI	SA	PSC 14	82 0	196 0	5469.7	VUL 010	551.0	14.90	6.10	0 397	10.00	573	720
7 EE	EOD TP4	SASI	SA	CEP 12 75	57.0	9.8	30000.0	VUL 506	256.0	11.01	11 80	0.807	10 00	262	291
210	EOD PN232	CLSI	SA	PSC 12	68.0	144.0	5690.9	CON 65	490.0	12.94	8.10	0 571	4.17	410	383
270	199.EOD	SA/SISA	SA	PSC12	46.0	144.0	5896.0	D30-23	479.7	13 86	5 63	0.573	2 92	302	363
83	BOR TP21	SASI	SA	PSC 36	63 0	487 0	6782.4	CON 300E	1390 0	33.97	5.60	0 473	8.00	864	1363
364	BOR TP3	CLSA	SA	PSC 18	102.0	324.0	6885 4	D46-32	1724.0	43.89	10.40	0.405	10 00	1264	2086
383	1P8	SA	SA	OCT 16.5"	92.0	225.0	0 06/9	D36-23	866.4	21.62	7.19	0.412	16.00	601	1095
220	EOD L	SA CL	SA	PSC 14 ⁻	47.5	196.0	6343.5	KOBE 25	600.0	12.19	6 08	0.416	10 00	372	568
216	EOD P7	SA CL	SA	PSC 14"	47.5	196.0	5898.5	KOBE 25	652.3	13.57	7.38	0 442	8.00	359	574
222	EOID P6	SA CL	SA	PSC 14"	47.5	196.0	5469.7	KOBE 25	67.9	11.45	8.09	0.405	6.42	287	490
322	B 70 P 5	ರ	SA	PSC24	66.7	5760	5888.9	V 020	987.4	19 38	3.83	0 353	41.67	1259	1235
271	293.BOR	SA/SASI	SA	PSC12	47.0	144.0	5896.0	D30-23	6:00:9	14.50	6 22	0.368	833	550	714
149	BOR PN12	SA	SA	CEP 16	0.77	16.8	30000.0	K25	359.0	14 07	11.10	0 727	70.83	411	456
219	EOD P11	SACL	SA	PSC 14"	42.5	196.0	6343 5	KOBE 25	604.3	9.66	7 12	0.299	15 00	383	634
ŝ	EOD PA8	AN	S A	PSC 18	114.0	324.0	5787.0	VUL 020	891.0	26.47	6.40	0.686	8.33	598	788
330	X TP1799	S	SA	PSC	20.0	96.5	6190.0	MKT DE33	2510	6.84	4 90	0 600	2.67	137	168
181	EOR PN1056	SASI	SA	PSC 14	77.0	196.0	6450.9	VUL 020	793.0	25 23	8.30	0 625	2.50	408	591
73	EOD PN398	CLSI	SA	PSC 20	78.0	400.0	5682.1	D46-32	1611.0	37.17	9 10	0.480	7.25	740	1444
214	EOD TP1	CLSI	SA	PSC 12	52.0	144.0	5655.4	CON 65	358.0	9.86	5.90	0.485	10.00	450	405
91	EOD TP21	SASI	SA	PSC 36	63 0	487.0	6782.4	CON 300E	1081.0	26.87	4 60	0.445	867	850	1151
165	BOR PN1056	CLSI	SA	PSC 14	77.0	196 0	5057 0	VUL 512	560.0	32 11	6.40	0 894	5 00	502	704
11	EOD PN392	CLSI	SA	PSC 20	79.0	400.0	5794 2	VUL 512	1252 0	27 85	7.20	0444	14 00	728	1297
6	PN1	¥	SA	CEP 16	670	12.4	0.0000E	D16-32	338.0	13 12	14.90	0 742	5 00	265	334

														CAPWAP	ENERGY
REF.	PILE	SKIN	10E	PILE		AREA		HANNUER	FMX	EMX E	Ň	XWQ	BLOWS/	Rull Vioc	APPROACH
į											2				
336	BOR TP4	SASI	SA	CEP 12.75	57.0	9.8	30000.0	VUL 505	264.0	11.95	12.70	0.841	11 00	267	308
377	151-BOR	CL,SC,S	SA	PSC12	42.0	144.0	6230.0	MKT DE40	490.5	10 40	69.69	0.332	19.00	486	650
164	BOR PN1040	CLSI	SA	PSC 14	77.0	196.0	6343.5	VUL 512	621.0	19 18	6.50	0.483	8.00	999	757
211	EOD PN244	CLSI	SA	PSC 12	66.0	144.0	5595.6	CON 65	450.0	12.00	7.30	0.637	3.08	300	00 0
221	EOD PB	SA CL	SA	PSC 14 ⁻	47.5	196.0	6343.5	KOBE 25	9 669	14.23	7.18	0.409	0 0.6	443	657
286	BOR TP2	CLSA	SA	PSC 14	38.0	196.0	5115.6	VUL 06	310.0	6.66	4.40	0.428	2.50	185	193
190	TP2 BOR	SVCL	SA	PSC 12"	65.0	144.0	5898 5	VUL.512	561.4	16.42	8.90	0.650	5.00	317	464
287	EOR TP1	CLSA	SA	PSC 14	48.0	196.0	5243.4	70L 06	345.0	6 0.8	6.00	0.606	2.83	193	202
217	EOD P7	SA CL	SA	PSC 14"	47.5	196.0	5898.5	KOBE 25	652.3	13.57	7.38	0.442	8.00	377	574
218	EOD P 10	SA CL	SA	PSC 14 ⁻	42.5	196.0	6253.2	KOBE 25	530.1	9.82	5.91	0.330	12.00	396	570
72	EOD PN396	CLSI	SA	PSC 20	79.0	400.0	5680.0	D46-32	1597.0	37 29	8.70	0.486	6 67	810	1407
331	151.EOD	CL/SA	SA	PSC12	42.0	144.0	6230.0	MKT DE40	531.7	12.41	17.1	0.402	10.58	499	600
150	EOD PN12	SA	SA	CEP 16	77.0	16.8	30000.0	K25	368.0	15 72	10.90	0.787	36.17	438	463
3 89	EOD TP1	SA SAT	SA SAT	CEPIPE 1	72.3	12.4	30000.0	D16-32	408.2	15.98	15.76	0 758	6.67	269	422
419	TP2 BOR	SA SI	SA SI	PSC24	96.0	576.0	7040 0	D80-23	1506.4	40 95	491	0.410	27.00	1630	2199
79 7	EOD-TP1	SA SI	SA SI	CEP16"	56.0	12 4	30000.0	D16-32	356.0	13.94	16.79	0.793	20.00	262	397
426	EOD FHAS	SA	SA TILL	CEP18	117.3	27.5	0 0000E	K45	550.5	26.98	10.73	1.034	150	346	381
368	BOR PN317	S	SAGR	CEP 14x0 37	124 0	161	30000.0	MH 35	523 0	34.18	15.30	1.169	10.00	513	646
369	EOD PN317	SI	SAGR	CEP 14x0.37	125.0	16.1	30000.0	MH 35	465.0	34.61	15 00	1.343	3.00	383	496
367	BOR PN243	S	SAGR	CEP 14x0.37	130.0	161	30000.0	MH 35	539.0	40.98	16.00	1.313	12.00	622	704
8	EOD PN15E	NA	SAGR	CEP 10.75	120 0	101	30000.0	VUL 06	209.0	12.27	11.38	0.925	6.67	235	274
335	X PN2031	SASI	SAROCK	PSC 12	57.0	144 0	5169.7	CON 65E5	481.0	15.34	0 6'8	0.628	00 6	342	498
350	EOD PN13	CLSI	SASI	PSC 12	30 0	144.0	4400.0	D30-23	556.0	19.12	8 60	1.083	1.33	132	250
4	BOR PM24	SASI	SASI	PSC 20	126.0	400 0	5947.0	VUL 020	925 0	23.50	5.60	0.476	7.00	558	911
32	BOR TP26	SASI	SASI	PSC 12	640	144.0	4660.6	D30-23	654.0	26.04	10.00	0.660	4.00	328	687
348	BOR PN43	CLSI	SASI	PSC 12	30.0	144.0	4400.0	D30-23	539.0	15 98	7.10	0.480	2.00	338	391
87	EOD PN3	SIGR	SASI	CEP 12x0 6	60.0	14.6	30000.0	ICE 520	296.0	10.00	10.00	0 695	10.67	312	304
82	BOR TP11	SIGR	SASI	PSC 36	47.0	487.0	6720 0	CON 300E	1287.0	32.00	5.50	0.464	10.00	878	1362
m	BOR PA35	SASI	SASI	PSC 20	126 0	400.0	5839.0	VUL 020	928.0	22.95	5.40	0 565	4 00	405	676
354	EOR PN310	CLSI	SASI	PSC 12	32.0	144.0	4400.0	D30-23	368.0	2.00	6.00	0.311	4 00	203	299
342	BOR PN13	CLSI	SASI	PSC 12	30 0	144.0	4400.0	D30-23	574.0	19.60	7.80	0.628	1.67	285	383
303	RES G37	SASI	SASI	PSC 14	83.0	196.0	53315	VUL 010	494.0	14.29	5.40	0.501	4 00	336	457
E	BOR PN19	CLSI	SASI	PSC 12	300	144.0	4400.0	D30-23	586.0	16.24	7.40	0.524	1.67	292	347
301	RES 858	SASI	SASI	PSC 14	85.0	196 0	5250.0	VUL 010	439.0	11.35	4.90	0.424	12 50	392	540
344	BOR PN218	CLSI	SASI	PSC 12	30.0	144.0	4400.0	D30-23	469.0	13.90	7.50	0471	2.00	294	344
¥	EOD TP26	SASI	SASI	PSC 12	62.0	144 0	4660 0	D30-23	592.0	23 21	10.20	0.995	1 25	206	310
92	EOD TP23	sigr	SASI	PSC 54	540	905.0	7420.0	CON 300E	2220.0	29.60	4.90	0.304	25.17	1182	2067
269	EOD IP3	SASI	SASI	PSC 12	65.0	144 0	5100.0	K25	618.0	19 00	066	0 670	5 83	400	542

REF	PILE Name	SKIN		PILE TYPF	LEN.	AREA IN2	E MOD KSI	HAMMER	FMX KIPS	EMX K-FT	VMX FT/S	XW M	BLOWS/	CAPWAP Rult KIPS	ENERGY APPROACH KUPS
					:		2				2				
351	EOD PN213	CLSI	SASI	PSC 12	32.0	144.0	4400.0	D30-23	477.0	18.34	8.40	1.275	1.33	118	217
302	RES F14	SASI	SASI	PSC 14	91.0	196.0	5200.0	VUL 010	505.0	15.74	5.90	0.519	4.33	352	504
349	BOR PN49	CLSI	SASI	PSC 12	30.0	144.0	4400.0	D30-23	440.0	9.41	5.30	0.316	2.17	335	290
347	BOR PN37	CLSI	SASI	PSC 12	32.0	144.0	4400.0	D30-23	513.0	12.92	6.80	696.0	2.50	402	403
8	EOD TP13	SIGR	SASI	PSC 36	46.0	487.0	7270.0	CON 300E	1109.0	23.51	4.70	0.348	27.33	026	1467
800	EOD TP9	SASI	SASI	PSC 18	56.0	324.0	5057.0	MKT DE70	554.0	9.71	3.90	0.369	4.17	264	383
353	EOD PN49	CLSI	SASI	PSC 12	30.0	144.0	4400 0	D30-23	436.0	16.54	8.00	1.111	1.00	127	188
352	EOD PN26	CLSI	SASI	PSC 12	32.0	144.0	4400 0	D30-23	470.0	20.33	8 70	1.366	1.08	110	213
346	BOR PN31	CLSI	SASI	PSC 12	32.0	144.0	4355.1	D30-23	555.0	14.92	0 £'' 1	0.367	2.83	9 60	484
79	BOR PN3	SIGR	SASI	CEP 12x0.6	60.09	14.6	0.0000E	ICE 520	401.0	15.18	12 60	0.714	19.00	376	475
6 9	EOD TP11	SIGR	SASI	PSC 36	47.0	487.0	6720.0	CON 300E	1068.0	23.99	4.50	0 413	19.58	786	1241
33	EOD TP114	SASI	SASI	PSC 12	64.0	144.0	4583.2	D30-23	591.0	28.02	0C 6	1.243	1 08	250	310
268	EOD IP1	SASI	SASI	PSC 12	65.0	144.0	5346.0	K25	543.0	16.39	8.60	0 566	6.25	409	542
345	BOR PN26	CLSI	SASI	PSC 12	32.0	144.0	4400.0	D30-23	504.0	13.69	6.70	0.397	2 67	385	426
8	BOR TP114	SASI	SASI	PSC 12	64.0	144.0	4583.2	D30-23	694 0	34.86	11.80	0 951	3.00	325	651
299	RES PN6	SASI	SASI	PSC 14	107.0	196 0	5171.2	VUL 506	422.0	10.34	4.90	0.418	12.00	495	495
31	BOR TP2	SASI	SASI	PSC 12	640	144.0	4583 2	D30-23	670 0	26.07	11.20	0.641	3.00	320	642
318	B9P3BOR	SACY	SACY	PSC18	41.0	324 0	5120.0	D4602	817.6	13 30	4.26	0.239	8.00	628	876
234	7S ABUT	Sa/Si/Gr	Sa/S//Gr	CEP 10.75	537	83	30000 0	VUL 50C	127 2	5 37	8 00	0.656	5.00	135	150
46	RES PN125	N	ĩ	PSC 24	82.0	4780	6062.0	D46-23	1553.0	33 67	7.10	0516	4.00	243	1055
49	RES PN125	SI	SI	PSC 24	82 0	4780	6062 0	D62-22	1008 0	26.81	4 90	0.810	3 00	287	563
370	RES TP13	S	SIGR	CEP 14x0.37	84.0	16 1	29990.9	ICE 422	284.0	6.85	7.60	0 369	40.00	323	417
371	RES TP6	S	SIGR	CEP 14x0.37	79.0	16.0	29990.9	ICE 422	326.0	7.59	7.60	0.402	78 00	368	439
320	P3 P1.R	SUSA	SUSA	PSC18	63.0	324.0	4672.8	D 46-02	757.2	10.75	4.35	0.226	5.00	532	606
323	PIER 7 P	SVSA	SUSA	PSC18	87.5	324.0	5244.5	D4602	1026.9	23.28	6 83	0.354	8.00	480	1166
T I W MS	NICDI ACEMENT	DI EC IN CI V													
40	BOR D12N	ರ	ರ	HP 10x42	46.0	12.4	30000.0	MKT 983	208.0	2.68	10.20	0 237	34.00	140	241
245	TP4 EOD	ರ	ರ	HP 14x73	27.0	21.4	30000.0	KOBE 25	398.4	9.43	9.78	0.478	4.00	226	311
152	BOR R1	CLSI	บ	HP 14x73	0.17	214	30000.0	D30	548.0	19.48	13.30	0 577	8.00	422	999
153	BOR R10	CLSI	ರ	HP 14x73	72 0	21.4	30000.0	020	583.0	21.44	13.80	0.602	10.00	456	733
246	TP4 BOR	ರ	ರ	HP 14x73	27.0	214	30000.0	KOBE 25	433.8	10.71	10.72	0 493	4.25	235	353
154	BOR T1	CLSI	ರ	HP 10x42	72.0	12.4	30000.0	D30	392.0	19.51	14.00	0.813	8.00	375	499
41	EOD D18E	ರ	ರ	HP 10x42	44.0	12 4	30000.0	MKT 9B3	223.0	2.41	10 50	0.220	23.00	117	220
160	EOD T1	CLSI	ರ	HP 10x42	72 0	12.4	30000 0	D30	336.0	17.99	13.20	0.793	2 83	230	377
158	EOD R1	CLSI	ป	HP 14x73	71.0	214	30000.0	D30	511.0	16.92	12.40	0.543	317	285	473
159	EOD R10	CLSI	ษ	HP 14x73	72.0	21.4	30000.0	020	498.0	14.75	12.00	0.495	6 67	316	549

			Tabl	e 24. Pilc/s	oil and	dynami	c measur	ements of e	lata set	PD (co	ntinue	d).			
														CAPWAP	ENERGY
REF.	PILE	SKIN	TOE	PILE	LEN.	AREA	E MOD	HAMMER	FMX	EMX	XW I	XIMO	BLOWS/	Ruit	APPROACH
g	NAME	sol	SOL	IYPE		ZNI	KSI		KPS	Ļ	FT/S	z	RCH	KIPS	KIPS
		đ	ð	00-4 A011	ļ	į			0000						
Į		ז ו	ני ני		47.1	71.4	0.0000		アカナ	17.1	0.0	0.402	8.5	507	741
424	BOR TP2	CL SI	CL SI	HP14x73	42.1	21.4	30000.0	K25	506.3	18.07	13.30	0.768	2.33	331	362
243	EOD TP3	CL SI	ព	HP 14X73	42.1	21.4	30000.0	K25	348.4	7 60	8.97	0.444	4.83	264	281
423	EOD TP2	CL SI	CL SI	HP14x73	42.1	21.4	30000.0	K25	506.3	18.07	13.30	0.768	2.33	326	362
185	EOD TP1	CLSA	CLSA	HP 10x42	49.D	12.4	30000.0	FEC 1500	288.0	8.10	11.50	0.624	1.75	145	163
274	BOR TP2	CL-SA	CL-SA	HP14	42.0	21.4	30000.0	K25	465 1	10.80	11.39	0.549	4.42	252	335
6 4	EOR D112E	ರ	ערר	HP 10X42	53.0	12.4	30000.0	MKT 983	238.0	2.74	10.90	0.219	60.00	201	279
187	EOD PN40	ರ	TILL	HP 12x74	47.0	21.8	30000.0	LB 520	473.0	12.28	11 50	0.486	26.00	517	562
4	EOR D6W	ರ	LILL	HP 10x42	59.0	12 4	30000.0	DROP	207.0	7 39	8.70	0.509	00.09	221	756
186	EOD PN38	ರ	LILL	HP 12X74	47.0	21.8	30000.0	LB 520	390.0	9.86	9.40	0.430	26.00	489	505
42	EOR D1105	ರ	עורר	HP 10x42	55.0	12.4	30000.D	DROP	224.0	7 23	06.9	0.475	20.00	256	351
SMAL	L DISPLACEMENT	PILES IN ROC	X												
273	69.EOD	SCORIA	CLAYSTN	HP10x57	43 3	16.7	30000.0	MKTDE50	522.4	24.65	15 66	0.810	6 00	473	606
272	177 EOD	SCORIA	CLAYSTN	HP10x57	43.3	16 7	30000	MKTDESO	4736	17 68	13.67	0 661	6.00	421	513
189	BOR PN224	SASI	COOPMAR	HP 12453	40.0	15.5	30000	K25	3740	11 72	11 90	0.526	10 00	305	449
427	BOR PN246	CLSA	COOPWAR	HP 14x73	54.0	214	30000	K25	5410	14 52	13.20	0.455	00 6	475	616
67	BOR PN280	CLSA	COOPMAR	HP 14x73	48.0	214	30000	K25	4430	11 91	10 60	0 431	7.00	383	498
188	BOR PN117	SASI	COOPINAR	HP 14×89	48.0	26 2	30000	K25	2665	12 44	12 00	0.369	18 00	504	203
18	EOD JB	CLSHALE	DOLOM	OEP 96	118.0	16 0	30000	RAY 5/0	417.0	22.04	12 50	0 933	55.00	511	556
5	BOR D418	CLSHALE	DOLOM	0EP 96	45 0	16 0	30000	RAY 150C	468 D	10 13	7.50	0 458	47 00	482	507
16	EOD D418	CLSHALE	DOLOM	OEP 9.6	45.0	16 0	30000.0	RAY 150C	454.0	12.26	9.70	0519	00:06	490	555
14	BOR J3	CLSHALE	DOLOM	0EP 9.6	113 0	16 0	30000.0	RAY 5/0	431.0	22.23	13.80	0.826	33 33	563	623
17	EOD J1	CLSHALE	DOLOM	0EP 9.6	162.0	16.0	30000.0	RAY 5/0	378.0	26.02	12.80	1.266	26.00	460	479
15	EOD D205	CLSHALE	DOLOM	0EP 9.6	77.0	16.0	30000.0	RAY 150C	378.0	14 84	06.6	0.740	36.00	380	464
193	DD PN355E6	SASI	LIMESTN	HP 10x42	24.0	12 0	30000.0	D12	311.0	6.94	12.60	0.448	2 00	222	282
180	EOD PN7	CLSA	LIMESTN	HP 10x57	72.0	16.8	30000.0	LB 520	314.0	10.86	8 80	0.580	4167	364	432
195	EOD PN375D2	SASI	LIMESTN	HP 10x42	32.0	12.0	30000.0	D12	304 0	6.87	12.60	0.445	6.00	308	270
67	BOR SHD1	CLSA	LIMESTN	HP 12x53	82	15.5	30000.0	ICE 640	271.0	11.92	8 30	0.721	2 00	217	234
233	EOD A-5-	CLAY	LIMESTN	HP10X57	67.0	16 7	30000.0	DEL D16-	453.7	13.61	15.84	0.640	10 00	409	441
8	EOD SHD1	CLSA	LIMESTN	HP 12x53	82.0	15.5	30000 0	ICE 640	224.0	11.27	7.20	0.920	1.33	148	162
<u>19</u>	DD PN375P1	SASI	LIMESTN	HP 10x42	32.0	12.0	30000.0	D12	306.0	7.13	12.90	0.472	12.00	240	306
232	BOR TP13	SA	LIMESTN	HP14X74	38.0	21.6	30000	MKT 9B3	394.4	4.72	10.99	0.223	41.67	C 6E	458
428	TP4EOD	SA	ROCK	HP12x74	36.3	218	30000	D25-32	520.9	19.31	13.42	0.730	5 00	297	498
249	EOD 4 DI	ե	ROCK	HP 12X53	58.0	15.6	30000 0	IHC S35	459.6	16 51	14 92	0 588	20 00	473	621
170	EOD PND10	CLTILL	ROCK	OEP 18	64.0	27.5	30000 0	ICE 1070	6010	17.90	11.30	0.685	5.00	531	485
248	PN28FOD	บี	ROCK	HP10x57	48.0	16.7	30000.0	KOBF K-1	354.0	8.04	10 27	0.428	166 67	348	445

.

						a y name			חקרם קרר			-/-			
														CAPWAP	ENERGY
	PILE	SKIN		PILE	Nj t	AREA		HAMMER	FMX	Ň,	AXX NX	XWO 4	BLOWS/	Ruft	
		SUL	SOL			N	2		2 Z		2	z		KIPS	CHIN
229	PILE 6 B	CL/SI	ROCK	HP14X117	97.0	34.4	30000.0	VUL 016	589.6	25.16	9.12	0.685	33,33	664	844
8	EOD PN12	ರ	ROCK	HP 10X57	57.0	16.8	30000.0	101 08	505.0	9.74	11.20	0.496	633.33	505	470
405	PN8 EOD	SA	SHALE	HP10x57	37.0	16.8	30000.0	V 06	275.6	5.62	6.83	0.369	22.33	302	311
408	EODTP2	M	SHALE	HP10x57	37.3	16.8	30000.0	90A	168.8	4.09	5.03	0.371	15.00	191	224
406	PN7 EOD	SA	SHALE	HP10x57	37.0	16.8	30000.0	80 N	275.6	5.62	6.83	0.389	22.33	302	311
SMALL	. DISPLACEMENT	. PILES IN SAN	g												
184	RES TP3	SI	GR	HP 10x12	88.0	12.4	30000.0	ICE 440	202.0	6.94	02.6	0.624	10 00	179	230
ŝ	BOR TP1	SA	SA	HP 14x73	87.0	21.4	30000.0	90 T/N	316.0	9 86	8.20	0.725	2.00	126	194
121	EOD TP2	SASI	SA	HP 12x179	38.0	67	30000	FEC 1500	204.0	11.07	13.80	0.868	5 00	225	244
151	EOD PN3	SASI	SA	HP 12x74	63.0	21.8	30000.0		511.0	21.64	13.50	0.832	6.00	415	520
420	TP1 EOD	SA CL	SACL	HP14x73	60.3	214	30000.0	K25	464.4	12.97	12.11	0.584	5 00	353	397
421	BOR TP1	SACL	SA CL	HP14x73	60.3	21.4	30000.0	K25	434.9	12 87	10 69	0.551	8.00	390	457
395	BOR-TP4	SA SI	SA SI	HP14x11	117.0	261	30000 0	K-45	5915	25 61	11.09	0.903	100	238	323
46	EOD E18	CLSI	SASI	OEP 12	33 O	81	30000	VUL 06	146 0	05 6	06.6	1.244	1 25	95	109
8	BOR PN4	SIGR	SASI	OEP 12x0 6	009	146	30000 0	ICE 520	339.0	11 88	11.80	0.677	15 00	318	383
88	EOD PN4	SIGR	SASI	OEP 12x0 6	60 0	146	30000	ICE 520	340 0	12 86	12.00	0:730	7.92	345	360
23	BOR PN9	SASI	SASI	HP 12x74	42 0	218	30000	VUL 01	2740	4.27	6.30	0 230	833 33	340	443
24	DD PN83	SASI	SASI	HP 12x53	58 0	15.6	30000	K13	422 0	11 30	13.70	0.553	1 92	177	252
26	X PN71	SASI	SASI	HP 12x53	112 0	15.6	30000 0	K13	343.0	9.24	11.80	0 540	3 58	204	271
25	DD PN83	SASI	SASI	HP 12x53	104 0	15.6	30000 0	K13	336.0	8 62	11.20	0.524	3.58	212	258
45	BOR A213	CLSI	SASI	OEP 12	86.0	81	30000	VUL 06	190.0	10 24	10.50	0.911	8 33	205	238
28	X PN80	SASI	SASI	HP 12x53	58.0	15.6	30000.0	K13	383.0	9.87	12.50	0.458	3 50	266	319
47	EOD J210	CLSI	SASI	0EP 12	67.0	8.1	30000.0	VUL 06	166.0	10.54	10.60	1.128	167	106	146
8 8	EOD PN2	SIGR	SASI	HP 10x42	56.0	12.4	30000.0	ICE 520	208.0	11 49	7.80	0.933	3.17	202	221
27	X PN80	SASI	SASI	HP 12x53	58.0	15.6	30000.0	K13	340.0	7.97	11.20	0.475	2 50	186	219
85	EOD PN2	SIGR	SASI	HP 10x42	56.0	12.4	30000.0	ICE 520	214.0	12.03	8.30	0.934	3.08	210	229
84	EOD PN1	SIGR	SASI	HP 12x74	56.0	21.7	30000.0	ICE 520	399.0	12.01	9.70	0.564	5 42	354	385
227	EOR PN12	SA	SACOBBL	HP14x89	66 4	26.1	30000.0	K25	623.6	13.48	10 69	0.440	7 25	471	560
228	EOD PN2	SA	SACOBBL	HP14X89	103.1	26.1	30000.0	K25	494.5	12.10	9.97	0.505	10 00	432	480
230	K7EOD	SA/SI	SA/SI	HP14X73	1545	21.4	30000.0	VUL010	508.6	21.04	13.32	0.832	16 00	391	564
384	BOR TP1	ŝ	SI	HP305MMX	68.8	17.0	30000.0	16T X 3M	391.2	14.67	10.82	0.582	4 00	290	423
231	K2BOR	SVSA	SVSA	HP14X73	1715	214	30000 0		540.5	21.21	14.35	0.838	20.83	420	575
MISCEI	LLANEOUS PILE 1	LYPES IN CLA	>												
396	BOR-PN43	S A	ರ	MONO 11"	38.0	4	30000	D16-32	406.7	13.81	14 97	0 603	16 00	358	498

						in the second se								CAPWAP	ENERGY
REF.	PILE	SKIN	TOE	PILE	LEN.	AREA	E MOD	HAMMER	FMX	EMX	XWX	DMX	BLOWS/	Ruh	APPROACH
ğ	NAME	SOIL	SOIL	TYPE	F	IN2	KSI		KIPS	K-FT	FT/S	Z	INCH	KIPS	KIPS
297	RES PN5	¥S	0	MONO 12	56.0	06	30000.0	ICE 520	309.0	9.62	007	0 622	40.00	310	357
373	RES TP2	e S	ರ	PIPE 12	0.88	6.9	30000.0	MKT DE30	162.0	7.55	14.00	0.743	20.00	172	228
292	BOR TP1	SA	ರ	MONO 12	62.0	0.6	30000 D	ICE 520	350.0	12.64	11.50	0.703	2.25	8	264
414	BOR PN3	ರ	ರ	PIPE24	85.1	54.8	30000.0	K35	969.8	19.38	10.00	0.659	11.00	60	620
241	E2 BOR	ರ	ರ	TIMBER	47.0	50.0	1602.0	MKT 1083	183.8	3.93	11.74	0.343	15.00	143	230
372	RES TP1	SA	ರ	PIPE 12	0.06	7.0	30000.0	MKT DE30	148.0	7.94	13.00	0.830	11.67	1 98	208
379	A4-21-EO	ರ	ರ	PIPE26	868.0	78.5	29700.0	D62	1597.6	35.71	11.56	0.564	4.17	821	1066
298	RES PN6	SA	ರ	MONO 12	51.0	9.0	30000.0	ICE 520	293.0	7.29	7.60	0.485	130.00	312	355
375	RES TP4	SA	ರ	PIPE 12	78.0	7.0	30000.0	MKT DE30	161.0	5.42	10.40	0.641	100 00	131	200
240	A2 EOID	ರ	ರ	TIMBER	42.0	50.0	1300.0	MKT 10B3	172.9	6.19	11.56	0.711	2.00	1	123
374	RES TP3	SA	ರ	PIPE 11 NU	92.0	16 1	30000.0	MKT DE30	2760	9.93	9.20	0.868	7 67	238	239
413	BOR PN3	ರ	ರ	PIPE24	85.1	54.8	30000.0	K35	943.9	25.74	10.24	0.882	00.6	374	622
296	RES PN4	SA	ರ	MONO 12	52 0	06	30000.0	ICE 520	320.0	9.75	9.20	0.602	13.00	346	345
157	BOR 14	CLSI	ರ	PIPE 12.75	66.0	19.2	30000.0	0:0	502.0	17.43	13.50	0.517	5 00	288	583
163	EOD T4	CLSI	ರ	PIPE 12.75	66 0	19 2	30000.0	D30	5140	16.72	13.00	0.531	2 50	244	431
2	RES PN6	SA	CLSI	MONO 11	380	8.1	30000.0	D22-02	316.0	20.56	12.80	1.037	317	271	365
108	EOD TP23	SA	CLSI	MONO14 NU	42 0	81	30000.0	D22	143.0	7.47	11.30	0.895	167	124	120
111	BOR TP24	SA	CLSI	MONO14 NU	42 0	8.1	30000.0	D22	262.0	15.91	12.70	668.0	283	224	305
12	EOD TP12	SA	CLSI	ONOM	44 0	81	30000.0	D16-32	282.0	11.45	14.30	0.741	2.50	246	241
110	BOR TP23	SA	CLSI	MONO14 NU	42.0	8.1	30000.0	D22	278.0	14.96	12.50	0.852	2.83	226	298
MISCE	LL ANEOUS PILE TY	PES SOIL T	YPE UNKNOV	N											
171	EOD B12	M	NA	PIPE 14	101.0	21.3	30000.0	D30-32	633 0	30 32	16.10	1.005	20.83	602	691
172	EOD B24	AN	M	PIPE 14	0.79	21.3	30000.0	D30-32	650.0	28.73	16.20	0 958	100	428	352
8	EOD TP4	¥	AN	PIPE 10.75	380	7.3	30000.0	FEC 1200	195.0	8.56	14.00	0.756	3.50	197	197
203	X PND161.1	Ą	MA	PIPE 12.75	0 .6E	9.8	30000.0	CON 65	229.0	6.72	7.90	0.655	12 00	221	218
202	X PNB141	¥	Ą	PIPE 12.75	<u> 39.0</u>	9.6	30000.0	CON 65	254.0	6.71	7.70	0.581	12.50	247	244
ansim	T ANECLIS BILE TV	DEC IN BUC	×												
			Ę												
201	X PN2	CLSI	COOPMAR	PSC/HP	118.0	478.0	5469.7	D46-23	1483.0	39.61	6.50	0.696	20.00	924	1274
128	BOR PN11B1	CLSA	COOPMAR	HP/PSC NU	142.0	477.0	6064.1	D46-23	882.0	17.66	3.70	0.361	63 33	850	1136
200	RES PN810	CLSA	COOPMAR	PSC/HP	112.0	478.0	5467.0	D46-23	1243.0	31.64	6.30	0.549	20.00	069	1268
179	BOR PNH218	N	COOPMAR	HP/PSC	123.0	268 0	5055 0	ICE 1070	682.0	17.94	5.80	0.460	12 00	622	792
<u>1</u> 30	BOR ET4	CLSI	COOPMAR	HP/PSC	119.0	4770	5467 0	D46-23	1275.0	31 40	5.80	0.641	10 00	864	1017
124	BOR ET2	CLSI	COOPMAR	HP/PSC24	119.0	4780	5470.0	D46-23	1070.0	26.15	5.10	0 608	56 00	718	1003
381	EOD PN3x011	CLSI	LIMESTN	PIPE	62.0	14 3	301112	LB 520	3180	10.37	10.90	0.604	40.00	348	396

			Tabl	e 24. Pile/so	il and	dynami	c measur	ements of	data set	PD (co	atinuc	d).			
														CAPWAP	ENERGY
REF.	PILE	SKIN	TOE	PILE	LEN.	AREA	E MOD	HAMMER	FMX	EMX	XMV	XWQ	BLOWS/	Ruft	APPROACH
ġ	NAME	SOIL	SOIL	түре	FT	IN2	KSI		KIPS	К. Г	FT/S	Z	INCH	KIPS	KIPS
Ę		0.0			6				0.000			001.0	01.0		
3 3							0.0000		0.002	8.7	0.0		8.5	8	C07
70					0.50				312.0	N C.21	0.00	0.032	B .5		
2	POR PN8964	¥0		P1PE 9.0	0.18	0	0.0000		0.662	56.0	13.00	0.6/3	8.00		907
382	EOD TP	CLSI	LIMESTN	PIPE	64.0	14.3	29839.9	LB 520	374.0	16.06	13.70	0.802	40.00	421	466
ŝ	EOD P4T1	SASI	ROCK	PIPE 12.75	37.0	14.6	30000.0	VUL 06	289.0	10.19	11.60	0.605	5.00	232	304
251	BOR J31	CL/SI	ROCK	PIPE 14"	91.0	31.2	30000.0	KOBE K-3	860.7	27.18	15.17	0.651	58.33	507	976
1 03	BOR PST1	SASI	ROCK	PIPE 12.75	75.0	14.6	30000.0	70L 06	276.0	60.6	10.30	0.554	8 33	248	324
į	EOD P311	SASI	ROCK	PIPE 12.75	75.0	14.6	30000.0	VUL 06	331.0	12.13	12.50	0.741	5.00	281	6 0£
250	EOD J31	CL/SI	ROCK	PIPE 14"	91.0	31.2	30000.0	KOBE K-3	681.7	35.24	11.65	0.999	20.00	705	806
263	EOD PN10.375	CLSI	SHALE	PIPE 7	42.0	10.5	30000.0	D25-32	267.0	15.94	12.50	1.310	0.83	128	152
261	DD PN10	CLSI	SHALE	PIPE 7	65.0	10.5	30000.0	D25-32	336.0	20.15	13.60	1.001	1 50	291	290
262	DD PN16	CLSI	SHALE	PIPE 7	64.0	10.5	30000.0	D25-32	358.0	18.35	14.00	0 932	1.58	274	282
MISCI	ELLANEOUS PILE T	YPES IN SAN	ē												
183	RES TP2	ิง	GR	PIPE 12	68 0	81	30000.0	ICE 440	164.0	8.04	9.40	0.802	20 00	190	226
182	EOD TP5	S	GR	PIPE 12	108.0	8.1	30000.D	ICE 440	157.0	8.90	8.80	0.981	20 0 0	174	207
237	BOR TP2	сı	SA	TIM. 12.	42 5	62 2	2400.0	LB 440	157.3	3 39	7.94	0.352	5.00	109	147
391	BORPNSO	SASI	SA	PIPE12 75"	76 8	15.7	0:0000E	706 V06	223.2	6.57	6.08	0.540	166 67	316	289
295	RES PN3	SA	SA	MONO 12	45.0	06	30000.0	ICE 520	371.0	11.33	9.10	0.541	19.00	311	458
416	BOR TP4	SA	SA	PIPE12	105.2	13.7	30000.0	D-22	443.8	18.46	15.81	0.857	4 00	304	400
114	BOR PN795	CLSA	SA	PIPE/PSC14	87 0	153.9	5565.4	VUL 010	590.0	20 12	8.50	0.519	10.58	558	787
178	X PNA17.5	SA	SA	PIPE 16	75.0	20.8	30000.0	K25	519.0	22.10	13.70	0.837	25 92	555	602
294	RES PN2	SA	SA	MONO 12	43.0	0.6	30000.0	ICE 520	304.0	7.27	10.00	0.454	18 00	306	342
115	BOR PN833	CLSA	SA	PIPE/PSC14	88.0	153.9	5043.4	VUL 010	480.0	18.79	8.40	0.588	7 83	492	630
376	G-16-1.B	CL/SA	SA	PIPE,10	63.4	67	30000.0	VUL 01	165.3	6.37	6.69	0.688	6 00	165	179
173	Х ТР7	SA	SA	MONO	47.0	8.1	30000.0	D16-32	328.0	16.83	17.90	0.840	2 58	283	329
117	BOR PN835	CLSA	SA	PIPE/PSC14	88.0	153.9	5067.0	VUL 010	439.0	15.29	7.20	0.536	83 33	502	670
112	BOR PN608	CLSA	SA	PIPE/PSC14	85 0	153.9	5773.7	VUL 010	605.0	21.69	8.60	0.607	8 33	632	716
118	BOR PN836	CLSA	SA	PIPE/PSC14	0.68	153.9	5565.4	VUL 010	370.0	13.19	5.40	0.517	16.08	567	547
113	BOR PN705	CLSA	SA	PIPE/PSC14	80 0	1539	5461.5	VUL 010	498.0	16.25	8.20	0.524	8 33	540	909
293	RES PN1	SA	SA	MONO 12	46.0	9.0	30000.0	ICE 520	320.0	9.17	8.40	0.580	50 00	339	367
116	BOR PN834	CLSA	SA	PIPE/PSC14	88 0	153.9	5220.1	VUL 010	533.0	20.73	8.30	0.583	6.25	511	670
239	TP2 BOR	SA	SA	TIMB.12.	42.5	62.2	2400 0	LINK 440	163.7	3.88	8.36	0.365	5.00	118	165
317	B15T EOD	SACY	SA	MONOTUBE	48.0	5.2	30000.0	K13	277.0	10.24	10.27	0.674	6.08	248	293
238	BOR TP2	SA CL	SA CL	TIMBER	42.5	62.2	2400.0	LB440	169.9	3.61	7.76	0.341	5 00	8	160
401	EOR D27	SA CL	SA CL	PIPE 14"	101 3	31 2	30000.0	K25	6.03.9	27.58	12.29	0.882	3 00	427	545
0 60	BOR TP1	SA SAT	SA SAT	EVCEPIPE	72.3	12.4	30000	D16-32	413.7	15.02	15.37	0.640	13.33	332	504

	KIPS	1096	318	471	340	244	297	587	362	173	294	243	240	275	337	277	263	443
CAPWAP	KIPS	742	320	\$	310	162	243	308	194	121	274	200	298	220	294	229	208	325
	BLOWS/	24.00	4.17	3.00	4.17	5.25	3 33	1.67	2 00	2.50	300	1.67	7.33	167	3 50	2 33	3 33	13 33
2	ž	0.728	0.567	1.173	0.563	0.500	0.692	0.659	1.448	0.638	0.730	0.948	0.447	0.963	0.710	0 784	0.754	0.635
200	FT/S	5.64	15.00	18.70	16.20	12.90	14.60	5.50	12.00	11.65	14.70	15.20	10.00	15.80	15.60	14.10	10.50	15.03
2	EWX	35.13	10.69	29.58	11.36	7.03	12.26	30.80	29.36	7.49	13.02	15.65	5.84	17.88	13.98	13.97	11.54	13.10
	KIPS	1140.7	306.0	429.0	374.0	265.0	289.0	1196.0	374.0	197.2	297.0	2740	253.0	302.0	325 0	284.0	244.0	345.3
	HAWWER	D46-23	HERA1500	D25-32	HERA1500	HERA1500	HERA1500	D46	D30-23	FEC 1500	D16-32	D16-32	D15	D16-32	D16-32	D16-32	D22-02	DEL 16 3
ç	2_	o,	0	0.0	0	0.0	0.0	0	5	63	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
		30000	30000	30000	3006	ğ	ğõ	5896	1857	1708	3000	800 000	800 000	0000	0000	3000	3000	3000
	akea emc IN2 KSI	26.0 30000	10.8 30000	9.8 30000	10.8 30000	10.8 30000	7.6 30000	478.0 5896	168.0 1857	89.0 1708	8.1 3000	8.1 3000	8.1 3000	8.1 3000	8.1 3000	8.1 3000	8.1 3000	4.4 3000
	LEN. AKEA EMC FT IN? KSI	122.5 26.0 30000	51.0 10.8 30000	52.0 9.8 30000	44.0 10.8 30000	57.0 10.8 30000	38.0 7.6 30000	119.0 478.0 5896	45.0 168.0 1857	18.0 89.0 1708	47.0 8.1 3000	38.0 8.1 3000	42.0 8.1 3000	38.0 8.1 3000	42.0 8.1 3000	43.0 8.1 3000	38.0 8.1 3000	43.0 4.4 3000
	PILE LEN. AKEA ENN. TYPF FT IN? KSI	COMPOSIT 122.5 26.0 30000	PIPE 14 51.0 10.8 30000	PIPE 12.75 52.0 9.8 30000	PIPE 14 44.0 10.8 30000	PIPE 14 57.0 10.8 30000	PIPE 12 38.0 7.6 30000	HP/PSC24 119.0 478.0 5896	TIMBER 45.0 168.0 1857	TIM 14" 18.0 89.0 1705	MONO 47.0 8.1 3000	MONO NU 38.0 8.1 3000	MONO 42.0 8.1 3000	MONO NU 38.0 8.1 3000	MONO 42.0 8.1 3000	MONO 43.0 8.1 3000	MONO 11 38.0 8.1 3000	MONO 43.0 4.4 3000
	I UE PILE LEN AKEA ENK SOII TYPF FT IN? KSI	SA SI COMPOSIT 122.5 26.0 30000	SAGR PIPE 14 51.0 10.8 30000	SAGR PIPE 12.75 52.0 9.8 30000	SAGR PIPE 14 44.0 10.8 30000	SAGR PIPE 14 57.0 10.8 30000	SAGR PIPE 12 38.0 7.6 30000	SASI HP/PSC24 119.0 478.0 5896	SASI TIMBER 45.0 168.0 1857	SA/GR TIM 14" 18.0 89.0 1708	SI MONO 47.0 8.1 3000	SI MONO NU 38.0 8.1 3000	SI MONO 42.0 8.1 3000	SI MONO NU 36.0 8.1 3000	SI MONO 42.0 8.1 3000	SI MONO 43.0 8.1 3000	SICL MONO 11 38.0 8.1 3000	SUCL MONO 43.0 4.4 3000
	SKIN I UE PILE LEN. AKEA EMC Soui Soui Type et inz KSI	SA SI SA SI COMPOSIT 122.5 26.0 30000	CLSI SAGR PIPE 14 51.0 10.8 30000	CLSI SAGR PIPE 12.75 52.0 9.8 30000	CLSI SAGR PIPE 14 44.0 10.8 30000	CLSI SAGR PIPE 14 57.0 10.8 30000	CLSI SAGR PIPE 12 38.0 7.6 3000	SI SASI HP/PSC24 119.0 478.0 5896	SASI SASI TIMBER 45.0 168.0 1857	SA SA/GR TIM 14" 18.0 89.0 1708	SA SI MONO 47.0 8.1 3000	SA SI MONO NU 38.0 8.1 3000	5A SI MONO 42.0 8.1 3000	SA SI MONO NU 38.0 8.1 3000	SA SI MONO 42.0 8.1 3000	SA SI MONO 43.0 8.1 3000	SA SICL MONO 11 38.0 8.1 3000	SI/CL SI/CL MONO 43.0 4.4 3000
	PILE SKIN I UE PILE LEN. AKEA EMC. NAME SOII SOII TYPF FT IN? KSI	1R B.120 SA SI COMPOSIT 122.5 26.0 30000	BOR PN17 CLSI SAGR PIPE 14 51.0 10.6 30000	BOR PN1 CLSI SAGR PIPE 12.75 52.0 9.8 3000	BOR PN26 CLSI SAGR PIPE 14 44.0 10.8 30000	EOD PN16 CLSI SAGR PIPE 14 57.0 10.8 30000	RES TP2 CLSI SAGR PIPE 12 38.0 7.6 30000	BOR PN2 SI SASI HP/PSC24 119.0 478.0 5896	BOR TB1 SASI SASI TIMBER 45.0 168.0 1657	PN#2 EOD SA SA/GR TIM 14" 18.0 89.0 1705	EOD TP5 SA SI MONO 47.0 8.1 3000	EOD TP27 SA SI MONO NU 38.0 8.1 3000	EOD TP12 SA SI MONO 42.0 8.1 3000	EOD TP26 SA SI MONO NU 38.0 8.1 3000	EOD TP2 SA SI MONO 42.0 8.1 3000	EOD PN2 SA SI MONO 43.0 8.1 3000	EOD PN6 SA SICL MONO 11 38.0 8.1 3000	TP22 BOR SVCL SVCL MONO 43.0 4.4 3000

1 ft = 0.305 m 1 in² = 645.2 mm² 1 kip/in² = 6895.1 kPa 1 kip = 4.448 kN 1 k-ft = 1.356 kN-m 1 in = 25.4 mm 1 blow/in = 0.039 blows/mm

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(I n)	(in)	(s/ft)	(s/ft)
_								
1	EOD PN6	SA	SICL	MONO 11	0,100	0.390	0.490	0.070
2	RES PN6	SA	CLSI	MONO 11	0.100	0.530	0.600	0.100
3	BOR PA35	SASI	SASI	PSC 20	0.100	0.300	0.200	0 170
4	BOR PM24	SASI	SASI	PSC 20	0.100	0.200	0.616	0.249
5	EOD PA8	NA	SA	PSC 18	0.100	0.280	0.040	0.340
6	EOD PN165	NA	NA	CEP 12.75	0 105	0.265	0.394	0.175
7	EOD PN210	NA	NA	CEP 12.75	0.110	0.230	0.270	0.200
8	EOD PN15E	NA	SAGR	CEP 10.75	0.090	0.180	0.299	0.350
9	PN1	NA	SA	CEP 16	0.100	0.420	0.350	0.473
10	BOR TP6	SASI	CLSI	PSC 12	0.070	0.070	0.091	0.773
11	EOD TP6	SASI	CLS	PSC 12	0 100	0.133	0.037	0.651
12	EOD TP12	SA	CLSI	MONO	0 080	0.080	1.200	0.050
13	BOR D418	CLSHALE	DOLOMITE	OEP 96	0 080	0.050	0.458	0.473
14	BOR J3	CLSHALE	DOLOMITE	OEP 96	0.100	0.080	0.550	0.600
15	EOD 0205	CLSHALE	DOLOMITE	OEP 96	0.100	0.080	1.025	0.555
15	EOD D418	CLSHALE	DOLOMITE	OEP 96	0 100	0.080	0.514	0.971
17	EOD J1	CLSHALE	DOLOMITE	OEP 9 6	0 100	0.100	0.700	0.550
18	ECD J8	CLSHALE	DOLOMITE	OEP 9.6	0.100	0.120	0.481	0.510
19	RES TN12	AGDITE	AGDITE	PCC 16	0.140	0.170	0.513	0.470
20	BOR CT	AGDITE	AGDITE	PCC 16	0 110	0.110	0.466	0.345
21	BORTN	AGDITE	AGDITE	PCC 16	0,160	0.190	0.296	0.500
22	EOD PN1	NA	NA	PSC 24 NU	0,100	0.350	0.338	0.225
23	BOR PN9	SASI	SASI	HP 12x74	0 096	0.080	1.439	0.194
24	DD PN83	SASI	SASI	HP 12x53	0.050	0.150	0.620	0.051
25	DD PN83	SASI	SASI	HP 12x53	0 040	0.040	0.486	0.042
26	X PN71	SASI	SASI	HP 12x53	0 050	0.050	0.447	0 035
27	X PN80	SASI	SASI	HP 12x53	0 040	0.150	0.270	0.101
28	X PN80	SASI	SASI	HP 12x53	0 060	0.060	1.201	0.052
29	BOR TB1	SASI	SASI	TIMBER	0 100	0.860	0.500	0.030
30	BOR TP114	SASI	SASI	PSC 12	0 100	0 600	0.547	0.065
31	BOR TP2	SASI	SASI	PSC 12	0.130	0.150	0.991	0.210
32	BOR TP26	SASI	SASI	PSC 12	0 1 00	0.350	0.887	0.109
33	EOD TP114	SASI	SASI	PSC 12	0.100	0.370	0.159	0.039
34	EOD TP25	SASI	SASI	PSC 12	0.100	0.200	0.250	0.037
35	EOD PN1	CL	TILL	CEP 10 75	0 080	0.300	0.400	0.300
36	DD PN25	NA	NA	PSC 30 NU	0.100	0.200	0.150	0.300
37	EOD PN25	NA	NA	PSC 30 NU	0 080	0.120	0.256	0.282
38	EOR PN30	NA	NA	PSC 30	0.143	0.255	0.011	0.354
39	BOR D12N	CL	CL	HP 10x42	0 100	0.115	0 571	0 638
40	EOD D18E	CL	CL	HP 10x42	0 100	0.130	0.292	0.574
41	EOR D110S	CL	TILL	HP 10x42	0 140	0.140	0.770	0.750
42	EOR D112E	CL	TILL	HP 10X42	0 100	0.120	0.722	0.646
43	EOR DOW	CL	TILL	HP 10x42	0.130	0.200	0.164	0.607
44	BOR A213	CLSI	SASI	OEP 12	0 100	0.150	0.439	0.112
45	EOD E18	CLSI	SASI	OEP 12	0 100	0.950	0.103	0.064
46	EOD J210	CLSI	SASI	OEP 12	0 100	0.520	0.252	0.053
47	RES PN125	SI	SI	PSC 24	0 060	0.450	0.481	0.162
48	RES PN125	SI	SI	PSC 24	0 091	0.400	0.182	0.011
		- ·	-					

Table 25. Side/tip quake and damping parameters of data set PD.

REF PILE DK.N TOE PILE QUAKE QUAKE CAMPING QMPING 49 ECOP PN12 CL RCK HP 10X57 0.060 0.050 0.300 1100 50 ECO BE11 CL TILL PSC 14 0.120 0.460 0.325 0.306 51 ECO PIT8NW CL TILL PSC 14 0.120 0.322 0.140 0.227 0.131 52 ECO PIT8NW CL TILL PCC 14 0.150 0.232 0.140 0.228 54 RES PN90 CL TILL PCC 14 0.167 0.211 0.500 0.285 56 DD TP15 CLSI CLSI PSC 16 0.100 0.420 0.150 0.501 57 DD TP16 CLSI CLSI PSC 14 0.060 0.250 0.560 0.200 58 DD TP16 CLSI CLSI PSC 14 0.070 0.120 0.560 0.200 <tr< th=""><th></th><th></th><th></th><th></th><th></th><th>SIDE</th><th>TIP</th><th>SIDE</th><th>TIP</th></tr<>						SIDE	TIP	SIDE	TIP
NO. NAME SOIL SOIL TYPE (in) (in) (s/ft) (s/ft) 49 EOD PN12 CL ROCK HP 10X57 0.060 0.050 0.300 1 100 50 EOD PRIW CL TILL PSC 16 0.130 0.400 0.325 0.306 51 EOD PRIW CL TILL PSC 14 0.120 0.320 0.140 0.250 53 RES PN30 CL TILL PCC 14 0.153 0.190 0.150 0.227 54 RES PN30 CL TILL PCC 14 0.167 0.211 0.500 0.283 56 DD TP15 CLSI CLSI PSC 16 0.100 0.420 0.150 0.100 58 DD TP16 CLSI CLSI PSC 14 0.070 0.200 0.140 0.030 59 DT TP16 CLSI CLSI PSC 14 0.070 0.100 0.510 0.340 50	REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
49 EOD PN12 CL ROCK HP 10X57 0.60 0.050 0.300 1.100 49 EOD PIT12 CL TILL PSC 16 0.130 0.400 0.385 0.189 51 EOD PIT6WW CL TILL PSC 14 0.120 0.320 0.222 0.181 53 RES PN9 CL TILL PCC 14 0.120 0.320 0.222 0.181 54 RES PN90 CL TILL PCC 14 0.167 0.211 0.500 0.288 55 RES PN90 CL TILL PCC 14 0.167 0.161 0.500 0.288 56 DD TP15 CLSI CLSI PSC 14 0.060 0.200 0.140 0.030 57 DD TP16 CLSI CLSI PSC 14 0.070 0.220 0.500 0.240 60 DD TP16 CLSI CLSI PSC 14 0.070 0.100 0.510 0.0404 510	NO.	NAME	SOIL	SOIL	TYPE	(in)	(IN)	(s/ft)	(s/ft)
49 ECO PN12 CL ROCK HP10X57 0.060 0.050 0.305 1100 50 EOD PRI10 CL TILL PSC 14 0.130 0.400 0.385 0.189 51 EOD PN10 CL TILL PSC 14 0.120 0.320 0.140 0.255 53 RES PN9 CL TILL PCC 14 0.120 0.320 0.272 0.1181 54 RES PN20 CL TILL PCC 14 0.160 0.420 0.150 0.277 55 BD TP15 CLSI CLSI PSC 16 0.100 0.420 0.150 0.030 54 DD TP15 CLSI CLSI PSC 14 0.060 0.200 0.140 0.030 55 DD TP16 CLSI CLSI PSC 14 0.100 0.510 0.044 60 DD TP16 CLSI CLSI PSC 14 0.100 0.450 0.350 0.050 61 RES PN122223 </th <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>									
50 ECO RH11 CL TILL PSC 16 0 130 0.400 0.325 0.159 51 ECO PIT6NW CL TILL PSC 14 0.120 0.460 0.225 0.326 53 RES PN0 CL TILL PCC 14 0.120 0.320 0.22 0.21 54 RES PN30 CL TILL PCC 14 0.150 0.216 0.150 0.281 56 DD TP15 CLSI CLSI PSC 16 0.100 0.420 0.150 0.100 57 DD TP15 CLSI CLSI PSC 14 0.070 0.220 0.500 0.036 59 DD TP16 CLSI CLSI PSC 14 0.070 0.150 0.560 0.200 61 RES PT19 SLI COOPERMARL PSC 24 0.170 0.450 0.330 0.050 62 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.350 0.050 <t< th=""><th>49</th><th>EOD PN12</th><th>CL</th><th>ROCK</th><th>HP 10X57</th><th>0 060</th><th>0.050</th><th>0.300</th><th>1 100</th></t<>	49	EOD PN12	CL	ROCK	HP 10X57	0 060	0.050	0.300	1 100
51 EOD PITENW CL TILL PSC 14 0.120 0.460 0.285 52 EOD PN10 CL TILL PCC 14 0.120 0.320 0.140 0.250 53 RES PN90 CL TILL PCC 14 0.153 0.199 0.150 0.283 56 DD TP15 CLSI CLSI PSC 18 0.100 0.420 0.150 0.100 57 DD TP15 CLSI CLSI PSC 18 0.100 0.420 0.150 0.100 58 DD TP16 CLSI CLSI PSC 14 0.070 0.220 0.500 0.044 50 DD TP16 CLSI CLSI PSC 14 0.070 0.100 0.510 0.040 51 DCDFERMARL PSC 24 0.070 0.100 0.541 0.030 52 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.350 0.050 56 DD TP2 ALLUVIAL	50	EOD 8611	CL	TILL	PSC 16	0 130	0.400	0 385	0 169
52 EOD PN10 CL TILL PCC 14 0 120 0 320 0 120 0 272 0 131 53 RES PN20 CL TILL PCC 14 0 130 0 323 0 272 0 131 54 RES PN20 CL TILL PCC 14 0 167 0 211 0 500 0 228 55 RES PN30 CL TILL PCC 14 0 167 0 211 0 500 0 228 56 DD TP15 CLSI CLSI PSC 18 0 100 0 320 0 150 0 050 59 DD TP16 CLSI CLSI PSC 14 0 100 0 150 0 500 0 044 50 DD TP16 CLSI CLSI PSC 14 0 100 0 150 0 500 0 440 0 611 51 COOPERMARL PSC 24 0 100 0 450 0 350 0 606 56 BOR PN120R9 SI COOPERMARL PSC 24 0 100 0 160 0 350 0 350 0 350	51	EOD PIT6NW	CL	TILL	PSC 14	0.120	0.460	0.295	0.306
S3 RES PN90 CL TILL PCC 14 0.120 0.320 0.272 0.131 S4 RES PN80 CL TILL PCC 14 0.153 0.199 0.150 0.237 S6 DD TP15 CLSI CLSI PSC 16 0.100 0.420 0.150 0.100 S7 DD TP15 CLSI CLSI PSC 14 0.060 0.200 0.140 0.030 S8 DD TP16 CLSI CLSI PSC 14 0.070 0.220 0.500 0.044 60 DD TP16 CLSI CLSI PSC 14 0.070 0.100 0.501 0.0404 61 RES PN19724 SI COOPERMARL PSC 24 0.070 0.100 0.450 0.350 0.060 63 RES PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.350 0.050 64 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.120 0.330 0.2	52	EOD PN10	CL	TILL	PCC 14	0 120	0.320	0.140	0.250
54 RES PN20 CL TILL PCC 14 0.153 0.193 0.193 0.287 55 RES PN50 CL TILL PCC 14 0.167 0.211 0.500 0.288 56 DD TP15 CLSI CLSI PSC 18 0.100 3.250 0.150 0.100 57 DD TP16 CLSI CLSI PSC 14 0.000 0.200 0.140 0.030 58 DD TP16 CLSI CLSI PSC 14 0.100 0.150 0.500 0.204 60 DD TP16 CLSI CLSI PSC 14 0.100 0.100 0.510 0.240 61 RES PN122823 SI COOPERMARL PSC 24 0.100 0.450 0.380 0.060 64 BOR PN12189 SI COOPERMARL PSC 24 0.100 0.120 0.800 0.360 0.360 0.360 0.360 0.360 0.360 0.360 0.360 0.360 0.360 0.360 0.360	53	RES PN9	CL	TILL	PCC 14	0.120	0.320	0.272	0.181
55 RES PN80 CL TILL PCC 14 0.167 0.211 0.500 0.288 56 DD TP15 CLSI CLSI PSC 18 0.100 0.420 0.150 0.100 57 DD TP16 CLSI CLSI PSC 14 0.060 0.200 0.140 0.030 59 DD TP16 CLSI CLSI PSC 14 0.070 0.220 0.500 0.0541 60 DD TP16 CLSI CLSI PSC 14 0.100 0.150 0.650 0.200 61 RES PN1922223 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.060 65 BOR PN121R9 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.060 66 BOR PN225 CLSI COOPERMARL PSC 18 0.100 0.120 0.820 0.1141 67 BOR PN225 CLSI COOPERMARL PSC 18 0.100 0.120 0.300 0.212	54	RES PN20	CL	TILL	PCC 14	0.153	0.199	0.150	0.287
56 DD TP15 CLSI CLSI PSC 18 0.100 0.420 0.150 0.100 57 DD TP15 CLSI CLSI PSC 18 0.100 0.350 0.150 0.100 58 DD TP16 CLSI CLSI PSC 14 0.060 0.200 0.500 0.254 60 DD TP16 CLSI CLSI PSC 14 0.070 0.200 0.501 0.0501 61 RES TP15 CLSI CLSI PSC 14 0.080 0.080 0.641 0.133 62 RES PN12282 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.661 64 BOR PN121R9 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.660 65 BOR PN25 CLSI COOPERMARL PSC 18 VU 0.100 0.180 0.895 0.131 66 BOR TP3 ALLUVIAL CL CEP 120.18 0.160 0.130 0.211 0.300	55	RES PN50	CL	TILL	PCC 14	0.167	0.211	0.500	0.268
57 DD TP15 CLSI CLSI PSC 18 0.100 0.350 0.150 0.100 58 DD TP16 CLSI CLSI PSC 14 0.060 0.200 0.140 0.030 59 DD TP16 CLSI CLSI PSC 14 0.070 0.220 0.500 0.261 60 DD TP16 CLSI CLSI PSC 14 0.070 0.150 0.641 0.133 62 RES PN119R24 SI COOPERMARL PSC 24 0.100 0.450 0.350 0.0661 63 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.350 0.0560 66 BOR PN200 CLSA COOPERMARL PSC 18 0.100 0.120 0.360 0.134 67 BOR PN225 CLSI SA PSC 20 0.901 0.100 0.120 0.300 0.214 0.353 0.320 68 DD TP2 ALLUVIAL CL CEP120.18 0.901	56	DD TP15	CLSI	CLSI	PSC 18	0.100	0.420	0.150	0.100
58 DD TP16 CLSI CLSI PSC 14 0 600 0 200 0.140 0.030 59 DD TP16 CLSI CLSI PSC 14 0 070 0 220 0 500 0.054 60 DD TP16 CLSI CLSI PSC 13 0 080 0 090 0.641 0.133 62 RES TP15 CLSI CLSI PSC 24 0 070 0 100 0.510 0.040 63 RES FN122B23 SI COOPERMARL PSC 24 0 100 0.450 0.350 0.060 64 BOR PN120R9 SI COOPERMARL PSC 24 0 100 0.450 0.350 0.060 66 BOR PN200 CLSA COOPERMARL PSC 24 0 100 0 120 0.820 0.111 68 BOR TP3 ALLUVIAL CL CEP 120.18 0 091 0.600 0.335 0.300 71 EOD PN392 CLSI SA PSC 20 0 100 0.300 0.211 0.433 </th <th>57</th> <th>DD TP15</th> <th>CLSI</th> <th>CLSI</th> <th>PSC 18</th> <th>0.100</th> <th>0.350</th> <th>0.150</th> <th>0.100</th>	57	DD TP15	CLSI	CLSI	PSC 18	0.100	0.350	0.150	0.100
58 DD TP16 CLSi CLSi PSC 14 0.070 0.220 0.500 0.054 60 DD TP16 CLSi CLSi PSC 14 0.100 0.150 0.650 0.200 61 RES PN119R24 SI COOPERMARL PSC 24 0.120 0.500 0.404 0.610 63 RES PN122R23 SI COOPERMARL PSC 24 0.120 0.500 0.404 0.661 64 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.660 65 BOR PN21R6 SI COOPERMARL PSC 14 0.100 0.120 0.820 0.131 67 BOR PN225 CLSI COOPERMARL PSC 10 0.000 0.180 0.666 0.550 68 BOR PT3 ALLUVIAL CL CEP 12:0.18 0.091 0.650 0.330 0.320 0.254 0.180 71 EOD PN366 CLSI SA PSC 20 0.120 0	58	DD TP16	CLSI	CLSI	PSC 14	0 060	0.200	0.140	0.030
60 DD TP16 CLSI CLSI PSC 14 0.100 0.150 0.650 0.200 61 RES TP15 CLSI DSC 16 0.080 0.080 0.641 0.133 62 RES PN19223 SI COOPERMARL PSC 24 0.120 0.500 0.404 0.661 63 RES PN122B3 SI COOPERMARL PSC 24 0.100 0.450 0.380 0.060 65 BOR PN120R SI COOPERMARL PSC 24 0.100 0.120 0.820 0.141 66 BOR PN200 CLSA COOPERMARL PSC 24 0.100 0.120 0.820 0.141 67 BOR PN200 CLSI COOPERMARL PSC 20 0.100 0.120 0.820 0.141 68 BOR PN23 ALLUVIAL CL CEP 12x0.18 0.091 0.650 0.330 0.211 0.433 71 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.221	59	DD TP16	CLSI	CLSI	PSC 14	0 070	0 220	0 500	0.054
61 RES PH19 CLSI CLSI PSC 13 0.080 0.090 0.641 0.133 62 RES PH119R24 SI COOPERMARL PSC 24 0.707 0.100 0.500 0.404 0.061 63 RES PH122R3 SI COOPERMARL PSC 24 0.100 0.450 0.350 0.060 64 BOR PN121R9 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.060 66 BOR PN250 CLSI COOPERMARL PSC 18 0.100 0.120 0.620 0.141 68 BOR TP3 ALLUVIAL CL CEP 12x0.18 0.001 0.180 0.696 0.550 69 DD TP2 ALLUVIAL CL CEP 12x0.18 0.091 0.500 0.310 0.221 0.300 71 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.221 0.300 72 EOD PN497 SIG SASI CEP 112x0 6 0.100	60	DD TP16	CLSI	CLSI	PSC 14	0.100	0.150	0.650	0.200
62 RES PN119R24 SI COOPERMARL PSC 24 0 070 0 100 0.510 0.040 63 RES PN12R83 SI COOPERMARL PSC 24 0 120 0.500 0.444 0 061 64 BOR PN12R9 SI COOPERMARL PSC 24 0 100 0.450 0.350 0.060 66 BOR PN226 CLSI COOPERMARL PSC 18 NU 0 100 0 120 0.820 0.114 67 BOR PN225 CLSI COOPERMARL PSC 18 NU 0 100 0 180 0.666 0.550 0.330 68 BOR P173 ALLUVIAL CL CEP 120.18 0 091 0.650 0.330 0.320 0.212 0.330 71 EOD PN396 CLSI SA PSC 20 0 160 0.330 0.320 0.224 0.390 73 EOD PN396 CLSI SA PSC 20 0 160 0.330 0.224 0.390 74 BOR PN4 SIGR SASI <	61	RES TP15	CLSI	CLSI	PSC 18	0.080	0.090	0.641	0.133
63 RES PN122B23 SI COOPERMARL PSC 24 0.120 0.500 0.404 0.061 64 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.060 65 BOR PN120R9 SI COOPERMARL PSC 24 0.100 0.450 0.360 0.060 66 BOR PN225 CLSI COOPERMARL PSC 18 NU 0.100 0.120 0.820 0.114 67 BOR PN225 CLSI COOPERMARL PSC 20 0.100 0.160 0.650 0.0350 0.111 0.433 68 DD TP2 ALLUVIAL CL CEP 120.18 0.091 0.300 0.111 0.433 71 EOD PN392 CLSI SA PSC 20 0.160 0.330 0.220 0.355 72 EOD PN392 CLSI SA PSC 20 0.160 0.300 0.212 0.390 73 EOD PN393 SIGR SASI OEP 12x0 6 0.100	62	RES PN119R24	SI	COOPERMARL	PSC 24	0 070	0,100	0.510	0.040
64 BOR PN121R9 SI COOPERMARL PSC 24 0 100 0.450 0.350 0.060 65 BOR PN121R9 SI COOPERMARL PSC 24 0 100 0.450 0.380 0.060 66 BOR PN225 CLSA COOPERMARL PSC 18 NU 0 100 0.120 0.820 0.134 67 BOR PN225 CLSI COOPERMARL PSC 18 NU 0 100 0.180 0.696 0.550 69 DD TP2 ALLUVIAL CL CEP 12x0.18 0.091 0.650 0.335 0.300 70 EOD PN392 CLSI SA PSC 20 0.160 0.330 0.312 0.359 71 EOD PN396 CLSI SA PSC 20 0.120 0.300 0.212 0.330 73 EOD PN396 CLSI SA PSC 20 0.100 0.330 0.212 0.330 74 BOR PN3 SIGR SASI CEP 12x0 6 0.100 0.100 0.250 0	63	RES PN122823	St	COOPERMARL	PSC 24	0.120	0.500	0.404	0 061
55 BOR PN121R9 SI COOPERMARL PSC 24 0 100 0.450 0.380 0.060 66 BOR PN280 CLSA COOPERMARL HP 14/73 0.060 0.090 0.995 0.134 67 BOR PN225 CLSI COOPERMARL PSC 18 NU 0.100 0.120 0.820 0.141 68 BOR TP3 ALLUVIAL CL CEP 120.18 0.100 0.160 0.330 0.111 0.4333 71 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.212 0.350 72 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.212 0.359 73 EOD PN396 CLSI SA PSC 20 0.150 0.300 0.254 0.198 74 BOR PN4 SIGR SASI OEP 120.6 0.160 0.330 0.277 0.370 75 BOR PN11 SASI SA PSC 36 0.160 0.330 0.2250	64	BOR PN120R9	SI	COOPERMARL	PSC 24	0 1 00	0.450	0.350	0.050
66 BOR PN280 CLSA COOPERMARL HP 14/73 0 080 0 090 0.995 0.134 67 BOR PN225 CLSI COOPERMARL PSC 18 NU 0 100 0 120 0.820 0.141 58 BOR TP3 ALLUVIAL CL CEP 12x0.18 0 091 0.650 0.355 0.300 70 EOD PN392 CLSI SA PSC 20 0 090 0.300 0.111 0.433 71 EOD PN396 CLSI SA PSC 20 0 160 0.320 0.320 0.250 72 EOD PN398 CLSI SA PSC 20 0 100 0.320 0.329 0.859 74 BOR PN3 SIGR SASI CEP 12x0 6 0 100 0.170 0.250 0.500 75 BOR PN4 SIGR SASI OEP 12x0 6 0 160 0.330 0.277 0.370 76 BOR TP11 SASI SASI PSC 36 0 160 0.330 0.260 0.344	65	BOR PN121R9	SI	COOPERMARL	PSC 24	0 100	0.450	0.360	0.060
67 BOR PN225 CLSI COOPERMARL PSC 18 NU 0 100 0 120 0.820 0.141 68 BOR TF3 ALLUVIAL CL CEP 12x0.18 0.100 0.160 0.650 0.550 69 DD TP2 ALLUVIAL CL CEP 12x0.18 0.091 0.650 0.355 0.300 70 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.320 0.212 0.390 73 EOD PN396 CLSI SA PSC 20 0.120 0.300 0.212 0.390 74 BOR PN3 SIGR SASI CEP 11x0.4 0.100 0.170 0.250 0.500 74 BOR PN3 SIGR SASI OEP 12x0.6 0.160 0.300 0.254 0.950 75 BOR TP11 SASI SA PSC 36 0.160 0.300 0.250 0.370 76 BOR TP11 SIGR SASI PSC 36 0.125 0.300 0.344 </th <th>66</th> <th>BOR PN260</th> <th>CLSA</th> <th>COOPERMARL</th> <th>HP 14x73</th> <th>0 080 0</th> <th>0.090</th> <th>0.995</th> <th>0.134</th>	66	BOR PN260	CLSA	COOPERMARL	HP 14x73	0 080 0	0.090	0.995	0.134
68 BOR TP3 ALLUVIAL CL CEP 12x0.18 0.100 0.180 0.696 0.550 69 DD TP2 ALLUVIAL CL CEP 12x0.18 0.91 0.650 0.355 0.300 70 EOD PN392 CLSI SA PSC 20 0.900 0.300 0.111 0.433 71 EOD PN398 CLSI SA PSC 20 0.160 0.300 0.212 0.390 73 EOD PN17 SI SHALE CEP 11x0.4 0.100 0.320 0.359 0.859 74 BOR PN4 SIGR SASI CEP 12x0.6 0.100 0.170 0.250 0.500 75 BOR PN4 SIGR SASI OEP 12x0.6 0.160 0.330 0.277 0.370 76 BOR TP11 SASI SA PSC 36 0.160 0.330 0.280 0.344 78 BOR TP21 SASI SASI HP 12x74 0.100 0.550 0.300 0.388	67	BOR PN225	CLSI	COOPERMARL	PSC 18 NU	0 1 00	0 120	0.820	0.141
68 DD TP2 ALLUVIAL CL CEP 12x0.18 0.091 0.650 0.355 0.300 70 EOD PN392 CLSI SA PSC 20 0.990 0.300 0.111 0.433 71 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.320 0.250 72 EOD PN396 CLSI SA PSC 20 0.100 0.320 0.254 0.390 73 EOD PNE17 SI SHALE CEP 11x0 4 0.00 0.300 0.254 0.198 75 BOR PN3 SIGR SASI OEP 12x0 6 0.100 0.170 0.250 0.500 76 BOR TP11 SASI SA PSC 36 0.160 0.330 0.277 0.370 77 BOR TP11 SIGR SASI PSC 36 0.125 0.300 0.388 0.300 79 EOD PN1 SIGR SASI HP 12x74 0.100 0.150 0.250 0.300	68	BOR TP3	ALLUVIAL	CL	CEP 12x0.18	0.100	0 180	0.696	0.550
70 EOD PN392 CLSI SA PSC 20 0.090 0.300 0.111 0.433 71 EOD PN396 CLSI SA PSC 20 0.160 0.330 0.320 0.250 72 EOD PN817 SI SHALE CEP 11x04 0.100 0.320 0.254 0.139 73 EOD PNE17 SI SHALE CEP 11x04 0.100 0.320 0.254 0.198 74 BOR PN3 SIGR SASI CEP 12x06 0.100 0.170 0.250 0.500 75 BOR TP11 SASI SA PSC 36 0.160 0.330 0.277 0.370 76 BOR TP11 SASI SA PSC 36 0.125 0.300 0.388 0.300 77 BOR TP11 SIGR SASI HP 12x74 0.100 0.150 0.250 0.300 76 BOR TP21 SASI SASI HP 12x74 0.100 0.150 0.250 0.300 77 EOD PN2 SIGR SASI HP 12x74 0.100 0.550 <td< th=""><th>69</th><th>DD TP2</th><th>ALLUVIAL</th><th>CL</th><th>CEP 12x0.18</th><th>0 091</th><th>0.650</th><th>0.355</th><th>0.300</th></td<>	69	DD TP2	ALLUVIAL	CL	CEP 12x0.18	0 091	0.650	0.355	0.300
71 EOD PN396 CLSI SA PSC 20 0 160 0.330 0.320 0 250 72 EOD PN398 CLSI SA PSC 20 0 120 0.300 0.212 0.390 73 EOD PNE17 SI SHALE CEP 11x0 4 0 100 0.320 0.359 0.859 74 BOR PN3 SIGR SASI CEP 12x0 6 0 150 0.300 0.254 0.198 75 BOR PN4 SIGR SASI CEP 12x0 6 0 160 0.170 0.250 0.500 76 BOR TP11 SIGR SASI PSC 36 0 160 0.330 0.277 0.370 77 BOR TP21 SASI SA PSC 36 0 125 0.300 0.388 0.300 79 EOD PN1 SIGR SASI HP 12x74 0 100 0.550 0.180 0.180 80 EOD PN2 SIGR SASI CEP 12x0 6 0 150 0.250 0.300 81	70	EOD PN392	CLSI	SA	PSC 20	0 0 90	0.300	0.111	0.433
72 EOD PN398 CLSi SA PSC 20 0 120 0.300 0.212 0.390 73 EOD PNE17 SI SHALE CEP 11x0 4 0 100 0.320 0.359 0.859 74 BOR PN3 SIGR SASI CEP 12x0 6 0 150 0.300 0.254 0.198 75 BOR PN4 SIGR SASI OEP 12x0 6 0 160 0.170 0.250 0.500 76 BOR TP11 SASI SA PSC 36 0 140 0.330 0.277 0.370 77 BOR TP11 SIGR SASI PSC 36 0.140 0.330 0.280 0.344 78 BOR TP21 SASI SA PSC 36 0.120 0.350 0.380 0.250 0.300 80 EOD PN1 SIGR SASI HP 12x74 0.100 0.550 0.160 0.180 81 EOD PN2 SIGR SASI CEP 12x0 6 0.120 0.230 0.250 0.300 <th>71</th> <th>EOD PN396</th> <th>CLSI</th> <th>SA</th> <th>PSC 20</th> <th>0 160</th> <th>0.330</th> <th>0.320</th> <th>0 250</th>	71	EOD PN396	CLSI	SA	PSC 20	0 160	0.330	0.320	0 250
73EOD PNE17SISHALECEP 11x0 40 1000.3200.3590.85974BOR PN3SIGRSASICEP 12x0 60 1000.1700.2500.50075BOR PN4SIGRSASIOEP 12x0 60 1000.1700.2500.50076BOR TP11SASISAPSC 360 1600.3300.2770.37077BOR TP11SIGRSASIPSC 360.1400.3300.2800.34476BOR TP21SASISAPSC 360.1250.3000.3880.30079EOD PN1SIGRSASIHP 12x740 1000.1500.2500.30080EOD PN2SIGRSASIHP 10x420 1000.5200.2570 25081EOD PN2SIGRSASICEP 12x0 60 1500.2500.1800 18082EOD PN3SIGRSASIOEP 12x0 60 1500.2500.3000.43084EOD PN4SIGRSASIOEP 12x0 60 1300.2500.3000.43085EOD PN4SIGRSASIPSC 360 1300.2500.3000.43086EOD TP23SIGRSASIPSC 360 1500.2300.3000.43086EOD TP23SIGRSASIPSC 360 1500.2000.3000.45088EOD TP41SIGRSASIPSC 300 1000.4100.3000.31189 </th <th>72</th> <th>EOD PN398</th> <th>CLSI</th> <th>SA</th> <th>PSC 20</th> <th>0 120</th> <th>0.300</th> <th>0.212</th> <th>0.390</th>	72	EOD PN398	CLSI	SA	PSC 20	0 120	0.300	0.212	0.390
74 BOR PN3 SIGR SASI CEP 12x0 6 0 150 0.300 0.254 0.198 75 BOR PN4 SIGR SASI OEP 12x0 6 0 100 0.170 0.250 0.500 76 BOR TP11 SASI SA PSC 36 0 160 0.330 0.277 0.370 77 BOR TP21 SASI SASI PSC 36 0.125 0.300 0.388 0.300 79 EOD PN1 SIGR SASI HP 12x74 0 100 0.150 0.250 0.300 80 EOD PN2 SIGR SASI HP 10x42 0 100 0.550 0.180 0.180 82 EOD PN3 SIGR SASI CEP 12x0 6 0 150 0.250 0.170 0.220 83 EOD PN4 SIGR SASI CEP 12x0 6 0 150 0.250 0.170 0.200 84 EOD TP11 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 <tr< th=""><th>73</th><th>EOD PNE17</th><th>SI</th><th>SHALE</th><th>CEP 11x0 4</th><th>0 100</th><th>0.320</th><th>0.359</th><th>0.859</th></tr<>	73	EOD PNE17	SI	SHALE	CEP 11x0 4	0 100	0.320	0.359	0.859
75 BOR PN4 SIGR SASI OEP 12x0 6 0 100 0.170 0.250 0.500 76 BOR TP11 SASI SA PSC 36 0 160 0.330 0.277 0.370 77 BOR TP11 SIGR SASI PSC 36 0.140 0.330 0.280 0.344 78 BOR TP11 SIGR SASI PSC 36 0.125 0.300 0.388 0.300 79 EOD PN1 SIGR SASI HP 12x74 0.100 0.150 0.250 0.300 80 EOD PN2 SIGR SASI HP 10x42 0.100 0.550 0.180 0.180 82 EOD PN3 SIGR SASI OEP 12x0 6 0.150 0.250 0.310 0.220 83 EOD PN4 SIGR SASI OEP 12x0 6 0.150 0.250 0.312 0.415 84 EOD PN4 SIGR SASI PSC 36 0.150 0.230 0.300 0.321	74	BOR PN3	SIGR	SASI	CEP 12x0 6	0 150	0.300	0.254	0.198
76 BOR TP11 SASI SA PSC 36 0 160 0.330 0.277 0.370 77 BOR TP11 SIGR SASI PSC 36 0.140 0.330 0.280 0.344 76 BOR TP21 SASI SA PSC 36 0.125 0.300 0.388 0.300 79 ECD PN1 SIGR SASI HP 12x74 0.100 0.550 0.250 0.300 80 EOD PN2 SIGR SASI HP 10x42 0.100 0.550 0.180 0.180 82 EOD PN3 SIGR SASI CEP 12x0.6 0.150 0.250 0.170 0.220 83 EOD PN4 SIGR SASI OEP 12x0.6 0.120 0.230 0.250 0.300 84 EOD TP11 SIGR SASI PSC 36 0.130 0.250 0.312 0.415 85 EOD TP13 SIGR SASI PSC 36 0.130 0.250 0.300 0.321	75	BOR PN4	SIGR	SASI	OEP 12x0 6	0 1 00	0.170	0.250	0.500
77 BOR TP11 SIGR SASI PSC 36 0.140 0.330 0.260 0.344 78 BOR TP21 SASI SA PSC 36 0.125 0.300 0.368 0.300 79 EOD PN1 SIGR SASI HP 12x74 0.100 0.150 0.250 0.300 60 EOD PN2 SIGR SASI HP 10x42 0.100 0.520 0.257 0.250 81 EOD PN2 SIGR SASI HP 10x42 0.100 0.550 0.180 0.180 82 EOD PN3 SIGR SASI CEP 12x0.6 0.150 0.250 0.170 0.220 63 EOD PN4 SIGR SASI OEP 12x0.6 0.130 0.250 0.300 0.430 84 EOD TP11 SIGR SASI PSC 36 0.130 0.250 0.312 0.415 85 EOD TP13 SIGR SASI PSC 36 0.170 0.300 0.360 0.321	76	BOR TP11	SASI	SA	PSC 36	0 160	0.330	0.277	0.370
76 BOR TP21 SASI SA PSC 36 0.125 0.300 0.388 0.300 79 EOD PN1 SIGR SASI HP 12x74 0 100 0.150 0.250 0.300 60 EOD PN2 SIGR SASI HP 10x42 0 100 0.520 0.257 0 250 61 EOD PN2 SIGR SASI HP 10x42 0 100 0.550 0.180 0.180 62 EOD PN3 SIGR SASI CEP 12x0 6 0 120 0.230 0.250 0.300 63 EOD PN4 SIGR SASI OEP 12x0 6 0 130 0.250 0.312 0.415 65 EOD TP11 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 66 EOD TP21 SASI SA PSC 36 0 150 0.200 0.300 0.650 68 EOD TP23 SIGR SASI PSC 30 0 108 0.254 0.030 0.311	77	BOR TP11	SIGR	SASI	PSC 36	0.1 40	0.330	0.260	0.344
79EOD PN1SIGRSASIHP 12x740 1000.1500.2500.30060EOD PN2SIGRSASIHP 10x420 1000.5200.2570 25061EOD PN2SIGRSASIHP 10x420 1000.5500.1600.18062EOD PN3SIGRSASICEP 12x0 60 1500.2500.1700.22063EOD PN4SIGRSASIOEP 12x0 6C 1200.2300.2500.30064EOD TP11SIGRSASIPSC 360 1500.2300.3000.43065EOD TP13SIGRSASIPSC 360 1500.2300.3000.43066EOD TP21SASISAPSC 360 1500.2000.3000.65068EOD PN7E3NANAPSC 300 1080.2540.0300.31189BOR PNPE28NANAPSC 300 0730.1430.5720 50090EOD TP4NANAPIPE 10 750 1000.4100.3000.45191BOR SHD1CLSALIMESTONEHP 12x530.0800.4500.3230.06793EOD ST1CLSALIMESTONEPIPE 140 1000.2500.1780.07695BOR PST1SASIROCKPIPE 12.750 0800.0800.2000.50096EOD P3T1SASIROCKPIPE 12.750 0800.2000.1200.400 <th>78</th> <th>BOR TP21</th> <th>SASI</th> <th>SA</th> <th>PSC 36</th> <th>0.125</th> <th>0.300</th> <th>0.388</th> <th>0.300</th>	78	BOR TP21	SASI	SA	PSC 36	0.125	0.300	0.388	0.300
60EOD PN2SIGRSASIHP 10x420 1000.5200.2570 25061EOD PN2SIGRSASIHP 10x420 1000.5500.1800.18062EOD PN3SIGRSASICEP 12x0 60 1500 2500.1700.22063EOD PN4SIGRSASIOEP 12x0 6C 1200.2300.2500.30084EOD TP11SIGRSASIPSC 360 1300.2500.3120.41585EOD TP13SIGRSASIPSC 360 1500.2300.3000.43086EOD TP21SASISAPSC 360 1700.3000.3600.32187EOD TP23SIGRSASIPSC 360 1500.2000.3000.65088EOD PN7E3NANAPSC 300 1080.2540.0300.31189BOR PNPE28NANAPSC 300 0730.1430.5720 50090EOD TP4NANAPIPE 10 750 1000.4100.3000.45191BOR SHD1CLSALIMESTONEHP 12x530.1000.4500.3230.06792EOD SHD1CLSALIMESTONEHP 12x530.0800.4500.3230.06793EOD ST1CLSALIMESTONEPIPE 140 1000.2500.1780.07695BOR PST1SASIROCKPIPE 12.750 0800.0800.2000.500<	79	EOD PN1	SIGR	SASI	HP 12x74	0 100	0.150	0.250	0.300
61 EOD PN2 SIGR SASI HP 10x42 0 100 0.550 0.180 0.180 82 EOD PN3 SIGR SASI CEP 12x0 6 0 150 0.250 0.170 0.220 83 EOD PN4 SIGR SASI OEP 12x0 6 0 130 0.250 0.300 84 EOD TP11 SIGR SASI PSC 36 0 130 0.250 0.312 0.415 85 EOD TP13 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 86 EOD TP21 SASI SA PSC 36 0 170 0.300 0.360 0.321 87 EOD TP23 SIGR SASI PSC 54 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 E	80	EOD PN2	SIGR	SASI	HP 10x42	0 100	0.520	0.257	0 250
82 EOD PN3 SIGR SASI CEP 12x0 6 0 150 0.250 0.170 0.220 83 EOD PN4 SIGR SASI OEP 12x0 6 0 120 0.230 0.250 0.300 84 EOD TP11 SIGR SASI PSC 36 0 130 0.250 0.312 0.415 85 EOD TP13 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 86 EOD TP21 SASI SA PSC 36 0 170 0.300 0.360 0.321 87 EOD TP23 SIGR SASI PSC 54 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 1075 0 100 0.410 0.300 0.451 9	81	EOD PN2	SIGR	SASI	HP 10x42	0 100	0.550	0.180	0.180
63 EOD PN4 SIGR SASI OEP 12x0 6 C 120 0.230 0.250 0.300 84 EOD TP11 SIGR SASI PSC 36 0 130 0.250 0.312 0.415 85 EOD TP13 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 86 EOD TP21 SASI SA PSC 36 0 170 0.300 0.360 0.321 87 EOD TP23 SIGR SASI PSC 36 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 10.75 0 100 0.410 0.300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042	82	EOD PN3	SIGR	SASI	CEP 12x0 6	0 150	0.250	0.170	0.220
84 EOD TP11 SIGR SASI PSC 36 0 130 0.250 0.312 0.415 85 EOD TP13 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 86 EOD TP21 SASI SA PSC 36 0 170 0.300 0.360 0.321 87 EOD TP23 SIGR SASI PSC 36 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PSC 30 0 073 0.143 0.572 0 500 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0 100 0.410 0.300 0.451 92 EOD SH01 CLSA LIMESTONE HP 12x53 0 080 0.450 0.323 0.067	83	EOD PN4	SIGR	SASI	OEP 12x0 6	C 120	0.230	0.250	0.300
85 EOD TP13 SIGR SASI PSC 36 0 150 0.230 0.300 0.430 86 EOD TP21 SASI SA PSC 36 0 170 0.300 0.360 0.321 87 EOD TP23 SIGR SASI PSC 36 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 10 75 0 100 0.410 0.300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0.080 0.450 0.323 0.067 <	84	EOD TP11	SIGR	SASI	PSC 36	0 130	0.250	0.312	0.415
86 EOD TP21 SASI SA PSC 36 0 170 0.300 0.360 0.321 87 EOD TP23 SIGR SASI PSC 54 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 10 75 0 100 0.410 0.300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0 080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.173 0.076	85	EOD TP13	SIGR	SASI	PSC 36	0 150	0.230	0.300	0.430
87 EOD TP23 SIGR SASI PSC 54 0 150 0.200 0.300 0.650 88 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 10 75 0 100 0.410 0 300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0 080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.173 0.076 <t< th=""><th>86</th><th>EOD TP21</th><th>SASI</th><th>SA</th><th>PSC 36</th><th>0 170</th><th>0.300</th><th>0.360</th><th>0.321</th></t<>	86	EOD TP21	SASI	SA	PSC 36	0 170	0.300	0.360	0.321
68 EOD PN7E3 NA NA PSC 30 0 108 0.254 0.030 0.311 89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 10 75 0 100 0.410 0.300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0 080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.173 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0 080 0.200 0.120 0.400 </th <th>87</th> <th>EOD TP23</th> <th>SIGR</th> <th>SASI</th> <th>PSC 54</th> <th>0 150</th> <th>0.200</th> <th>0.300</th> <th>0.650</th>	87	EOD TP23	SIGR	SASI	PSC 54	0 150	0.200	0.300	0.650
89 BOR PNPE28 NA NA PSC 30 0 073 0.143 0.572 0 500 90 EOD TP4 NA NA PIPE 10 75 0 100 0.410 0.300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0 080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.173 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0 080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0 080 0 200 0.120 0.400	88	EOD PN7E3	NA	NA	PSC 30	0 108	0.254	0.030	0.311
90 EOD TP4 NA NA PIPE 10 75 0 100 0.410 0.300 0.451 91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0.080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0.100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0.100 0.250 0.173 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0.080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0.080 0.200 0.120 0.400	89	BOR PNPE28	NA	NA	PSC 30	0 073	0.143	0.572	0 500
91 BOR SHD1 CLSA LIMESTONE HP 12x53 0.100 0.300 0.319 0.042 92 EOD SHD1 CLSA LIMESTONE HP 12x53 0.080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0.100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0.100 0.250 0.173 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0.080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0.080 0.200 0.120 0.400	90	EOD TP4	NA	NA	PIPE 10 75	0 1 00	0.410	0.300	0.451
92 EOD SHD1 CLSA LIMESTONE HP 12x53 0 080 0.450 0.323 0.067 93 EOD ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.173 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0 080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0 080 0 200 0.120 0.400	91	BOR SHD1	CLSA	LIMESTONE	HP 12x53	0.100	0.300	0.319	0.042
93 EOD ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.323 0.154 94 EOR ST1 CLSA LIMESTONE PIPE 14 0 100 0.250 0.178 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0 080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0 080 0 200 0.120 0.400	92	EOD SHD1	CLSA	LIMESTONE	HP 12x53	0 0 80	0.450	0.323	0.067
94 EOR ST1 CLSA LIMESTONE PIPE 14 0.100 0.250 0.173 0.076 95 BOR PST1 SASI ROCK PIPE 12.75 0.080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0.080 0.200 0.120 0.400	93	EOD ST1	CLSA	LIMESTONE	PIPE 14	0 100	0.250	0.323	0.154
95 BOR PST1 SASI ROCK PIPE 12.75 0.080 0.080 0.200 0.500 96 EOD P3T1 SASI ROCK PIPE 12.75 0.080 0.200 0.120 0.400	94	EOR ST1	CLSA	LIMESTONE	PIPE 14	0 100	0.250	0.178	0.076
96 EOD P3T1 SASI ROCK PIPE 12 75 0 080 0 200 0.120 0.400	95	BOR PST1	SASI	ROCK	PIPE 12.75	0 0 80	0.080	0.200	0.500
	96	EOD P3T1	SASI	ROCK	PIPE 12 75	0 080	0 200	0.120	0.400

	Table 25. Side/tip	o quake and	damping parameters	ofd	lata set P	D	(continued)
--	--------------------	-------------	--------------------	-----	------------	---	-------------

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(in)	(i n)	(s/ft)	(s/ft)
97	EOD P4T1	SASI	ROCK	PIPE 12.75	0.100	0.250	0.030	0 330
98	BOR PN123	SI	COOPERMARL	PSC 24	0.050	0.700	0 070	0 099
99	BOR PN123R7	SI	COOPERMARL	PSC 24	0.050	0.500	0.106	0 012
100	EOD TP23	SA	CLSI	MONO 14 NU	0.150	0.500	0.300	0.250
101	BOR TP1	SA	SA	HP 14x73	0.100	0.330	0.250	0 050
102	BOR TP23	SA	CLSI	MONO 14 NU	0.110	0.300	0.637	0.044
103	BOR TP24	SA	CLSI	MONO 14 NU	0.130	0.400	0.589	0.069
104	BOR PN608	CLSA	SA	PIPE/PSC14	0.098	0.244	0.250	0.350
105	BOR PN705	CLSA	SA	PIPE/PSC14	D.100	0.220	0.196	0.480
106	BOR PN795	CLSA	SA	PIPE/PSC14	0.100	0.260	0.200	0.530
107	BOR PN833	CLSA	SA	PIPE/PSC14	0.100	0.110	0.146	0.400
108	BOR PN834	CLSA	SA	PIPE/PSC14	0.100	0.120	0.156	0.500
109	BOR PN835	CLSA	SA	PIPE/PSC14	0.080	0.100	0 200	0.400
110	BOR PN836	CLSA	SA	PIPE/PSC14	0.080	0.080	0.350	0.500
111	BOR PN11	CLSA	LIMESTONE	PSC 12	0.100	0.220	0.100	0.520
112	BOR PN115	CLSA	LIMESTONE	PSC 12	0.100	0.150	0.125	0.700
113	EOD TP2	SASI	SA	HP 12x179	0.112	0.391	0.040	0.240
114	BOR PN2	SI	SASI	HP/PSC24	0.120	0.080	0.504	0.001
115	EOD PN22SE	SA	SANDSTONE	PSC 30	0.090	0.250	0.172	0.250
116	BOR ET2	CLSI	COOPERMARL	HP/PSC24	0.092	0.043	0 230	0.100
117	BOR ET2	CLSI	COOPERMARL	PSC 24	0.082	0.425	0.362	0.148
118	EOD TP1	SASI	ROCK	PSC 12	0 470	0.240	0 200	0.540
119	EOD TP2	SASI	ROCK	PSC 12	0 050	0.290	0.120	0.650
120	BOR PN11B1	CLSA	COOPERMARL	HP/PSC NU	0.128	0.120	0.480	0 021
121	BOR ET3	CLSI	COOPERMARL	PSC	0 096	0.350	0.150	0 350
122	BOR ET4	CLSI	COOPERMARL	HP/PSC	0 091	0 080	0.250	0.150
123	RES PN120	CLSI	COOPERMARL	PSC	0.080	0 360	0.497	0.057
124	EOD PN6	NA	NA	PSC	0.100	0.280	0.180	0.180
125	EOD PN7	NA	NA	PSC NU	0.070	0.100	0.100	0.130
126	EOD PN16	CLSI	SAGR	PIPE 14	0.060	0.300	0.400	0.550
127	RES TP2	CLSI	SAGR	PIPE 12	0.080	0.310	1.100	0.070
128	BOR PN17	CLSI	SAGR	PIPE 14	0.071	0.160	0.810	0.605
129	BOR PN26	CLSI	SAGR	PIPE 14	0.050	0.100	0.749	0.425
130	BOR PN7E	SA	SASTONE	PSC 30	0 090	0.080	0.200	0.628
131	BOR PN7N6	SA	SASTONE	PSC 30	0.078	0.070	0.507	0.625
132	EOD PN6E25	SA	SASTONE	PSC 30 NU	0.100	0.240	0.175	0.200
133	EOD PNSE18	SA	SASTONE	PSC 30 NU	0.100	0.270	0.160	0.220
134	BOR PN5E19	SA	LIMESTONE	PSC 30	0.060	0.190	0.681	0.336
135	BOR PN5E22	SA	LIMESTONE	PSC 30	0.070	0.170	0.800	0.320
136	BOR PN5E25	SA	LIMESTONE	PSC 30	0.060	0.100	0.527	0.253
137	BOR PN5	SA	SASTONE	PSC 30	0.053	0.260	0.181	0.305
138	EOD PN4	SA	SASTONE	PSC 30	0.100	0.250	0.060	0.260
139	EOD TP26	SA	SI	MONO NU	0.100	0.600	0.410	0.062
140	EOD TP27	SA	SI	MONO NU	0.090	0.700	0.451	0.042
141	SOR PN12	SA	SA	CEP 16	0.060	0.100	0.462	0.759
142	EOD PN12	SA	SA	CEP 16	0.060	0.160	0.620	0.348
143	EOD PN3	SASI	SA	HP 12x74	0.170	0.250	0.120	0.420
144	BOR R1	CLS	CL	HP 14x73	0.110	0.110	1.644	0.198
				· · · · · · · · · · · · · · · · · · ·				

Table 25. Side/tip quake and	damping parameters of	f data set PD	(continued).
------------------------------	-----------------------	---------------	--------------

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(in)	(in)	(s/ft)	(s/ft)
145	BOR R10	CLSI	CL	HP 14x73	0.150	0.100	1.669	0.050
146	BOR T1	CLSI	CL	HP 10x42	0 086	0.100	2.058	0.301
147	BOR T2	CLSI	CL	PSC 12	0.150	0.200	1.400	0 080
148	BOR T3	CLSI	CL	PSC 14	0 069	0.360	0.700	0.400
149	BOR T4	CLSI	CL	PIPE 12.75	0.110	0.100	1.298	0.055
150	EOD R1	CLSI	CL	HP 14x73	0.140	0.120	0.898	0.196
151	EOD R10	CLSI	CL	HP 14x73	0.160	0.100	1.508	0.052
152	EOD T1	CLSI	CL	HP 10x42	0.100	0.270	1 631	0.047
153	EOD T2	CLSI	CL	PSC 12	0.220	0.200	0 838	0.032
154	EOD T3	CLSI	CL	PSC 14	0.060	0.310	0.750	0.247
15 5	EOD T4	CLSI	CL	PIPE 12.75	0.120	0.180	0.950	0.025
156	BOR PN1040	CLSI	SA	PSC 14	0.060	0.100	0.404	0.231
157	BOR PN1056	CLSI	SA	PSC 14	0.100	0.360	0.311	0.400
158	BOR PN896A	SA	LIMESTONE	PIPE 9.6	0.200	0.310	0.220	0.634
159	BOR PN898B	SA	LIMESTONE	PSC 14	0.050	0.050	1.347	0.213
160	EOD PNB1	TILL	ROCK	CEP 18	0.100	0.280	0.250	0.200
161	EOD PNB4	TILL	ROCK	CEP 18	0.100	0.320	0 080	0.701
162	EOD PND10	CLTILL	ROCK	OEP 18	0.100	0.320	0.279	0.492
163	EOD 812	NA	NA	PIPE 14	0.250	0.330	0.349	0.319
164	EOD B24	NA	NA	PIPE 14	0.140	0.380	1.300	0.283
165	X TP7	SA	SA	MONO	0.050	0.120	0.993	0.027
168	EOD TP12	SA	SI	MONO	0.040	0.040	1.200	0.150
167	EOD TP2	SA	SI	MONO	0.050	0.050	0.800	0.120
168	EOD TP5	SA	SI	MONO	0.050	0.050	0.749	0.093
169	EOD PN2	SA	SI	MONO	0.050	0.170	0.800	0.080
170	X PNA17.5	SA	SA	PIPE 16	0.060	0.230	0.150	0.453
171	BOR PNH218	SI	COOPERMARL	HP/PSC	0.150	0.150	0.300	0.450
172	EOD PN7	CLSA	LIMESTONE	HP 10x57	0.106	0.090	0.350	0.534
173	EOR PN1058	SASI	SA	PSC 14	0.080	0.250	0.313	0. 095
174	EOD TPS	SI	GR	PIPE 12	0.150	0.220	0.600	0.350
175	RES TP2	SI	GR	PIPE 12	0.150	0.170	0,751	0.436
176	RES TP3	SI	GR	HP 10x12	0 100	0,165	0,600	0.220
177	EOD TP1	CLSA	CLSA	HP 10x42	0.098	0.200	0.250	0.150
178	EOD PN38	CL	TILL	HP 12X74	0.128	0.100	0 900	0.450
179	EOD PN40	CL	TILL	HP 12x74	0.130	0.100	1.000	0.500
180	BOR PN117	SASI	COOPERMARL	HP 14x89	0.089	0.100	1,494	0.254
181	BOR PN224	SASI	COOPERMARL	HP 12x53	0.090	0.250	1,010	0.154
182	TP2 BOR	SI/CL	SA	PSC 12	0.120	0.250	0.250	0.210
183	BOR TP1235	SA	CLSA	PSC 14	0 100	0,150	0.355	0.748
184	BOR TP1259	SA	CLSA	PSC 14	0 100	0.160	0.514	0 530
185	DD PN355E8	SASI	LIMESTONE	HP 10x42	0 100	0.210	0.320	0.430
186	DD PN375P1	SASI	LIMESTONE	HP 10x42	0 100	0,210	0.297	0.492
187	EOD PN375D2	SASI	LIMESTONE	HP 10x42	0,100	0,160	0.320	0.618
188	X TP3	CLSA	LIMESTONE	CEP	0.099	0.070	0.450	0,630
189	BOR PN14	SASI	SA	PSC 14	0.070	0.050	1.009	0.211
190	BOR PN74	SASI	SA	PSC 14	0.070	0.050	0.820	0 100
191	C-41 BOR	COOP MAR		PSC 18	0.059	0.150	1.026	0.289
192	RES PNATO	CLSA	COOPERMAR	PSC/HP	0.000	0.140	0.250	0.200
		~*~	**** BUUMUUN					0.200

Table 25. Side/tip quake and damping parameters of data set PD (continued).

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(i n)	(in)	(s/ft)	(s/ft)
				-				
193	X PN2	CLSI	COOPERMARL	PSC/HP	0.157	0.186	0.264	0.042
194	X PNB141	NA	NA	PIPE 12.75	0.100	0.200	0.010	0.400
195	X PND161.1	NA	NA	PIPE 12.75	0.100	0.260	0.008	0.582
196	PN884380	COOP MAR	COOP MAR	18 OCT.	0.160	0.400	0.390	0.050
197	PN25BOR	COOP MAR	COOP MAR	24 OCT.	0.275	0.340	0.833	0.211
198	BOR 14B1	COOP MAR	COOP MAR	PPC24OCT	0.250	0.300	0.620	0.140
1 99	PN9B BOR	MARK	MARK	24 OCT.	0.250	0.400	0.720	0.122
200	BOR21B12	COOP MAR	COOP MAR	PPC24OCT	0.320	0.370	0.414	0.149
201	DD PN69	CLSI	SA	PSC 12	0.100	0.200	0.020	0.554
202	EOD PN232	CLSI	SA	PSC 12	0.100	0.250	0.105	0.369
203	EOD PN244	CLSI	SA	PSC 12	0.100	0.340	0.012	0.256
204	EOD PN316	CLSI	SA	PSC 12	0 100	0.190	0.086	0.516
205	EOD PN332	CLSI	SA	PSC 12	0.100	0.460	0.075	0.185
208	EOD TP1	CLSI	SA	PSC 12	0.096	0.190	0.040	0.398
207	BOR P5	SA CL	SA	PPC 14	0.098	0.150	0.120	0.407
208	EOD P7	SA CL	SA	PPC 14	0.060	0.310	0.250	0.340
209	EOD P7	SACL	SA	PPC 14	0.060	0.270	0.250	0.340
210	EOD P 10	SA CL	SA	PPC 14	0.080	0.180	0.255	0.400
211	EOD P11	SA CL	SA	PPC 14	0.100	0.100	0.670	0.200
212	EOD P9	SACL	SA	PPC 14	0.100	0.265	0.300	0.150
213	EOD P8	SA CL	SA	PPC 14	0.080	0.200	0.300	0.200
214	EOID P6	SA CL	SA	PPC 14	0.060	0.110	0.350	0.250
215	BOR PN110	SACL	CLSI	PSC 16	0.080	0.090	0.920	0.540
216	BOR PN111	SACL	CLSI	PSC 16	0.100	0.110	1.100	0.100
217	EOD PN110	SACL	CLSI	PSC 16	0.070	0.140	0.550	0.150
218	EOD PN111	SACL	CLSI	PSC 16	0.044	0.072	0.840	0.170
219	EOR PN12	SA	SA/COBBL	HP14x89	0.100	0.145	0.782	0.483
220	EOD PN2	SA	SA/COBBL	HP14x89	0.125	0.200	0.563	0.433
221	PILE 6 B	CL/SI	ROCK	HP14x117	0.100	0.100	0.700	0.150
222	K7EOD	SA/SI	SA/SI	HP14x73	0.050	0.250	1.400	0.200
223	K2BOR	SI/SA	SI/SA	HP14x73	0.047	0.100	1.800	0.220
224	BOR TP13	SA	LIMESTON	HP14x74	0.060	0.055	0.218	0.650
225	EOD A-5-	CLAY	LIMESTON	HP10x57	0.060	0.060	0. 700	0.720
226	7S.ABUT	SA/SI/GR	SA/SI/GR	CEP 10.7	0.100	0.250	0.105	0.456
227	BOR L-8	SI/CL	CL SI	PPC 12	0 140	0.210	0.150	0.490
228	BOR 36	CLAY	LIMESTON	PPC	0.140	0.150	1.050	0.550
229	BOR TP2	CL	SA	TIM, 12	0.040	0.040	1.117	0.350
230	BOR TP2	SA CL	SA CL	TIMBER	0.040	0.040	1.400	0.416
231	TP2 BOR	SA	SA	TIM8.12.	0.050	0.100	1.043	0.215
232	A2 EOID	CL	CL	TIMBER	0.200	0.200	0.620	0.021
233	E2 BOR	CL	CL	TIMBER	0 045	0.130	1.200	0.200
234	PN#2 EOD	SA	SA/GR	TIM 14	0.100	0.350	0.680	0.029
235	EOD TP3	CL SI	CL SI	HP 14x73	0.060	0.170	0.374	0.240
238	TP3 20FT	CL	CL	HP14x73	0.050	0.250	0.340	0.200
237	TP4 EOD	CL	CL	HP 14x73	0.060	0.200	0.330	0.090
238	TP4 BOR	CL	CL	HP 14x73	0.050	0.350	0.400	0.150
239	TP22 BOR	SI/CL	SI/CL	MONO	0.070	0.100	1.700	0.204
240	PN28EOD	CL	ROCK	HP10x57	0.095	0.060	0.310	1.050

.

Table 25. Side/tip	quake and	damping	parameters of	of da	ata set	PD	(continued).	•
--------------------	-----------	---------	---------------	-------	---------	----	--------------	---

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(in)	(in)	(s/ft)	(s/ft)
								ورواي في التركي في التركي
241	EOD 4 DI	CL	ROCK	HP 12x53	0.080	0.060	0.648	0.765
242	EOD J31	CL/SI	ROCK	PP14	0 090	0.385	0.359	0.081
243	BOR J31	CL/SI	ROCK	PP14	0 078	0.229	0 550	0.250
244	DD J31	CLAY TIL	CLAY TIL	CEPIPE 1	0.100	0 550	0.150	D.316
245	BOR PN20	SA/SI	LIMESTON	PSC 14	0.050	0.130	0.240	0.280
246	B1P26BOR	SA	LIMESTON	PSC18-SQ	0.145	0.215	0.193	0.215
247	B11P50BO	SA	LIMESTON	PSC18"SQ	0.044	0.246	0.120	0.320
248	EODP.26	SANDSTON	LIMESTON	PSC18"SQ	0.080	0.130	0.190	0.380
249	B13P48BO	SA	LIMESTON	PSC18*SQ	0.044	0.246	0.120	0.320
250	EOD 258	ALLUVIUM	TILL ALL	CEPIPE 1	0.185	0.300	0.531	0.406
251	BOR 174	ALLUVIAL	ALLUVIAL	CEPIPE 1	0.114	0.100	0.800	0.570
252	BOR PN1	CLSI	SAGR	PIPE 12.75	0.100	0.220	0.578	0.775
253	DD PN10	CLSI	SHALE	PIPE 7	0.080	0.120	0.550	0.150
254	DD PN18	CLSI	SHALE	PIPE 7	0.080	0.100	0.982	0.050
255	EOD PN10.375	CLSI	SHALE	PIPE 7	0 080	0.800	0.284	0.020
256	EOD IP1	SASI	SASI	PSC 12	0 100	0.242	0.550	0.240
257	EOD IP3	SASI	SASI	PSC 12	0 094	0.364	0.416	0,156
258	199.EOD	SA/SISA	SA	12"PSPC	0.055	0.320	0.382	0.075
25 9	293.BOR	SA/SASI	SA	12"PSPC	0 083	0.141	1.849	0.161
260	177.EOD	SCORIA	CLAYSTON	HP10x57	0 090	0.207	0.964	0.285
261	69.EOD	SCORIA	CLAYSTON	HP10x57	0 090	0.303	0.705	0.261
262	BOR TP2	CL-SA	CL-SA	HP14	0 070	0.230	0.350	0.250
263	BOR PNF2	CLSA	CLSA	PSC 12	0 100	0.350	0.522	0.134
264	BOR PNH20	CLSA	CLSA	PSC 12	0 150	0.160	0.509	0.115
265	EOD PNH20	CLSA	CLSA	PSC 12	0 500	0.620	0.258	0.059
268	BOR K521C2	NA	NA	PSC 14	0 110	0.110	1.084	0.381
267	BOR M27A1	NA	NA	PSC 14	0 100	0.150	1.065	0 237
268	BOR M29C3	NA	NA	PSC 14	0 170	0.160	1.164	0.105
2 69	BOR M29C3	NA	NA	PSC 14	0 080	0.275	0.532	0.275
270	DD SD91	CLSA	ROCK	PSC 10 NU	0 080	0.100	0.400	1.000
271	EOD AB345	CLSA	ROCK	PSC 10	0 060	0.100	0.1 73	0.273
272	BOR PNA4W	CLSA	LIMESTONE	PSC 14	0 090	0.090	0.184	0.438
273	EOD TP3	CLSA	LIMESTONE	PSC 10 NU	0 100	0.110	0.202	0.750
274	BOR TP2	CLSA	SA	PSC 14	0 058	0.350	0.120	0.200
275	EOR TP1	CLSA	SA	PSC 14	0 098	0.450	0.020	0.180
276	BOR PN234A3	SASI	COOPERMARL	PSC 18	0 100	0.500	0.106	0.062
277	BOR TP1	SA	CL	MONO 12	0 078	0.050	1.316	0.107
278	RES PN1	SA	SA	MONO 12	0 1 1 9	0.080	1.962	4.277
279	RES PN2	SA	SA	MONO 12	0 036	0.020	1.250	0.250
280	RES PN3	SA	SA	MONO 12	0 105	0.172	1.130	0.242
281	RES PN4	SA	CL	MONO 12	0 090	0.060	1.801	0.297
282	RES PN5	SA	CL	MONO 12	0 047	0.025	7.500	0 393
283	RES PN6	SA	CL	MONO 12	0 056	0.030	1.241	0.443
284	RES PN6	SASI	SASI	PSC 14	0.060	0.060	1.430	0.138
285	EOD TP9	SASI	SASI	PSC 18	0 080	0.240	0.294	0.177
286	RES 858	SASI	SASI	PSC 14	0 0 90	0.165	1.082	0.410
267	RES F14	SASI	SASI	PSC 14	0 090	0.240	0.623	0.147
2 88	RES G37	SASI	SASI	PSC 14	0 1 10	0.300	0.482	0.105

Table 25. Side/tip quake and damping parameters of data set PD (continued).

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(in)	(in)	(s/ft)	(s/ft)
289	RES PN2	NA	NA	PCC 18	0.100	0.240	0.287	0.115
290	RES PN7-61	NA	NA	PCC 18 NU	0.164	0.300	0.367	0.022
291	BOR TP1	SA	SANDSTONE	PSC 14	0.120	0.100	1 621	0.049
292	BOR TP10	SA	SANDSTONE	PSC 14	0.100	0.116	1.473	0.138
293	BOR TP3	SA	SANDSTONE	PSC 18	0 134	0.118	0.965	0.172
294	BOR TP5	SA	SANDSTONE	PSC 14	0 186	0.200	0.841	0.183
295	SOR TP6	SA	SANDSTONE	PSC 18	0.130	0.224	0.530	0.250
296	BOR TP7	SA	SANDSTONE	PSC 14	0.140	0 140	0.999	0.143
297	BOR TP8	SA	SANDSTONE	PSC 14	0.150	0.140	0.927	0.208
298	BOR TP9	SA	SANDSTONE	PSC 14	0.080	0 114	1.447	0.450
299	EOD TP1	SA	SANDSTONE	PSC 14	0.100	0.500	0.157	0.065
300	EOD TP3	SA	SANDSTONE	PSC 18	0.100	0.260	0.102	0.254
301	EOD TP8	SA	SANDSTONE	PSC 14	0.100	0.340	0.311	0.135
302	B15T EOD	SAVCY	SA	MONOTUBE	0 100	0.070	1.594	0.684
303	B9P3BOR	SAICY	SA/CY	18"PCP	0.080	0.125	1.167	0.098
304	S4PC N20	CL	CL	30 PPC	0.110	0.550	0.373	0.023
305	P3 P1.R	SI/SA	SI/SA	18 PPC	0.050	0.100	2.049	0.022
306	83E 2 RE	CL	CL	18 PPC	0.150	0.400	0.156	0.063
307	B70 P5	CL	SA	24 PPC	0.110	0.106	0.099	0 454
308	PIER 7 P	SI/SA	SI/SA	18 PPC	0.080	0.220	1.103	0.035
309	X PN201E2	SASI	COOPERMARL	PSC 18	0 150	0.670	0.040	0.040
310	X PN205E3	SASI	COOPERMARL	PSC 18	0 200	0.750	0.080	0.080
311	X PN209E3	SASI	COOPERMARL	PSC 18	0.120	0.700	0.060	0.110
312	X PN213E2	SASI	COOPERMARL	PSC 18	0.200	0.900	0.080	0.090
313	EOD TP1799	SASI	CL	PPC	0.040	0.620	0.250	0.080
314	RES TP1799	SASI	CL	PPC	0 100	0.100	0.580	0.330
315	X TP1799	SI	SA	PPC	0 050	0.450	0.250	0.150
316	151.EOD	CL/SA	SA	12 PPC	0 085	0.179	0.350	0.350
317	X PN25BK	NA	NA	CEP 20x0 5	0 1 40	0.140	0.600	0.650
318	X PN29K	NA	NA	CEP 20x0 5	0 260	0.210	0.350	0.550
319	X PN30K	NA	NA	CEP 20x0 5	0 2 00	0.150	0.300	0.350
320	X PN2031	SASI	SAROCK	PSC 12	0.1 00	0.320	0.207	0.261
321	BOR TP4	SASI	SA	CEP 12 75	0 1 00	0.250	0.183	0.707
322	EOD TP4	SASI	SA	CEP 12 75	0 060	0.320	0.320	0.481
323	BOR PN13	CLSI	SASI	PSC 12	0 060	0.400	0.470	0.025
324	BOR PN19	CLSI	SASI	PSC 12	0 050	0.330	0.450	0.050
325	BOR PN218	CLSI	SASI	PSC 12	0 210	0.180	0.300	0.220
326	BOR PN26	CLSI	SASI	PSC 12	0.150	0.170	1.000	0.050
327	BOR PN31	CLSI	SASI	PSC 12	0 1 30	0.300	0.700	0.250
328	BOR PN37	CLSI	SASI	PSC 12	0 150	0.180	0.400	0.440
329	BOR PN43	CLSI	SASI	PSC 12	0 1 30	0.550	0.550	0.040
330	BOR PN49	CLSI	SASI	PSC 12	0 100	0.100	0.650	0.200
331	EOD PN13	CLSI	SASI	PSC 12	0 080	0.300	0.250	0.030
332	EOD PN213	CLSI	SASI	PSC 12	0 050	0.400	0.150	0.050
333	EOD PN26	CLSI	SASI	PSC 12	0.070	0.300	0.180	0.070
334	EOD PN49	CLSI	SASI	PSC 12	0.150	0.500	0.200	0.030
335	EOR PN310	CLSI	SASI	PSC 12	0.130	0.145	0.250	0.550
336	X PNA3	CLSA	TILL	CEP 14x0 5	0.070	0.070	0.700	0.500

Table 25.	Side/tip	quake and	damping	parameters of	of data set	PD (continued).

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(in)	(in)	(s/ft)	(s/ft)
337	X PNB5	CLSA	TILL	CEP 14x0.5	0.030	0.050	0.952	0.400
338	X PNG3	CLSA	TILL	CEP 14x0.5	0.200	0.120	0.750	0.200
339	BOR PN128	CL	TILL	PSC 14	0.326	0.218	0.350	0 530
340	BOR PN177	CL	TILL	PSC 14	0.318	0.174	0.320	0.650
341	EOD PN128	CL	TILL	PSC 14	0.366	0.331	0.300	0.514
142	FOD PN177	CI	TIL I	PSC 14	0.120	0.340	0.100	0.500
242	BOR TP1	CISA	CL	PSC 18	0.060	0 170	0.580	0.130
344	BOR TP2	CLSA	CL	PSC 24	0.120	0.180	0.500	0.336
345	BOR TP3	CLSA	SA	PSC 18	0.067	0.090	0.650	0.400
146	BOR TP4	CISA	CI	PSC 18	0.065	0.200	0.000	0.080
347	BOR TPS	CLSA	CL	PSC 24	0.130	0.369	0.350	0.000
347	DOR 1FU	CL3A CL3A	SACP	CED 14-0 27	0.144	0.130	0.000	0.700
340	BOR PN243	31	SAGR	CEP 14x0.37	0.194	0.150	0.800	0.700
349	SOD DN317	51	SAGR	CEP 14X0.37	0.150	0.155	0 600	0.735
350	EOD PN317	51	SAGR	CEP 14X0.37	0.150	0.430	0.550	0.160
351	RE8 (F13	51	SIGR	CEP 1480.37	0.095	0.060	0.950	0.150
372	RESIPO	51	SIGR	CEP 14X0.37	0.050	0.050	1.100	0.090
353	RESIPI	SA			0.120	0.120	1.150	0.500
354	RES TP2	SA	CL CL	PIPE 12	0.200	0.160	0.700	0,700
355	RES TP3	SA	CL	PIPE 11 NU	0.077	0.069	0.400	0.270
356	RES TP4	SA	CL	PIPE 12	0.150	0.150	1.000	0.539
357	G-16-1.B	CL/SA	SA	PIPE 10	0.100	0.152	0.550	0.872
358	151-BOR	CL,SC,S	SA	12 PPC	0 085	0.142	0.380	0.385
359		NA	NA	PPC 12	0.075	0.150	0.360	0.380
360	A4-21-EO	CL	CL	26 PIPE	0.150	0.100	0.700	0.500
361	EOD PN3x011	CLSI	LIMESTONE	PIPE	0.167	0.198	0.250	0.400
362	EOD TP	CLSI	LIMESTONE	PIPE	0.140	0.200	0.353	0.677
363	TP8	SA	SA	OCT16.5	0.160	0.225	1.150	0.450
364	BOR TP1	SI	51	HP305MMX	0.080	0.300	1.800	0.100
365	80R 843P	OOPERMAR	COOPERMARL	18 OCT.	0.140	0.220	0.865	0.234
366	BOR B12P	SA/SI	LIMESTON	PSC 18	0.035	0.080	0.174	0.798
367	EORPN20	SA	LIMESTON	PSC18"SQ	0.110	0.150	0.400	0.137
368	EODPN23	SA	LIMESTON	PSC18"SQ	0.110	0.150	0.400	0.137
369	EOD TP1	SAND SAT	SAND SAT	CEPIPE 1	0.100	0.250	0.640	0.440
370	BOR TP1	SA SATUR	SA SATUR	E\CEPIPE	0.080	0.080	1.265	0.346
371	BORPN50	SA/SI	SA	PP12.75	0.044	0.048	1.800	0.700
372	EOD-15-3	SA SI	LIMESTON	PSC10	0.060	0.100	0.201	0.463
373	BOR-TP4	SA SI	SA ŜI	HP14x11	0.050	0 060	0.720	0.030
374	BOR-PN43	SA	CL	MONO 11	0.075	0.150	3 800	G.400
375	EOD-TP1	SA SI	SA SI	CEP16	0.090	0.240	0 157	0 974
376	BOR-F6/4	CI	LIMESTON	PSC1650	0.080	0.100	0.465	0.395
377	BOB 20MB	NA	NA	PBC 24	0.000	0.160	0.254	0.643
279	10 9 120	110 64 61	54.51	COMPOSIT	0.100	0.160	0.204	0.090
370	EOB 027	54.01		DD14	0.197	0.100	0.000	0.050
3/8		SACL	SA UL	24750	0.000	0.400	0.43/	0.313
380	0.27°.105	SACL	LIMESTON	29 50.	0.090	0.300	0.270	0.200
367	EUDIPZ	NA	SHALE	HP10x5/	0.083	0.165	0.330	0.707
382	BUR 1386	OVERBURD	MARL	24-Oct	0.274	0.200	0.545	0.141
383	BOR 58.5	MARL	MARL	18 OCT.	0.180	0.180	1.208	0.071
384	F20 5 MI	SI	LIMESTON	24 PPC	0.100	0.180	0.058	0.244

Table 25. Side/tip quake and damping parameters of data set PD (contin	ued).
--	-------

					SIDE	TIP	SIDE	TIP
REF	PILE	SKIN	TOE	PILE	QUAKE	QUAKE	DAMPING	DAMPING
NO.	NAME	SOIL	SOIL	TYPE	(in)	(in)	(s/ft)	(s/ft)
			· · · _ _					
365	EOD FHA5	SA	SA TILL	CEP18	0.100	0.320	0.250	0.170
386	BOR PN248	CLSA	COOPERMARL	HP 14x73	0.080	0.080	1.010	0.316
387	TP4EOD	SA	ROCK	HP12x74	0.080	0.340	0.100	0.480
388	PN8 EOD	SA	SHALE	HP10x57	0.090	0.127	0.234	0.651
389	PN7 EOD	SA	SHALE	HP10x57	0.090	0.127	0.234	0.651
390	EOD PN7	CL SA	CL SA	CEPIPE24	0.075	0.330	0.240	0.116
391	BOR PN7	CL SI	CL SI	CEPIPE24	0.075	0.330	0.240	0.116
392	BOR PN3	CL	CL	24 PP	0.147	0.672	0.200	D.087
393	BOR PN3	CL	CL	24 PP	0.170	0.340	0.188	0.250
394	EOD PN1	SA	LIMESTON	18 PSC	0.075	0.220	0.281	0.133
395	BOR TP4	SA	SA	12 PP	0.250	0.340	0.550	0.450
396	BOR2/59	MARL	MARL	PPC18OCT	0.110	0,110	0.720	0.102
397	BOR4/58	MARL	MARL	PPC18OCT	0.120	0.200	0.800	0.103
398	TP2 BOR	SA SI	SA SI	24 PSPC	0.140	0.140	1.200	0.285
399	TP1 EOD	SA CL	SA CL	HP14x73	0.080	0.280	0.250	0.750
400	BOR TP1	SA CL	SACL	HP14x73	0.080	0.170	0.550	0.838
401	PN B EOD	SA	LIMESTON	24 PSC	0.100	0.150	0.250	0.290
402	EOD TP2	CL SI	CL SI	HP14x73	0.090	0.320	0.240	0.208
403	BOR TP2	CL SI	CL SI	HP14x73	0.100	0.320	0.400	0.276

Table 25. Side/tip quake and damping parameters of data set PD (continued).

1 in = 25.4 mm 1 s/ft = 3.281 s/m

.

.

REFERENCES

- American Society for Testing and Materials, 1992. "Standard Test Method for Piles Under Static Axial Compressive Loads," D-1143-81, Vol. 4.08, Sec. 4, pp. 195-205.
- Abe, S., G. Likins, and C.M. Morgan. 1990. "Three Case Studies on Static and Dynamic Testing of Piles," *Geotechnical News*, December 1990, pp. 26-32.
- Benjamin, J.R. and C.A. Cornell. 1970. Probability, Statistics, and Decision for Civil Engineers. McGraw-Hill.
- Bernardes, G.D.P. 1989. Dynamic and Static Testing of Large Model Piles in Sand, Eng. Ph.D. Dissertation, Dept. of Civil Engineering, Norwegian Institute of Technology.
- Bowles, J.E. 1988. Foundation Analysis and Design. McGraw-Hill, 4th ed.
- Briaud, J.L. and C.M. Tucker. 1988. "Measured and Predicted Axial Response of 98 Piles," American Society of Civil Engineers Journal of Soil Mechanics and Foundations, Div. 114(9): 984-1001.
- Bustamante, M.G. and L.P. Weber. 1988. "Dynamic and Static Measurements of Steel H-Pile Capacities," 3rd International Conference of Stress-Wave Theory in Piles, Ottawa, Canada, pp. 579-589.
- Butler, H.D. and H.E. Hoy. 1977. Users Manual for the Texas Quick-Load Method for Foundation Load Testing. Federal Highway Administration, Office of Development, Washington, DC. Report No. FHWA-IP-77-8, 59 pp.
- Chellis, R.D. 1961. Pile Foundations. McGraw-Hill, 2nd ed.
- Cheng, S.S.M. and S.A. Ahmad. 1988. "Dynamic Testing Versus Static Loading Test: Five Case Histories," 3rd International Conference of Stress-Wave Theory in Piles, Ottawa, Canada, pp. 477-489.
- Chernauskas, L.R. 1993. Dynamic Analysis of Plugged Piles in Clay, Master of Science Thesis submitted to the Department of Civil Engineering, University of Massachusetts-Lowell, 1993.
- Cummings, A.E. 1940. "Dynamic Pile-Driving Formulas," Boston Society of Civil Engineers Journal 1925-1940, January 1940, pp. 392-413.

Preceding page blank

- Davisson, M.T. 1972. "High Capacity Piles," Proceedings, Soil Mechanics Lecture Series on Innovations in Foundation Construction, American Society of Civil Engineers, Illinois Section, Chicago, March 22, 1972, pp. 81-112.
- DeBeer, E.E. 1970. "Proefondervindellijke bijdrage tot de studie van het grandsdraagvermogen van zand onder funderinger op staal." English version, *Geotechnique*, Vol. 20, No. 4, pp. 387-411.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-Proposed Cimarron River Bridge U.S. 64, Buffalo, Oklahoma: FHWA Bridge and Demonstration Projects Divisions, February 1989.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-Replacement Bridge State Route 115 over Missouri River, Bridgeton/St. Charles, Missouri: FHWA Bridge and Demonstration Projects Divisions, April 1989.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-I-80/480 Interchange, Omaha, Nebraska: FHWA Bridge and Demonstration Projects Divisions, October 1989.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-State Highway 77 Fore River Bridge Replacement, Portland-South Portland, Maine: FHWA Bridge and Demonstration Projects Divisions, April 1990.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-Mo-Pac Railroad Overpass Route LA, 415, West Baton Rouge, Louisiana: FHWA Bridge and Engineering Applications Divisions, November 1990.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-Hartford Bridge BRZ1444, White River Junction, Vermont: FHWA Technology Applications and Bridge Divisions, January 1991.
- Dynamic Pile Monitoring and Pile Load Test Report Demonstration Project 66-State Route 15, Section 63M, Tioga County, Pennsylvania: FHWA Technology Applications and Bridge Divisions, December 1991.
- DiMaggio, J. 1986. <u>Dynamic Pile Monitoring and Pile Load Test Report-Proposed</u> <u>Relocated U.S. 61 over the Peosta Channel</u>, Dubuque, Iowa: FHWA Bridge Division and Demonstration Projects Division, December 1986.
- DiMaggio, J. 1991. <u>Dynamic Pile Monitoring and Pile Load Test Report Demonstration</u> <u>Project 66-County Route 18 over Minnesota River</u>, Bloomington, Minnesota: FHWA Engineering Applications and Bridge Divisions, January 1991.

- Dumas, C. 1993. <u>Dynamic Pile Monitoring and Pile Load Test Report Demonstration</u> <u>Project 66-Central Bridge over the Ohio River-U.S. 27</u>, Campbell County, Kentucky: FHWA Technology Applications and Bridge Divisions, January 1993.
- Edde, R.D. and B.H. Fellenius. 1990 "Static or Dynamic Test Which to Trust," *Geotechnical News*, December 1990, pp. 28-32.
- Fellenius, H.B. 1989. Guidelines for the Interpretation and Analysis of the Static Loading Test, Deep Foundations Institute.
- Fellenius, H.B. 1991. "Pile Foundations," Chapter 13 of Foundation Engineering Handbook, edited by Hsai-Yang Fang. Van Nostrand Reinhold. New York, NY. 2nd edition.
- Flaate, K. 1964. An Investigation of the Validity of Three Pile-Driving Formulae in Cohesionless Material, Norwegian Geotechnical Institute, Publication 56, Oslo, Norway.
- Fox, E. 1932. "Stress Phenomena Occurring in Pile Driving," Engineering Journal, London, England, Vol. 134.
- FPDS-3 Foundation Pile Diagnostic System-3, TNO Building and Construction Research, Delft, The Netherlands, 1993.
- Graff, K.F. 1975. Wave Motion in Elastic Solids, Ohio State University Press, Columbus, Ohio.
- Goble, G.G., G. Likins, and F. Rausche. 1970. Dynamic Studies on the Bearing Capacity of Piles - Phase III, Report No. 48, Division of Solid Mechanics, Structures, and Mechanical Design. Case Western Reserve University.
- Goble, G.G., G. Likins, and F. Rausche. 1975. Bearing Capacity of Piles from Dynamic Measurements, Final Report, Ohio Dept. of Trans., OHIO DOT-05-75.
- Goble, G.G., R.H. Scanlan, and J.J. Tomko. 1967. Dynamic Studies on the Bearing Capacity of Piles, Phase II, Volumes I and II, Case Institute of Technology.
- Goble, G.G., F. Rausche, and G. Likins. 1980. "The Analysis of Pile-Driving: A State of the Art," Proceedings from the 2nd Conference on the Application of Stress-Wave Theory on Piles, Stockholm, Sweden, June 1980, pp. 1-34.
- Gravare, C.J., G.G. Goble, F. Rausche, and G. Likins. 1980. "Pile-Driving Construction Control by the Case Method," Ground Engineering, pp. 21-24.

- Highway Research Record No. 167. 1967. Bridges and Structure, Highway Research Board, Washington, DC.
- Highway Research Record No. 333 1970. Pile Foundations, Highway Research Board, Washington, DC.
- Holtz, R.D. and W.D. Kovacs, 1981. An Introduction to Geotechnical Engineering. Prentice-Hall, Inc., Englewood Cliffs, New Jersey.
- Housel, W.S. 1965. "Michigan Study of Pile-Driving Hammers," Proceeding of the ASCE Journal of Soil Mechanics and Foundations, Volume 91, No. SM5, September 1965, pp. 37-64.
- Housel, W.S. 1970. "Pile Load Capacity: Estimates and Test Results," Proceeding of the ASCE Journal of Soil Mechanics and Foundations, Volume 92, No. SM4, July 1966, pp. 1-30.
- Issacs, D. 1931. "Reinforced Concrete Pile Formula," Transactions of the Institute of Engineers, Australia, Vol. 12, pp. 312-323.
- Iwanowski, T. 1987. "Stress-Wave Testing of Piles," Proceedings of the International Conference on Foundations and Tunnels, University of London, March 1987, pp. 262-266.
- Kazmierowski, T. and M. Devata. 1978. Pile Load Capacity Evaluation, HWY 404 Structures, Report EM-20, Ontario Ministry of Transportation and Communications, Engineering Materials Office, Soil Mechanics Section. July 1978.
- Kazmierowski, T. and M. Devata. 1983. Evaluation of Selected Piles Under Axial and Lateral Loading Conditions - C.N.R./C.P.R. Structures Northwest Metro Arterial, Ontario Ministry of Transportation and Communications, Engineering Materials Office, Soil Mechanics Section. Feb. 1983.
- Lambe, T.W. and R.V. Whitman, 1969. Soil Mechanics. John Wiley and Sons Inc., New York, NY.
- Likins, G., F. Rausche, and M. Hussein, 1990. "Introduction to the Dynamics of Pile Testing," *Geotechnical News*, December 1990, pp. 21-23.
- Lowery, L.L., T.J. Hirsh, T.C. Edwards, H.M. Coyle, and C.H. Samson, 1969. *Pile-Driving* Analysis - State of the Art Research Report 33-13 (Final), Texas Highway Department, Research Study No. 2-5-62-33.

- McDonnell, J. 1991. The Energy Approach Method for the Prediction of Pile Capacity, A Research Project Submitted in Partial Fulfillment of the Requirements for the Degree of Master of Science (Unpublished).
- Michigan State Highway Commission. 1965. A Performance Investigation of Pile-Driving Hammers and Piles.
- Middendorp, P. and P.J. van Weel. 1986. "Application of Characteristic Stress-Wave Method in Offshore Practice." Proceedings of the 3rd International Conference on Numerical Methods in Offshore Piling.
- Ohio Department of Transportation. 1975. Bearing Capacity of Piles from Dynamic Measurements, Research Report OHIO-DOT-05-75, Final Report.
- Olsen, R.E. and K.S. Flaate. 1967. "Pile-Driving Formulas for Friction Piles in Sand," ASCE Journal of Soil Mechanics and Foundations. Vol. 92(6): pp. 279-297.
- Olsen, R.E. and N. Dennis. 1982. "Actual Capacity of Driven Pipe Piles in Clay," Proceedings of the Symposium on State of the Art in Offshore Geotechnical Engineering, American Society of Civil Engineers.
- Paikowsky, S.G. 1982. Use of Dynamic Measurements to Predict Pile Capacity Under Local Conditions, M.Sc. Thesis, Dept. of Civil Engineering Technion-Israel Institute of Technology, July 1982.
- Paikowsky, S.G. 1984. "Use of Dynamic Measurements for Pile Analysis," including PDAP-Pile-Driving Analysis Program, GZA Inc., Newton, Massachusetts, 1984.
- Paikowsky, S.G., R.V. Whitman, and M.M. Baligh. 1989. "A New Look at the Phenomenon of Offshore Pile Plugging," Marine Geotech., 8(3): 213-230.
- Paikowsky, S.G. 1990. Investigation of Pile Foundation Behavior, Northern Avenue Bridge, Boston, MA, Consulting/Research Report submitted to GZA, Inc.
- Paikowsky, S.G. and R.V. Whitman. 1990. "The Effects of Plugging on Pile Performance and Design," *Canadian Geotechnical Journal*, Vol. 27, No. 3, August 1990.
- Paikowsky, S.G. and L.R. Chernauskas. 1992. "Energy Approach for Capacity Evaluation of Driven Piles," 4th International Conference on the Application of Stress-Wave Theory to Piles, The Hague, The Netherlands, pp. 595-601.

Pile Dynamics Inc. 1990. Model GCPC Pile-Driving Analyzer, Manual.

- Poulos, H.G. and E.H. Davis, 1980. Pile Foundation Analysis and Design, Robert E. Krieger Publishing Co., 2nd edition.
- Prakash, S. and H.D. Sharma. 1990. *Pile Foundations in Engineering Practice*. John Wiley and Sons Inc., New York, NY.
- Rausche, F., F. Moses, and G.G. Goble. 1972. "Soil Resistance Predictions from Pile Dynamics," ASCE Journal of Soil Mechanics and Foundations. Vol. 98, No. SM9, Sept. 1972, pp. 917-937.
- Rausche, F., G.G. Goble, and G.E. Likins, 1985. "Dynamic Determination of Pile Capacity," Journal of Geotechnical Engineering, ASCE, Vol. III, No. 3, pp. 367-383.
- Reiding, F.J., P. Middendorp, R.P. Schoenmaker, F.M. Middeldorp, and M.W. Bielefeld. 1988. "FPDS-2, A New Generation of Foundation Pile Diagnostic Equipment," 3rd International Conference on Stress-Wave Theory in Piles, Ottawa, Canada, pp. 123-134.
- Riker, R.E. and B.H. Fellenius. 1988. "Case Method Capacity Estimates for Piles in Glacial Soils," 3rd International Conference on Stress-Wave Theory in Piles, Ottawa, Canada, pp. 565-578.
- Ryan, T.P. 1989. "Linear Regression," Chapter 13 of Handbook of Statistical Methods for Engineers and Scientists, H.M. Wadsworth, editor. McGraw-Hill.
- Schmertmann, J. 1991. "The Mechanical Aging of Soils," The 25th Terzaghi Lecture, Journal of Geotechnical Engineering, ASCE, Vol. 117, No. 9, Sept. 1991, pp. 1288-1303.
- Selby, K.G., M.S. Devata, P. Prayer, and D. Dundas, 1989. "Ultimate Capacities Determined by Load Test and Predicted by the Pile Analyzer," Proceeding of the 12th International Conference on Soil Mechanics and Foundation Engineering, Rio De Janeiro, August 13-18, 1989, pp. 1179-1182.
- Smith, E.A.L. 1960. "Pile-Driving Analysis by the Wave Equation," Journal of Soil Mechanics and Foundations, American Society of Civil Engineers, August 1960, pp. 35-61.
- Stokes, W.L. and D.J. Varnes, 1955. Glossary of Selected Geologic Terms, Colorado Scientific Society Proceedings, Volume 16.
- Taylor, Donald W. 1948. Fundamentals of Soil Mechanics, John Wiley and Sons, Inc. 12th edition.

- Terzaghi, K. 1942. "Discussions of the Progress Report of the Committee on the Bearing Value of Pile Foundations," *Proceedings ASCE*, Vol. 68:311-323.
- Texas Highway Department. 1973. Bearing Capacity for Axially Loaded Piles, Research Report 125-8-F, Sept. 1967-Aug 1973, pp. 134.
- Thompson, C.D. and M. Devata. 1980. "Evaluation of Ultimate Bearing Capacity of Different Piles by Wave Equation Analysis," Proceedings from the 2nd Conference on the Application of Stress-Wave Theory on Piles, Stockholm, Sweden, June 1980, pp. 1-33.
- Thompson, C.D. and G.G. Goble. 1988. "High Case Damping Constants in Sand," 3rd International Conference on Stress-Wave Theory in Piles, Ottawa, Canada, pp. 464-555.
- Trow Report. 1978. <u>Research Project: Dynamic Behavior of Foundation Piles and</u> <u>Driving Equipment.</u> The Trow Group Ltd., Ontario, Canada.
- Vanikar, S.N. 1984. <u>Dynamic Pile Monitoring and Pile Load Test Report-I-90, Third</u> <u>Lake Washington Bridge</u>). Seattle, Washington: FHWA Office of Highway Operations Demonstration Projects Program, April 1984.
- Vanikar, S.N. 1987. <u>Dynamic Pile Monitoring and Pile Load Test Report Proposed</u> <u>Alsea River Bridge (Oregon Coast Highway 101)</u>, Walport, Oregon: FHWA Bridge and Demonstration Projects Divisions, June 1987.
- Vanikar, S.N. 1987. <u>Dynamic Pile Monitoring and Pile Load Test Report Bridges on</u> <u>Colorado S.H. 55 (BRS 0055(4))</u>, Crook, Colorado: FHWA Bridge and Demonstration Projects Divisions, August 1987.
- Vesic, A.S. 1977. Design of Pile Foundations, National Cooperative Highway Research Program. Synthesis of Highway Practice, Publication No. 42.

Veneziano, D. 1993. Personal Communication.

-.